Rethinking mobile learning for development:

Using the Capability Approach and a mixed-methods systematic review to conceptualise the application of mobile technologies as an educational tool in Low- and Middle-Income Countries

> Submitted for the degree of Doctor of Philosophy

> > Laurenz Langer

University College London

October 2017

Declaration

I, Laurenz Langer, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Word count (excluding references and appendices): 95,901.

[signature removed for online repository]

Laurenz Langer

Abstract:

Mobile technologies are now ubiquitous in Low- and Middle-Income Countries (LMICs). Education in LMICs is among the latest disciplines set out to explore how mobile technologies' particular affordances can be used to support programme interventions and practice. However, neither the effectiveness of mobile learning programmes, nor the causal mechanisms and contexts through which teaching and learning with mobile technologies is assumed to support education and development in LMICs, have been systematically reviewed and conceptualised. Notwithstanding, learning and teaching with mobiles is regularly attributed as having the potential to change and improve education in LMICs. This thesis assesses the effectiveness and underlying theory of change of mobile learning programmes in LMICs through a mixed-methods systematic review, including meta-analysis and thematic synthesis. Building on the findings of the systematic review, it presents the Capability Approach as a theoretical lens through which to conceptualise the effects of mobile technologies on education and development in LMICs.

Reviewing the evidence-base and theory of change of the application of mobile technologies as an educational tool in LMICs, I find little evidence to support claims to mobile learning's potential to support development outcomes. These systematic review findings are then expanded in a qualitative case study of a mobile learning project in rural South Africa exploring teachers' use of mobile technologies from the perspective of the Capability Approach. The case study finds that teachers' use of mobile technologies can best be understood as an expansion in four dimensions of capabilities: informational, educational, societal, and economic capabilities, which taken together can enhance teachers' well-being and human development. I use the combined case study and systematic review findings to reposition mobile learning's role in international development. I argue that conceptualising mobile learning for development through the Capability Approach supports a focus on an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of actors in the education system. In the absence of evidence supporting mobile learning's impact on development outcomes, a focus on the role of mobile technologies to expand teachers' and learners' valued functionings and capabilities is presented as an alternative conception of the links between mobile technologies, their use for educational purposes in LMICs, and development outcomes.

Table of contents

| Chapter 1. Introduction to the thesis | 11 |
|---|------------|
| 1.1 Research background | 13 |
| 1.2 Research outline and questions | 19 |
| 1.3 Contribution of the thesis | 23 |
| 1.4 Structure of the thesis | 25 |
| 1.5 Key terminology | 21 |
| <u>PART I:</u> a systematic review of the empirical and conceptual evidence-base mobile learning's contribution to development outcomes | for 28 |
| Chapter 2. Literature review 1: Mobile Learning for Development | 29 |
| Introduction | 29 |
| 2.1 Mobile learning | 31 |
| 2.1.1 An introduction to mobile learning | 31 |
| 2.1.2 Definitions of mobile learning | 32 |
| 2.1.3 Mobile learning's pedagogical contribution | 34 |
| 2.2 Mobile learning in LMICs | 36 |
| 2.2.1 An introduction to mobile learning in LMICs | 36 |
| 2.2.2 Defining mobile learning for development | 38 |
| 2.2.3 Mobile learning's assumed contribution to development intervention | |
| 2.3 1 Critique 1: Mobile learning's empirical performance in LMICs | 4 5 |
| 2.3.2 Critique 2: Simplistic causalities between mobile technologies, education, and | |
| development | 47 |
| 2.3.3 Critique 3: A techno-centric approach to ML4D and its risks | 49 |
| Conclusion | 52 |
| Chanter 3 Literature review 2: the Canability Approach | 53 |
| Introduction | 53 |
| 3.1 Introducing the Capability Approach to ML4D | 54 |
| 3.1.1 The Capability Approach: theory and key concepts | 54 |
| 3.2 Investigating technology and development from a capabilities perspective | 60 |
| 3.3 Capabilities, mobile technologies, and education in LMICs | 64 |
| 3.4 ML4D and the Capability Approach | 67 |
| Conclusion | 68 |
| Chapter 4. Research design and methodologies | 69 |
| Introduction | 69 |
| 4.1 Research design: overall thesis | 70 |
| 4.2 Research design: systematic review | 73 |
| 4.2.1 Introduction to systematic review as a research methodology | 73 |
| 4.2.2 Mixed-methods reviews | 75 |
| 4.2.3 Mixed-methods review design | 77 |
| 4.2.4 Summary of systematic review steps | 80 |
| 4.2.5 Ethics in conducting the systematic review research | 88 |
| 4.2.6 Limitations of the systematic review | 88 |
| 4.3 Kesearch design: case study research | |
| 4.3.1 Introduction to case study research as a research methodology | |
| 4.3.2 Qualitative case study design | ອາ ດວ |
| | 92 |

| 4.3.4 Ethics in conducting the case study research | 99 |
|--|------------|
| 4.3.5 Limitations of the qualitative case study | 101 |
| Conclusion | 103 |
| Chapter 5. The aggregate effects of mobile technologies on education in | |
| LMICs | 104 |
| Introduction | 104 |
| 5.1 What research evidence do we have on mobile learning in LMICs? A descrip | otive |
| map of the evidence | 106 |
| 5.1.1 What research evidence did I identify? (systematic search results) | 106 |
| 5.1.2 Socio-economic contexts within the available evidence-base | . 108 |
| 5.1.5 Types of mobile learning interventions within the evidence-base | . 113 |
| 5.1.5 The design of ML4D interventions | 114 |
| 5.1.6 Assumptions about how ML4D interventions facilitated educational change | 118 |
| 5.2 Meta-analysis findings: aggregate effects of ML4D interventions | 122 |
| 5.2.1 How trustworthy is the quantitative evidence? | 122 |
| 5.2.2 Do mobile learning interventions improve learning outcomes in LMICs? | 123 |
| 5.2.3 Do mobile learning interventions improve teaching practice in LMICs? | 134 |
| 5.2.4 Do mobile learning interventions improve EMIS in LMICs? | . 130 |
| 5.2.5 Do mobile learning interventions a cost-effective approach to support educat | 137 ion |
| in LMICs? | 139 |
| Conclusion | 142 |
| Chanter 6. Configuring a theory of change for the use of mobile technologi | A 6 |
| to support education in LMICs | 146 |
| Introduction | |
| 6.1 Thematic synthesis: contexts and mechanisms configuring the impact of M | L4D |
| interventions | 148 |
| 6.1.1 An overview of the included qualitative evidence | 148 |
| 6.1.2 How trustworthy is the qualitative evidence (critical appraisal of studies)? | 149 |
| 6.1.3 Thematic synthesis findings: What mechanisms and contexts mitigate and | 151 |
| 6 2 A theory of change for ML4D drawing on my mixed-methods synthesis | 161 |
| 6.2.1 Developing a theory of change for MI 4D interventions | 168 |
| 6.2.2 Putting the theory of change to work: assessing Mobile Learning 'for' | |
| Development? | 171 |
| 6.3 What have we learnt from the evidence-base? Discussion of systematic revi | iew |
| findings | 179 |
| 6.3.1 A mobile revolution? Claims to ML4D's impact | 179 |
| 6.3.2 The limitations of access to technologies | . 181 |
| 6.3.4 Education as a broken service? | 103 185 |
| Conclusion | 188 |
| | |
| PART II: Developing a capabilities conceptualisation of mobile learning's | 400 |
| contribution to development | 190 |
| Chapter 7. Using mobile technologies to expand teachers' capabilities: A | |
| qualitative case study of a mobile learning programme in rural South Africa | a 191 |
| Introduction | 191 |
| 7.1 Overlap between the systematic review findings and the CA | 193 |
| 7.3 Investigating the case study research | 190 |
| perspective | 199 |

| 7.3.1 Informational capabilities | 199 |
|---|-------------|
| 7.3.2 Educational capabilities | 204 |
| 7.3.3 Economic capabilities | 208 |
| | 212 |
| Conclusion | 219 |
| Chapter 8. Conceptualising mobile learning for development from a | |
| capabilities perspective | .220 |
| Introduction | 220 |
| 8.1 A framework of teachers' use of mobile technologies as an educational tool | |
| from a capabilities perspective | 222 |
| 8.1.1 Introducing the capabilities framework | 222 |
| 8.1.2 How to interpret and apply the framework | 226 |
| 8.2 Development as freedom and ML4D | 228 |
| 8.2.1 Vignette: Building the nation | 232 |
| 8.2.2 Vignette: Alternative ways to conceptualise what being a teacher means | 234 |
| 8.2.3 Vignette: We are moving with time—we are in demand | 236 |
| 8.3 Towards a capabilities conception of ML4D | 238 |
| Conclusion | 243 |
| Chapter 9. Rethinking ML4D—positioning mobile learning in international | |
| development | 245 |
| Introduction | 245 |
| 9.1 Comparing the capabilities conception of ML4D with my systematic review | |
| findings | 246 |
| 9.2 Finding a space for mobile learning in international development | 249 |
| 9.2.1 Opportunity-focused ML4D | 250 |
| 9.2.2 People-focused ML4D | 251 |
| 9.2.3 Education-focused ML4D | 251 |
| 9.2.4 Transformation-focused ML4D | 252 |
| 9.3 Defending the argument for my conception of ML4D | 255 |
| 9.3.1 Transformation from within, undermining structural changes? | 255 |
| 9.3.2 Mobile learning pedagogies and development as freedom | 259 |
| 9.3.3 Meta capabilities: A quest for the magic bullet capability? | 262 |
| 9.3.4 Revisiting the gap in existing capabilities frameworks | 264 |
| 9.3.5 In defence of human capital and economic growth | 267 |
| Conclusion | 269 |
| Chapter 10. Conclusion of the thesis | 270 |
| Introduction | 270 |
| 10.1 Limitations of the research | 270 |
| 10.2 Summary of findings and contribution of the thesis | 270 |
| 10.2.1 Summary of findings and argument | 271 |
| 10.2.2 Review of thesis contributions | 273 |
| 10.3 Implications of the research findings for policy and practice | 274 |
| 10.3.1 Implications of my systematic review findings | 274 |
| 10.3.2 Implications of my capabilities conception of ML4D | 276 |
| 10.4 Implications of the thesis for future research | 279 |
| Conclusion | 281 |
| References | .283 |
| Annondiaas | 200 |
| Appendices Annendix 4 1: Search strategy and report | .∠30 200 |
| Appendix 4.1. Search strategy and report Annendix 4.2: Mixed-methode critical annraisal tool | 215 |
| Appendix 4.3. Effect size calculations | 326 |
| Appendix 4.4: Interview guide | 328 |
| · · · · · · · · · · · · · · · · · · · | |

| Appendix 4.5: Observation schedule | 330 |
|--|-------|
| Appendix 5.1: List of included studies | 332 |
| Appendix 5.2: Overview of included studies | 338 |
| Appendix 5.3: Detailed overview of studies included in the meta-analysis | 349 |
| Appendix 5.4: More detailed analysis of mobile learning pedagogies | 356 |
| Appendix 6.1: Overview of mechanism and context themes | 360 |
| Appendix 6.2: Example of analysis in thematic synthesis | 367 |
| Appendix 6.3: A narrative summary of the best available aggregative evidence | ə 373 |
| Appendix 7.1: On the absence of negative case study findings | 377 |
| Appendix 7.2: Exhaustive list of all capabilities and functionings | 379 |

List of figures

| Figure 1.1 Overlap of ML4D with related bodies of literature | 15 |
|--|----------|
| Figure 4.1 Visual overview of the research design and methodologies | 72 |
| Figure 4.2 Systematic review approach | 78 |
| Figure 4.3 Geographical location of the ICT4RED | 94 |
| Figure 4.4 Examples of ICTs provided to schools and teachers | 95 |
| Figure 4.5 ICT4RED professional teacher development course | 96 |
| Figure 5.1 Search results | 107 |
| Figure 5.2 Geographical location of included studies | 110 |
| Figure 5.3 Design characteristics of ML4D programmes (technological/educational) | 116 |
| Figure 5.4 Targeted pedagogies and mobile technologies applied | 120 |
| Figure 5.5 Overview of risk of bias assessment | 123 |
| Figure 5.6 Forest plot: Average effect of mobile learning programmes on learning ou | ıtcomes |
| in LMICs | 124 |
| Figure 6.1 Overview of critical appraisal: qualitative research | 151 |
| Figure 6.2 A theory of change for ML4D | 170 |
| Figure 8.1 A framework of teachers' use of mobile technologies as an educational to | ool from |
| a capabilities perspective | 224 |
| Figure 8.2 Capabilities configuration for 'building the nation' | 232 |
| Figure 8.3 Capabilities configuration for 'alternative ways to conceptualise what bein | ıg a |
| teacher means' | 234 |
| Figure 8.4 Capabilities configuration for 'we are moving with time—we are in deman | d' 236 |

List of tables

| Table 5.1 Overview of mobile learning programme types | 112 |
|---|-----|
| Table 5.2 Overview of mobile devices' technological affordances | 114 |
| Table 5.3 Sensitivity analyses | 127 |
| Table 5.4 Moderator analyses | 130 |
| Table 5.5 Overview of cost-effectiveness calculations | 139 |
| Table 5.6 Overview of meta-analysis results | 144 |
| Table 6.1 Summary of mechanism themes | 153 |
| Table 6.2 Summary of context themes | 161 |

Acknowledgements

My PhD was a beautiful journey.

It would not have been possible if not for the two most amazing and supportive travel guides, Ruth Stewart and Niall Winters, without who's guidance, motivation, kindness, humour, and intellect, I'd still be wondering where to go.

This journey also would not have been possible if not for my parents and grandparents, whose travels and sacrifices in life created the foundation for my own journey and adventure.

Along the journey, I met many travel companions whose ideas and advice manoeuvred me out of dead ends and helped me sustain the joy in the journey—thank you colleagues at the Africa Centre for Evidence and UCL EPPI-Centre for providing an intellectual home throughout.

I went on this journey because I believe that research can make a difference. I found a wealth of existing knowledge, the producers of which I am heavily indebted to. My research would have not been possible without the daily work of educational policy-makers and practitioners in South Africa and elsewhere. I am immensely grateful to the teachers, policy-makers, and ICT geeks who made time to meet and talk to me, and allowed me access to their schools and classrooms.

Of course, any journey is only worth it if there is someone to listen to the stories and adventures, to walk with me on particular tricky roads, and to provide a home to come back to once the journey ends—thank you, Londeka Mahlanza, for providing each of these and much more over the last four years.

Chapter 1. Introduction to the thesis

Mobile technologies are an integral part of our daily lives. Mobiles' ubiquity and penetration of most aspects of social and economic structures have left them the defining technology of the Information Age in the early 21st century—to an extent that a Mobile Age or Mobile Area has been proposed (Castells & Himanen 2014; UNESCO 2014). In this thesis, I am concerned with mobile technologies in one particular setting, education in Low- and Middle-Income Countries (LMICs), and to one particular end, their conceptualisation as a tool that can support development outcomes such as poverty reduction. As I will show in more detail below, there is a diverse set of interests that, for better or worse, have positioned mobile technologies as an intervention with vast potential to improve education and subsequently development in LMICs. It is my objective to examine the empirical validity of these claims, that is that mobile technologies as an educational tool can bring about development; and to propose a conceptualisation of the links between mobile technologies, education, and development in LMICs. To do so, I used the tools of research synthesis and Amartya Sen's Capability Approach (Sen 1999) to navigate the thesis's contribution in this contested area. In this, I position the thesis to approach the above three themes from a development perspective. That is, this thesis's angle of investigation is grounded in the domain of international development and its associated literature and theories, rather than in the domain of technology studies or educational research. Throughout the thesis, I will use the term 'Mobile Learning for Development' (ML4D) to underline the thesis's positioning and to place boundaries around its empirical investigation and contribution.

The point of departure for this thesis were a series of reports by international development and mobile vendors organisations operating in LMICs that claimed large-scale development and education outcomes through the use of mobile technologies to support teaching and learning¹. To give a flavour of the tone of the narrative on the educational use of mobile technologies in LMICs, UNESCO (2014) ascribed a 'revolution' in education in LMICs to be caused by mobiles, while the GSMA (2010) identified mobile technologies as 'teachers in pockets' and 'classrooms at learners' fingertips'. The narrative of learning and teaching with mobile technologies in LMICs was (and is) framed in highly optimistic terms with strong confidence that mobile technologies make an important contribution to education and

¹ For example, UNESCO (2012a; 2012b), GSMA (2012a; 2012b), USAID (2014), WEF (2012).

development in LMICs. National governments in LMICs, too, have announced large-scale investments in educational mobile technologies, including in South Africa where I live and work.

At the same time, however, researchers concerned with the pedagogical underpinning of using mobiles to support teaching and learning as well as researchers concerned with the use of Information and Communication Technology (ICT) for development more broadly raised doubts about this strongly positive framing of the contribution of mobile technologies to education and development in LMICs². My thesis enters these debates both from an empirical and conceptual angle aiming to understand the evidence-base behind the claims for mobile learning's contribution to education and development in LMICs as much as how such a potential contribution of mobile technologies can be conceptualised and positioned.

This inquiry has gained much currency throughout the time of research. From Facebook.org's rejected attempt to connect millions of Indians to the internet, to the spread of Bridge International Academies across Africa using scripted lesson plans that teachers access and read of tablet devices, and major development funder's experimentation with classrooms delivered in a box full of digital technologies to learners in LMICs, the perceived promise of mobile technologies, largely driven by top-down initiatives, continues to prevail³. Through this thesis, I contribute a more empirically- and conceptually-grounded perspective to this debate with a particular focus on using mobile technologies in educational settings in LMICs.

 ² For example, Winters (2013; 2015), Traxler (2013a; 2013b), Kleine (2013), Gigler (2015).
³ Facebook.org: <u>http://tinyurl.com/rejectfborg</u>; Bridge International Academies:

http://tinyurl.com/bridgeacademies; Classroom in a box: http://tinyurl.com/solar-classroom

1.1 Research background

As of 2017, there are 6.1 billion mobile phone subscriptions in LMICs resulting in 99 subscriptions per 100 people in LMICs (ITU 2017). While this does not necessarily mean that each individual in LMICs has access to a mobile phone or another mobile device, it indicates the ubiquity of mobile devices in LMICs more broadly. This is further underlined when taking into consideration the historic trend of these numbers. A mere 10 year ago, mobile subscriptions in LMICs stood at 39 subscriptions per 100 people and in 2002 only 2.2 billion subscriptions were recorded in LMICs (ibid). This rapid growth of mobile technologies has also affected the nature of the mobile devices and their technological abilities. For example, since 2015 the majority of phones sold in LMICs is made up of feature and smart phones (GSMA 2017) contributing to a more powerful application of mobile devices. This reflects in numbers such as ITU's recent announcement that 67% of youth in LMICs are able to accesses the internet (ITU 2017).

While stark inequalities in access and use to mobile devices remain, this trend of increasing access and use is reaching the poorest part of the world's population too. In the most disadvantaged quintile of households in LMICs, close to 7 out of 10 households own a mobile phone; and these households are more likely to own a mobile device than to have access to sanitation or clean water (World Bank 2016). Again, this increasing access to mobiles is correlated with more powerful applications of the technology and 30% of mobile users in the group of least developed countries can access the internet using their devices (ITU 2017; GSMA 2017). Though, these numbers do not control for within country inequality in internet access (Donner 2016).

Against the background of rapidly increasing access to and use of mobile devices in LMICs, mobile technologies have become a popular platform to host various types of development interventions (World Bank 2012; Donner 2008). By just one count, this includes different development priorities such as using mobiles to support health outcomes (mHealth), financial inclusion (mFinance), gender empowerment (mEmpower/mWomen), agriculture (mFarm), and good governance (mGovernance) (World Bank 2012). This plethora of different development applications is often justified by the endogenous uptake of mobile devices, which are portrayed as technologies of choice by users in LMICs. Seminal research such as Jensen's (2007) observation of Indian fishermen's use of mobile phones to obtain

the best market prices for their daily catch⁴, or the World Bank's finding that almost threequarters of Kenyan mobile phone users would skip a meal in order to finance mobile phone usage (World Bank 2012), fuel this narrative that mobile technologies are already integrated and valued in people's daily lives and therefore provide potent tools to host development interventions. Further, there is a range of well-covered success stories of mobile applications that have supported development outcomes such as Kenya's mobile payment systems MPESA⁵ and the use of SMS reminders to enhance HIV/AIDS medication adherence (e.g. Lester et al 2010).

Given education's high profile within international development debates (Unterhalter et al 2015; Pritchett 2013), it is little surprise then that the education sector in LMICs has been positioned as to benefit from the potential of mobile technologies too. For example, the UNESCO is hosting an annual mobile learning week and has invested in a portfolio of work around the potential of learning and teaching with mobiles in LMICs⁶. USAID and other major development donors maintain an mEducation alliance aiming to reduce barriers to "access appropriate, scalable, and low-cost mobile technologies to help improve learning outcomes in formal and non-formal education" (mEducation Alliance 2017).

This high-level interest of development actors into the educational use of mobile technologies in LMICs has nurtured a vibrant and diverse body of development projects and programmes that aim to apply mobiles to support education and development in LMICs. As of 2015 the mEducation's Alliance database of mobile learning programmes in LMICs comprised 123 entries⁷. Some of these projects have rapidly gained scale. For example, the One Laptop Per Child (OLPC) programme claims to have reached over 2 million children and teachers in 42 countries and the Worldreader programme estimates to have provided 6.3 million people in LMICs with access to e-books on mobile devices (OLPC 2017; Worldreader 2017). Throughout 2013-2014, national governments in a range of LMICs (Thailand, Turkey, South Africa, Kenya) also announced investments in tablet devices at scale in order to support their education systems (Tamim et al 2015).

From an academic perspective, the introduction of mobile learning into the domain of international development overlaps with a number of existing discourses and research areas on education, development, and ICTs (Figure 1.1). First, ML4D enters existing conversations

⁴ This portrayal persists despite ongoing critiques of this narrow and linear framing of technology usage and socio-economic outcomes (e.g. Donner 2016; Srinivasan& Burrell 2013; Vrasidas et al 2009)

⁵ MPESA: <u>http://tinyurl.com/7vgi6hx</u>

⁶/₇ UNESCO Mobile Learning: <u>http://tinyurl.com/unesco-ml</u>

⁷ Mobiles for Education Alliance resources: <u>http://www.meducationalliance.org/?page_id=243</u>

around development and education. Education and development form a symbiotic relationship in which enhanced education both contributes to development processes and presents a development outcome in its own right (Unterhalter 2015; Pritchett 2013). Education constituted a major development priority in the former Millennium Development Goals (MDGs) and continues to do so in the 2015 Sustainable Development Goals (SDGs). Moreover, supporting global progress in education presented an international standalone goal pursued by the UNESCO-led Education for All (EFA) movement and its six EFA goals, which is now taken forward in the Incheon Declaration as a framework for action for education in 2030 (UNESO 2016).



Figure 1.1 Overlap of ML4D with related bodies of literature

ICT4D=Information & Communication Technology for Development ICT4E=Information & Communication Technology for Education edTech=Educational technology

The global development policy interest on education has supported an unprecedented expansion in access to education (Evans & Yuan 2017). In 2015, primary education enrolment rates in LMICs stood at 91% (EFA 2015). Learners' average years of schooling have also increased significantly up from 3.9 years in 1980 to 7.5 years in 2010 (Lee & Lee 2016). This hides, however, regional and population-based inequalities in access to education with large segments of learners in the poorest quintile of household in LMICs and learners in conflict regions being much more likely to remain out-of-school than their peers in more affluent and secure contexts (EFA 2015).

What is more, despite the large progress, on aggregate, to increase access to schooling, improvements in education outcomes such as learning gains are lagging behind. A learning gap or learning crisis has been diagnosed to describe the discrepancy between a rise in access to schooling and a stagnation in learning outcomes (e.g. EFA 2014; Pritchett 2013; Evans & Yuan 2017). For example, it was estimated in 2014 that of the 650 million children attending primary school, at least 250 million did not gain basic proficiency in reading and mathematics (EFA 2014). Increased access to schooling thus has not automatically translated into improved learning outcomes and if the current rate of progress on learning outcomes is maintained, LMICs will require at least a century to achieve similar learning outcomes as high-income countries (Evans & Yuan 2017; Robinson & Winthrop 2016).

A complex and overlapping set of causes for this learning gap has been suggested such as a lack of funding for education systems (EFA 2014), teacher education and professional conduct (Chaudhury 2006), insufficient learning environments and pedagogical approaches (Westbrook et al 2014), and systematic challenging such as school governance and leadership (Carr-Hill et al 2016). In this context, mobile technologies are positioned as a means to remedy unequal access to education and to improve learning outcomes more broadly (e.g. UNESCO 2012a; 2012b). Learners without access to formal education services, for instance, are assumed to be able to access such services through their mobile devices. And, the integration of learning content, for example videos and e-books, as well as instant access to information on mobiles are framed as tools to support learners' educational attainment (e.g. USAID 2014; GIZ 2015)

Second, the use of mobile technologies to support teaching and learning also relates to ongoing debates about the educational use of ICTs in LMICs (edTech). Investments in ICTs have featured prominently on the educational agenda of policy-makers in LMICs driven by the assumption that IT skills are required for countries to meaningfully participate in the information age (infoDev 2012; UNESO 2009; Vrasidas et al 2009). As a result, a majority of LMICs government have adopted and implemented national ICT in education policies (Hennessey et al 2010; World Bank 2013). Some of these initiatives even extend to a regional scope such as the New Partnership for Africa's Development e-schools initiative (NEPAD 2013) and ICTs are also positioned as a tool for achieving the education SDG targets related to numeracy and literacy (World Education Forum 2016).

However, there remains controversy on whether investment in ICT in education in LMICs presents a cost-effective approach to support learning outcomes (Trucano 2013; Evans 2017). Impact evaluations and systematic reviews of ICT in education programmes

16

contribute mixed results to technologies' impact and cost-effectiveness (e.g. Snilstveit 2016; McEvan 2015; Kremer et al 2013). This includes a range of evaluations of the OLPC programme at scale, which find learning outcomes to vary from insignificant effects (e.g. Cristia et al 2012) to small positive effects (e.g. Mo et al 2013)⁸. Reviews and syntheses of evidence on the barriers and facilitators to the effective use of ICT in education in LMICs, though, show more agreement. These suggest factors such as appropriate pedagogy, institutional readiness, teacher training competencies, leadership and management, and long-term financing models are important variables for the effective integration of ICTs into education systems (e.g. Sharmin et al 2017; Hennessey et al 2010; Gulati 2008).

Mobile technologies and mobile learning enter this debate on educational technology in LMICs as a sub-set of a more developed discourse. In this discourse, mobiles are assumed to contribute new technological abilities that fixed ICTs cannot contribute such as mobility, connectivity and social networking, and personalisation (Unwin 2015). The affordability of mobiles also means that educational actors are more familiar with their use and that technology investments are less cost-intensive. Mobiles are also seen as supporting a set of unique mobile-specific pedagogies such as context-crossing (Wali et al 2004) and ubiquitous learning (Ogata 2004), which cannot be implemented using fixed ICTs.

Third, the research and practice on the educational use of mobile technologies in LMICs relates strongly to an existing body of literature exploring the role and contribution of ICTs to development more broadly. Under the umbrella term Information and Communication Technologies for Development (ICT4D) (Unwin 2009), a diverse set of research strands has explored the conceptual and empirical relations between access to and use of ICTs in LMICs and development outcomes (e.g. Avgerou 2010; Zheng 2009; Kleine 2013; Heeks 2010). This literature and discourse also includes mobile ICTs, in particular mobile phones (e.g. Donner 2008; 2016; Galperin & Viecens 2017; Porter et al 2016).

A key feature of this body of literature is a vibrant debate about the nature of the contribution of ICTs to development in LMICs. A range of scholars largely frame ICTs' contribution in terms of economic outcomes such as increased productivity and rates of economic growth. The 2016 World Development Report, for example, investigates the digital dividends of ICTs understood as ICTs' ability to enhance "growth, jobs, and services" (World Bank 2016: 5).

⁸ Despite laptops technically constituting mobile technologies, the OLPC programme has been largely discussed within the wider literature on edTech in LMICs rather than the specific literature on ML4D. There is substantial debate around the relevance, appropriateness, and effectiveness of the programme (see e.g. Annay & Winters 2007; James 2010; Varly 2010). My thesis does not aim to contribute to these debates, which are largely programme specific.

This framing is in line with earlier much-cited research on the economic contribution of ICTs in LMICs in terms of an 1.38% increase in GDP growth for every 10% increase in broadband connectivity and access to ICTs (Qiang et al 2009; World Bank 2009). More recent research subscribes to a similar line of argument in order to conceptualise the use of ICTs to support development in LMICs (e.g. Galperin & Viecens 2017; Cirera et al 2015).

This economic framing of development in ICT4D has been challenged on empirical and conceptual grounds. Systematic reviews and in-depth investigations of ICT4D programmes at a more micro-level have found mixed effects of ICT4D programmes (e.g. Geldof et al 2011; Brown & Skelly 2017; Adera et al 2014). Conceptually, too, scholars have developed a richer understanding of how ICTs might support development outcomes in LMICs taking into account structural inequalities and people's abilities to use ICTs in a self-determined manner and to self-determined ends (e.g. Kleine 2013; Andersson et al 2012; Zheng 2009). The latter discussions are of particular relevance to the literature on ML4D and the conceptual rigour of debating ICT4D's positioning within theories of development contributes a range of analytical lenses in order to position the educational use of mobile technologies in relation to development in LMICs.

1.2 Research outline and questions

As indicated above, this thesis commences on the premise that there is large interest in the use of mobile technologies to support education and development in LMICs, but that proponents of ML4D have not offered convincing empirical evidence or conceptual models for the claimed transformational potential of mobiles on education in LMICs and subsequent development. Rather, the potential of mobile learning is understood in a techno-centric manner in which the wide-spread access and use of mobiles in LMICs and mobiles' powerful technological features justify mobile learning's positioning as a development intervention. However, neither the effectiveness of mobile learning programmes, nor the causal mechanisms and contexts through which teaching and learning with mobile technologies is assumed to support education and development in LMICs, have been systematically reviewed and conceptualised.

This absence of a convincing evidence-base on mobile learning's effectiveness and of a thorough logic model for how the use of mobiles technologies can contribute to education and development presents a serious challenge to the field of ML4D. Mobile technologies cannot uncritically be positioned as a benevolent input in education in LMICs with guaranteed positive impacts on poverty reduction and human development. First, any social intervention or policy has as much potential to do good as to cause harm (Chambers 2002; Oakley 2000). This naturally applies to the educational use of mobile technologies in LMICs too, which might disrupt education processes and cause conflict among educational actors. Investments into mobile technologies also divert scarce public resources away from other educational inputs in LMICs, which could leave education systems at a net loss if mobile technologies do not contribute proportionally to learning outcomes. Unequivocally advocating for the use of mobiles devices in education without a detailed evidence-base is thus not ethical and runs the risk of hindering, rather than supporting, development processes.

Second, a linear and techno-centric understanding of the use of mobiles to support teaching and learning in LMICs risks introducing reductionist notions of education and development, which carries negative implications for the actors and institutions involved in both. Positioning mobile technologies as all-powerful tools to fix long-standing issues such as underperforming education systems risks marginalising other educational inputs such as training teachers and investing in curricula. Similarly, a techno-centric focus on access to technologies risks defining education and development progress in terms of inputs and easy-to-measure outputs. Such a focus on access to educational inputs might overlook

19

structural inequalities that mitigate how different education actors can translate access to inputs into meaningful educational experiences and outcomes.

The above are but two illustrations of why generating an evidence-base and detailed conceptualisation of ML4D cannot be an afterthought in the pursuit of providing teachers and learners in LMICs with mobile technologies. To be clear, I am not negating that mobile learning can make a positive and important contribution to development in LMICs⁹. Rather, I am questioning whether the current thinking and modus operandi in the international development domain regarding the educational use of mobile technologies have sufficiently engaged with the more challenging aspects of mobile learning; in particular, whether the practice and understanding of ML4D has gone beyond the question of access and beyond strong claims to positive impacts based on anecdotes, good intentions, and simple theories.

As a result, the main research question for my thesis and subsequent sub-questions refer to:

1. How can the use of mobile technologies as an educational tool in LMICs be conceptualised as to contribute to development outcomes?

- 1.1 What are the effects of the educational use of mobile technologies in LMICs on learning outcomes, teaching practice, and empowerment?
- 1.2 What is the programme-level theory of change for mobile learning for development?
- 1.3 Does the Capability Approach serve as an effective analytical framework to investigate ML4D?

In order to answer the above research questions, I divide my thesis into two parts: in part 1, I investigate the empirical and conceptual evidence-base for mobile learning's impact on development outcomes; in part 2, having found an absence of such evidence in part 1, I then develop my own conceptualisation of mobile learning's contribution to development using the CA. Both parts of the thesis are concerned with the same overall objective of investigating how the use of mobile technologies as an educational tool in LMICs can be conceptualised as to contribute to development outcomes. In part 1, I assess how the *existing* empirical and conceptual research on ML4D has positioned mobile learning's contribution to development and whether this positioning is empirically justified given the existing evidence-base. In part 2, having rejected the existing claims to mobile learning's contribution to development outcomes based on a lack of evidence, I then assess

⁹ For example: Leach (2005); Aker et al 2010; Kaleebu et al (2013); Vu Henry et al 2016).

prospectively whether and how an alternative conception of mobile learning's contribution to development from a capabilities perspective can be constructed. Combining both parts of the thesis, i.e. the assessment of ML4D's empirical and conceptual evidence-base and the development of an alternative conception of ML4D, allows me to suggest a rethinking of mobile learning's contribution to development.

I use a range of different research methods across both parts of the thesis. Taken together these research methods give rise to three main investigations: (1) a mixed-methods systematic review to appraise and synthesise the existing evidence-base on mobile learning to assess its effects on development outcomes; (2) a qualitative case study of a mobile learning programme in rural South Africa using the CA as an analytical lens; and (3) a conceptual inquiry to develop a conceptualisation of ML4D from a capabilities perspective. Each investigation makes its distinct contribution to my rethink of ML4D as outlined below.

My research commences with the mixed-methods systematic review of existing primary research that has investigated the educational use of mobile technologies in LMICs. The review serves two purposes: examining the impact of mobile learning on education and development outcomes and examining how ML4D programmes have outlined the processes and causal links between the use of mobiles, education, and development in LMICs. Taken together, the systematic review therefore allows me to conclude whether empirical and conceptual claims to mobile learning's contribution to development are warranted. The review thereby constitutes my first key empirical component in rethinking ML4D based on an assessment of its current evidence-base.

Empirically speaking, the mixed-methods systematic review of ML4D interventions presents the departure point of this thesis. However, while its findings support an assessment and deconstruction of current conceptions of mobile learning's impact on education and development, the review is bound to assessing *existing* research on mobile learning. As a result, the systematic review findings cannot support me in developing a *new* conceptualisation of mobile learning's role and contribution in international development.

For this inquiry—the development of a new conceptualisation of ML4D—I first require an explicit conceptual framework that could define and outline the contribution of mobile learning to development; and then after having identified such a conceptual framework, I

21

need to test it through new primary research for its validity and relevance¹⁰. I identified Amartya Sen's Capability Approach as a potent conceptual framework to guide a conceptualisation of ML4D. The choice for the CA was informed by its application as a conceptual lens in related bodies of research, the ICT4D domain in particular (c.f. Kleine 2013; Zheng 2009).

Having identified the CA, I then use a qualitative case study of a mobile learning programme in rural South Africa to apply the CA as a conceptual lens in practice. This serves to test whether the CA can be operationalised as a conceptual framework in research on ML4D. I then apply the findings of my case study research to develop an explicit conceptualisation of ML4D based on the CA. This enquiry is purely of conceptual nature but uses the empirical validation of the CA as a useful analytical lens to assess mobile learning's contribution to development outcomes as a starting point. The developed capabilities conception constitutes my second key component in rethinking ML4D based on a small empirical case study and subsequent conceptual inquiry.

In a last step, I then combine my developed capabilities conception of ML4D with my systematic review findings of the empirical evidence on ML4D. This constitutes my rethink of ML4D and explores how a conceptualisation of ML4D based on the CA and the empirical evidence-base on ML4D's impacts changes the understanding and positioning of mobile learning's contribution to development.

¹⁰ An alternative approach could have been to rely on the theoretical validity and relevance of a new conceptual framework rather than testing it empirically. However, given the detailed review of existing empirical evidence to justify the need for a new conceptual framework, this approach does not seem justified.

1.3 Contribution of the thesis

This thesis offers a range of contributions to the research and practice of using mobile technologies to support education and development in LMICs. In terms of empirical research findings, this thesis contributes knowledge on what works, how and why in using mobile technologies as an educational tool in LMICs. It systematically collects, appraises, and synthesises the available primary research evidence on the educational use of mobile devices in LMICs. The thesis thereby collates an evidence-base on ML4D and then uses this evidence-base to assess the impact of ML4D. This investigation goes beyond individual impact evaluations of mobile learning programmes in LMICs and allows for conclusions to be drawn on mobile learning's effectiveness that are based on the body of evidence available.

The collated evidence-base further allows me to systematically unpack the specific manner and approach in which mobile learning programmes in LMICs have attempted to contribute to education and development outcomes. That is, I can investigate the processes, mechanisms, and contexts that facilitate the effects of mobile learning programmes. Based on this, my thesis constructs a theory of change for the use of mobile technologies as an educational tool in LMICs. This theory of change makes the assumed links between mobile technologies, education, and development explicit allowing for a detailed investigation of their empirical and conceptual validity. This contrasts with most of the literature on mobile learning and development produced by international development and mobile vendor organisations in which the link between technologies, education, and development is largely implicit and not detailed.

The thesis also contributes empirical primary research findings on the investigation of a mobile learning programme from a capabilities perspective. To my knowledge, my case study presents the first attempt to analyse the educational use of mobile technologies in LMICs using the CA. I use the case study findings to construct an analytical framework for teachers' use of mobile technologies based on the CA. The case study research primarily illustrates that the CA can be operationalised to investigate mobile learning programmes in LMICs and how the results of such an investigation present a novel perspective on the educational use of mobiles.

In terms of conceptual findings, my thesis contributes the development of a conceptualisation for ML4D. This contribution is in three parts. First, I review existing

literature on the understanding and positioning of using mobiles to support education and development in LMICs. This investigation highlights a range of empirical and conceptual gaps, which I then fill through my systematic review findings. Based on this, I reject existing suggestions that mobile learning, in its current form and practice, can be positioned to contribute to development outcomes. Second, I then use the CA to develop a detailed definition and conceptualisation of how the educational use of mobile technologies could contribute to development in LMICs. Third, I review this capabilities conception of ML4D against the existing evidence-base and highlight how it changes the positioning of mobile learning in international development towards a more opportunity-, people-, education-, and transformation-focused understanding of ML4D.

In terms of methodological developments, my thesis contributes to a nascent body of work on mixed-methods systematic reviews of development policies and programmes (e.g. Langer & Stewart et al 2014; Snilstveit & Langer 2016; Oya et al 2017). I conduct a twomodule mixed-methods review comprising distinct review modules to synthesise quantitative and qualitative primary research. My thesis also makes a minor methodological contribution to the operationalisation of the CA as an analytical lens by illustrating the CA's versatility in a body of research in which it had not been applied before.

24

1.4 Structure of the thesis

The remainder of the thesis is structured as follows:

PART I: A systematic review of the empirical and conceptual evidence-base for mobile learning's contribution to development outcomes, including chapters 2–6.

Chapter 2 presents a literature review of ML4D to indicate the gaps in the empirical and conceptual evidence-base on the use of mobile technologies to support education and development in LMICs. It outlines how ML4D has been positioned as a development intervention and questions whether the claimed development impacts are reflected in the available research evidence. It also illustrates that mobile learning's contribution to development is largely framed in a techno-centric manner and motivates the need for a detailed conceptualisation of the links between using mobiles in LMICs, education outcomes, and development impacts.

Chapter 3 presents an introduction to the CA as a theoretical framework to conceptualise the contribution of mobile technologies to education and development in LMICs. It outlines the CA and reviews literature on its application in ICT4D and education in LMICs in order to justify the relevance of the CA to guide a conceptualisation of ML4D.

Chapter 4 presents the research design and methodologies applied in this thesis. It outlines the methods followed for my mixed-methods systematic review of the effects of using mobiles to support education and development in LMICs as well as the methods followed for my qualitative case study of a mobile learning programme in rural South Africa.

Chapter 5 presents the first set of findings of my mixed-methods systematic review. It provides a systematic map of the empirical research on the application of mobiles in LMICs to support education and development outcomes in order to describe the nature and design of ML4D programmes. I then synthesise the quantitative effects of these programmes using statistical meta-analysis to understand the impact of mobile learning on education and development in LMICs.

Chapter 6 presents the second set of findings of my mixed-methods systematic review. It outlines the results from my thematic synthesis of qualitative evidence on the contexts and mechanisms influencing the educational use of mobiles in LMICs. I then combine these

contexts and mechanisms with the results from the systematic map and meta-analysis to construct a theory of change for the use of mobile technologies to support education and development. Based on the combined systematic review results, I then question the claims laid to ML4D's impacts in the existing literature and how this perceived contribution of mobile learning to development has been framed.

PART II: Developing a capabilities conceptualisation of mobile learning's contribution to development, including chapters 7–9.

Chapter 7 presents the results of my qualitative case study of a mobile learning programme in rural South Africa. The case study uses the CA as an analytical lens to explore teachers' perceptions of how the educational use of tablet devices has supported their professional and personal development and agency. The case study introduces four dimensions of capabilities that teachers valued and explored using the tablets.

Chapter 8 presents an analytical framework for teachers' use of mobile technologies to illustrate the interplay between the four identified dimensions of capabilities. The chapter then develops the case study findings further into a conceptualisation of ML4D that is informed by the CA. This capabilities conception of ML4D is then proposed as a foundation to define mobile learning's contribution to development in LMICs.

Chapter 9 presents a synthesis of my systematic review findings, my case study findings, and the capabilities conception of ML4D. It reviews the consistency between my systematic review findings and the developed conceptualisation of ML4D. It then highlights how the positioning of mobile learning's role in international development changes following my capabilities conception. This rethink of ML4D is then defended against selected lines of possible critique.

Chapter 10 concludes the thesis. It recounts the main findings presented and arguments developed before highlighting some of the implications thereof. It also indicates areas for future research.

1.5 Key terminology

I briefly introduce key terminology used in this thesis. This excludes definitions of mobile learning and mobile learning for development, which are discussed in detail in chapter 2, section 2.1–2.2, as well as definitions related to the CA, which are provided in chapter 3, section 3.1.1. In this thesis I define 'development' following Sen's (1999) definition of development as freedom or, in other words, as an expansion of people's capabilities to live a life they have reason to value. However, this definition does not prevail in most of the literature on ML4D, which defines development largely in terms of reductions in national poverty rates and other national and international economic and social indicators. There is therefore a mismatch between my conceptualisation of development and definitions of development in the secondary literature reviewed and synthesised in this thesis. In chapters 2–6, I rely largely on the prevailing definitions of development in the reviewed literature, while chapters 7–9 rely on my definition of development are used and how these affect the development of my argument.

When referring to country's levels of development, I adopt the World Bank's classification of economies which groups countries according to four groups of economic development (World Bank 2017). This terminology is preferred over alternative terms such as developing/developed countries or global north/global south as it seems most empirically grounded and more nuanced in terms of possible grouping of countries. The terms 'impact' and 'effectiveness' are adopted in the context of evaluating programme and policy effects and connote a methodological understanding of using counter-factual evaluation designs to establish causality between the applied intervention and the observed effects (White 2009). The term mobile learning 'programme' is used interchangeably with the term mobile learning 'intervention' and refers to a deliberate design of an activity or set of actives that directly intervene and affect participants lives. Lastly, I use the female pronoun throughout the thesis rather than mixed expressions such as she/he. This is to support clarity and consistency of the text.

27

PART I:

A SYSTEMATIC REVIEW OF THE EMPIRICAL AND CONCEPTUAL EVIDENCE-BASE FOR MOBILE LEARNING'S CONTRIBUTION TO DEVELOPMENT OUTCOMES

Chapter 2. Literature review 1: Mobile Learning for Development

Introduction

The literature review is presented in two chapters: the first focuses on mobile learning and the second on the Capability Approach (CA). The aim of the two chapters is to outline the wider discourse on the use of mobile technologies as an educational tool in Low- and Middle-Income Countries (LMICs). The literature review discusses relevant research and concepts related to mobile learning for development (ML4D) and the CA in order to set up the space for the specific contribution of the thesis in the context of the current knowledge base on ML4D. In doing so, the chapters specifically detail conceptual and empirical gaps in the current literature on mobile technologies, their application to support education in LMICs, and their relationship to development outcomes¹¹.

The literature review on ML4D presented in this chapter provides the background to how research and practice on the educational application of mobile technologies have developed¹². It covers relevant research on the empirical performance of the use of mobiles to support education and further discusses how the educational use of mobile technologies has been justified and understood. In investigating this understanding, I pay particular attention to how different actors have positioned and outlined mobile learning's potential as a development intervention. I then contextualise these different positionings in the light of the empirical research and discuss their implications for the practice of mobile learning to support education and development. This then sets up the space for the CA as an analytical framework to guide a conceptualisation of ML4D.

This chapter is structured as follows. I commence with an introduction to mobile learning in general before detailing how mobile learning is applied in the context of education in LMICs. I track the historical development of the educational use of mobile technologies and present

¹¹ Unless specified otherwise, all of the discussions that follow relate to the context of low- and middle-income countries and I am therefore not making explicit references to education and development *in LMICs* throughout the text.

¹² This chapter does not cover a detailed review of the wider use of mobiles to support development outcomes outside the education domain.

different definitions and understandings of this use. I next outline three approaches to conceptualise the contribution of mobile learning to development and how they vary in their positioning of the developmental potential of teaching and learning with mobiles. This discussion includes examples of applied interventions and their relation to development initiatives such as Education for All. I lastly illustrate three key critiques that explore a range of gaps and shortcomings in the understanding and positioning of ML4D as a development intervention. The chapter concludes with a brief summary to set up for the review of the CA.

2.1 Mobile learning

2.1.1 An introduction to mobile learning

Mobile learning in its broadest sense refers to the use of mobile technologies to support educational processes, in particular learning and teaching activities (Ally 2009; Sharples et al 2007). The research domain has largely been driven by academics and practitioners based at US and European research institutions, which influences its research tradition and concepts (Traxler 2013a; Winters 2013). The idea of the educational use of hand-held devices that combine advanced computing powers and communication abilities has been conceptualised as early as 1972 (Kay 1972), but only gained currency with the near universal uptake of mobile technologies since the early 2000s (Sharples et al 2007; Pachler et al 2010; Unwin 2015). As mobile technologies have become ubiquitous and started to influence most aspects of our lives, in particular how we communicate and access information, the presumed potential to use these technologies to support education has grown (Castells & Himanen 2014; Toyama 2015; Donner 2016). Though, this presumed positive potential of mobiles is certainly not without critique (e.g. Turkle 2016; Elliot & Urry 2010).

Learners' social realities are increasingly shaped by constant connectivity, instant access to information, and a contextualised and customised experience of using this information (Sharples 2007; Traxler 2009). Consequently, children's learning activities are already being influenced by mobiles, regardless of formal education interventions (Ally 2009). The ubiquity, appeal, and wide applicability of mobile devices therefore necessary affect established processes of learning and teaching (Pachler et al 2010). The theory and practice of mobile learning, at its core, aims to investigate how these mobile technologies can be used to support informal and formal learning activities and how the technological affordances¹³ of mobiles can be integrated within teaching and learning processes (Sharples et al 2007). However, this acknowledges that learners differ in their individual abilities and social opportunities to access and apply mobile technologies for educational purposes. Socio-economic contexts and structures and practices do shape mobile learning and need to be carefully considering when designing teaching and learning opportunities with mobiles (Pachler et al 2010; Wali et al 2008).

¹³ The term and concept of 'affordances' has some important shortcomings as discussed by Oliver (2005) and other terms such as 'abilities' and 'functionalities' have been suggested. However, the technical language of the CA refers to capabilities and functionings. I therefore retain the term affordances throughout the thesis, which should be understood as a linguistic substitute for functionalities.

Learning processes, to some extent, have always been mobile (Wali et al 2008). Learners have read books inside and outside of classrooms and interactions between teachers and learners have rarely been fixed to a single context. Humans have for the most part of our history developed tools and artefacts to allow us to learn 'on the move' or access information across different contexts (Wali et al 2008; Unwin 2015). What is new then about current understandings of mobile learning is the increased technological abilities that mobile technologies provide; and how these abilities can be harnessed to support learning and teaching activities.

As a result, conceptions of mobile learning distinguish it from other educational technologies by means of the applied devices. For example, whereas initial e-learning activities using desktop computers required fixed educational infrastructure such as computer labs, mobile technologies can be used without special educational infrastructure being in place. The most common referenced unique affordances associated with mobile technologies compared to other educational technologies are: mobility of devices (size and battery life), constant connectivity (either using fixed or mobile internet connections), ubiquity (based on their accessibility and ease of use), and mobiles being more personally adaptable than fixed educational technologies (e.g. Unwin 2015; Ally 2009; Winters 2007). These affordances have led mobile phones, tablets, personal digital assistants (PDAs), and laptop computers to be positioned as the main technologies associated with mobile learning (Wali et al 2008; Ally 2009).

2.1.2 Definitions of mobile learning

Definitions of mobile learning have changed over time from a device-based, techno-centric conception towards a learning activity-based, pedagogical conception (Pachler et al 2010; Winters 2013; 2015; Traxler 2013a). Changing definitions of mobile learning also influenced the practice and research agenda of using mobile technologies to support learning and teaching, which evolved in line with definitions and conceptions of mobile learning. Pachler and colleagues (2010) identify three broad phases in the evolution of mobile learning: (1) a focus on the technical feasibility of mobile devices in an educational context from the mid-1990s; (2) a focus on the use of devices to guide learning 'anytime, anywhere' outside the classroom from the mid-2000s; and (3) a current focus on the mobility of the learner and the learning activities supported by mobiles.

In the first phase, mobile learning in essence was equated with the mere use of mobile devices to support educational activities. Research and practice of mobile learning focused

narrowly on assessing the technical feasibility of the educational use of emerging devices such as PDAs and the first mobile phones without much attention being paid to the pedagogical underpinnings of such use. This reflected in techno-centric definitions of mobile learning such as Traxler and Leach (2006: 1): "Mobile learning' is the term increasingly being applied to the use of small, portable, handheld and lightweight electronic devices used for educational activities in classrooms, in fieldwork, at home, at work and when travelling".

With the increase in the technological abilities of mobiles starting in the early 2000s, this conception of mobile learning shifted. Mobile devices now allowed for a much broader range of applications, and research interest and practice clustered around the use of smart phones and PDAs to support learning activities outside the classroom such a field trips. Mobile learning came to be understood as a "personal, unobtrusive, spontaneous, 'anytime, anywhere' way to learn and to access educational tools and materials" (Kukulska-Hulme & Traxler: 2005: 1). However, this understanding of mobile learning remains techno-centric merely shifting the emphasis on a new set of technological affordances of the devices (Wali et al 2008).

In the third and current phase, a range of scholars provided strong critiques of device-based, techno-centric definitions of mobile learning, which were seen as, first, unstable and in need of constant updating due to the pace of technological innovation (e.g. Traxler 2009; Kukulska-Hulme 2009), and, second, were not integrated with formal theories of learning (e.g. Sharples 2007; Wali et al 2008). Many of these critiques can be traced back to Roschelle's (2003) seminal paper on unlocking the learning value of mobiles. Investigating the educational use of mobile technologies from an explicit pedagogical perspective, he showed that while mobiles "preform a small, well-defined function uniquely well, much of the rest of teaching and learning is left to social practice" (Rochelle 2003: 8-9). His critique emphasised the potential of these small and well-defined functions that mobile technologies do uniquely well, but stressed that these functions only become powerful educational interventions through their interplay with and embeddedness in social practices.

On the back of this critique a diverse body of literature has offered different models and theories of how mobiles' particular affordances can best be integrated with theories of learning to design powerful educational mobile learning interventions. Sharples and peers (2007) for example propose a theory of learning for the mobile age in which they draw on Laurillard's (2002) Conversational framework for learning and Engestroem's (1987) expansive activity model. This leads them to define mobile learning as "the processes of coming to know through conversations across multiple contexts among people and personal

interactive technologies" (Sharples et al 2007: 4). This definition presents a large shift away from applying a device-based conception of mobile learning focusing on the role of mobiles to support conversation, interactions, and context-crossing which are positioned as the channels through which learning occurs. Similar shifts in definitions can be observed in Pimmer's (2010) conception of mobile learning from a socio-cognitive perspective and Bell's (2010) use of actor network theory to conceptualise mobile learning.

For the purpose of my thesis, I will follow Wali and colleagues' (2007: 55) definition of mobile learning as: "learning that occurs as a result of pursuing learning activities that are directed towards achieving the same objective across multiple contexts (both physical and social)". This definition refines Sharples and colleagues' theory of learning in a mobile age. It defines mobile learning in terms of context-crossing allowing for a shift away from the use of mobile technologies towards a focus on the forms of learning practices supported by mobiles and how these practices are mediated by social and physical contexts. In the context of my thesis on ML4D, however, the intricacies of current definitions of mobile learning are less of a concern. What is more important here, is the outlined shift of definitions of mobile learning from a device-based, techno-centric understanding towards a learning activity-based, pedagogical understanding.

2.1.3 Mobile learning's pedagogical contribution

The progression of mobile learning's definition towards a learning-activity, pedagogical understanding of mobile learning has led to an exploration of a range of rich mobile learning pedagogies. I define pedagogies here in line with Watkins and Mortimor (1999: 3) as: "any conscious activity by one person designed to enhance learning in another". There is now consensus among mobile learning scholars in the educational domain that mobile learning can best be understood as a socio-cultural form of practice (Winters et al 2017; Pimmer et al 2014; Kearney et al 2012; Frohberg et al 2009; Koole 2010). Based on a review of 102 Mobile Learning projects, Frohberg and peers (2009), for example, identify the following underlying pedagogical approaches to mobile learning as most frequently applied: constructivist, situated, collaborative, and informal and life-long learning. However, this hides a rich field of specialised mobile learning approaches that have emerged such as 'context-aware ubiquitous mobile learning' (Chen & Li 2010) and 'RFID-supported immersive ubiquitous mobile learning' (Liu et al 2009).

The benefits that learners might derive from these mobile learning pedagogies, that is mobile learning's pedagogical contribution, have been outlined in different ways (e.g.

Winters 2007; Kukulska-Hulme 2009; Ally 2009). Kearny and colleagues' (2010) framework for viewing mobile learning from a pedagogical perspective provides a helpful synthesis of these benefits. The authors define the three distinctive features of mobile learning as fostering personalisation, authenticity, and collaboration in the learning process across time and space. Each of these three features can enrich the learning experience of students. Personalisation of learning supported by mobiles is assumed to enhance learners' choice, agency, self-regulation and customisation. Authenticity of the learning is linked to an educational use of mobiles to facilitate learning tasks of real world relevance and personal meaning to the learner. Collaboration, lastly, links to learning fostered by social interaction and conversation as assumed by socio-cultural perspectives on learning as referenced above. Notable sub-domains from Kearney and colleagues' framework refer to mobile learning approaches using mobiles to support game-based learning, augmented reality learning, and learning analytics (Kearney et al 2010).

All in all, Kearney and colleagues' framework reflects a wider agreement in the literature that mobile learning can be positioned to support more learner-centred educational activities (e.g. Traxler 2009; Pachler et al 2010; Winters et al 2017). From an educational perspective, mobile learning has centred around how mobile technologies can support learners to create and acquire knowledge through an interaction with their socio-cultural contexts and practices. This positioning is perhaps best captured by Pachler and colleagues (2010: 6): "Mobile learning—as we understand it—is not about delivering content to mobile devices but, instead, about the processes of coming to know and being able to operate successfully in, and

and knowing how to utilise our everyday life-worlds as learning spaces. Therefore, in case it needs to be stated explicitly, for us mobile learning is not primarily about technology."

across, new and ever changing contexts and learning spaces. And, it is about understanding

2.2 Mobile learning in LMICs

I next discuss the application and development of mobile learning in LMICs. In this, it is important to keep in mind that the origins of the research and practice on mobile learning are predominately from US and European institutions. Concepts and models of mobile learning are thus contextualised by this research tradition and it is crucial to keep this in mind when investigating the application of mobile learning in LMICs. Within HICs themselves, there are large inequalities in learners' abilities to use the full pedagogical potential of mobile learning. This is even more so in LMICs and discourses around 'connected learners' and learners developing agency to take control of learning activities, for example, need to be carefully interrogated for their applicability across contexts. The values embedded in mobile learning and its assumed educational contribution cannot be uncritically positioned to apply to marginalised learners with different educational needs. Mobile learning needs to be reflexive of its origins; and its theory and practice requires adaptation in LMICs contexts, in particular when attempting to influence development outcomes. My thesis is only concerned with the latter aspect of mobile learning's conceptual adaptation in LMICs-i.e. to support development outcomes-with the conceptualisation of a pedagogical theory of mobile learning adapted for LMICs being beyond my scope.

2.2.1 An introduction to mobile learning in LMICs

As mobile technologies have become ubiquitous in most parts of the world, so has the rationale to use mobile devices to support education. The initial uptake of mobile learning in LMICs followed the above-outlined conceptual development (Traxler 2013a; 2013b). Early mobile learning projects applied a device-based and techno-centric approach. The first mobile learning projects reported in LMICs in the mid-2000s focused on providing access to mobile devices in order to enhance the reach of education programmes, distance learning in particular (e.g. Brown 2005; Matthee & Liebenberg 2007; Traxler 2005). However, in line with conceptual developments in HICs, mobile learning projects in LMICs soon began to apply more pedagogically-rich project designs exploiting the whole range of technological affordances provided by mobile devices. Prominent examples of early mobile learning programmes in LMICs with rich pedagogical underpinnings can be found in Zurita and Nussbaum's research in Chile on mobile computer-supported collaborative learning systems based on Activity Theory (Zurita & Nussbaum 2004; 2007); Kim's mobile learning model of literacy development for underserved migrant indigenous children in Peru and Bolivia (Kim
2009); Sahni and peers' Digital StudyHall mobile learning programme drawing on critical pedagogy to support situated and participatory learning approaches (Sahni et al 2008); and Leach's Digital Education Enhancement Project using mobiles to support teacher professional development in reference to more learner-centred teaching approaches (Leach et al 2005).

However, in parallel to this pedagogical development of mobile learning as an educational intervention, both mobile technologies and education received increased attention in the international development sector. First, mobile technologies fuelled by success stories such a Kenya's MPESA mobile finance application and Jensen's (2007) research on fishermen's use of mobile phones to access market prices came to be understood as endogenous platform on which development interventions could be hosted (World Bank 2008; Donner 2009; Banks 2014). Mobile technologies due to their ubiquity in LMICs were ascribed vast potential to support various development sectors leading to a plethora of development domains aiming to 'maximise mobile' such as mHealth, mFinance, mPowerment, and mFarm (World Bank 2012; 2016; Donner 2009; 2016).

At the same time, education itself received renewed policy interest in LMICs due to an enhanced emphasis on supporting education systems as a key mechanism in international development (Unterhalter 2015; McCowan & Unterhalter 2015). This renewed policy interest in education is evident in large-scale multilateral development initiatives such as the former MDGs and EFA targets. The rise of mobiles for development as well as the increased focus on education in international development therefore began to contextualise the conceptual development of mobile learning in LMICs. This contextualisation led to a shift in understandings of mobile learning, which from a development perspective was increasingly referred to as the use of mobile technologies to support education, rather than an explicit pedagogical education intervention (Traxler 2013a; 2013b; Kinuthia & Marshall 2014; Winters 2013; 2015). This drifting understanding led to different foci of mobile learning practice and research depending on whether the topic was approached from an education or a development perspective (Traxler 2013b). This resulted in a somewhat silo nature of ML4D in which existing educational research on mobile learning (in particular when developed in HICs) was often not considered to inform programme and policy design (Kinuthia & Marshall 2013; Winters 2013).

This shift in an understanding of mobile learning in international development is also evident in a change in terminologies used to refer to the educational application of mobile devices. Starting with the first use of the term 'Mobile learning for development' to describe a set of workshops run in Nigeria in 2010 (Winters et al 2010), ML4D has been adopted mainly by the educational research community. Examples of this can be found in Kinuthia and Marhsall's (2013) volume 'On the move: Mobile learning for development', the work of John Traxler (Traxler 2013a; 2013b; 2013c) and Niall Winters (Winters 2013; 2015; 2017). Development actors, on the other hand, have preferred a more device-focused language such as UNESCO's (2010) 'Mobiles, learning, and development' and USAID's 'mobiles for education for development' (USAID 2010). Here, mobile education is distinguished from mobile learning to position it "as an extension of mobile learning defined as the exploitation of ubiquitous handheld technologies, together with mobile and wireless networks, to facilitate, support, enhance or extend the reach of teaching and learning" (GSMA 2011: 3).

This presents a deliberate shift away from the pedagogical understanding of mobile learning towards a techno-centric definition emphasising the technology over the learning processes and contexts. This is even the case for publications by development actors that adopt the term 'mobile learning'. UNESCO's Working Paper Series on Mobile Learning of 2012, for example, defines mobile learning as "modern ways to support learning process through mobile devices, such as handheld and tablet computers, MP3 players, smartphones and mobile phones" (UNESCO 2012d: 6). This implies a conception of mobile learning in line with very early understandings of the theory and practice of mobile learning in the educational domain. From a pedagogical perspective, therefore, mobile learning in LMICs is essentially going backwards neglecting a rich and long-standing body of educational research on mobile learning (see 2.3.3). Given this discrepancy between the positioning of mobile learning in international development by education researchers and development actors, Traxler (2013b; 2013c) goes as far as to identify two distinct schools of thought on ML4D.

2.2.2 Defining mobile learning for development

From the above discussion, it emerges that the term ML4D is likely to be interpreted differently depending on the research and policy background of scholars. For an educational audience, the term indicates an explicit use of mobile learning pedagogies to support education in LMICs (e.g. Winters 2015; Traxler 2013a). For a development audience, it indicates the use of mobile technologies to support education. Neither of the two audiences has proposed an explicit definition of ML4D. In my thesis, I therefore define ML4D as:

'the process of both teaching and learning with mobile technologies in LMICs, building on the unique affordances of mobile devices (Roschelle 2003) to provide learners and teachers with an educational tool that can be used to address some of the specific educational challenges

faced by LMICs.'

I thus position myself alongside pedagogical definitions of mobile learning and further aim to distinguish my conception from the mere use of mobile technologies in education by emphasising the use of mobiles to address some of the specific educational challenges faced by poorer countries. This also aims to position myself in relation to mobile learning *in* LMICs, which I would argue should not be equated with Mobile Learning *for* Development (see 2.2.3). Lastly, in my definition, the process of learning and teaching with mobile devices can assume diverse forms and is not restricted to formal education.

However, within the wider field of international development current understandings and practices of ML4D are more in line with a techno-centric conception. Of the 123 mobile education projects listed on the online database of USAID's Mobiles for Education Alliance, only 18 projects referenced an explicit pedagogical approach as part of the project design. Likewise, while there is ongoing innovative mobile learning research and practice investigating the links between mobile learning pedagogies and their support to development processes and outcomes (e.g. Vu Henry et al 2016; Pimmer 2016; Piper 2016; Hennessey et al 2015), large-scale mobile technology programmes in LMICs are not designed around explicit mobile learning pedagogies. For example, Worldreader, an NGO claiming to have supported literacy development of over 6 million people in LMICs, focuses on the provisioning of reading materials through e-books and mobile phone apps (Worldreader 2017). Worldreader does not design the use of their mobile technologies around any explicit theory or framework of learning. The same holds true for other frequently cited mobile education programmes such as TeacherMate (e.g. Masperi 2008) and Project ABC (Aker et al 2012; 2015). Lastly, from a public sector perspective, there also is a strong interest in the use of mobile technologies in education. But, national-scale mobile technology programmes in education in countries such as Kenya, Turkey, and Thailand, only reference an investment in the supply of technologies without an explicit educational framework for how this technology is supposed to provide benefits to and be integrated with existing teaching and learning activities (Tamim et al 2015).

2.2.3 Mobile learning's assumed contribution to development

The definitions and understandings of ML4D have important implications on the assumed potential of teaching and learning with mobiles to support development outcomes¹⁴. As explained above, understandings affect programme design and practice, which in return

¹⁴ In the below, I rely on the definitions of development adopted in the ML4D literature and group these into three categories; note that these do not correspond to Sen's definition of Development as freedom, which I suggest as an alternative definition of development.

determines the conceptual linkages between the use of mobiles, education, and development. Within both understandings of ML4D, there is consensus that the use of mobiles is linked to development outcomes through enhancing educational processes and outcomes (Traxler 2013a; Traxler & Vosloo 2014; UNESCO 2012a). However, there is no clarity or conceptual model on what the exact contribution of teaching and learning with mobiles is to educational processes and outcomes that allows mobile learning programmes to be linked to development outcomes such as reductions in poverty and inequality. Some commentators assume that any mobile learning project in LMICs inherently carries a developmental agenda and contribution (e.g. Brown 2005; Agence Française de Developpement 2014; ITU 2015). However, albeit education being a main factor contributing to development, it cannot be equated with development. Questions of access to and quality of education, for example, negate a linear pathway from the provisioning of educational services to socio-economic development (Tikly & Barrett 2011; Lewin 2009; Unterhalter 2015). Similarly, just because a mobile learning project is conducted in an LMIC does not mean it involves disadvantaged learners and teachers. South Africa, an upper-middle income country, for example, has a vibrant mobile learning research and practice community, but much of its work is focused on well-equipped urban schools in affluent suburbs (Batchelor 2007; Ford et al 2014). As a result, mobile learning's contribution to development needs to be investigated and understood in detail in order to allow for careful programme design and consistent conceptualisation.

Mobile learning's pedagogical contribution to development

From an educational perspective, mobile learning's contribution to development is defined through its pedagogical value (Winters 2013; 2015; Unwin 2015; Traxler & Vosloo 2014). That is, by supporting new teaching and learning practices that can cater to the specific educational challenges in LMICs, mobile learning assumes developmental potential. Such challenges can refer to large classroom sizes with few individual and tailored learning opportunities, frontal and teacher-centred lesson designs, a lack of locally-relevant educational content, among other (Westbrook et al 2015; McCowan & Unterhalter 2015). Applying Kearney and peers' (2010) three-level framework of mobile learning's pedagogical value, a range of scholars have outlined how the particular pedagogies supported by mobile devices might be well-suited to address some of the specified educational challenges in LMICs (e.g. Asabere 2013; Motlik 2008; Iqubal 2012; Rao 2013)¹⁵.

In terms of personalisation, mobile learning is assumed to be able to support more self-

¹⁵ This overlaps with related work in the domain of mHealth, which is however outside the scope of my thesis (c.f. Winters 2017; Pimmer 2013)

directed learning under the agency of learners and tailored to their preferences (Asabere 2013; Masperi 2008). Frias-Martinez and colleagues (2013), for example, show how a game-based mobile learning tool in both formal and informal settings at a low-income school in Peru facilitated a more engaging and learner-centred educational experience. In terms of authenticity, mobile learning is linked to the facilitation of more contextualised and less-bounded learning activities. Resources for field trips and school excursions in many LMICs are limited and educational content is often not tailored to the ethnic and linguistic diversity of learners. Ekayanke (2013), for instance, illustrates how science teachers in Sri Lanka explored mobile phones to relate subject knowledge to authentic locations and activities during teaching. In terms of collaboration, reference has already been made to Zurita and Nussbaum's experiments in Chile with further examples of collaborative mobile learning pedagogies aiming to support more learner-centred teaching strategies in LMICs being presented by Jere-Folotiya and peers (2014) and Kumar and colleagues (2012).

Mobile learning's contribution to international development goals

A second way in which mobile learning's contribution to development has been framed is through its assumed contribution to international development goals such as the MDGs (now SDGs) and EFA targets. This is perhaps most strongly articulated by UNESCO's (2012) working papers series on mobile learning (UNESCO 2012a; 2012b; 2012c). Based on a synthesis of six regional papers on mobile learning, UNESCO (2012c) positioned mobile learning to support the MDGs and EFA targets through four key mechanisms: the use of mobiles to (1) increase access to education; (2) build lifeskills; (3) support gender equality; and (4) enhance learning outcomes. Access to education in this conception is one-dimensional and refers to the use of mobiles to extend educational opportunities to learners currently not integrated in the formal education system. It also relates to creating access to educational content on mobiles such as Text2Teach's video recordings of model lessons provided to teachers (UNESCO 2012b; Deriquito & Domingo 2012). This conception is in line with a supply-driven emphasis of educational policy in international development during MDGs (McCowan & Unterhalter 2015).

UNESCO's focus on lifeskills is largely defined in relation to the former MDGs and EFA targets on increasing numeracy and literacy rates in LMICs. Here again, there is an emphasis on extending the reach of educational opportunities through mobile devices to learners currently not able to access the formal education system. A similar reasoning underpins UNESCO's positioning of mobile learning in relation to gender equality. Gender empowerment is framed as a function of a lack of access to educational opportunities and relevant information for women. For example, access-orientated mobile learning projects

such as Niger's adult literacy programme for women, Project ABC, or Text4baby, a free mobile health education for women who lack access to the internet and other sources of health information are positioned to support gender empowerment (UNESCO 2012c). It is only in relation to the fourth contribution of enhancing learning outcomes in which UNESCO for the first time references the need for explicit pedagogical models of mobile learning in relation to the MDGs and EFA targets. Alas, this is not further specified and the report concludes that "at mobile learning's present stage of development, what is most important is offering new kinds of learning opportunities that were not available previously, rather than just making marginal improvements to traditional education" (UNESCO 2012c: 25). However, what these new kind of learning opportunities and improvements refer to is not discussed.

UNESCO's positioning of mobile learning's contribution to the former MDGs and EFA is echoed in subsequent reports of international development organisations (e.g. GIZ 2015; USAID 2015; AFD 2015). Further refinement of UNESCO's positioning towards the end of the MDGs added an emphasis on lifeskills for workforce development (Traxler & Vosloo 2014) as well as a calls for a greater emphasis on quality education (EFA 6) rather than a mere access to education (Unwin 2015). Lastly, a separate body of literature on schoolbased decision-making and community-based monitoring also has begun to explore the use of mobile devices in reference to the EFA goals (e.g. Cilliers 2015; Aker et al 2015). Scholarly or policy work on mobile learning's relevance to the 2016 SDG goals has not yet emerged.

Revolutionising education in LMICs through the use of mobiles

A third body of literature defines mobile learning's contribution to development through a fundamental disruption and change of current education systems. In this conception, mobile learning is assumed to foster a "revolution" in education in LMICs breaking with existing practices and facilitating learning opportunities and outcomes of unprecedented scale. This conception is particularly expressed by mobile vendor organisations, foremost the GSMA, but can also be identified in selected UNESCO and USAID reports. For example, mobile learning's potential to "revolutionise" learning in LMICs is explicitly referenced in Banks (2013: 6), UNESCO (2014: 83; 2015), and GSMA (2012a: 5).

The assumed potential of mobile technologies to fundamentally disrupt and alter education in LMICs is most comprehensively outlined in three GSMA reports on mobile learning in 'emerging markets' (GSMA 2010; GSMA 2012a; GSMA 2012b). In the report mLearning: A Platform for Educational Opportunities at the Base of the Pyramid, the vendor organisation claims that mobile learning programmes can "give our most vulnerable children, a teacher in their pocket, a classroom in their hand, and a future at their fingertips" (GSMA 2010: 3). The report estimates that mobile learning could enhance future income of learners by as much as 20% and that a failure to transform education through innovative approaches such as mobile learning risks leaving youth "imprisoned by poverty, immobilised by insecurity, and beyond the reach of traditional aid" (ibid). In a subsequent report GSMA goes as far as claiming that 180 million children will have the opportunity to stay in school between 2013 and 2017 due to advancements in mobile learning interventions (GSMA, 2012b)¹⁶. The organisation thus urges educational policy-makers in LMICs to transform learning through mEducation as "few (education) services are as rich with potential as mLearning or have the potential to create such positive change (on life prospects of young people)" (GSMA 2012a: 36-37). This GSMA position on mobile learning in LMICs, however, has been critiqued strongly (Winters 2013) and needs to be contextualised with the vendor organisation's commercial interest.

Yet, the GSMA is by far not alone it the creation of a narrative that positions mobile learning to carry the potential to revolutionise education in LMICs in order to support development. The UNESCO, too, describes mobile learning's impact on education as to be "without precedent" (UNESCO 2012b: 10) proclaiming the already mentioned mobile reading revolution in LMICs. The World Economic forum likewise assumes that "mobile learning provides a level of scope, reach and flexibility that is largely unattainable through traditional classroom environments" (WEF 2012: 5–6), attributing it the potential to be able to "systemically redefine the way that individuals and communities can contribute to society" (ibid). And, neither are academics and NGOs immune to subscribe to this revolutionary narrative of mobile learning in LMICs. For example, Kim and peers (2008) refer to their mobile learning interventions in Latin America as a 'Pocketschools' and Worldreader has described its provisioning of eBooks to community libraries in Ghana as to provide a clear link between the development of literacy skills and positive effects on poverty, health, gender equality, and social mobility (Worldreader 2015: 10).

What is particular striking in the revolutionary narrative of mobile learning's contribution to development is the strong emphasis on access to devices. A mobile phone is positioned as a classroom at hand or in the pocket; potential increases in access to educational content such as e-books and access to informal educational opportunities are linked to increases in life earnings and social mobility. Little reference is made to how exactly access to mobile

¹⁶ I am not aware of a follow-on report or evaluation of this claim.

technologies can be translated into these massive assumed development outcomes. It might also be asked whether such big claims to revolutionary impacts are backed up by equally strong research evidence (Langer et al 2014; Traxler 2013a; 2013c). In the next section, I therefore aim to unpack the outlined potential contributions of mobile learning to development illustrating to what extent scholars agree with them and what critiques have been offered.

2.3 Critiques of mobile learning's positioning as a development intervention

Mobile learning's positioning as a development intervention, that is the assumption that mobile learning programmes in LMICs can contribute to development outcomes, is not without critique. Winters (2013: 402), for example, posits that "*despite many years of research, and countless implementation projects, there has been little advancement in understanding the contribution mobile learning can make to development goals*". The three propositions above of how mobile learning can contribute to development thus need to be carefully interrogated. Below, I provide a synthesis of three key challenges levelled against current understandings of mobile learning as a development intervention identified in the literature: (1) a lack of evidence of effects; (2) a focus on simplistic causalities; and (3) an overly techno-centric conception of ML4D.

2.3.1 Critique 1: Mobile learning's empirical performance in LMICs

A range of scholars have expressed concerns that proponents of mobile learning in LMICs have provided little empirical evidence that allows for the attribution of the educational use of mobiles to claimed development impacts¹⁷. The 2016 World Development Report describes the current evidence-base on mobile learning's impact as weak (World Bank 2016: 146). Trucano (2013) as well as Traxler (2013b) claims that ML4D's proponents might subscribe to a 'faith-based' assessment of mobiles' educational and development impact, rather than a research-based view. Isaacs (2012: 7) concurs with this assessment explicitly illustrating a "dearth of evidence-based research and the limited credibility and trustworthiness of available information on mobile learning in the Africa and Middle East region".

This perceived lack of an absence of rigorous research evidence on mobile learning's effects on education and development outcomes applies equally to all three contributions of mobile leaning outlined above. Traxler (2013b), for example, highlights that outside of small-scale pilot studies the ability of mobile learning to fundamentally alter teaching and learning approaches in LMICs has not been assessed. Likewise, there has been no explicit evaluation of mobile technologies' educational contribution to either the MDGs or EFA targets¹⁸. Large-scale mobile technology programmes in the education sector in LMICs, however, have been terminated with neither the government of Kenya, Thailand, or Turkey

¹⁷ Development impact here refers to prevailing definitions of development in the literature, e.g. changes in poverty rates, the Human Development Index, economic growth rates, and the MDGs.

¹⁸ Unwin (2015) provides an assessment of the evolution and prospects for the use of mobile technologies to improve education access and learning outcomes for the 2015 Education Global Monitoring Report though.

completing the roll-out of the nation-wide mobile technology programmes in schools (Tamim et al 2015).

The need for rigorous evidence of mobile learning's impact is arguable most acute for proponents of a mobile learning revolution in LMICs. Given the magnitude of their claims, the burden of proof weights particular heavily on these actors¹⁹. It is concerning then that UNESCO's claims to mobile learning's revolutionary effects on education in LMICs are based on an online survey of the Worldreader programme, which found that female participants read seven minutes per day on their mobiles, while male participants used their mobiles to read for a sole minute per day (UNESCO 2014: 30). Albeit, this being an extreme example, it pays testimony to a wider pattern of ambitious claims for ML4D's potential and impact that are rarely grounded in reliable research evidence. Donner (2016: 89) dubbed this pattern as mobile learning's 'enthusiasm gap' and Tamim et al (2015: 2) conclude that launches of government-led tablet initiatives in 11 countries was, by and large "driven by the tablet hype rather than by educational frameworks or research-based evidence".

In recognition of this thin evidence-base, an increased investment in quantitative and qualitative impact evaluations of ML4D programmes can be observed since the early 2010s (Langer et al 2014; Traxler 2013a)²⁰. However, the results of these evaluations do not provide a coherent picture of mobile learning's effects. While some evaluations of individual mobile learning pilot programmes find evidence of improvements in learning outcomes (e.g. Aker et al 2010; Kaleebu et al 2013), other studies fail to replicate such effects (e.g. Cole et al 2012; Chen et al 2010) and even identify potential for harmful effects (e.g. Potter et al 2016). This situation of a growing and diverse evidence-base provides an opportunity for methods of research synthesis such as systematic reviews to be applied.

However, there has been no systematic investigation of the impact of mobile technology's provision as an assumed educational tool in LMICs at a research synthesis level. While a number of high-profile meta-analyses and systematic reviews (Petrosino et al 2012; McEwan 2013; Snilstveit et al 2016) of educational interventions in LMICs have identified *ICTs* as an effective tool, these synthesis studies have not included mobile technologies. There have been a number of literature reviews and landscape reviews on the use of mobiles to support education in LMICs (e.g. Valk et 2010; Wagner et al 2014; GIZ 2015), but

¹⁹ For a detailed discussion on the relationship between claimed impacts for an intervention (in any area) and the size and nature of evidence required to support these claims, see Guyatt et al (2011), White (2015), and Glennester (2017).
²⁰ This acknowledges important contributions on the challenges of evaluating the effects of mobile learning

²⁰ This acknowledges important contributions on the challenges of evaluating the effects of mobile learning programmes (Vavoula & Sharples 2009).

these do not conduct a formal methodological synthesis of mobile learning intervention effects. The only studies to do so are presented by Tamim and peers (2015) and Hassler and colleagues (2016), but both reviews are only focused on tablets and do not limit their scope to LMICs exclusively.

In sum, there is a strong challenge that claims of ML4D's effectiveness are rarely based on reliable systematic evidence and that there is a current gap in ML4D's evidence-base at a research synthesis level. Despite a growing number of rich and diverse primary research studies, no attempts have been made to analyse or synthesise the combined knowledge in the domain using formal methods of research synthesis such as systematic review and meta-analysis. This evidence gap is particular concerning given the magnitude of claims to mobile learning's large and disruptive impact on education and development in LMICs (Traxler & Vosloo 2014; Unwin, 2015; Langer et al 2014).

2.3.2 Critique 2: Simplistic causalities between mobile technologies, education, and development

A frequent explanation offered for the absence of evidence on mobile learning's effects relates to a second critique on mobile learning's assumed contribution to development: that ML4D proponents have adopted overly simple causalities between access and use of mobile technologies, education outcomes, and development (Wagner 2014; Traxler 2013b; Unwin 2015). This critique is particularly targeted at the assumed integration of ML4D with the former MDG an EFA targets (Traxler 2013b; Leach 2008). It holds that ML4D proponents such as UNESCO have not sufficiently conceptualised either the use of mobiles, educational outcomes, nor development. First, all three components of ML4D remain under-defined in most policy documents on ML4D (Traxler 2013b; Unwin 2015). Second, the causal links and pathways between the three components are outlined in a narrow linear fashion. Endogenous access to and use of mobile devices is assumed to lead to educational applications of the technologies; these applications in return are assumed to lead to improved educational outcomes; which then somehow translate into a black box of development impact (Traxler 2013a; Wagner 2014). In this linear pathway, key questions such as why access to and the use of mobiles is able to circumvent systemic challenges in LMICs, which negated other educational inputs desired educational outcomes, are not addressed. Unfortunately, neither the introduction of technology, nor the process of education, let alone development, seems to follow such a linear pathway in practice (Leach 2008; Unwin 2015).

As a result, ML4D seems to lack a viable theory of change (Wagner 2014). Assuming direct causalities between three complicated to complex problems in their own right (i.e. effective use and integration of mobile technologies in educational processes; enhancing educational performance and outcomes; and reductions in poverty and inequalities) is described as 'naïve' (Unwin 2015) or as 'a failure to think clearly' (Traxler 2013b). In this pathway, mobile technologies essentially act as a magic bullet in education in LMICs which 'ought to be successful' seeing the impact mobile technologies had on other development domains (ibid). If mobile learning were to be able to support such a wide range of development outcomesimproving educational attainment, building lifeskills, supporting gender empowerment, and integrating disadvantaged groups into the educational system (UNESCO 2012c)-mobile learning requires multiple and overlapping pathways linking technological inputs to causal changes on these diverse education and development outcomes. For example, if mobile learning is embedded within existing educational structures that are subject to systemic limitations—say an exclusion of certain groups of learners—ML4D's theory of change needs to be explicit about the processes and outcomes that mobile learning triggers which are different from the educational status quo and therefore allow the assumption that these systemic limitations will be circumvented. In short, while mobile devices might feature intrinsic affordances that could facilitate transformative pedagogical innovations, mobile learning programmes—and since ML4D interventions as well—need to be explicitly designed for this objective.

This critique is thus not so much about whether mobile learning's contribution to development follows effective or meaningful pathways; rather it is about the absence of an explicit definition of these pathways, for example in form of a theory of change for ML4D programmes (Wagner 2014; Traxler 2013a; Unwin 2015). There is thus much conceptual development left for ML4D scholars and Traxler (2013b) further cautions that the development of a programmatic theory and logic model is just the first of many steps in ML4D conceptual development, flagging the need to unpack mobile technologies inherent ideologies and the impact of political realities in education in LMICs.

Yet, a range of scholars have expressed concerns with this line of argument (e.g. Ally 2009; Asabere 2013; Leach 2008; ICCD 2013; Donner 2016; Kinuthia & Marshall 2013). While not disputing that mobile learning programmes in LMICs suffer from simplistic causalities, they dispute that ML4D has no theoretical grounding at all. Leach (2008) and Donner (2016) in particular make a strong case that many mobile learning programmes in LMICs supported by development agencies can be seen as to follow a human capital approach (Becker 1964) to both education and development. This argument seems to be able to at least partially explain how development actors such as the UNESCO assume an implicit link between education outcomes and development. Framing education with a narrow purpose of contributing to increased human capital allows for a linear link between increased education outcomes, such as skills, and economic definitions of development, such as increases in productivity and Gross Domestic Product (GDP).

2.3.3 Critique 3: A techno-centric approach to ML4D and its risks

A third line of critique on mobile learning's positioning in relation to education and development unpacks the implications of following a techno-centric approach to ML4D. This critique is broader than the above critique of simplistic causalities, which is an inherent issue in device-based, techno-centric conceptions of ML4D²¹. In a narrow techno-centric conception of ML4D, access to mobile devices or access to educational content on mobile devices becomes the primary contribution of mobile learning to education and development in LMICs. I have shown above that such a focus on access was particular prevalent in the positioning of mobile learning to contribute to formal development goals and in the positioning of mobile learning as to disrupt and revolutionise education in LMICs. This focus on access and the provisioning of technologies as the key component of mobile learning interventions is problematic in two ways: one, it neglects mobile learning's pedagogical potential and underpinning, and, two, it fuels a deficit model of education in LMICs risking to undermine its current structures through the creation of parallel educational processes.

First, as outlined in 2.1.3 and 2.2.3 the innovation and educational contribution of mobile learning rests on the different pedagogies, often more learner-centred educational approaches, that the affordances of mobiles can support. From an educational perspective, the mere provision of mobile devices and educational content on mobile devices excludes this innovation and contribution. The progression of definitions of mobile learning in the educational sector has for good reasons moved away from a device-based, techno-centric understanding of mobile learning towards a learning activity-based, pedagogical understanding. In the development sector, however, this progression has stalled and both Traxler (2013b) and Winters (2013) highlight that current definitions of mobile learning in international development are directly at odds with definitions in the educational sector. This

²¹ To be clear, this is not to suggest that techno-centric approaches to ML4D by design are inherently problematic. For example, in cases where the educational environment is stable and the applied device has a close fit with the educational need, the provision of technologies might be all that is needed as an intervention design (e.g. Aker et al 2010; Velghe 2014). I would argue, however, that these conditions are rarely given in the educational contexts of LMICs.

outdated techno-centric understanding of mobile learning has negative implications for the potential of ML4D programmes. For example, in a review of the application of mobile learning in LMICs during the course of the Millennium Development Goals, Unwin (2015) identified a techno-centric approach as the main obstacle to a transformative role of mobile learning. This overlaps with similar conclusions reached by an earlier review of teachers' roles in mobile learning in LMICs (Isaacs 2012) and experiences of selected pedagogical-rich ML4D programmes (e.g. Buckner & Kim 2014; Ekanyanke 2013).

Second, a techno-centric approach to ML4D runs the risk of fuelling a deficit model of education in LMICs and to undermine its current structures through the creation of parallel educational processes (Winters 2013; 2015; Traxler 2013b; Smith et al 2011). Introducing the mere provisioning of mobile technologies or content on mobiles as on a par with or even as a solution to existing educational processes requires a reframing of education in LMICs as a series of absences; for example, a lack of text-books, a lack of access to information, a lack of motivated teachers, and so forth. Only through reframing education in this matter can simple inputs such as access to mobile technologies be positioned as a solution to educational structures and systems as 'dysfunctional', 'broken', or 'in crisis' can the need for mobile technologies as an appropriate solution be delineated (Traxler 2013b; Winters 2013). Access-driven positionings of ML4D such as the need for a 'teacher in the pocket' or 'a classroom in your hand' seem to rapidly lose their appeal once existing classrooms and practicing teachers were framed in a more positive way.

By presenting existing educational structures and the actors shaping them as part of a dysfunctional education system, a techno-centric approach to ML4D allows mobile learning programmes in LMICs to be positioned as an adjunct to the formal education system, rather than an integral part of it (Winters 2013; 2015). The introduction of mobile technologies then serves to construct alternative, inherently more effective structures and, in doing so, instead of transforming the existing system, sets up parallel educational structures. This parallel set-up has clear negative implications for the existing educational system and its actors, which at best are viewed as non-relevant, and at worst regarded as obstacles to be removed in order to not hinder educational 'progress' (Traxler 2013b). The consequences of framing education in terms of access to and delivery of educational content on sophisticated mobile devices are particularly grim for cost-intensive investments into teaching staff and qualifications, educational infrastructure, and curriculum development. For example, if applications on mobile devices (e.g. the Worldreader app) are attributed the power to 'revolutionise' reading and thus literacy in LMICs, there is little need to invest scarce public

resources into the training of teachers to more effectively facilitate literacy lessons. To be clear, the above narrative is observed in practice. UNESCO and UNICEF are pioneering the development of a 'Digital school in a box', a literal box featuring a solarpowered multimedia kit built around a laptop, projector, document camera, and speaker, that is assumed to deliver formal education in remote rural areas (UNESCO/UNICEF 2017). A similar concept is applied by the DFID-funded solar classroom in the box (SolarClassRoom 2017). Tamim and peers' (2015) also report on an OLPC-inspired tablet programme in rural villages in Ethiopia which lacked access to formal schooling. Here, "tablets were dropped off in the villages in boxes taped closed, with no instruction, and the children were followed to investigate how much they could learn without training or teacher support" (Tamin et al 2015: 18). These practical examples reflect attempts to develop exogenous mobile education solutions that could provide an easy fix to bypass or revolutionise the existing, cost-intensive education system.

Subscribing to a techno-centric approach to ML4D thus carries the risk to undermine existing educational structures in LMICs. Rather than supporting these structures and the actors shaping them, mobile learning is positioned at best as adjunct to them and at worst as in direct conflict to them. It is in the context of this conflict that the need for an explicit and conceptualisation of the role and contribution of mobile learning in international development has emerged.

Conclusion

This chapter has reviewed the existing research and practice of ML4D. In particular, it has investigated how the use of mobile technologies to support education in LMICs is conceptually linked to development outcomes. It thereby outlines the rationale for my main research questions evidencing the conceptual and empirical gap on mobile learning's contribution to development. Following a review of the pedagogical grounding of using mobiles technologies to support learning processes, that is mobile learning, I contrasted this pedagogical understanding of mobile learning with its understanding in the development literature. This juxtaposition illustrates that mobile learning is largely understood in a technocentric manner in the context of development with an emphasis placed on the access to mobiles and the educational content hosted by them. I also illustrate that mobile learning's contribution to development outcomes is often framed as to disrupt and revolutionise education in LMICs with the ability to generate vast gains in learning outcomes.

However, I then provide three key critiques in the literature that question the conceptualisation of mobile learning as a development intervention. First, there is a disconnect between the assumed vast development potential of ML4D and the empirical evidence-base on the effects of using mobile technologies to support education in LMICs. This justifies the needs for my systematic review of the overall effectiveness of ML4D interventions. Second, there is a lack of an explicit theory of change for how the use of mobile technologies is assumed to support education in LMICs and subsequent development outcomes. Current ML4D programmes and policy positions assume simplistic causalities and under-define both education and development. This gap justifies my second systematic review objective of developing a theory of change of ML4D. Third, due to its techno-centric positioning, ML4D interventions run a risk of circumventing and undermining the existing education system. This positioning presents ML4D as exogenous to existing educational efforts and is at odds with a people-focused perspective of education and development in LMICs. Based on the above, I then conclude that the positioning and conceptualisation of ML4D as a development intervention is doubtful. An explicit conceptual framework is required to outline and define mobile learning's contribution to development, and I propose the CA as such a framework, which is discussed next.

Chapter 3. Literature review 2: the Capability Approach

Introduction

This chapter presents the second of my two literature review chapters and introduces the Capability Approach (CA) as conceptual framework to analyse development processes and outcomes. It outlines the rationale for introducing the CA as a conceptual framework to mobile learning for development (ML4D) based on the empirical and conceptual gap on mobile learning's role and contribution to international development illustrated in the previous chapter. It also reviews the theoretical foundation of the CA. The chapter's introduction of the CA aims to explore whether there are sufficient synergies between the CA's conceptual devices and the needs of my targeted conceptualisation of ML4D; and whether such synergies would provide a space in which to apply the CA as a conceptual framework to ML4D.

To explore these conceptual synergies, I first review literature on the theoretical and conceptual foundations of the CA itself. This aims to facilitate a basic understanding of the CA and its contribution to development theory and practice. This introduction is then followed by an investigation of the operationalisation of the CA in two bodies of research closely related to ML4D: ICT4D and education in Low- and Middle-Income Countries (LMICs). These serve to underline my proposition that the CA indeed presents a relevant conceptual framework to define and outline the contribution of mobile learning to education and development in LMICs. The chapter concludes with a brief summary.

3.1 Introducing the Capability Approach to ML4D

The conceptual space for the introduction of the CA to ML4D arises from four gaps and shortcomings in current understandings of ML4D reviewed above:

- (1) A narrow view of access to mobile technologies.
- (2) A reductionist notion of the processes and outcomes of education in LMICs.
- (3) A reductionist notion of the processes and outcomes of development in LMICs.
- (4) An exogenous positioning of mobile learning as an adjunct rather than integral part of the education system in LMICs.

Using the CA as a conceptual lens, I show how ML4D can be moved away from a technocentric, access-based view of mobile technologies towards a view that prioritises what educational actors can do with these technologies in their respective contexts. I illustrate how a CA perspective on ML4D guides a more holistic conception of education outcomes and definitions of quality education in LMICs. This directly relates to a similar conceptualisation of what constitutes 'development' in ML4D using the CA's definition of 'Development as Freedom' (Sen 1999).

My assumption that the CA could be operationalised in this pursuit is based on its existing application in two bodies of literature closely related to ML4D: ICT4D (e.g. Kleine 2013; Zheng 2009; Zelezny-Green 2017) and education in LMICs (e.g. Walker &Unterhalter 2007; Unterhalter 2005). Below, I first provide a brief introduction to the CA and its underlying theory and concepts, before then describing the application of the CA in ICT4D and education in LMICs. Having reviewed its application as a conceptual lens, I then justify my assumption that the CA can similarly guide a conceptualisation of the role of mobile learning in international development.

3.1.1 The Capability Approach: theory and key concepts²²

The Capability Approach is a normative and evaluatory framework to conceptualise the space in which human well-being and development should be assessed (Sen 1992; 2009; Alkire 2005; Zheng 2009). It was developed by Amartya Sen beginning with his 1979 Tanner Lecture on Human Values titled 'Equality of What?' (Sen 1982) and subsequently refined over the last four decades (Sen 1985a; 1985b; 1992; 1999; 2009). Much of Sen's work on

²² The purpose of this section is not to provide a comprehensive introduction to the complexities and subtleties of the CA. Rather, it aims to provide an introduction to the core concepts and theoretical underpinnings of the CA with a particular focus on how they have been applied in the context of technology and development.

the CA is highly economic and philosophical and not easily accessible to a wider development studies audience (Kleine 2013). As a result, his comprehensive overview of a capabilities perspective on international development, "Development as Freedom" (Sen 1999) is most associated with the CA by a development audience. However, Sen's work, as well as the CA itself, has applicability to human development and well-being in any research domain. Sen's development of the CA bore out of a dissatisfaction of different approaches to conceptualise human well-being and development, most notably utilitarism (Bentham 1789; Mill 1861), resource-based approaches such as income-, wealth-, and asset-based measures (e.g. GDP and other indicators of national accounts including monetary poverty lines as applied by the World Bank), and means-orientated evaluatory approaches such as John Rawl's 'primary goods' (Rawls 1972) (Sen 1999; 2009). His aim was to provide an alternative informational space that could better capture an individual's real opportunity to choose and pursue a live she has reason to value; which he argues is not sufficiently provided for when focusing on individual's levels of utilities, income, primary goods, etc. (Sen 1999).

It is important to consider though that Sen is explicit that the CA is not a theory but an approach that constitutes of a range of analytical devices that can be applied in different ways (Sen 1999). Some scholars therefore prefer the term 'capabilit*ies* approach' to reflect the plurality of perspectives comprising the CA's analytical devices (e.g. Kleine 2013; Venkatapuram 2011). A key iteration of the CA refers to Nussbaum's development of the CA from a moral and political philosophy perspective, the intricacies of which are not discussed here. It is important to recognise, however, that the CA's association with the concept of human development in the field of international development, which results from its contribution to UNDP's Human Development Approach' (Nussbaum 2010; Venkatapuram 2011). Lastly, before turning into more detail to the analytical devices of the CA itself, it is important to acknowledge that the CA remains deliberately open-ended and receives ongoing conceptual development (Sen 2009). Key developments in the broad field of international development (Sen 2009). Key developments in the broad field of international development (2003).

Capabilities and functionings

As an analytical and ethical framework, the CA asserts that the right space or informational basis on which to assess human well-being and development are capabilities and functionings. Functionings refer to "the various things a person may value doing or being" (Sen 1999: 75). For example, valued being and doings in the context of ML4D could refer to

'being connected' or 'generating content'. Capabilities, then, refers to "the various combinations of functionings (beings and doings) that the person can achieve (Sen 1992: 40). This positions capability as a freedom, the freedom of a person to "lead one type of life or another" (ibid). In short, "a functioning is an achievement, whereas a capability is the ability to achieve" (Sen 1987: 36). Taken together, capabilities and functionings are thus the key analytical devices of the CA.

The CA's core inquiry is what each person is able to do or be. Such an investigation is necessarily focused on choice or freedom and Sen asserts that it is crucial to make a distinction between a freedom or opportunity to be and do certain things (i.e. capability) and the achieved outcome of being and doings these things (i.e. functionings). To illustrate, a teacher in an affluent and well-resourced school and who has been trained in technology usage may have the same educational functioning in terms of being promoted for achieving certain pass rates as a teacher in a remote school without resources and professional development. However, the capability set or opportunity of both teachers to achieve this functioning is not equal.

The normative dimension of the CA then posits that the focus of social interventions and social arrangements should be firmly on the expansions of people's capabilities. It holds that "the crucial good society should be promoting for their people is a set of opportunities, or substantial freedoms which people then may or may not exercise in action" (Nussbaum 2011: 20). This is the rationale for Sen's definition of development as "the process of expanding the real freedoms that people enjoy" (Sen 1999: 3)²³. In this conception, the expansion of effective freedoms to live the life one has reason to value becomes both the principal means and primary end of development. That is, an expansion of human freedoms has intrinsic value and the assessment of development has to be informed by it (freedom's constitutive role); but an expansion of people's freedoms (i.e. capability) to pursue their valued being and doings is likewise a key mechanism to support development outcomes (freedom's instrumental role). This conception of development differs strongly from a definition and assessment of development in terms of income, consumption, or economic activity.

Resources, commodities, and conversion factors

In the CA, capabilities and functionings do not exist in a black box. Capabilities are generated by inputs, which refer to resources or commodities and the characteristics thereof

²³ This includes the removal of major sources of 'unfreedoms' such as tyranny, poor public facilities, repressive social norms (Sen 1999: 3).

(Sen 1992). Mobile technologies provide a good example of a resource. In the language of the CA, resources present the 'means to achieve' (Sen 1992; Rosebyn 2005). However, the ability of 'means to achieve' to translate into capability crucially depends on a person's conversion factors. For example, a person might have access to mobile technologies, but might not explore the capability to be able to connect with friends through the device due to a lack of network reception. Sen identified three different types of conversion factors— personal, social, and environmental—which mitigate the extent to which people can generate capabilities from resources provided. This conception reduces the importance attributed to the provisioning of resources such as mobile technologies, which only become meaningful developmental inputs through the capabilities that people are able to generate of them.

If a person is able to transform access to a resource through her conversion factors into a capability, she has gained the 'freedom to achieve' (Sen 1992; Rosebyn 2005). The freedom to achieve thus represent her set of available capabilities that her conversion factors allow her to explore based on the provided resources. However, freedom to achieve is distinct from the actual achievement or valued functioning. For a range of reasons, a person might choose to not translate a freedom to achieve into an actual functioning. As an example, a teacher might be provided with a mobile device (resources) and possess the skills (conversion factor) to use it to generate more learner-centred lesson designs (educational capability), but chose not to do so in practice (functioning) as she might prefer a different type of lesson design. This distinction between freedom to achieve and actual achievement is important to note as it highlights the need for choice and agency to be taken into consideration in addition to the focus on resources and conversion factors as the only determinants of capabilities and functionings (Sen 1992; Haenssgen & Ariana 2017).

Agency and well-being

The full set of capabilities of a person represents her freedom to achieve well-being and agency in Sen's CA. Sen derives his definition of agency in terms of an agent's ability or freedom to pursue and shape her own destiny and perception of 'the good life'. This leads to a people-focused (or, more formally, agent-orientated) perspective in which "people have to be seen as being actively involved—given the opportunity—in shaping their own destiny, and not just as passive recipients of the fruits of cunning development programmes" (Sen 1999: 53). This central importance of agency underlines the need for people themselves to be able to define a life they have reason to value including the ability to explore different capabilities and to choose whether or not to translate them into functionings.

57

Well-being then is constituted of the actual opportunities or real freedoms a person enjoys and her agency and choice to translate these into the achievement of valued beings and doings. In short, capabilities are the basis for the achievement of functionings, subject to people's agency and choice, from which they may derive well-being and human development. The language of human development here is closer to Nussbaum's interpretation of the CA. She holds that it is not necessary to make a distinction between agency-well-being, and capabilities-functionings—in particular in relation to the further distinction of well-being and agency freedom vs. achievement—as the latter (capabilities-functionings) sufficiently incorporate the notion of former (Nussbaum 2011: 200; Oosterlaken 2015)²⁴. Her language of human development, though, essentially captures the same process and outcomes as described by Sen as well-being²⁵.

Concluding remarks on the theory and concepts of the CA

To conclude this abstract introduction to the theory and key concepts of the CA, I provide below an example of their application related to mobile learning in LMICs. Gigler (2015) proposes to evaluate the provisioning of ICTs in development in terms of informational capabilities. Using the CA, this would investigate the capability or opportunity of a recipient of these ICTs to transform access to ICTs into human agency and real opportunities in society to achieve the things she values doing or being, such as learning online. The CA just allows us to investigate how a person's access to ICT can be converted into her ability to integrate the use of ICTs into the pursuit of her own conception of a good life. It also highlights important differences in people's ability to transform access into effective opportunities and valued actions and outcomes such as social norms and power relations.

It is also important to remember that the CA is not a prescriptive theory of development. That is, the CA does not intend to explain poverty, inequality, agency, and well-being or what type of programmes to design in order to address them (Robeyns 2005; Oosterlaken 2015; Zheng 2009). Rather, it presents a "tool and framework with which to *conceptualise* and *evaluate* these phenomena" (Robeyns 2005: 94, emphasis in original). In relation to my investigation of a conceptualisation of ML4D, the CA is thus not applicable to provide guidance on the detailed practical causal pathways between using mobile technologies in an educational context, how this use might support learning outcomes, or how these learning outcomes might be linked to development processes. It only enters my efforts to

²⁴ This is sometimes also referred to as a distinction between a narrow or broad application of the CA. A narrow application only focuses on the capabilities-functioning aspect as a concept of well-being, whereas a broad application includes issues of agency, procedural fairness, justice, etc. (Oosterlaken 2015; Robeyns 2011)

²⁵ Throughout the later chapters, I will use well-being and human development interchangeably given the association of the CA in international development with the Human Development Index and the often vague and contested nature of the term well-being.

conceptualise ML4D as a broad conceptual framework providing me with analytical devices required to unpack and investigate these detailed practical causal pathways. In short, the CA provides me with an analytical space or informational basis on which to guide the conceptualisation of what constitutes development in ML4D. As there is currently no existing theoretical and empirical work linking the CA to mobile learning in LMICs, I next present a short detour into the rich literature on the use of the CA in the related research area of ICT4D and education in LMICs. This serves to support my conceptual case for the application of the CA to ML4D providing empirical examples of how the CA has been operationalised to guide similar conceptual inquiries in related bodies of research.

3.2 Investigating technology and development from a capabilities perspective

The role of technology in development is contested and a rich body of evidence testifies to both its creative and destructive potential (Castells & Himanen 2014; Zheng 2007; Oosterlaken 2015; Haenssgen & Ariana 2017). Modernisation theories to development present technological change as a key driver of productivity and therefore economic growth, which is assumed to lead countries onto a pre-defined way towards socio-economic 'progress' (Rostow 1960). In this process, technology is an exogenous input to be adopted in LMICs following the path of technological innovation in HICs. Examples of this model of technology transfer include industrial machinery, green revolution technologies, and most recently ICTs (Castells & Himanen 2014). Scholars investigating the discourse of the latter in development in particular claimed technologies' impact to be overly framed in economic terms. Kleine (2013), for example, shows how the internet was positioned in the context of development as a neutral instrument that would allow economies in LMICs to 'leapfrog' stages of development reaching the pre-defined desirable notion of progressing into 'knowledge economies'. ICT4D scholars have criticised this techno-centric and growthfocused approach as overlooking issues of injustice in access and use, ownership and participation, as well as social and cultural impacts of technology adoption (Oosterlaken, 2015; Kleine 2013; Zheng 2009; Zheng & Stahl 2011). ICT4D initiatives—it was claimed did not pay sufficient attention to what people can effectively do with the introduced technologies and whether they had the power to decide on the terms and desired outcomes of technology usage and access.

To synthesise and conceptualise these critiques into a coherent framework, the CA has been suggested as a normative lens to investigate the nexus of technology and development by a number of ICT4D scholars (e.g. Oosterlaken 2015; Kleine 2013; Zheng 2009; Haenssgen & Ariana 2017; Andersson et al 2012)²⁶. In this, using the CA as a conceptual framework to investigate the impact of ICT4D interventions is positioned to shift attention from the technological artefact towards the individual's effective opportunities to transform the access and usage of the artefact into valuable beings and doings that enhance her well-being. A number of in-depth case studies have shown how this changes the narrative of ICT4D in practice.

For example, Gigler (2015) assesses the impact of an ICT for indigenous development programme in rural Bolivia on ICT users from a capabilities perspective. Entitled

²⁶ See Oosterlaken (2012) for an extensive overview of the literature.

'Development as freedom in the digital age', he conducts a mixed-methods case study on how indigenous people in Bolivia perceive, access, and use ICTs; and how such use influences their livelihoods. ICTs refer to desktop computers mainly and the spread of the internet in rural areas more broadly. He applies the CA to conceptualise the impact of ICTs in rural livelihoods in order to track how access to ICTs can translate into changes in indigenous people's well-being and human development. His research finds that ICTs can significantly enhance indigenous peoples' human and social capabilities, which translated into increased well-being and human development. However, the catalyst for this process are people's 'informational capabilities', which Gigler argues should be understood as a catalytic capability akin to literacy or health. Approaching ICT4D from a CA perspective, he urges to "deemphasizes the role of technology" in favour of individual's and communities' abilities to appropriate and transform new technologies into their socio-cultural contexts and aspirations about life (Gigler 2015: 27).

Kleine (2013) also uses the CA in a related setting examining how access to telecentres in Chile and a new national ICT policy affect the lives of rural entrepreneurs. Her case study supplements the CA with DFID's Livelihoods' framework (DFID 1999) and Aslop and Heinson's (2005) empowerment framework in order to operationalise the CA as a tool to evaluate the development outcomes associated with access and use of ICTs in Chile. Using the CA, she develops a 'choice-focused' view of ICT and development showing how ICTs might function as multipurpose technologies which—depending on users' choices—can support (and prevent) a variety of social and economic outcomes. She then applies this choice framework in her empirical case study and argues that ICTs made a contribution to human development and well-being for Chilean rural entrepreneurs when applied in a bottom-up fashion to support entrepreneurs' freedom and agency to explore ICTs in line with the live they had reasoned to value. As much as expanding choice and agency, the investigated ICT policies also carried the potential to constrain choice and agency highlighting the need for a careful design and interrogation of facilitating access and use of ICTs to further a people-centred pursuit of development.

To turn to an example of mobile technologies and the CA in ICT4D, Oosterlaken (2015) highlights how an iPod-facilitated agricultural extension programme in rural Zimbabwe expanded a range of functionings and capabilities of importance to farmers. Investigating the 'Local content, local voice' project, she outlines how farmers integrated the provided digital extension opportunities to pursue their well-being goals such as increased income from their livestock. She illustrates how many of the well-being achievements of farmers were not initially targeted by the project design and how a CA investigation highlighted the importance

61

of conversion factors in farmers' well-being achievements. However, Oosterlaken also cautions that technological design choices, such as not opting for constant connectivity and communication affordances, limited the agency achievement of the project. The mobile technology design chosen for the project thus limited its ability to contribute to people's self-determined pursuit of other well-being goals not related to the project objectives.

Lastly, Poveda and Roberts (2017) provide an ICT4D example using the CA that is somewhat related to education. The authors merge critical theories (critical pedagogy and critical feminism) with Sen's CA approach in two cases studies of ICT4D programmes in Zambia and Brazil. They find that in both case studies participants were able to appropriate technologies to support their practical needs, but further to also develop a critical conscious of the structural inequalities surrounding them and how they could use the technologies to uproot these. They use these findings to argue that the CA complemented with critical theories can provide an effective means to guide a conceptualisation of ICT4D that focuses on people's critical agency to unlock ICTs' transformational potential in relation to existing social structures.

These four cited examples show that in the research area of ICT4D the CA has served as an effective conceptual framework to gain a richer understanding of the relations between technology and human development. There are clear parallels in the rationale for using the CA between the research area of ICT4D and its proposed usage to guide the framing of mobile learning programmes in LMICs: a doubtful positioning of access and usage of technology as an end rather than a means; narrow conceptions of intended benefits of technology usage; and a neglect of local realities and participation in the design of programmes and technologies, among other.

However, from a mobile learning perspective the current work of ICT4D and the CA has a number of important gaps. First, there is little work on mobile technologies and the potential capabilities that users might develop from their particular functionalities. Most work on ICT4D has focused on ICT policies and fixed infrastructure such as telecentres, desktop computers, and access to the internet in general. This overlooks important aspects of mobile technologies, which are far more widespread in terms of access, and personal in terms of use, than other ICTs. Frankly, mobile technologies might be relevant to a more diverse set of valued opportunities and functionings. Second, technologies are not discussed in an educational setting, and neither is the role of education, or pedagogies within education, investigated. Education might be presented as required to build basic human capabilities, but is not discussed as a dimension of capabilities in its own right.

Third, as ICTs often cannot assume such a diverse role in users' pursuit of a good life as mobile devices can, the capabilities associated with the use of ICTs are at times framed overly sectoral. For example, Gigler (2015) sees one set of capabilities, i.e. informational capabilities, to serve as a catalyst for another set, i.e. social capabilities. Mobile technologies due to their constant and personalised use in all areas of life might be able to help users explore multiple types of capabilities simultaneously. On account of these gaps, conceptual frameworks operationalising the CA in ICT4D such as Kleine's (2013) Choice Framework cannot simply be transferred to guide the conception of ML4D. Mobile learning is both an educational and a technological intervention and its investigation from a capabilities perspective is required to reflect this.

3.3 Capabilities, mobile technologies, and education in LMICs

The observation that mobile technologies constantly connected to the internet might occupy a special space in the nexus of technology and human development is not novel to this thesis and has been raised in the literature before (Sen 2010; Oosterlaken 2015; Smith et al 2011; Kleine 2013). Most prominently, in a brief article Sen himself presents mobile phones to be "generally freedom-enhancing" (2010: 2). Elsewhere, mobile technologies are seen as driving "one of the greatest expansions of human capabilities in known history (...)" (Smith et al 2011: 77) and have been described as "technologies of choice" (Kleine 2013: 8).

While this should not be read as portraying mobile technologies as anything but mere means, mobile devices seem to combine a number of particular affordances that might allow them to relate to different types of opportunities valued by users (Kleine 2013; Oosterlaken 2015). Mobiles' ease of use and affordability have made access to the technology ubiquitous and allow a majority of people the opportunity of personal and contextualised usage. Mobile users might then be able to translate this usage into different information and communication opportunities, which in return might support human development, for example allowing users to build social networks and to explore alternative livelihood opportunities. It thus might be worth exploring whether the combination of these unique affordances of mobiles—in particular accessibility, personalisation, contextualisation, connectivity, access to information, and networking effects—might allow mobile technologies to assume a multidimensional role in the expansion of people's capabilities and well-being.

Mobile learning as an educational approach builds on these particular affordances of mobile devices and the pedagogical innovations and opportunities they can create for learners and teachers (Roschelle 2003; Wali et al 2008). As explained above, a major criticism on the current application of mobile learning in LMICs is the focus on the technological artefact rather than its pedagogical value (Winters 2013; Traxler 2013a). Mobile learning as a socio-cultural educational approach might be well-suited to support some of the specific challenges of education systems in LMICs, for example frontal teaching methods, foreign curricula, and low levels of agency for teachers and learners alike. It is due to this specific contribution that mobile learning has been positioned to be of particular relevance to support education in LMICs.

However, it is important to unpack what is meant when referring to 'quality education'. In this remit, the CA has found wide application too (e.g. Unterhalter 2005; Walker & Unterhalter 2006; Vaughan & Walker 2012). Concerning education and international development, the

CA has been used as a tool to generate a richer understanding of the perceived 'low quality' of education in some LMICs (Unterhalter 2015; Walker 2006; Tao 2013; Tikly & Barrett, 2011). Framing 'quality education' in terms of capabilities and functionings shifts the discourse from educational inputs and resources to the agency of teachers and learners to own and determine the educational process. This shift might re-define quality education as an education that supports what each person has reason to value and that provides equality of opportunity rather than equality of resources and outcomes (Walker & Unterhalter 2006; Tikly & Barrett 2011).

A capabilities perspective on education therefore provides an alternative to a range of evaluatory educational approaches such as assessing education in terms of learners' attainment, human capital, or structural and power issues. Unterhalter (2005) outlines how a CA perspective on education highlights two key issues in education in LMICs: one, teachers' and learners' conversion of educational inputs and resources and the inequalities and obstacles in this conversion; and, two, that educational outcomes cannot be narrowly defined in terms of test scores or investments in inputs. As much as being an instrumental capability, education itself is of intrinsic value and, if aimed at supporting development, learners in LMICs require equal educational capabilities, not equal educational inputs (Unterhalter 2005).

A range of practical applications show how a CA perspective alters the conception of quality education in LMICs. Walker (2006), for example, highlights the implication of a capabilities view on education for girl learners in South Africa. She identifies eight capabilities valued by learners (e.g. being able to have choices; being able to participate in learning; bodily and emotional integrity) and shows how these expand the nature and scope of what to consider when aiming to facilitate 'quality education' in LMICs. In the context of the Implementing Education Quality in Low Income Countries (EdQual) programme, Tilky and Barrett (2011) also apply the CA for a practical assessment of education interventions. They propose to marry the CA with a social justice perspective to highlight three dimensions of quality education: inclusion, relevance, and democracy, which are assumed to reflect the capabilities that educational actors have reason to value.

Lastly, the CA has also been used to assess the educational realities of teachers in LMICs (e.g. Buckler 2015; 2016; Tao 2013; Cin & Walker 2013). The role of teachers in supporting quality education in LIMCs has been particularly contested and educators' behaviours, such as absenteeism, lack of preparation, and rote teaching, are often claimed as main causes of quality issues (Carr-Hill & Ndalichako 2005; Chaudhury et al 2006). Using the CA, Tao

(2013) explains how these behaviours can be reframed as "products of the constraints on teachers' valued functionings inside and outside of schools" (2013: 7). She identifies a number of valued personal and occupational functionings (e.g. being able to take care of family; being able to help students learn) from whose achievement teachers are hindered by surrounding structures and working conditions. Teachers' behaviour and performance are mitigated by navigating these constraints in the pursuit of their valued beings and doings. Lastly, Buckler's research on teachers' professional ambitions and desires in five Sub-Saharan countries comes to similar conclusions (2015; 2016). She identifies 16 capabilities of teachers that overlap to a large extent with Tao (2013). Her work illustrates how teachers own conception of quality teaching is shaped by a range of educational capabilities and functionings, but directly at odds with educational policy-makers' definitions of quality teaching.

The CA has thus been operationalised as an analytical lens in a range of educational settings in LMICs. None of these, however, have focused on the use of technology—either mobile or fixed—as an educational input. There seems to be good reason to believe that learners and teachers can generate different types of capabilities and valued functionings of the conversion of mobile technologies as an educational resource. For example, mobile learning from a pedagogical perspective is associated with the ability to cross contexts and to personalise and tailor learning and teaching. As explained above, mobiles' particular affordances might allow for a range of different educational opportunities to be created should teachers and learners wish to do so. Current work on the CA in education in LMICs has not explored these opportunities and there thus remains a gap on investigation mobile learning in LMICs from a capabilities perspective.

3.4 ML4D and the Capability Approach

In sum, there seems to be reason to believe that the CA can be applied as an effective conceptual framework to investigate mobile learning in LMICs. Theoretical and empirical work in the related field of ICT4D has shown that the CA can be operationalised to yield a richer understanding about the role of technologies in human development. Further, there seems to be synergies in the pedagogical underpinning of mobile learning and the perception of mobile technologies as potentially supporting a variety of human capabilities simultaneously. The CA has also emerged as an important voice in debates on improving 'quality' of education in LMICs and the role of teachers in this process. In short, using a CA conception of ML4D can go beyond a simplistic understanding of access to technology and content as mobile learning's main contribution to development. It also enhances the conception of both education and development, which I argued above as currently being under-defined in ML4D.

While these synergies can justify the application of the CA to ML4D in general, an important gap remains as the literature has not investigated the interplay between mobile technologies, education, and capabilities. Capability frameworks and lists of capabilities or functionings developed in the context of ICT4D and education in LMICs cannot simply be transferred to ML4D (e.g. Kleine 2013; Gigler 2015; Walker 2006; Rao 2013). The nature of using mobile technologies to support education and development in LMICs seems to intersect across a range of relevant areas of capabilities as it presents both a technological and educational input. It thus needs to be investigated in its own right rather than adopting of-the-shelf conceptual frameworks developed elsewhere. To conclude, mobile learning might be able to support the expansion of a different set of capabilities, which thus far have been overlooked in previous research applying the CA in education and technology.

Despite the vacant conceptual space in ML4D to allow the introduction of the CA as an analytical lens, there is, to my knowledge, no existing empirical or theoretical work making the case to apply the CA to ML4D. A small body of work on the educational use of mobiles in LMICs references the CA, but does not use its analytical devices in practice (Andersson & Hatakka 2010; Balasubramanian et al 2010; Sahni et al 2008). My thesis aims to contribute to this an explicit empirical exploration of a ML4D programme from a capabilities perspective in order to investigate the operationalisation of the CA as a conceptual framework for ML4D.

Conclusion

This chapter has reviewed the CA as a conceptual framework to understand and define the contribution of mobile learning to development. It has introduced the theory and concepts underlying the CA and outlined its application in bodies of research closely related to ML4D, namely ICT4D and education in LMICs. Based on this review and the synergies between the CA and the needs of my targeted conceptualisation of ML4D, I conclude that there is indeed a sufficient conceptual space in which to apply the CA as a conceptual framework for ML4D. I conclude that the CA with its alternative conception of development as freedom and consequential focus on the notions of capabilities, functionings, and agency as means to live the life one has reason to value (Sen 1999) does present an effective framework to guide a conceptualisation of ML4D in which the link between mobile learning and development might hold.

In summary, the combined two literature review chapters have set up the empirical and conceptual gaps that my thesis is trying to contribute to. In chapter 2, I have indicated how neither the empirical evidence-base nor the current conceptual approaches of ML4D justify the claims and positioning of mobile learning to support education and development in LMICs. In this chapter, I have proposed the CA a relevant conceptual tool to address this lack of conceptualisation regarding mobile learning's role and contribution in international development. Taken together, my literature review chapters therefore set up the thesis's research questions and contributions to the literature. In the next chapter, I will outline the research design and methodologies applied in addressing these questions before presenting my empirical research findings in chapters 5–9.

Chapter 4. Research design and methodologies

Introduction

This chapter presents the research design and the applied research methodologies. It provides a detailed description of the research process and the data collected and analysed. I also discuss the ethics involved in these processes. In brief, this thesis employs a mixed-methods systematic review followed by a qualitative case study. The chapter justifies how and why the two applied research methodologies were chosen in order to answer the thesis's research questions.

The chapter is structured as follows. I first discuss the overall research design of the thesis before discussing the two respective research methodologies in return. I begin with the design of my mixed-methods review and emphasise the rationale to adopt a mixed-methods approach. I then provide detail on each step of the review process. Thereafter, I discuss the design of the qualitative cases study. I discuss the chosen case of the Information Communication Technology for Rural Education Development (ICT4RED) programme and describe the primary research process in detail. Finally, for both types of research I flag key ethical considerations.

4.1 Research design: overall thesis

My thesis combines a secondary research methodology with a primary research methodology. The secondary research methodology is presented by my mixed-methods systematic review and the primary research methodology is presented by my qualitative case study. Within the systematic review I combine qualitative and quantitative review approaches whereas the case study follows a purely qualitative approach. I use the findings from both research methodologies to answer my overall research question of how the use of mobile technologies as an educational tool in Low- and Middle-Income Countries (LMICs) can be conceptualised as to contribute to development outcomes. Both research methodologies, however, make distinct contributions in answering this question. Figure 4.1 below provides a visual guide to how the different methodologies contribute differently to answering my research question.

My research commences with the mixed-methods systematic review of existing primary research that has investigated the educational use of mobile technologies in LMICs. The review serves two purposes: examining the impact of mobile learning on education and development outcomes and examining how mobile learning for development (ML4D) programmes have outlined the processes and causal links between the use of mobiles, education, and development in LMICs. Taken together, the systematic review therefore allows me to conclude whether empirical and conceptual claims to mobile learning's contribution to development are warranted.

In order for my systematic review to do so, I need to apply a mixed-methods systematic review design. The mixed-methods design allows me to not only assess and synthesise quantitative evidence on the impact of ML4D programmes, but to also assess and synthesise qualitative evidence on the mechanisms and contexts configuring this impact. Only by combining the synthesised quantitative and qualitative evidence on mobile learning programmes in LMICs can I construct a detailed theory of change for mobile learning and plot the identified review findings against this theory of change. The theory of change thus combines the results from the existing quantitative and qualitative research evidence to assess the empirical and conceptual claims to ML4D's impact.

Empirically speaking, the mixed-methods systematic review of ML4D interventions presents the departure point of this thesis. However, while its findings support an assessment and deconstruction of current conceptions of mobile learning's impact on education and development, the review is bound to assessing *existing* research on mobile learning. As a result, the systematic review findings cannot support me in developing a *new* conceptualisation of mobile learning's role and contribution in international development. For this inquiry—the development of a new conceptualisation of ML4D—I first require an explicit conceptual framework that could define and outline the contribution of mobile learning to development; and then after having identified such a conceptual framework, I need to test it through new primary research for its validity and relevance. I identified Amartya Sen's Capability Approach (CA) as a potent conceptual framework to guide a conceptual station of ML4D. This decision was based on the application of the CA as a conceptual framework in related bodies of research, namely ICT4D and education in LMICs. In chapter 3, I further reviewed the synergies between the CA and the needs of a conceptualisation of ML4D in more detail to justify my choice of the CA as a conceptual lens.

I then use my qualitative case study research of a mobile learning programme in rural South Africa to apply the CA as a conceptual lens in practice. This serves to test whether the CA can be operationalised as a conceptual framework in research on ML4D. And, if so, to further develop an analytical framework for the educational use of mobile technologies based on the CA. It is important to keep in mind that my case study research does not aim to assess the impact of a ML4D programme, but rather attempts to explore how programme participants themselves experienced the use of technologies as an educational tool, and then to investigate how these experiences can be understood from the perspective of the CA. The main purpose of the case study is of instrumental nature to develop a conceptualisation of ML4D based on the CA.

In a last step, I then conduct a brief conceptual inquiry by combining my capabilities conception of ML4D with my systematic review findings of the empirical evidence on ML4D. This constitutes my rethink of ML4D and explores how a conceptualisation of ML4D based on the CA and the empirical evidence-base on ML4D's impacts changes the understanding and positioning of mobile learning's contribution to development.

Figure 4.1 Visual overview of the research design and methodologies

Research questions

1. How can the use of mobile technologies as an educational tool in LMICs be conceptualised as to contribute to development outcomes?

- 1.1 What are the effects of the educational use of mobile technologies in LMICs on learning outcomes, teaching practice, and empowerment?
- 1.2 What is the programme-level theory of change for mobile learning for development?
- 1.3 Does the Capability Approach serve an effective analytical framework to investigate ML4D?

| Secondary research: Mixed-methods systematic review | Primary research: Qualitative case study |
|---|---|
| The review contributes the following empirical data sets to answer the research questions: (A) Quantitative evidence on the effects of mobile learning in LMICs (B) Qualitative evidence on the mechanisms and themes unpacking the effects of mobile learning in LMICs (C) Configuring a theory of change for mobile learning in LMICs based on (A) + (B) | The case study contributes the following empirical data sets to answer the research questions: (A) Qualitative data on teachers' use of mobile technologies analysed using the CA (B) A synthesis of the qualitative data in (A) into a analytical framework for ML4D from a capabilities perspective |
| The review and its findings are needed to: Empirically assess claims to ML4D's impact Empirically unpack the assumed links between the use of mobiles and education and development Conclude whether empirical and conceptual claims to mobile learning's contribution to development are warranted | The case study and its findings are needed to: Test the operationalisation of the CA to investigating ML4D Develop an explicit conceptualisation of ML4D based on the CA |

Combined systematic review and primary research findings

Rethinking ML4D based on:

- a capabilities conceptualisation of mobile learning's contribution to development
- the empirical evidence-base on mobile learning's impact on education and development
4.2 Research design: systematic review

I first provide a justification and outline of the research design applied in my mixed-methods systematic review.

4.2.1 Introduction to systematic review as a research methodology

Systematic review is a research methodology to systematically and transparently identify, access, appraise, and synthesise all available research studies on a given research question or topic. Systematic review is therefore a methodology of secondary or meta research, that is a methodology to conduct empirical research based on data collected from existing research. Formally, systematic review can be defined as "a review of research literature using systematic and explicit, accountable methods" (Gough et al 2012: 2). The methodology's guiding principles are to be transparent and systematic at any step of the research process (Oliver 2014; Petticrew & Roberts 2006; Stewart 2014). This translates into a set of methodological characteristics associated with any systematic review: a clearly stated set of objectives with pre-defined eligibility criteria for inclusion of primary studies; an explicit, reproducible methodology; a systematic search that attempts to identify all studies that would meet the eligibility criteria; an assessment of the validity of the findings of the included studies; a systematic presentation and synthesis of the characteristics and findings of the included studies (Higgins et al 2011; Gough et al 2102; Stewart 2014).

Different types of systematic reviews and review methodologies have been developed. In addition to conventional systematic reviews, different types of reviews include evidence maps and systematic maps (e.g. Snilstveit 2016); rapid evidence assessments (e.g. Thomas et al 2013) and review of reviews (e.g. Becker & Oxman 2008). Sometimes these reviews are confused with literature reviews. While literature reviews use existing research as a unit of analysis too, they usually lack an explicit empirical research methodology to identify, appraise, and synthesise research knowledge. Systematic review as a research methodology has been developed with direct reference to addressing the shortcomings of literature reviews as a way to draw conclusions based on a body of research (Chalmers 2002; Petticrew & Roberts 2006). In this, systematic review methodology aims to address two key biases within the process of research synthesis: one, biases in the trustworthiness and relevance of the included primary research studies themselves; and two, biases in the process of bringing together and synthesising the included primary research studies, that is biases in the review process itself. If unaccounted for, both biases undermine any type of research synthesis as neither the underlying research used in the synthesis, nor the

empirical process of synthesising research is controlled for.

The underlying strength and contribution of systematic review as a research methodology is to provide research findings that are based on the full body of existing research rather than on single studies or on selected groups of studies. However, there are many different systematic review approaches and methodologies in order to synthesise different bodies of research and to answer different types of research questions. Historically, systematic reviews have been most widely conducted in the healthcare sector in order to assess the effectiveness of medical interventions (Higgins & Green 2011; Chalmers 2002; Gough et al 2012). Such effectiveness reviews are based on synthesising quantitative impact evaluations using statistical techniques to synthesise the reported effects, such as metaanalysis. However, systematic review is now widely applied in the social sciences too which has led to a range of iterations and adaptations of the initial methodology (Petticrew & Roberts 2006; Gough et al 2012; Stewart 2014; Langer & Stewart 2014)²⁷. Current systematic review methodologies therefore range from more quantitative reviews to more qualitative reviews and the spectrum of review methodologies features a more diverse body of reviews such meta-ethnographies (Nolbit & Hare 1998), meta-narrative reviews (Greenhalgh et al 2005), realist reviews (Pawson 2006) or critical interpretive synthesis (Dixon-Woods et al 2005).

A way to classify and group this plethora of different review methodologies is to investigate them on a spectrum from reviews that aim to aggregate primary research findings to answer a review question and reviews that aim to configure and arrange primary research findings to answer a review question (Gough & Thomas 2016; Oliver 2014; Voils et al 2008; Sandelowski et al 2011). *Aggregative reviews* are associated with reviews that include predominately quantitative data as their main aim is to aggregate or add up findings from multiple, similar research studies. *Configurative reviews* are associated with reviews that include predominately qualitative data as their main aim is to configure or arrange findings from multiple, heterogeneous research studies. Aggregative reviews commonly are thought of as a more deductive review approach as they answer narrow and specified questions using a priori defined quantitative methods to test theory based on empirical observations (Gough & Thomas 2016; Sandelowski et al 2011). Configurative reviews, then, ask more open review questions with less pre-specification of concepts and apply qualitative methods that leave room for iteration and allow for interpretation of specific cases. Configurative

²⁷ In fact, the earliest attempts to use structured and transparent methods to make sense of a body of research evidence to inform decision-making is recorded in the US education sector in the 1970s (Oakley 2000). Systematic reviews have a longer than usual assumed history in the social sciences. For examples, the UCL Social Science Research Unit was funded in 2000 already to set up an education review facility.

reviews thus tend to explore and generate theory rather than test it in line with an inductive review approach (ibid). However, aggregation and configuration in reviews should be understood as different ends of a spectrum of review approaches rather than dichotomous concepts.

4.2.2 Mixed-methods reviews

Mixed-methods or mixed-knowledge systematic reviews are a growing innovation in research synthesis (Gough & Thomas 2016; Pluye & Hong 2014). They aim to synthesise diverse data sets featuring both quantitative and qualitative research evidence. Based on the concept of mixed-methods primary research, these reviews can address complex questions, drawing on broader types of evidence, and thus a larger pool of research findings. Mixed-methods reviews are further an effective tool to "explore implicit or partially developed theories" due to their ability to arrange different types of knowledge along a gradually developing framework (Oliver et al 2012: 77). Mixed-methods systematic reviews therefore present an adaptation to both, traditional effectiveness reviews and qualitative research syntheses. Mixed-methods reviews have the ability to concurrently test, explain, and develop theories. In reference to programmatic considerations, they therefore contribute knowledge not only on 'what works' or only on 'how and why does it work', but on both sets of questions.

Conceptually, the mixed-methods approach to systematic review presents a middle ground between aggregative and configurative reviews. Mixed-methods reviews have been operationalised in different forms depending on the specific review question. Van der Kaap and peers (2008), for example, advocate to first conduct an effectiveness review to test the impact of an intervention and subsequently to apply a realist synthesis to understand in what context and for whom the intervention worked. A more fluid mixed-methods approach is taken by Thomas and colleagues (2003), who develop different review questions, review these distinctively, and bring together the findings in a combined synthesis.

Mixed-methods review are particular attractive for reviewing the effects of social interventions (Sandelowski et al 2011; O'Mara-Eves & Thomas 2016; Langer & Stewart 2014). This is based on the assumption that a large body of experimental research in most areas of social science is not available; that interventions often differ in design across contexts and that the underlying mechanisms through which they work are complex and fluid; that implementation considerations and behavioural effects are crucial; and that contexts plays a larger role in determining intervention effects. As a result, mixed-methods

reviews have seen an increase in application over the last five years, though they still present a small minority of conducted systematic reviews overall (Gough et al 2017; O'Mara-Eves & Thomas 2016).

In international development, mixed-methods reviews are increasingly popular (Langer & Stewart 2014; Snilstveit 2012; White & Waddington 2012). Most development interventions refer to social interventions and the rationale for mixed-methods reviews outlined above applies. The Campbell Collaboration International Development Co-ordinating Group²⁸ actively advocates for the production of mixed-methods reviews and different methods for their production have been conceptualised by Snilstveit (2012). Her suggestion of conducting 'Effectiveness Plus' reviews has found wide-spread adoption and the majority of international development reviews—that is, systematic reviews of development interventions—conducted in the last five years have incorporated an assessment of qualitative evidence linked to the reviewed intervention (e.g. Oya et al 2017; De Buck et al 2017; Carr-Hill et al 2016).

The particular nature of my research questions as well as the encountered evidence-base of ML4D justifies the application of a mixed-methods systematic review design. In order to investigate not only 'what works' in ML4D, but further 'why and how' mobile technologies lead (or do not lead) to education and development outcomes, a broad range of research evidence is required. Further, different types of synthesis are needed as I aim to both explicate the theory of change underlying ML4D interventions and to test which steps of this theory of change are empirically supported. This requires processes of configuration as much as aggregation of research data. As a result, on its own, neither a conventional effectiveness review, nor a qualitative research synthesis, would have allowed me to meaningfully synthesise the diverse range of research evidence required to answer my research questions.

²⁸ IDGC guidelines: <u>https://www.campbellcollaboration.org/resources-international/guidelines-for-reviewers.html</u>

4.2.3 Mixed-methods review design

I follow a mixed-methods systematic review design in this thesis. This mixed-methods design is operationalised through a two-module review approach²⁹. That is, the mixed-methods systematic review features a distinct aggregative review module and a second configurative review module. This two-module design was informed by the need to meet three research objectives in my systematic review:

- (1) Generate evidence of ML4D's impact (or lack of impact).
- (2) Generate analytical themes on mechanisms and contexts explaining this impact (or lack of impact).
- (3) Construct an evidence-informed theory of change of 'what works, how and why' in ML4D.

In order to construct an evidence-informed theory of change of ML4D I require a synthesis of two types of information: (1) evidence of ML4D's effectiveness, defined as aggregative data measuring the impact of mobile learning in LMICs and (2) analytical themes of mechanisms and contexts to explain this impact (or lack of impact). These analytical themes will allow for a configuration of the potential impacts of ML4D, unpacking the black box of how the interventions might have led to the established outcomes. I therefore design two distinct review modules to be able to generate both types of information and to synthesise them in a mixed-methods synthesis. Each module follows its own logic and design applying an aggregative and a configurative approach respectively. This two-module mixed-methods review approach is presented in Figure 4.2.

²⁹ This two-module review approach is inspired by Thomas et al (2003).

Figure 4.2 Systematic review approach

AGGREGATIVE REVIEW MODULE



CONFIGURATIVE REVIEW MODULE

The contribution of both review modules is of equal weight in generating the evidenceinformed theory of change. The relation between the aggregative and configurative module can best be described as a 'sequential explanatory' design (Pluye & Hong 2014), with the configurative module arranging the aggregative findings. Yet, the combination of both types of knowledge to generate an evidence-informed theory of change thereafter is more adequately characterised as a 'convergent' design (ibid). There is a danger that the sequential approach might affect the identification of inductive themes in the configurative module, but this risk seems justified keeping in mind that the module's main mandate is the arrangement and unpacking of the aggregative findings. As a result, after a combined systematic search and application of inclusion criteria, the review commenced with the conduction of the aggregative module. Both review modules used distinct critical appraisal tools and methods to synthesise research findings.

The aggregative module mirrors a traditional effectiveness review. It only includes rigorous impact evaluations of mobile learning programmes in LMICs, judges their quality thoroughly through a risk of bias assessment, and conducts a statistical meta-analysis of the included studies in order to yield a numerical value of ML4D's effectiveness. This pooled effect size is interpreted as evidence of ML4D's impact and since assumed to answer the thesis's first sub-research question. Meta-analysis as a method to establish 'what works' in ML4D was chosen because the method presents the most rigorous tool in synthesising quantitative impact measures (Borenstein et al 2009).

The configurative module resembles a systematic review of qualitative research. It is not limited to a particular study design, places a greater focus on relevance and context in its critical appraisal of research, and applies a thematic synthesis in order to establish analytical themes on contexts and mechanisms at play when teaching and learning with mobiles in LMICs. It thereby aims to arrange the aggregative knowledge with the help of the analytical themes on context and mechanism configurations—a process required to design ML4D's theory of change. Thematic synthesis (Thomas & Harden 2008) was identified as the most rigorous and transparent approach to extract codes and descriptive themes from the included studies, and further, to use these themes to identify mechanisms and contexts in ML4D.

Lastly, the findings of both review modules will be brought together in a mixed-methods synthesis to construct the evidence-informed theory of change of ML4D in graphical and narrative format. In this, the findings generated in the configurative review module allow me to unpack the links between the provision of mobiles, educational process and outcomes,

and subsequent changes in socio-economic development. Having outlined these links and relations, I then plot the results of the aggregative module against this theory of change to highlight for which steps and processes in the educational use of mobiles in LMIC there is evidence of effects.

4.2.4 Summary of systematic review steps

This section presents a summary of my systematic review protocol. The full protocol can be accessed as an online appendix³⁰ and outlined all methodological steps pertaining to the conduct of my mixed-methods review a priori.

Acknowledging the importance of an explicit systematic review protocol to ensure transparency and rigour in the review process, I would have preferred to register my review protocol with a systemic review body. Alas, the only open-access register of systematic review protocols, Prospero, is limited to health systematic reviews. The umbrella body for social science reviews, the Campbell Collaboration, does not cater for the conduct of mixedmethods reviews with an independent qualitative research synthesis. Keeping this caveat in mind, I present below a summary of the most important methodological steps and decisions in the design of my systemic review.

Inclusion/exclusion criteria

I defined a set of pre-defined and explicit inclusion criteria to determine what type of research studies were included in my systematic review. Studies not meeting one of the below criteria were excluded from the review:

Region: All studies conducted in Low- and Middle-Income Countries (LMICs) were eligible for inclusion in the review³¹.

Population: No criteria related to the study population were applied. That is, study populations could relate to individuals (e.g. teachers or learners), any other form of human organisation (e.g. communities), and administrative groupings (e.g. schools).

Intervention: As discussed, the term 'mobile learning for development' is not clearly defined in the literature. A broad operating definition of ML4D interventions was since applied and I included all studies referring to the use of mobile technologies to support education in

³⁰ Online appendix 1: <u>https://africacentreforevidence.org/project-outputs-3/</u>

³¹ This follows the World Bank classification of economies as of 14 February 2014. Categorisations were applied based on the date of data collection in the primary studies: <u>https://datahelpdesk.worldbank.org/knowledgebase/articles/906519-world-bank-country-and-lending-groups</u>

LMICs. Mobile technologies were understood as any ICT device capable of mobile usage³². The reference of explicit mobile learning events was not part of the inclusion criteria, but studies had to collect empirical data in reference to the use of mobiles in an educational context. Studies collecting data not linked to the active use of mobiles, for example general surveys of perceptions on mobile technologies not linked to a mobile learning intervention, were excluded.

Outcomes: The review focused on four primary outcomes of interest:

- (1) *Learning outcomes*: were broadly defined as any change in the process, perception, and outcome of learning. This ranged, for example, from attendance to motivation and performance of learners.
- (2) *Teaching practice*: was broadly defined as any change in the process, perception, and outcome of teaching. This ranged, for example, from changes in lesson designs to teachers' professional behaviour and indicators of teaching quality.
- (3) *Education Monitoring and Information Systems (EMIS)*: referred to changes in the way information is collected, administered, and analysed in education systems.
- (4) Empowerment: might either be an outcome or a process. As an outcome it was broadly defined to embrace various themes such as critical thinking, agency, opportunity, confidence, etc. Empowerment outcomes could be observed at any level (e.g. individual, school, community).

Study design: Two different eligibility criteria for quantitative and qualitative study designs were applied. Quantitative studies were required to use at least a quasi-experimental designs, reporting pre- and post-data of two experimental groups. Qualitative studies were not limited to a particular study design given the absence of accepted standardisations of qualitative research methodologies. In lieu of an inclusion criterion linked to an accepted terminology of qualitative research designs, eligible studies needed to at least report on the following: evidence of an explicit applied research design, and evidence of collected primary data, sampling strategy, and methods of data analysis.

³² This excludes all studies assessing the impact of the OLPC programme. These were excluded for two reasons: (i) laptop devices at the time of the conception of the review in 2014 were still vastly superior in computing power and affordances to other mobile technologies; (ii) the OLPC programme has been evaluated rigorously elsewhere already (e.g. Beuerman et al 2015; de Melo 2014) and including these evaluations in my review would have inflated the sample of includes and led the review to become an evaluation of OLPC + other ML4D interventions.

Systematic search strategy

In order to identify all relevant literature in this review a transparent, systematic, and exhaustive search for research studies was designed. The search strategy was deliberately formulated to be over-inclusive and as exhaustive as possible to cater for the unstructured nature of the ML4D evidence-base and to reduce single reviewer bias. These two key features of being over-inclusive and exhaustive inform both the design of the search terms as well as the choice of search sources. The full search strategy and report is provided in Appendix 4.1. and was reviewed by an information scientist at the UCL Institute of Education.

Search terms: I combined three search concepts and related terms with each other using Boolean language (Box 4.1). This refers to the intervention concept of mobile learning, the outcome concept of education and development, and the population concept of LMICs. In order to be overinclusive the intervention concept could be

Box 4.1: Example of search strings

- (1) Intervention terms: (e.g. mobile learning OR mobile educational OR mobile technology) AND
- (2) Outcomes terms (e.g. education OR learning OR development) OR
 (3) Population terms (e.g. South Africa OR developing countr*)

combined either with the outcome or the population term. With a similar objective in mind, no concept related to study design was applied. For each concept different terms were collected and wild cards were applied. A snapshot of the search string is provided below with the master string being reported in Appendix 4.1. The master string was adapted for use in each scientific database and its core concepts were used to guide the Grey literature search.

Search sources: I consulted a total of 108 search sources. This included both academic and Grey literature sources. The latter were in particular important given that a large body of the ML4D research is conducted by IGOs, NGOs, and vendor organisations. The consulted types of search sources are listed below with a full record being provided in Appendix 4.1.

Academic sources

- Scientific databases, e.g. EbscoHost, Web of Science;
- Hand-search of key journals, e.g. World Development, Information Technology for Development;
- Conference proceedings, e.g. elearning Africa; UNESCO Mobile Learning Week.

Grey literature sources

- Organisational websites, e.g. DFID Research for Development, World Bank;
- ML4D project websites, e.g. Worldreader, English in Action;
- Google scholar;
- Forward and backward citation searches of primary and secondary research studies;
- Contacting experts;
- Twitter searches.

I encountered a challenge in conducting the systematic search early during my first year of PhD registration in 2014 as this date meant that my searches would be out of date by submission of my thesis in 2017. To mitigate this issue, I experimented with using automatic search alters with the scientific databases used in the search. That is, I saved my search string in these databases and received an email update every time an item was added to the database meeting my string's parameters. I then screened the incoming citations on an ongoing basis. In practice, however, I realised quickly that I was not altered to all relevant studies using this technique. I therefore conducted a full search update two years post the initial search in April 2016.

Screening and application of inclusion criteria

All search hits were imported into EPPI-Reviewer software (v 4.6.4.1) and screened initially on title and abstract against the developed inclusion criteria. No information on reasons for exclusion were collected at this stage. Full-texts of studies included at title and abstract were then sought and screened. I recorded the reason for exclusion for each study at this stage. No inter-reviewer reliability assessment was conducted as I conducted this review as a single reviewer.

Coding and data extraction

I developed a detailed coding tool to extract relevant data from the included studies (Online appendix 2³³). EPPI-reviewer software was used to generate coding sets and to facilitate data management. The coding strategy can best be described as 'mixed coding' (Oliver & Sutcliffe 2012). Codes for the aggregative review module were pre-defined, whereas codes for the configurative module were open codes. However, for the configurative review module I also predefined a list of descriptive themes, which the literature suggested to be of relevance to the construction of a theory of change for ML4D. It was required to define these

³³ Online appendix 2: <u>https://africacentreforevidence.org/project-outputs-3/</u>

deductive codes in order to assess an absence of evidence on these themes.

Critical Appraisal

Critical appraisal refers to the process of assessing the trustworthiness and relevance of the studies included in a systematic review. It is a required review step in order to ensure that the conducted synthesis is based on reliable research results. Due to the diversity of configurative and aggregative data a critical appraisal tool that can cater for both qualitative and quantitative studies was required. I developed a critical appraisal tool drawing on Pluye and colleagues' (2011) Mixed-methods assessment tool (MMAT) as well as Sterne and colleagues' (2013) risk of bias tool for non-randomised studies. My critical appraisal tool uses the MMAT's basic structure but extends its criteria to assess the quality of the qualitative and qualitative studies. To ensure the comparability of the appraisal of both types of studies, six domains of appraisal judgments were developed for each study type. The full tool is provided in the Appendix 4.2.

Critical appraisal of quantitative studies: For quantitative studies, the developed critical appraisal tool assessed the rigour of the impact evaluation design to establish the reliability of the reported aggregative effect. The tool needed to be able to assess both randomised and non-randomised impact evaluations. In this remit, a Cochrane risk of bias tool for non-randomised studies was adapted (Sterne et al 2013). The tool assessed six domains of bias: (1) selection bias; (2) bias due to baseline confounding; (i) bias due to ineffective randomisation³⁴; (3) bias due to departures from intended interventions; (4) bias due to missing data; (5) outcome reporting bias; and (6) bias in selection of reported results. Studies were judged on a scale from critical to low risk of bias, and studies of critical risk of bias were excluded from the synthesis.

Critical appraisal of qualitative studies: For qualitative studies, the developed critical appraisal tool was based on the underlying principles of rigour (in the research conduct) and relevance (contribution to the research question). These were broken down into six domains of: (1) research is defensible in design; (2) research features an appropriate sample; (3) research is rigorous in conduct; (4) research findings are credible in claim; (5) research attends to contexts; (6) research is reflexive (CASP 2006; Dixon-Woods et al 2005). Diverting from the tool for quantitative studies, no scaled appraisal scale was applied to rate the qualitative studies. A study was either included as making a reliable contribution to the research question or excluded as not rigorous or not relevant. No distinction was then made

³⁴ This domain of bias was only applicable to RCTs, which by design can account for domain (1) and (2).

between the quality of the respective contributions.

Methods of synthesis

I applied two different methods of synthesis for the aggregative and the configurative review module. Statistical meta-analysis was used in the aggregative review module and thematic synthesis was used in the configurative module.

Meta-analysis: In the aggregative review module, I conducted a statistical meta-analysis to investigate the impact of mobile technologies on education in LMICs. Meta-analysis is the most rigorous method to synthesise quantitative research studies (Lipsey & Wilson 2001; Borenstein et al 2009). As a statistical approach, it aggregates the numerical effect sizes of research results to report a pooled overall numerical value. This numerical value—the pooled effect size—expresses the overall finding derived from the combined primary research results. The pooled effect size reflects the direction and magnitude of the observed primary effect sizes, which are allocated different weight in the analysis depending on sample sizes and variance.

I report calculated effect sizes in tabular format as well as using forest plots. Where sufficient contextual homogeneity prevailed, effect sizes were averaged across studies by using an inverse variance weighting of the individual effect size. This weighting results in the individual effect sizes of studies with larger study samples being given more weight in the combined, pooled effect size. The meta-analysis was carried out using random effects statistical models.

Statistical synthesis of educational outcomes needs to take into consideration the diversity of study designs and outcome measures. I envisaged that learning outcomes would be the main outcome reported, and likely be reported in form of test scores or grades. This presents a continuous outcome, and outcomes measures and scales were expected to differ between studies. As a result, the meta-analysis calculated standardised mean differences (SMD) in order to yield comparable effect sizes. Cohen's *d* as well as Hedge's *g* were calculated for each included study, with *g* being the effect size used in the meta-analysis due to its ability to adjust for small sample bias prevailing in *d* (Deeks et al 2001). SMDs express the measure of effect in a change of standard deviations making the effect size difficult to interpret. For ease of interpretation, I report SMDs in this thesis alongside the corresponding percentage change in the intervention group over the control group. All formulae for effect size calculations are reported in Appendix 4.3.

Given the assumed heterogeneity in the true effects of studies across, for example,

geographical reach, educational systems, and socio-economic contexts, a random effect model was applied. In addition, I applied moderator and sensitivity analyses to test the robustness of the meta-analysis and to explore the expected large heterogeneity. Acknowledging the limitations of a quantification of heterogeneity and the different strengths of statistical approaches, I conducted the following test for heterogeneity: calculation of the Q statistic as a statistical test of heterogeneity (Hedges & Olkin 1985); calculation of the i² and tau² statistic to provide estimates of the magnitude of the variability across study findings caused by heterogeneity (Higgins & Thompson 2002; Higgins et al 2003). More indepth discussions around the importance of single effect sizes, choice of outcome measures, heterogeneity testing, as well as sub-group analysis can be accessed in the review protocol (Online appendix 1) and appendix 5.4.

Thematic synthesis: In the configurative review module, I conducted a thematic synthesis to configure the impact of mobile technologies on education and development in LMICs. The findings of qualitative research studies were synthesised in form of analytical themes on intervention mechanisms and contexts to unpack how and why learning and teaching with mobiles might (or might not) have an impact. I followed Thomas and Harden's (2008) approach to thematic synthesis. They suggest three key stages in thematic synthesis based on thematic analysis in primary research: coding text; developing descriptive themes; and generating analytical themes.

In **stage one**, the reported research findings of the included qualitative studies were subject to line-by-line coding. Findings would ideally have referred to the primary data of each included study (e.g. interview excerpts), but due to limited reporting of this information, authors' analyses and conclusions represented study findings and the unit of analysis in my thematic synthesis. The line-by-line coding feature in EPPI-reviewer was applied to guide and manage the coding of the reported analyses and conclusions. Guidelines to thematic analysis, as applied in qualitative primary research, informed this process of creating thematic codes from the included studies.

In **stage two**, the identified codes were then grouped into descriptive themes. In addition to the inductive creation of descriptive themes from studies' codes, a number of pre-defined (deductive) descriptive themes were introduced in the synthesis and controlled for during line-by-line coding (see review protocol, online appendix 1). These themes relate to common claims and statements in the literature on ML4D, for example that females are discriminated against in access to mobiles. I needed to

86

introduce these deductive descriptive themes in order to identify a possible absence of evidence on these themes, which would have not emerged in a purely inductive thematic synthesis.

In stage three of the thematic synthesis, I translated the descriptive themes into analytical themes. This translation is the key process in generating new data in the thematic synthesis. In the context of the thesis' research questions, analytical themes were formulated exclusively around mechanisms and contexts that can configure ML4D's impact or lack of impact. I therefore used a mechanism-context framework to quide my translation of descriptive themes into analytical themes. In this, I adapted Pawson's (2006) definitions of mechanisms and contexts in realist synthesis. I define mechanisms as 'changes caused by the use of mobile technologies that influence its impact'. For example, using mobile phones might change learner's ability to access information, which then might lead to better learning outcomes. Access to information is thus not part of the actual intervention, but a change induced by it that supports its effect. Subsequently, I define contexts as 'variables exogenous to the ML4D intervention that influence its impact'. For example, using mobile phones might only change learners' abilities to access information if a reliable internet connection exists. The contextual factor of a reliable network connection, which is not part of the actual intervention, thus also determines its effect.

I report the findings of the thematic synthesis in narrative tables of all identified analytical themes, divided into mechanisms and contexts, and illustrate the underling descriptive themes.

Mixed-methods synthesis: In a last step of my mixed-methods review, the findings of the aggregative meta-analysis and the configurative thematic synthesis are then brought together in a mixed-methods synthesis to construct the evidence-informed theory of change of ML4D. This construction involved three stages. First, it entailed the plotting of the intervention-to-outcomes pathways in the included ML4D interventions. In this, I used standard templates for developing theories of change in international development (Vogel 2012; Valters 2014). These suggest to break down intervention-to-outcome pathways into: inputs, immediate changes (or outputs), outcomes (on a spectrum from intermediate to final, or short-term to long-term), and impact. Second, I plotted the results of the meta-analysis against the outcomes represented on the theory of change. This provided a visual breakdown for what steps in the theory of change of ML4D there is reliable empirical

evidence of effects. Third, I also plotted the mechanism and context themes against the outlined intervention-to-outcome pathways. This aimed to indicate conditions that are associated with the observed effects. Taken together, the theory of change thus presents a visual representation of (i) the assumed links between the provision of mobile devices, their impact on educational outcomes, and subsequently their link to development outcomes; (ii) the evidence of effects for each step in this process of an assumed link between technologies, education, and development; and (iii) the underling mechanisms and contexts that might explain these links and the observed effects. I then expanded in narrative on the theory of change highlighting what implications the combined systematic review findings have on the conception and positioning of ML4D.

4.2.5 Ethics in conducting the systematic review research

Ethics approval for the conduct of the systematic review was obtained from the UCL Institute of Education. Ethical concerns in systematic reviews relate to the nature of the underlying data used, for example, patient datasets in healthcare research. However, I did not collect or request any additional human data aside from the data published in included primary research studies. I therefore only used data in my systematic review that already was in the public domain.

4.2.6 Limitations of the systematic review

Despite following accepted guidelines and methodological protocols for gold standard systematic reviews in social sciences (Gough et al 2017; Campbell Collaboration 2015), the nature of conducting a systematic review as part of a PhD thesis necessarily limits some technical aspects of the review process. First, as a single reviewer technical quality assurance processes such as double-screening and double-coding could not be conducted. Second, the registration of the review with an umbrella review organisation such as the Campbell Collaboration was not possible as I could have no longer controlled the timelines of conducting and concluding the review. Third, as the timelines of the review and its publication are set by the period of PhD registration, the review is unlikely to be as up-to-date as usually expected for publication, in particular if conducted at the beginning of the PhD. This limitation applies to my systematic review in particular and despite experimenting with different methods to keep the search hits up-to-date, my review only includes primary research until June 2016 when I ran the last full search update.

In addition, I conducted a mixed-methods systematic review, a review approach which

presents a minority among the systematic reviews in the social sciences (Snilstveit 2012; Langer & Stewart 2014). While this meant that I address common limitations of effectiveness systematic reviews, such as only including a narrow range of research and only investigating the question of intervention effects, on the downside, there is no agreement on methodological approaches to the precise conduction of mixed-methods reviews. I therefore developed a range of review steps more iteratively and while formulating an a priori review protocol, did not submit this protocol for publication. For the same reason, I did not report my systematic review following PRISMA reporting guidelines (though all PRISMA items are covered between chapters 5 and 6, and appendices 4.1–4.3 and 5.1–5.3) and did not conduct a strength of the evidence and recommendation assessment, such as the GRADE or CERQUAL frameworks.

4.3 Research design: case study research

I next provide a justification and outline of the research design applied in my qualitative case study.

4.3.1 Introduction to case study research as a research methodology

The second part of my empirical research presents a qualitative case study, following explicit methodological guidelines for case study research developed by Merriam (1988) and Yin (2003; 2014). Case study as a research methodology presents an in-depth investigation of a particular social phenomenon using multiple sources of data to construct an understanding of the essence of the phenomenon under investigation. Formally defined, the case study research method can be understood as "an empirical inquiry that investigates a contemporary phenomenon within its real-life context; when the boundaries between phenomenon and context are not clearly evident; and in which multiple sources of evidence are used" (Yin 2014: 18). One of the methodology's key strength is therefore that it is able to include and bring together multiple perspectives on a single phenomenon and its context allowing for a conversation and interaction between perspectives to emerge. Analytical insights are then gained from this conversation and interaction to configure a rich construct of the case under investigation.

Case study as a research methodology is most widely applied in social sciences and often associated with a qualitative research paradigm, though this is not necessarily a feature of the methodology but rather a result of the pattern of its adoption (Yin 2003; 2014; Stake 1995; Baxter 2008). Likewise, case studies are often associated with a social constructivist research design due the method's ability to facilitate the "social construction of meaning insitu" (Stark & Torrance 2005: 33); again, however, this might be a reflection of how the methodology has been applied and there are also more deductive applications of the methodology, for example to evaluate the effects of public policies (e.g. Patton 2002; Vogel 2016). Thus, case study research can include both quantitative and qualitative data, follow different analytical strategies, and differ in the researcher's positioning vis-à-vis the case under investigation.

Different types of case study research exist. The most common types refer to: intrinsic, instrumental, and collective case studies (Yin 2014) and exploratory, explanatory, and descriptive case studies (Stake 1995). Each of these types differ according to the purpose of the conducted research, which subsequently influences the case study research design.

All different types of case studies can further vary between a focus on a single case or multiples cases. However, there are a range of common characteristics between each type of case study design. The most frequently cited characteristics refer to: that the boundaries of the case under investigation are defined; that the unit of analysis in the case (e.g. individuals, policies, social behaviours) is clearly stated; that there is an explicit attempt to reflect the wholeness, unity, and integrity of the case; and that multiple sources of data and/or analysis are used (Yin 2014; Punch 2014).

4.3.2 Qualitative case study design

My case study research focuses on the case of the Information Communication Technology for Rural Education Development (ICT4RED) programme, a ML4D intervention in rural South Africa. The single case is bounded by the schools taking part in the programme and my unit of analysis are the teachers involved in the ICT4RED. I apply two main research instruments to collect qualitative data to construct my investigative case: semi-structured qualitative interviews and structured classroom observations. These are supplemented by focus group discussions and observations of policy roundtables and professional development courses.

I adopt a qualitative case study design subscribing to an instrumental approach to case study research for the purpose of my research. An instrumental case study approach seems most relevant in the context of my research question. My case study aims to explore teachers' own perceptions of how the use of mobile technologies in an educational context might support their pursuit of valued beings and doings. I am thus interested if teachers' use of mobiles can be investigated from a capabilities perspective. As a result, the case itself in my case study becomes secondary and I am more interested in using the case to provide insights into the feasibility of a CA investigation into teachers' use of mobiles. The case of the ICT4RED programme thus could be substituted for a different mobile learning programme and is used in an instrumental sense in my case study research.

I further position my case study in a social-constructivist research paradigm. I do not assume there to be a singular conception of the educational use of mobile technologies from a capabilities perspective. Teachers will vary in their use of the mobile devices to support their own exploration of valued being and doings. I am interested in how individual teachers themselves construct these usages in relation to their own definitions of the 'good life'. The case study therefore attempts to unpack these different perspectives and configures and contrasts them without losing their constructed meaning by teachers themselves. This nature of inquiry requires me to collect rich qualitative data on how teachers perceive and apply the mobile technologies as part of the ICT4RED programme. Lastly, I considered the conduct of an ethnography rather than a case study in my research. Ethnography as a research methodology seemed to be equally able to generate sufficient data and analysis to answer my primary research question; and, arguably would have yielded a richer data set. Resources constraints were the main factor in guiding my decision to conduct a qualitative case study.

In summary, I conducted an instrumental case study of the ICT4RED programme employing a range of research instruments to collect qualitative data on teachers' own perception on using mobile technologies in an educational setting. The ICT4RED programme was identified as an *instrumental* case to study a mobile learning programme from a capabilities perspective (more information on the programme is provided below). The aim of the study was to capture a bottom-up view on what teachers valued about technology usage and to then explore to what extent this view can be conceptualised from a capabilities perspective. In this, I chose a case study research design due to its ability to develop an in-depth understanding of the teachers' use of technologies which is situated within their own social and professional contexts and therefore allows me to explore multiple constructions of how technologies might support teachers' pursuit of valued beings and doings.

4.3.3 Overview of the primary research process

The case study research took place between July 2014 and March 2015, with the main period of data collection extending from January to March 2015. Research activities in 2014 were limited to piloting of the research instruments, introducing myself to all programme stakeholders, understanding the programme context, and setting up the logistics for the main research period in 2015. In total, four research visits were conducted in 2014, each lasting between three to four days.

In 2015, I resided in the research area in order to conduct the main data collection. In this research period, I accompanied teachers from five schools involved in the ICT4RED project during their daily teaching. At each school I first conducted an interview with the principal of the school on the first day of my visit. This was then followed by interviews with four to five teachers of the school over the next couple of days. I attempted to pair each teacher interview with an observation of a taught lesson of this teacher with the interview taking place after the observed lesson. In practice, this proofed challenging and the majority of interviews were conducted before observing the teachers' lesson. This resulted from the

teachers' strong preference to conduct interviews in the mornings before they commenced their teaching duties. In total, I conducted 25 individual in-depth interviews with teachers, four principal interviews, and observed 18 lessons facilitated by the interviewed teachers.

In addition to the main data collection of individual in-depth interviews and classroom observations, I further conducted two focus groups including six and seven participants respectively. These were held as part of the teachers' monthly professional development classes facilitated by the ICT4RED programme. I attended all ICT4RED professional development courses facilitated during January to March 2015 as an observer, which comprised four days of training of the trainers' classes and five days of teacher training classes. Lastly, I also attended two ICT4RED graduations (phase 1 in 2014 and phase 2 in 2015), two policy roundtables (both in Johannesburg), and one policy site visit.

Following the data collection period, I cleaned and analysed my research data throughout 2015 with a first presentation of preliminary findings in September 2015 at the 17th Human Development and Capabilities Association annual conference in Washington. Upon completion of the first draft of the thesis in 2017, I returned to schools involved in my research to discuss my findings.

The case: the ICT4RED programme³⁵

The investigated case—the Information Communication Technology for Rural Education Development initiative (Herselman & Botha 2015)—is a mobile learning programme conducted in the Cofimvaba school district, an administrative area within the South African Eastern Cape province (Figure 4.3). The Eastern Cape is the most impoverished of South Africa's nine provinces with 46.7 per cent of the population falling below the national poverty line (StatsSA 2015). The province also has the lowest educational attainment indicated by a provincial matriculation rate of only 65.4 per cent (Department of Basic Education 2015). The Cofimvaba district is categorised as a rural area and more than half of the schools in the district report not having reliable access to water and electricity (CSIR 2012).

³⁵ More detailed programmatic information on the ICT4RED programme can be accessed here: <u>https://ict4red.co.za/</u> Additional academic work on the ICT4RED programme has been published including teachers' perception of the provided tablets (Phiri et al 2010; Nkula & Krauss 2014); the pedagogical approach to mobile learning (Botha et al 2014; Botha & Herselman (2015b); programme implementation (Ford et al 2014; Botha & Herselman 2015a); among others.



Figure 4.3 Geographical location of the ICT4RED

The ICT4RED programme is a large-scale mobile learning intervention with a teacher- and learner-centred programme component. The programme started as pilot in 2012 and over the course of three phases (2013; 2014; 2015) involved by 2015 a total of 26 schools within the district, ranging from primary to senior secondary schools. To be eligible to be part of the programme, schools have to be classified as quintile 5 schools³⁶. ICT4RED is entirely funded by the South African government through a collaboration between the Department of Basic Education (DBE) and the Department of Science and Technology (DST).

In terms of teachers, the programme aims to improve rural educators' teaching methods through the provision of personal tablet devices³⁷ and integrated professional development. Following the provision of a tablet device, teachers are trained over a 14-week professional development course on different learner-centred teaching strategies (e.g. role play, jigsaw, gaming) and trained on how the tablets can be used to support these strategies. Teachers successfully participating in the training earn further IT equipment for their schools, stepwise attaining projectors and whiteboards, WIFI connectivity, and tablet devices for their learners. From each involved school, a minimum of five teachers are required to attend and complete the professional development course and the provision of the IT equipment is further dependent on a number of management structures being in place, including: a school

³⁶ DBE uses a quintile system to group schools in relation to the social deprivation of the communities in which they are located. Schools in more deprived areas are allocated a larger percentage of the public budget.
³⁷ Different tablet devices were used throughout the programme in different phases. All tablets were 10-inch

Android-based devices.

ICT policy; a school ICT committee; explicit consent of the school governing body to the use of technologies; and providing a reserved space for the storage of the tablets, the charging device, and the WIFI server³⁸.

Figure 4.4 Examples of ICTs provided to schools and teachers



The professional development course is the integral programme component of ICT4RED and is primarily targeted at supporting learner-centred teaching strategies rather than technology usage. The full course curriculum is available online³⁹. The course was held every second Saturday during the school terms and therefore extended roughly over nine months. Technology ownership by teachers was phased in through an 'earn as you learn' approach (Herselman & Botha 2015). This entailed that teachers initially received the tablets on loan and only gained ownership through the continued participation in the professional development course. Learners could not obtain ownership of the tablets. However, a 1:1 model of tablet provision for learners was targeted so that each learner would be able to work of their own device in school.

I identified the ICT4RED programme as a rich case to contribute primary data to the thesis for a number of reasons. First, the programme is a rare example of an ML4D intervention that is not technology-centred, but rather starts from a position of what value technology can add in the merit of improving teaching practice in deprived contexts. This directly links to the thesis's interest in rethinking ML4D. Second, ICT4RED targets teachers as the main users of the mobile technology. Teachers receive the tablets before the learners, and only after teachers are comfortable in using the technology, and have acquired detailed skills on how to integrate it in their teaching, is the programme expanded into the next phase. The focus of the ICT4RED project since is placed firmly on the improvement of the quality of rural education through equipping teachers to apply more learner-centred teaching (Herselman & Botha 2015). This mechanism is of particular interest to thesis as it diverts from the learner-centred focus of most ML4D programmes.

 ³⁸ No pictures depict research participants. All pictures were taken with consent of individuals depicted.
 ³⁹ ICT4RED teacher professional development course curriculum: <u>https://ict4red.co.za/courses/12-component-</u>

<u>model/</u>

Third, and linked to the focus on teachers, the ICT4RED entails an explicit attempt to build teachers as professionals. The programme actively aims to build teachers' confidence with and control over the tablet devices to portray rural teachers as 'tech champions'. This comes in the context of low perceptions of the teaching profession within South Africa. This focus on personal and professional development is of interest to the thesis as it overlaps with some of the theoretical underpinnings of the CA. Fourth, ICT4RED is one of the few ML4D interventions in Africa that have been applied at scale and are driven by the public sector. Lessons-learned from this programme might therefore be of particular relevance to other education departments. It also allows for the investigation of a ML4D intervention that is firmly positioned within, and not adjunct to, the existing education system.

Figure 4.5 ICT4RED professional teacher development course



Research sample

The five schools were chosen through a stratified-random sampling procedure. This was guided by pre-defined criteria in order to ensure that the sample schools provided an adequate representation of the total 26 schools involved in the project. Of the five sampled schools, four were Junior Secondary Schools (Grade 1-7) and one school was a Senior Secondary (Grade 7-12). All schools were ranked according to South Africa's quintile system to qualify for the government's school feeding scheme, indicating the high levels of deprivation within the schools' local communities. The average number of pupils per school was 382 (range: ~180–552). The teacher to pupil ratio ranged roughly between 30 and 50 learners per teacher.

As indicated above, all schools fell into the lowest quintile of the South African schooling system and are therefore located in the most deprived locations of the country. Schools further varied in terms of basic infrastructure and support provided by DBE. Two schools were located more than one hour away from the next tarred road and had the lowest level of infrastructure support with not all classroom having built structures and only sporadic access to water and sanitation. None of the five schools had received any ICT equipment prior to

the introduction of ICT4RED⁴⁰. Consequently, none of the schools involved had teaching staff trained in ICT usage or policies and governance structures in place regarding the integration of ICTs in teaching and learning. In general, despite some variances between schools, it is accurate to describe all five schools as to belong to the most deprived educational institutions in the country; which was an explicit selection criterion for schools in the ICT4RED programme design (Herselman & Botha 2015).

Each school had successfully completed phase two of the programme and the interviewed teachers since owned their tablet devices having used them for over a year in their teaching. Within each school, I discussed with the principal and the chairperson of the ICT committee which teachers could be involved in the research as well as the logistics of my visits. The final sample consisted of 25 teachers of which all but two were females. All teachers were part of a teachers' union and all but one teacher were South African nationals. Only ten teachers were isiXhosa speaking residents of the local communities with most teachers being seconded from the Western Cape or Gauteng to their posts in the Eastern Cape⁴¹.

Of the 25 teachers involved, only three were below the age of 30 years with the majority of teachers indicating to be between the age of 30–40 years (n=8) and 50–60 years (n=11). Three teachers were of age 60 and above. Teachers had on average 18 years of teaching experience; though only ten teachers had taught for more than five years at their current school. Nineteen teachers indicated that their formal teaching training lasted for two years or less indicating that the majority of teachers had obtained teaching diploma from dedicated teaching colleges rather than university-based formal teaching degrees, which require at least four years of studies. The most frequent subject expertise cited by teachers referred to english, isiXhosa, mathematics, natural sciences and life orientation. All teachers indicated to teach subjects that they were not trained for. Lastly, in terms of technologies use, none of the teachers owned a tablet device prior to joining the ICT4RED, but all teachers owned smart phones.

 ⁴⁰ Teachers, however, had access to personal mobile technologies including laptop PCs that they sometimes reported to have used for professional administrative purposes.
 ⁴¹ Following completion of teacher training degrees, DBE assigns teachers to schools country-wide. Gauteng and

⁴¹ Following completion of teacher training degrees, DBE assigns teachers to schools country-wide. Gauteng and Western Cape are South Africa's most developed provinces and host the largest number to teachers graduating annually. A large number of teachers based in Gauteng and the Western Cape is thus allocated to schools in other provinces, in particular schools that struggle to fill their teaching positions otherwise due to their rural location.

Research instruments

Individual in-depth interviews: I developed a semi-structured interview guide that was piloted during the 2014 research visits (Appendix 4.4). The guide focused on conversational topics to explore teachers' usage of the tablets in their professional and social lives, and what they valued about this usage. The interviews lasted 30 to 75 minutes and were conducted at the schools' facilities. I aimed to conduct at most two interviews per day, which I adhered to with the exception of one occasion on which three interviews were conducted.

Classroom observation schedule: I developed a semi-structured observation schedule that was piloted during the 2014 research visits (Appendix 4.5). The classroom observations lasted for the duration of the class facilitated by the teacher and concerned the teacher's pedagogical approach and technology use.

Focus group discussions: The focus group discussions took place during the training of trainers' sessions of the ICT4RED professional development course, which was held outside regular school hours on the weekends. The groups lasted for 30 minutes and I did not facilitate or participate in any discussions apart from providing the topic of the conversation: what do you value about the use of tablets; what don't you value?

In-depth interviews and focus groups were audio-recorded and transcribed using Dragon Dictate software for Mac.

Coding and analysis of data

The Capability Approach was applied as a conceptual lens to inform the coding and analysis of the collected data. However, I did not apply a specific framework linked to the CA such as the Choice Framework (Kleine 2013). Rather, I used the CA's analytical devices of functionings and capabilities as a lens through which to guide an inductive coding and thematic analysis of the data⁴². In this process of a CA-framed thematic analysis, I initially coded interview data for teachers' valued functionings using *In Vivo* codes (Saldaña 2013). These initial *In Vivo* codes (e.g. 'the tablet is always going with me') were then thematically grouped into descriptive themes with each theme being represented by a single *In Vivo* code (e.g. 'I like the tablet because it is mobile'). Following this iteration, the data was coded again using only the higher-level thematic *In Vivo* codes. This was required in order to assess overlap between the themes and to merge repetitive and similar codes.

⁴² This extends to associated concepts of opportunities, freedoms, choice, values, agency, and development.

After this second stage of coding, the remaining descriptive themes were configured into higher-level analytical themes to develop the final list of valued functionings expressed by teachers (e.g. 'to be mobile'). In a last step, these functionings were then organised into corresponding capabilities. Each capability was thus derived inductively from the teachers' valued functionings. This inductively produced capabilities then provided the foundation to develop an overall framework for teachers' use of mobile technologies as an educational tool. EPPI-reviewer software was used to facilitate the coding process and the thematic analysis. The data from the teacher observations was used to cross-reference the reported valued functionings expressed by the teachers.

4.3.4 Ethics in conducting the case study research

My case study research did take into account a range of ethical issues, which were fully outlined in a detailed ethics application (Online Appendix 3)⁴³. Below, the key considerations are discussed.

My case study research required ethical approval at multiple layers. First, I sought and obtained approval from the UCL Institute of Education. However, as this research was conducted in South African schools, ethical approval was also required by the South African Department of Education, which controls researchers' access to engage in research on the formal education system. In addition, I also required ethical approval from the institution implementing the ICT4RED programme, the Council for Scientific and Industrial Research (CSIR). Ethical approval from each body was obtained. In addition, I informed the local teachers' unions about my research as well as education officials at the district and ward level. This level of scrutiny was required as education in South Africa, and in the Eastern Cape in particular, is a highly charged political arena. External researchers are often perceived as a threat by school administrators and teacher unions due to the often negative results of educational research studies. This is particular acute in the Eastern Cape in which the NGO Equal Education has successfully taken the provincial education department to court on a couple of occasions using research and M&E data to support their case. Without careful abiding to all formal structures at all levels of governance, the research could have presented a threat to the professional careers of the teachers and principals involved.

Participation in the research was voluntary and I sought free and informed consent from each research participant. Two teachers did not want to be interviewed when being chosen by the school principals, who then nominated a different teacher. Throughout the research

⁴³ Online appendix 3: <u>https://africacentreforevidence.org/project-outputs-3/</u>

period, I was particular concerned about the additional workload I was adding to teachers and the extent to which my research disrupted the regular schooling activities. I therefore ensured to closely co-ordinate all logistics with the school's principal and the involved teachers. All research visits were set around dates offered by the schools. A linked issue was that, due to the prominence of the programme, a number of research activities were engaging the ICT4RED schools already including an internal and external evaluation, ongoing M&E, and a number of South African post-graduate projects. Research fatigue among the teachers and learners of these schools was since a serious threat. To mitigate this issue somewhat, I decide to focus my research in the schools involved in stage 2 of the programme, which unlike stage 1 and 3 schools were only subject to the external evaluation at the same time as my research and not to other research activities mentioned above.

I also grappled with the question of what level of anonymity to apply in my research. While all teacher- and school-level data presented in this thesis is fully anonymised, the research area and the ML4D programme under investigation is not. I took this decision for two reasons. First, the particular design of the ICT4RED programme and context in which it is applied is important in order to understand the data collected. It did not seem feasible to remove all design- and context-related information that could have potentially deanonymised the research without removing vital information to contextualise the data. Second, ICT4RED is the educational technology flagship programme of DBE and DST. In a South African context, it is therefore not feasible to anonymise the programme due to its popularity and media coverage.

The research interaction between the research participants and myself as a researcher are subject to a range of power imbalances. Gender, age, ethnicity, and language all affected my positionality as a researcher. This is reinforced by the historical context of apartheid in South Africa. Throughout the research I was sensitised to my behaviour and interactions as a mid-twenty white male associated with a UK institution in a context in which most of the research participants where black African females in their forties. I was aided in this somewhat based on my previous experience of working in rural Eastern Cape on community development programmes for four years and my basic proficiency of isiXhosa. However, my positionality as a researcher is likely to have influenced the interactions and data collection to some extent.

Lastly, there is also a question whether desirability bias has influenced some of the teachers' responses. There is a curious absence of negative themes in my qualitative case study research and even when probed for what they did not value about using the tablets,

teachers reported little feedback that could be seen as a criticism of the ICT4RED. I discuss this issue in more detail in appendix 7.1 linked to chapter 7.

4.3.5 Limitations of the qualitative case study

My qualitative case study research is subject to a range of limitations too. These stem from the representativeness of the identified case, the nature of the research process, and the analytical framework chosen. In terms of the representativeness of the identified case, there is a risk that the Information Communication and Technology for Rural Education Development (ICT4RED) programme does not reflect the average mobile learning programme in LMICs. The programme's design places a large emphasis on pedagogies rather than on the mobile technologies; it is focused on teachers and takes a systemic approach; and it is implemented at scale by the public sector. Each of these attributes suggest that the ICT4RED might present the case of a mobile learning programme in LMICs that subscribes to an innovate approach to teaching and learning with mobiles that does not present the norm. In addition, my research was only focused on teachers and not on learners involved in the ICT4RED and I further zoomed in on teachers partaking in phase 2 of the 3-phase programme.

In terms of the nature of the research process, my positionality as a white male researcher also is likely to influence and limit my case study's findings. There might be themes and narratives that research participants did not feel comfortable sharing with me and my presence might have been associated with the formal programme evaluation and implementation team as discussed in appendix 7.1. Limited research funds also set boundaries around the number of schools and teachers that I could access as well as the time spent in the Eastern Cape for fieldwork.

In terms of the analytical framework chosen for my case study, a key limitation of the CA is its perceived methodological or ethical individualism. A range of scholars hold that the CA is too focused on individuals and their functionings and capabilities and that this focus neglects structural issues such as power relations, social constraints on individual's choices, and group interactions and identities (e.g. Stewart & Deneulin 2002; Ibrahim 2006; Evans 2002). Sen does address these critiques to some extent, for example, taking into account adaptive preferences, different social environments (through conversion factors) and providing an evaluatory device that can highlight inequalities in opportunities and not just outcomes (the capability-functioning distinction).

However, there is consensus that CA remains incomplete in regard to theorizing how structural issues and power relations influence individual's and group's freedoms and agency (Robeyns 2005; Zheng 2009; Gigler 2015). In such instances the CA needs to be supplemented with additional conceptual tools or complementary theoretical frameworks. Examples of this refer to the use of critical theories (e.g. Zheng & Stahl 2011; Podova & Roberts 2017) and the development of the concept of collective capabilities (e.g. Ibrahim 2006; Stewart 2004). The proposed CA conception in my thesis is naturally subject to the same limitations and the qualitative case study and capabilities conception derived of it could be challenged as neglecting structural issues hindering effective and meaningful human development and agency. In chapter 9, I will propose Kullman & Lee's (2013) conception of 'liberation within' as a justification for my focus on individual's functioning and capabilities, but acknowledge that further conceptual and empirical work is required in this frontier issue of the CA.

A second often claimed limitation of the CA is that is does not provide practical advice on what to do in order to advance human's capabilities and agency (e.g. Robeyns 2005; Sen 2009b.) While accurate, this challenge somewhat misses the point of the CA, which is not intended as a normative framework or theory of development (Sen 2009b). The CA serves only a normative function insofar as to prescribe the space in which to evaluate development policies and interventions, that is capabilities and functionings. This sentiment is perhaps most succinctly captured by Robeyns's (2005: 94) explanation that "the capability approach is not a theory that can explain poverty, inequality or well-being; instead, it rather provides a tool and a framework within which to conceptualise and evaluate these phenomena".

In the context of my thesis it is thus important to acknowledge that my capabilities conception does not claim to provide practical guidance on the specific design and implementation of mobile learning programmes in LMICs. Further, I do not claim that the capabilities and valued beings and doings of teachers identified in my case study research are a unique product of the application of the CA or necessarily have to be expressed using its vocabulary. Rather, the unique contribution of the CA is its ability to "bring to the surface, systematically and coherently, a set of key concerns for scrutiny based on an explicit philosophical foundation" (Zheng 2009: 74). It is in this function that the CA enters the discourse on mobile learning's role and contribution in international development. In order to develop specific ML4D policy and programme recommendations and to guide decisionmaking, my capabilities conception of ML4D will have to be supplemented with additional explanatory theories.

Conclusion

This chapter has outlined and justified my applied research design and methodologies. In return it has discussed the design and conduct of the mixed-methods systematic review and the qualitative case study. The systematic review follows a mixed-methods approach in order to synthesise both quantitative and qualitative studies investigating the effects of ML4D interventions in practice. The review design therefore aims to provide empirical findings of what works, how and why for ML4D and to then plot these findings against a theory of chance for the use of mobiles to support education and development in LMICs. I apply meta-analysis and thematic synthesis as the method of synthesis to aggregate and configure the effects of ML4D programmes.

The qualitative case study follows as an instrumental case study design and generates qualitative data on teachers' perceptions on the use of mobiles in an educational context. It aims to assess the feasibility of applying the CA as a conceptual lens to frame ML4D. It uses the ICT4RED, a ML4D programme in rural South Africa, as a case to explore whether teaching with mobile technologies is perceived to support valued beings and doings and to enhance opportunities of educators. I use in-depth individual interviews, classroom observations, and focus group discussions as the main instruments for data collection.

The provided alternative conceptualisation of ML4D based on the CA is of conceptual nature. It draws from both empirical datasets. I first use the results of the mixed-methods systematic review to highlight gaps and contradictions in current conceptions of ML4D (chapters 5 and 6). I then use the results of the case study to explore the feasibility of an alternative conception of ML4D from the capabilities perspective (chapters 7 and 8). In a last step, I combine both these findings in a conceptual inquiry to propose a rethink of ML4D guided by the empirical evidence-base established by the systematic review research and the conceptual framework developed from the case study research (chapter 9).

Chapter 5. The aggregate effects of mobile technologies on education in LMICs

Introduction

This chapter presents the first set of findings from my mixed-methods systematic review. The systematic review set out to investigate the effects of mobile technologies' application by students and educators in Low- and Middle-Income Countries (LMICs) on learning outcomes, teaching practice, empowerment, and EMIS. Following a mixed-methods systematic review approach, the review combines quantitative and qualitative data on mobile technologies' impact using two distinct review modules (see chapter 4). This chapter only reports findings on the detailed aggregate effect of mobile technologies on observed educational outcomes, for example changes in learners' test scores, self-efficacy, and applied teaching strategies. The findings of the configurative review module on contexts and mechanisms that unpack the aggregate effects reported here will be illustrated in chapter 6.

This chapter is concerned with what works for mobile learning in LMICs, to what extent, and modified by what variables. First, I provide a descriptive map of the included research evidence on the use of mobile technologies to support education in LMICs. This map describes the characteristics of the available evidence-base. It comments on overall patterns in the included evidence on mobile learning for development (ML4D), for example what type of programmes have been evaluated, technologies used, educational approaches followed. This discussion includes an assessment of the trustworthiness of the available evidence-base. The objective of the descriptive map is to contextualise the subsequent meta-analysis of the best available research evidence against the broader features of the evidence-base, its overall trustworthiness, and relevance to the research question.

Second, this chapter discusses the findings of the statistical meta-analysis on mobile technologies' effects on educational outcomes in LMICs. This meta-analysis synthesises quantitative effect sizes derived from the included research evidence to estimate the overall pooled effect of mobile learning interventions. The conducted meta-analysis, in addition, interrogates this pooled effect using moderator and sensitivity analyses to illustrate the

relative effects of key attributes of mobile learning such as the applied pedagogy, educational context, and existing use and perceptions of mobile technologies. I also assess the robustness of the meta-analysis findings to factors associated with the available evidence, in particular heterogeneity across mobile learning interventions and educational contexts, and the sensitivity of calculated effect sizes to research designs and outcome measures. The chapter ends with an overview of the key findings presented and a brief conclusion.

5.1 What research evidence do we have on mobile learning in LMICs? A descriptive map of the evidence

This systematic review employed exhaustive and reproducible methods to identify all available research evidence presenting an empirical account of what works, why, and how when using mobile technologies to support education in LMICs. Applying the inclusion criteria outlined in chapter 4, I first map the characteristics of the included research evidence. This serves to unpack the features of the available evidence-base and to identify descriptive patterns and structures for further analysis in the synthesis.

5.1.1 What research evidence did I identify? (systematic search results)

This section reports on the results of the exhaustive search for relevant research evidence. I designed a combined search strategy to identify quantitative and qualitative research studies. The search strategy was deliberately over-inclusive, that is the applied search techniques aimed to identify as much relevant research studies as possible, aiming to reduce the chance of missing any relevant studies. I ran searches from February to April 2014 and conducted a full search update between February to March 2016 (Appendix 4.1). Figure 5.1 below presents a detailed outline of the search results. The exhaustive search yielded a total of 32,002 hits from 108 unique sources. As anticipated the majority of search hits (31,069) was not relevant to the review topic and consequently excluded. As figure 5.1 shows, after the removal of these irrelevant hits, 933 papers were left for screening on abstracts and I excluded 656 further papers with reasons for exclusion listed below.

This left 277 studies that were accessed at full-text and screened against the detailed inclusion criteria. As a result, a total of 80 unique studies, reported in 99 papers, were included in the review (Appendix 5.1). These 80 studies included 52 configurative studies and 34 effectiveness studies. Six mixed-methods studies included both configurative and effectiveness studies. All identified studies were published in English even though search hits included Spanish and French citations⁴⁴. The following descriptive analysis of the evidence-base draws on all 80 included studies. Appendix 5.2 provides an overview about the key characteristics of the included 80 studies.

⁴⁴ While I translated all citations at abstract level using Google Scholar, I did not have capacity to screen foreign language studies at full-text.

Chapter 5: Impact of mobile learning in LMICs



5.1.2 Socio-economic contexts within the available evidence-base

The identified evidence-base featured a wide geographic spread (Figure 5.2). A majority of studies was conducted in Asia (n=37), closely followed by Africa (n=34). Only eight studies focused on mobile learning in Latin America, and a sole study was conducted in Oceania. The review did not find any research on ML4D in a middle-income country in Europe. India (n=10), Taiwan (n=9), and South Africa (n=9) were the single most represented countries. Research from East Asia, due to the strong presence of the Asian Tigers group of countries, contributed almost a quarter of the studies (n=19). I therefore caution that there is heterogeneity across the socio-economic contexts in which the reviewed mobile learning interventions were applied. The setting of a ML4D mobile learning programme in urban Malaysia (Ismail 2013⁴⁵), for example, will differ from a programme in rural Tanzania (Enge 2012). I analyse the sensitivity of the review findings to countries' economic status as low-/lower-middle income or upper-middle income in the moderator analysis in section 5.2.2.

The majority of the reviewed interventions were conducted in urban areas (n=43) and only 17 ML4D programmes were implemented solely in rural areas. Twenty interventions reported mixed or peri-urban settings. More than two-thirds of programmes were conceived and driven by research interest rather than by policy actors or practitioner (n=55). National education institutions, on the other hand, only initiated eleven programmes, leaving government institutions in a clear minority compared to external entities—that is IGOs, NGOs, and corporates—who conceived 23 mobile learning interventions⁴⁶. A vast majority of ML4D programmes were implemented in a formal educational setting (n=65). Only 15 studies assessed mobile learning in an informal educational environment (e.g. extension services, NGO-run literacy campaigns). In the sample of formal education, primary schools presented the most targeted level of schooling (n=32). They are followed by high schools (n=20) and universities (n=19). Of the university programmes only three studies featured research in a distance education context⁴⁷. These overall school classifications, however, hid a clear pattern. ML4D in urban areas featured a significantly higher number of programmes conducted in high schools and universities (62% of the sample). The corresponding number in rural areas is a mere 18%. Rural ML4D programmes are predominately implemented in primary schools (36%) and informal education programmes

⁴⁵ When reporting on the systematic review findings, studies included in the review are only referred to by last name of the first author and year of publication.

⁴⁶ These numbers do not add up to 80 as some interventions were initiated through collaboration between different entities.

⁴⁷ The review identified a larger body of literature on mobile learning in distance education. Yet, most of these studies did not meet the methodological inclusion criteria.
run by NGOs and other non-state actors (27%). These patterns indicate that ML4D programmes by design follow existing structural inequalities in LMIC education systems (e.g. McCowan & Unterhalter, 2015; Motala et al 2013; Unterhalter 2015).

This unequal pattern further holds when assessing the existing resources and infrastructure ML4D programmes could draw from. Reliable access to electricity, stable network coverage, and safe storage facilities were reported in over three-quarter of the studies, which again hints at an implementation of mobile learning interventions in less resource-constrained educational contexts. Further, 42% of interventions were implemented using educators' and learners' own mobile devices. There was also evidence of an existent systematic use of ICTs in education in 23 interventions, all but one of which were conducted in urban areas.

Figure 5.2 Geographical location of included studies



Participants within the reviewed ML4D interventions similarly were unlikely to represent the most marginalised groups within communities or societies. A clear minority of programmes (n=7) targeted illiterate learners. Semi-literate learners—i.e. primary school learners in each instance—were a focus of 24 interventions, while the majority of 51 ML4D programmes targeted literate participants. Most studies were inclusive of gender and thus reported a balanced sample. Only four studies reported findings of female-only ML4D programmes (Balasubramanian 2010; Pimmer 2014 Velghe 2014; Zelezny-Green 2014) and in a single programme the use of mobile technologies was reserved for males (Nawi 2015). Participants' age correlated with the above-outlined schooling classifications⁴⁸. Using technologies to support adult learning, a group for example often at a higher risk of illiteracy in LMICs (Nussey 2015, Maddox & Esposito 2011), was only reported in 11 studies. There was no experimental research reported on mobile learning programmes to support people with disabilities in LMICs.

5.1.3 Types of mobile learning interventions within the evidence-base

The identified 80 studies assessing the use of mobile technologies to support education in LMICs applied a diverse set of mobile learning programmes. The review identified 70 independent mobile learning programmes that aimed to improve educational outcomes in LMICs. The TeacherMate programme (e.g. Kim 2011), Worldreader (e.g. UNESCO 2014), Bridge International Academies (e.g. Enge 2011), as well as English in Action (e.g. Sohel 2012) each were applied in multiple interventions. I retrospectively assigned ML4D interventions into programme categories (Table 5.1). These categories thus iteratively emerged from the included studies and were formulated along stated intervention objectives. Categories were not mutually exclusive.

'Mobile-assisted language learning' (MALL) comprises the largest category in the identified ML4D programmes (n=31). MALL as a sub-set of mobile learning interventions applied mobile technologies with the explicit aim to improve language acquisition. In 20 programmes English as a foreign language was taught using mobile technologies. The remainder of programmes focused on literacy development in the learners' home language, predominantly Spanish (e.g. Zurita & Nussbaum 2004). Sixteen ML4D programmes were categorised as 'mobile assisted mathematics learning' (MAML). These programmes used mobile technologies explicitly to improve numeracy or more advanced mathematical

⁴⁸ Studies pre-dominantly reported the age of participants by the average national age for a certain grade. This hides the widespread problem (particularly rural areas) of learners accessing schools later in life or in disrupted patterns.

subjects. I assessed in the aggregative module whether MALL and MAML, given the approaches' popularity, are particular effective subgroups of ML4D interventions.

I classified 19 studies as exploring 'perceptions & adoption' of ML4D programmes. These studies were solely included in the configurative module as they explored the context of ML4D. This exploration of context, for example, could include a survey investigation whether teachers were willing to use mobile technologies as an educational tool. In nine of these studies factors of adoption in line with the technology acceptance model (Davis 1989; Venkatesh & Davis 2000; Venkatesh et al 2003) were analysed. A minority of only 10 studies were identified as 'teacher development'. Teacher development interventions used mobile technologies to primarily improve teachers' educational skills. Interestingly, none of these programmes were embedded in official teacher training courses. Mobile learning in teacher development was exclusively used as a form of on-job training and only in one instance linked to an accredited professional development course (Leach et al 2004). A specific sub-set of studies from East Asia (n=9) applied mobile technologies to facilitate contextualised learning, for example during field trips to national parks or heritage sites. Termed as 'experiential / inquiry-based' these studies presented a homogenous programme category. Alas, the external validity of this category is doubtful due to the complexity of the applied technologies. The studies nevertheless used and reported on rich mobile learning pedagogies in LMICs and I therefore categorised them as feasibility experiments and relevant for inclusion in the systematic review. The remaining categories referred to mHealth, agricultural extension, EMIS, and distance learning.

| Types | # of studies |
|---------------------------------|--------------|
| MALL | 31 |
| (English as a foreign language) | (20) |
| (Native language) | (11) |
| MAML | 16 |
| Perception & adoption | 19 |
| Teacher development | 10 |
| Experiential / inquiry-based | 9 |
| mHealth | 3 |
| Distance learning | 3 |
| Agricultural extension | 2 |
| EMIS | 2 |

 Table 5.1 Overview of mobile learning programme types

5.1.4 Mobile technologies used in mobile learning interventions

The reviewed mobile learning programmes applied an array of mobile technologies in order to support education in LMICs. Mobile phones, used in 66 programmes, were the dominant technology by a large margin. This dominance of phones as the most prevalent technology in ML4D programmes was independent of context. Mobile phones were the most applied technology in both rural and urban settings, foremost in each geographical sample and type of schooling, and featured as the technology of choice in each ML4D intervention category. Smart phones and basic/feature phones were used to the same extent (n=33/n=33). There was nevertheless an unsurprising trend to use more sophisticated phone models in more recent programmes. This is in line with the advancement in mobile technologies and the growing diffusion and affordability of more sophisticated devices.

Tablets and PDAs presented the second most applied technologies (n=14). Regarding PDAs, this finding was somewhat surprising as the technology itself is increasingly outdated and production and development of devices has been discontinued. The review did not record a use of PDAs in a mobile learning intervention post 2012. On the contrary, the use of tablets in mobile learning programmes was a recent trend. Most of these programmes have been conducted since 2013 (n=11) and of the included studies published in 2015 and 2016, 57% reported the use of tablet devices. All but three tablet-based mobile learning programmes were conducted in Africa. Additional mobile technologies identified in the review included gaming devices (n=4), mp3/iPod (n=2), and radios (n=1).

Mobile learning interventions further exploited a diverse range of mobile devices' unique affordances. Table 5.2 gives an overview of the main affordances used.

| Affordance | # of studies |
|---------------------------|--------------|
| Connectivity | 57 |
| Educational software | 31 |
| (Incl. educational games) | (21) |
| (Incl. apps) | (11) |
| Camera/video | 29 |
| SMS | 18 |
| GPS | 14 |
| e-books | 11 |
| Voice/Audio | 9 |
| Social networks | 6 |
| Mobility ⁴⁹ | n/a |

Table 5.2 Overview of mobile devices' technological affordances

A large majority of programmes applied or intended to apply a 1:1 ratio of device ownership (n=69). This finding can be explained with the underlying ubiguity of in particular mobile phones in LMICs, which allowed 56 programmes to use mobile technologies that participants reported as having used before. In addition, where reported (n=20), over 90% of participants on average had a positive perception of the applied technologies at baseline. However, mobile learning programmes in rural areas were less likely to use participants' own devices and relied on providing mobile technologies as part of the intervention design.

5.1.5 The design of ML4D interventions

The reviewed ML4D interventions followed a range of different designs in order to support education in LMICs. A large majority of mobile learning interventions was learner-centred (n=68). Only 18 programmes focused primarily on teachers as the unit of implementation⁵⁰. The same pattern held true for the assessed educational outcomes. While 51 studies measured changes in learning achievements, only 12 attempted to record changes in teaching practices. This finding reveals a bias of mobile learning programmes to bypass teachers, despite teachers serving as the principal agent in and gatekeeper to high quality education in LMICs (Hassler 2011; Leach 2005; Piper 2015; Rao 2014). While teachers usually were not excluded from the 68 learner-centred interventions (e.g. they could use a similar device), the interventions were designed around learners' needs and preferences

 ⁴⁹ The inherent mobility of devices presented an inclusion criterion in the review.
 ⁵⁰ Six studies reported on interventions that targeted learners and teachers simultaneously.

and aimed to primarily support this group rather than educators. Ten teacher-centred programmes placed a focus on teacher professional development (e.g. acquisition of more advanced teaching skills) (Leach 2005). Six teacher-centred interventions merely supplied mobile technologies as a support tool to allow for a more convenient teaching experience (e.g. Valderama 2010). Lastly, two studies investigated the potential of mobile monitoring to increase teachers' attendance (Aker 2015; Cilliers 2014).

The reviewed evidence identified a clear trend to embed ML4D programmes into a larger educational effort. That is, ML4D programmes were for most parts not standalone interventions in which the facilitation of mobile learning was the only educational activity. The large majority of interventions were embedded in formal education structures (n=65), and an additional seven programmes were implemented in conjunction with a comprehensive informal education programme, such as an NGO-led adult literacy campaign. Only eight reviewed interventions could be regarded as independent programmes (e.g. the Worldreader's mobile app). This observed design of ML4D interventions indicates that most interventions assumed a supplementary educational role. By itself, learning and teaching with mobiles, let alone the mere provision of devices, appeared to not present a viable intervention design in ML4D.

Technological components of ML4D interventions

The review also assessed the detailed design characteristics of ML4D interventions by breaking down interventions according to their technological and educational programme components (Figure 5.3)⁵¹. On the technology side (panel A), the most applied component referred to the provision of digital curricula and educational content on mobile devices. A total of 66% of the interventions applied this programme component. Provision referred to making available pre-designed curricula or subject materials *without* input from the programme participants. This top-down mode of programme delivery presented the most applied intervention component. Only 55% of the ML4D interventions provided mobile technologies to the programme participants, which was the second most applied intervention component. This runs counter to famous examples of perceived technology-led learning as advocated by Sugata Mitra's Hole in the Wall project and subsequent development of minimally invasive education (Mitra & Vana 2001; Mitra 2003). The provision of mobile devices and third of interventions offered training on device use to the

⁵¹ Only the six most prominent components are reported below.

programme participants (n=22) and a similar number of programmes (n=20) developed tailored new educational content for the applied intervention.

The design of mobile learning interventions was dependent on the socio-economic context of the ML4D programme. Figure 5.3 compares the design patterns of intervention components in rural and urban areas. Keeping in mind that, overall, there were fewer interventions conducted in rural areas than in urban areas, the data shows that interventions conducted in rural contexts were more likely to provide mobile technologies as well as to engage in user training regarding the correct application of the device. This different programme approach seems to reflect existing inequalities in technology usage. Rural programmes were further less likely to develop new tailored content, an intervention component that requires more sophisticated user input. However, there was no difference in the provision of infrastructure as one might have suspected given rural areas' resourceconstrained contexts.







Educational components of ML4D interventions

Panel B of figure 5.3 indicates the applied educational intervention components. Educational components refer to the educational input that the programme facilitated. It is important to recall that, as shown above, the majority of interventions were embedded into ongoing educational interventions and that the use of the mobile technology therefore was mainly supplementary to the existing educational events and actions. First, mobile-supported lessons were the most applied educational component (n=41). Of these, ML4D programmes most commonly provided structured lesson plans in which the technology use was incorporated (n=21). Direct lessons were facilitated using video technology (n=13), SMS (n=9), and audio/voice (n=5). Lessons represented a structured and targeted string of learning activities to meet a predefined learning objective. Second, mobile-facilitated quizzes and exercises equally presented a popular educational input in ML4D interventions and a total of 34 programmes made use of this input.

Third, educational software such as the TeacherMate (Kim 2012) or MobileMath (Voigt 2010) applications featured as a fast-growing educational component in the reviewed interventions (n=31). This component provided the most comprehensive educational environment with the ability to include for example lesson plans, exercises, and tutorial videos within one programme. Educational software components were, however, dependent on the availability of devices featuring more sophisticated operating systems. Fourth, in the

most rudimentary approach, a number of ML4D programmes relied on supplying participants with additional educational information such as access to lexica or homework reminders (n=16). This component mainly increased the ability of participants to receive information of interest or relevance to their learning and teaching practice. Lastly, the use of e-readers or tablets to store e-books as an educational component replacing traditional text-books presented a minor intervention approach in the reviewed ML4D programmes (n=15).

As in the case of technological programme components, the review identified a structural impact of the socio-economic context on the design of educational components. The reviewed evidence, overall, indicated that the educational design of rural ML4D programmes subscribed to a less complex educational structure. Programmes components that required less pedagogical input (i.e. quizzes and exercises; extra information; and e-books), were overrepresented in rural ML4D interventions. Pedagogical rich components (i.e. educational software; and lessons) were, on the other hand, underrepresented in the rural programmes. Again, these findings provide evidence that mobile learning programmes in LMICs might fail to counter existing socio-economic inequalities in education systems.

5.1.6 Assumptions about how ML4D interventions facilitated educational change

The above data shows the diversity in the design of mobile learning interventions and the manner in which mobile technologies are used to support education in LMICs. In addition, the review investigated how the mobile learning interventions were positioned in their ability to influence educational change in LMICs. I therefore systematically coded the included research studies for data on the assumed educational change that the introduction of mobile technologies aimed to facilitate. Extracting data from the reported mobile learning interventions served to unpack what kind of changes in the educational environment (both teaching and learning related) were targeted in the interventions. This information then allowed comparisons as to whether these changes in learning and teaching through the use of mobile devices in fact referred to mobile learning's particular pedagogical strengths and contribution.

In this remit, I first assessed whether the mobility affordances of the technological devices were exploited in the interventions. A main attribute of mobile learning is the ability to allow learning in different contexts due to the mobility of the device. This feature allows the fluid interaction of learning events between formal and informal contexts to reinforce learning. Yet, despite this rationale only half of the applied ML4D interventions allowed an unrestricted

use of the devices across contexts $(n=32)^{52}$. That is, half of the programmes implemented restrictions to the use of devices in informal contexts. In most instances (n=27), this translated into a limitation to only apply the mobile technologies in the classroom. In case where no limitations to device use were applied, the most common usage was described as learning with mobiles 'anytime, anywhere' (n=32), 'at home' (n=14), and 'at work' (n=17). Mobile phone-based interventions were strongly correlated with an unrestricted usage, whereas programmes using PDAs or tablet devices were by a large majority limited to use in formal contexts. I assessed in the aggregative review module (section 5.2.2) the possible impact of these limitations in device usage on the effects of ML4D programmes.

Reporting on the underlying pedagogies that informed the ML4D interventions, as well as how the programmes aimed to change teaching or learning was limited to information received at face value⁵³. That is, I used the terms studies used to self-identify their applied pedagogies. Therefore, I only considered pedagogies that were *explicitly expressed* as such in the included studies and refrained from attempting to imply from the programme description what pedagogical approach might have informed its design. Facilitating more learner-centred educational approaches was a main change that was targeted through the introduction of mobile devices (n=52). Yet, more often than not, it was thereafter not evident how the applied intervention changed the existing educational situation to assume a more learner-centred character. For example, it was not clear how reading e-books and taking notes on a tablet device was inherently more learner-centred than reading and note-taking using traditional learning materials (e.g. Worldreader 2012). The review therefore identified a large discrepancy between stated educational approach and practical intervention design.

In addition to the overall theme of learner-centred pedagogies, a number of more focused educational strategies were stated in the identified ML4D programmes (Figure 4.4). The single most prominent pedagogy referred to context-aware learning (n=27) in which learners were assumed to gain a more relevant, personalised, and practical educational experience. This group included the most sophisticated mobile learning approaches such as a 'context-aware ubiquitous mobile learning system' (Chen & Li 2010) and 'RFID-supported immersive ubiquitous learning environment' (Liu 2009). Nineteen mobile learning interventions applied game-based learning strategies which went hand-in-hand with the design of educational

⁵² The sample size is 61 studies as some studies did not implement a ML4D programme but assessed prevailing perceptions and contexts towards ML4D.

⁵³ I attempted to code studies as follows: prior educational approach, learner / teacher interaction, teacher skill level, teacher / pupil ratio. There was sufficient reliable information reported within the included studies to populate these codes consistently.

gaming devices or educational software. Game-based learning was mainly targeted at primary school children. Collaborative learning was stated by 16 ML4D programmes as the underlying pedagogical approach. These programmes designed group activities in which the mobile technology served as tool to ensure effective collaboration took place between learners. For instance, devices were used to enforce participation of each group member, preventing more advanced or assertive learners to take over the group's task. Lastly, six studies mentioned an explicit transformational agenda, for example critical pedagogies as developed by Freire (1972), to inform the ML4D programme design. Unfortunately, again, it was challenging to follow how the stated critical pedagogy objective was reflected in the intervention design (see also section 6.2.2 and 9.2.3 and 9.3.2).



Figure 5.4 Targeted pedagogies and mobile technologies applied

The systematic review of the included interventions established a clear correlation between types of mobile technologies and corresponding pedagogies. Out of the 21 studies that did not state any educational approach to inform the programme design, 18 used basic/feature mobile phones. Of the 27 studies with the most sophisticated mobile learning approach all used mobile devices with advanced computing and visualisation affordances such as smartphones or PDAs. Similarly, only a marginal number of game-based and collaborative learning ML4D interventions were conducted using basic/feature phones (n=2). Given the above-reported pattern that mobile learning interventions in rural areas applied more basic

technologies (5.1.4 - 5.1.6), this therefore further translated into a lower likelihood of rich mobile learning pedagogies to underpin interventions in rural areas. This finding is also in line with the data presented above (5.1.5 - 5.1.6) that rural ML4D interventions followed a less complex programme design. In sum, there is thus further evidence of structural inequalities being reproduced by the choice of mobile technologies and applied educational approaches in ML4D interventions.

5.2 Meta-analysis findings: aggregate effects of ML4D interventions

I next present the findings of the statistical meta-analysis that synthesises the research evidence included in the aggregative review module. To recall, the aggregative review module aimed to answer the question of mobile learning's overall impact on learning outcomes, teaching practice, EMIS, and empowerment. Research evidence included was therefore required to make use of a rigorous study design able to evaluate the causal relationship between the provision of mobile learning interventions and the changes in the above-mentioned outcomes. The applied synthesis methods—i.e. meta-analysis—then aimed to aggregate the reported causal effects into one pooled numerical effect size reflecting the direction and magnitude of mobile learning's impact on educational outcomes in LMICs.

5.2.1 How trustworthy is the quantitative evidence (risk of bias assessment of studies)?

I included 34 rigorous impact evaluations of ML4D interventions in the aggregative review module that met the minimum methodological requirements. These studies were then subject to a detailed risk of bias assessment to establish the trustworthiness of the reported research results (Appendix 4.2). Fourteen studies were found to be of critical risk of bias and therefore excluded from the synthesis. Consequently, 20 studies were eligible to feed effect size data into the meta-analysis⁵⁴. The detailed results of the risk of bias assessment are reported in figure 5.5. The assessment results indicated clearly that baseline confounding and missing data were the main causes to exclude studies from the synthesis based on a critical risk of bias. Baseline confounding resulted from an inadequate control for observable and unobservable characteristics between experimental groups. In critical-judged studies, I found clear information that experimental groups were inherently not comparable and that any findings reported in the study were at risk to result from the differences between groups rather than from the applied ML4D intervention. For example, in the evaluation of the Worldreader programme in Ghana (Worldreader 2012), baseline test scores between experimental groups were significantly different and intervention schools were handpicked by local officials whereas control schools were chosen at the convenience of the evaluation team. Missing data, most commonly caused by a loss of control units (e.g. schools) rather than the loss of data from individual pupils, presented a second major factor of critical risk of

⁵⁴ A detailed overview of these 20 studies reporting data on applied interventions-to-outcome configurations and effect size data is presented in Appendix 5.3.

bias ratings. Enge's (2012) evaluation of the Bridge International Academies, for instance, did not report data from more than half of their initial control schools at endline.



The remaining 20 studies included in the aggregative synthesis were heterogeneous in trustworthiness. I judged six studies of low and moderate risk of bias respectively. The six low risk of bias studies presented five RCTs (Aker 2012; 2015; Cole 2012; He 2008; Pitchford 2014) and one rigorous matched quasi-experimental study (Kaleebu 2013). The studies were of considerable scale featuring an average sample size of 2514 participants (range: 88–5317) as well as an average period of follow-up of six months (range: 2–12).

5.2.2 Do mobile learning interventions improve learning outcomes in LMICs?

This section reports the overall impact of the educational use of mobile technologies in LMICs on learning outcomes. The reported meta-analysis (Figure 5.6) combines the learning outcomes of 19 individual mobile learning programmes reported in 17 studies⁵⁵. The meta-analysis thereby comprises data from 17,909 participants involved in the included studies. Learning outcomes were measured in validated assessment tools such as the Early Grade Reading Assessment and tailored pre- and post-intervention tests. Eleven studies administered learning assessment as extra curricula tests, while six studies drew data from

⁵⁵ Piper and colleagues (2015) PRIMR intervention tested three different mobile learning programmes.

regular curricula tests. No study relied on administrative records of learners' test scores. The choice and application of the reported outcome measures was judged as a low risk of bias for each of the included studies in the meta-analysis. While intermediate learning outcomes such as student attendance and motivation were reported by individual studies, these intermediate outcomes were rarely measured consistently across both experimental groups and were therefore not included in the meta-analysis.

The meta-analysis of standardised mean differences indicates a positive and significant effect of mobile learning of 0.47 (0.35, 0.58) SMD on learning outcomes. This pooled effect size constitutes evidence of the impact of mobile learning interventions in LMICs. It expresses a clear and statistically significant increase in learning that can be attributed to the application of mobile technologies as an educational tool. The systematic review since presents evidence of the effectiveness of ML4D interventions as a means to increase learning outcomes in LMICs.



Figure 5.6 Forest plot: Average effect of mobile learning programmes on learning outcomes in LMICs

Heterogeneity: Q=127; df=18; p=1.68E-18; l²=85.9%; tau²=0.0349

The identified pooled effect size of 0.47 (0.35, 0.58) SMD can be considered a large effect size. Using the conversation formula outlined in appendix 4.3, the effect size translates into a 23.2% larger improvement in learning outcomes reported in the group of learners using

mobile technologies as compared to learners not using mobiles. Comparing the established overall effect size of 0.47 SMD to related meta-analyses assessing the impact of educational interventions in LMICs, it emerges that mobile technologies seem to present a more effective approach to improve learning outcomes than other educational interventions. A meta-analysis by Petrosino and colleagues (2012), for example, estimated the overall impact of educational interventions at 0.14 (0.10, 0.23) SMD; albeit interventions to improve language learning score a higher effect size of 0.25 (0.19, 0.35) SMD. A meta-analysis of only RCTs on interventions that aimed to improve learning in primary schools in LMICs further found ICTs to be among the most promising interventions with a pooled effect of 0.15 SMD⁵⁶. Finally, the most comprehensive systematic review and meta-analysis of interventions for improving learning outcomes LMICs identified structured pedagogy to have had the largest effect on education of 0.23 (0.13, 0.34) SMD (Snilstveit et al 2015). Snilstveit and peers' review also included 19 studies investigating the impact of computer-assisted learning which were found to feature an overall positive effect size of 0.07 (0.02, 0.11) SMD. The findings of my meta-analysis are since in line with the positive impact of ICTs on education in LMICs in general. Yet, I identified a larger effect size for mobile technologies and the mobile learning programmes which these technologies facilitate.

The identified pooled effect size of mobile learning interventions is statistically significant with a narrow confidence interval. It therefore presents robust evidence of the overall effectiveness of mobile learning as a tool to improve learning outcomes in LMICs. However, I caution to narrow down the question of mobile learning's impact to a single effect size and further analysis is required to understand and interpret this effect size. To being with, there is large heterogeneity within the conducted meta-analysis (Q=172, p<0.05; i^2 =85.9%).

A visual overview of the forest plot in figure 5.6 reveals that, while the combined impact of mobile learning is statistically significant, of the individual included studies only a minority of six studies features a narrow confidence interval, and over a quarter of the studies (n=6) cannot rule out the possibility that the study's findings are due to chance. However, this observation can be explained by the limited sample size of these studies, which include on average 34 learners per experimental group. Such small-scale studies are by design often underpowered for the purpose of statistical analysis (Thomas et al 2012; Borenstein et al 2013). Meta-analysis as a methodology is in particular designed to allow for a combination of samples and results from individual studies in order to increase the statically power of the overall analysis. I am since confident that the overall effect size, which is statistically

⁵⁶ There is no overlap in studies included between this meta-analysis and the reported studies.

significant and carries a narrow confidence interval, is a more accurate reflection of the magnitude of mobile learning's impact than the reported individual effects.

The high degree of heterogeneity reported in the meta-analysis was anticipated⁵⁷ and I therefore a priori identified a range of variables to that might explain the diversity in the identified effects. These sensitivity and moderator analyses presented below control that there were no systematic differences across the included studies that generate the variation in effect sizes. If systematic differences would prevail, the reported differences in effect sizes would not be related to the effectiveness of the applied interventions but rather to some external factors; in which case the calculated overall effect size would fail to present a true reflection of mobile learning's impact.

Assessing the quality of my meta-analysis: are the findings sensitive to heterogeneity in study design?

I first assessed whether the variance in effect sizes might be caused by factors related to the applied impact evaluation design. For example, a more rigorous impact evaluation design might systematically yield different effect sizes from a less robust design. I therefore investigated the sensitivity of the pooled effect estimate to a number of possible design variables (Table 5.3). It is, however, important to note that this table merely presents an observational overview to uncover possible sensitivities that I then formally assessed statistically as reported below.

⁵⁷ For example, Petrosino and colleagues' (2012) meta-analysis reports heterogeneity statistics of Q= 325.60, df=.24, p<.001; l^2 =92.62, tau²=.019.

| Variable | SMD | 95% CI | Q | Tau ² | l ² | P-value | Sample |
|------------------------------|-------|--------------|------|------------------|----------------|---------|--------|
| | | | | | | | |
| Mobile learning: all studies | 0.465 | 0.348, 0.583 | 127 | 0.035 | 85.9% | 0.0000 | 17 |
| Study type: | | | | | | | |
| Randomised controlled trial | 0.496 | 0.096, 0.896 | 39.4 | 0.107 | 94.9% | 0.0000 | 5 |
| Quasi-experimental design | 0.661 | 0.390, 0.931 | 32.6 | 0.121 | 69.3% | 0.0003 | 12 |
| Risk of bias: | | | | | | | |
| Low risk of bias | 0.359 | 0.137, 0.586 | 43.8 | 0.041 | 93.2% | 0.0000 | 5 |
| Moderate risk of bias | 0.730 | 0.242, 1.220 | 14.0 | 0.219 | 71.4% | 0.0074 | 6 |
| High risk of bias | 0.667 | 0.376, 0.959 | 3.01 | 0.000 | 0% | 0.5560 | 6 |
| Removal of outliers: | | | | | | | |
| Aker 2012 | 0.679 | 0.421, 0.937 | 77.4 | 0.174 | 84.5% | 0.0000 | 16 |
| +/- 0.5 SD from the mean | 0.317 | 0.201, 0.432 | 60.5 | 0.015 | 76.8% | 0.0000 | 13 |
| Period of follow-up: | | | | | | | |
| above 1 month | 0.422 | 0.210, 0.634 | 49.2 | 0.043 | 89.8% | 0.0000 | 8 |
| 1 month or less | 0.712 | 0.380, 1.040 | 16.9 | 0.131 | 58.6% | 0.0179 | 9 |
| Sample size: | | | | | | | |
| above n=100 | 0.183 | 0.071, 0.294 | 8.97 | 0.007 | 77.7% | 0.0113 | 4 |
| n=100 or less | 0.833 | 0.515, 1.115 | 30.1 | 0.188 | 66.7% | 0.0084 | 13 |

Table 5.3 Sensitivity analyses

In the combined meta-analysis, I pooled studies of randomised controlled and quasiexperimental evaluation designs. The pooling of these study designs is a continuous debate within the systematic review community (Oakley 2000; Gough et al 2012; White & Phillips 2013) as quasi-experimental designs are often regarded as a less rigorous form of evaluation. In table 5.3 one can observe that RCTs and quasi-experimental studies indeed yield different pooled effect sizes with the RCTs reporting a smaller pooled effect. However, there was still considerable heterogeneity within the two sub-groups. Conducting a formal sub-group analysis (Online appendix 4)⁵⁸, I found that the different study designs could not explain the heterogeneity within the meta-analysis (Q=0.443, p= 0.51; heterogeneity explained: 0%). I therefore ruled out study design as an explanation for heterogeneity and the meta-analysis results are not sensitive to which impact evaluation design was applied.

In addition to study design, I also used the risk of bias ratings as an indicator of studies' rigour in the sensitivity analysis. I did not find any significant explanation for heterogeneity between low- and moderate-rated studies (Q=1.84, p=0.18; heterogeneity explained: 21%); between low- and high-rated studies (Q=2.72, p=0.10; heterogeneity explained: 15%); and between moderate- and high-rated studies (Q=0.05, p=0.83; heterogeneity explained: 0%). The pooling of studies with different risk of bias judgments since can neither explain the

⁵⁸ All formal sensitivity and moderator sub-group analyses reported in this section can be found in online appendix 4: <u>https://africacentreforevidence.org/project-outputs-3/</u>

identified heterogeneity.

As a next step, I analysed the sensitivity of the meta-analysis results to the removal of outliers. Following Borenstein and peers (2013), outliers are defined as studies diverting from the pooled effect size (0.47) by 0.5 standard deviations. This resulted in the exclusion of four studies from the analysis (Kumar 2012; Liu 2009; Pitchford 2014; Wu 2012). I found that the removal of outliers marginally reduces heterogeneity in the meta-analysis (Q=60.5; df=14; p<0.000; I^2 =76.8%; tau²=0.015). Using formal sub-group analysis, it emerged that the inclusion of these four outlying studies accounted for 57% of the prevailing heterogeneity (Q=53.5, p<0.05). However, as illustrated, the removal of outliers did only marginally reduce heterogeneity in the remaining sample of studies.

I also hypothesised that the period of follow-up could explain heterogeneity across the effect sizes in the meta-analysis. The duration between baseline and endline measurements seems to have had a small influence on heterogeneity (Q=2.56, p=0.11; heterogeneity explained: 58%). This finding, however, was not statistically significant. Lastly, I assessed whether the sample size of experiments can explain the variance in the identified effects. I found a significant impact of sample size on heterogeneity (Q=14.3, p=0.0001; heterogeneity explained: 61%). This finding explains that studies with a sample size above 100 participants have significantly smaller effect sizes than studies featuring a smaller sample.

Taken together, I identified two design related explanations for the large heterogeneity within the meta-analysis findings. The first explanation relates to sample size and to a lesser extent, secondly, to the period of follow-up. Both these factors relate to the degree of control researchers had over the implementation of the mobile intervention. Studies that were more closely supervised by the research team due to the smaller sample size and shorter periods of follow-up had significantly larger effect sizes. On the other side, more longitudinal studies with samples in the thousands (e.g. n=5014, Aker 2012) were under less control of the research teams. The degree of control arguably might have provided more room for immediate programme adaptation, targeting of inputs, supply of resources, etc. In contrast to study design and risk of bias, degree of control is a significant cause of heterogeneity. The larger the degree of control, the larger the impact of mobile learning programmes. Lastly, regardless of the controlled sensitivity variables, I highlight that the overall results of the analysis remain unchanged. In each analysis, the identified positive pooled effect size of mobile learning interventions remains robust and significant. I was unable to run a sensitivity analysis on the influence of the variance in outcomes measures applied in different studies as the included studies used a homogenous range of outcome measures judged at a low

risk of bias.

Assessing the quality of my meta-analysis: are the findings moderated by characteristics of the applied mobile learning interventions and their contexts?

Aside from variables related to study design, variables related to the applied mobile learning interventions themselves and the context in which these were conducted could have systematically influenced the differences in effect sizes. The meta-analysis included 17 studies, which applied a variety of programme approaches, were implemented in diverse settings, used different technologies, and so forth. It was since expected that the true effects of the interventions would vary across these programme- and context-related variables. This assumption was confirmed by the large heterogeneity identified in the review. In the following, I assessed possible variables moderating the identified impact of learning and teaching with mobile technologies on learning outcomes. Using the same structure as in the sensitivity analysis, I first constructed a descriptive overview table of all possible moderator variables (Table 5.4), before engaging in formal sub-group analysis.

| Variable | SMD | 95% CI | Q | Tau ² | l ² | P-value | Sample |
|------------------------------|-------|-----------------|------|------------------|----------------|---------|--------|
| | | | | | | | |
| | | | | | | | |
| Mobile learning: all studies | 0.465 | 0.348, 0.583 | 127 | 0.035 | 85.9% | 0.0000 | 17 |
| Applied technology: | | | | | | | |
| Basic/feature phones | 0.183 | 0.072, 0.294 | 8.97 | 0.007 | 77.7% | 0.0011 | 4 |
| Smart phones/PDAs | 0.726 | 0.454, 0.999 | 17.2 | 0.089 | 47.5% | 0.0464 | 11 |
| Tablets | 0.564 | 0.356, 0.772 | 29.1 | 0.035 | 89.7% | 0.0000 | 2 |
| 1:1 model | 0.571 | 0.386, 0.755 | 86.2 | 0.064 | 84.9% | 0.0000 | 17 |
| Shared ownership | n/a | no observations | | | | | |
| Socio-economic contexts: | | | | | | | |
| Urban | 0.796 | 0.393, 1.200 | 29.0 | 0.254 | 75.9% | 0.0001 | 9 |
| Rural | 0.200 | 0.068, 0.332 | 14.2 | 0.011 | 78.8% | 0.0027 | 5 |
| LICs ⁵⁹ | 0.224 | 0.177, 0.271 | 38.2 | 0.049 | 94.8% | 0.0000 | 4 |
| LMICs | 0.585 | -0.649, 1.820 | 6.57 | 0.687 | 84.8% | 0.0104 | 3 |
| UMICs | 0.690 | 0.410, 0.969 | 15.6 | 0.087 | 48.7% | 0.0485 | 10 |
| Formal education | 0.770 | 0.450, 1.090 | 59.5 | 0.228 | 81.5% | 0.0000 | 14 |
| Informal education | 0.162 | 0.110, 0.214 | 3.08 | 0.060 | 67.5% | 0.0792 | 3 |
| Primary School | 0.710 | 0.346, 1.070 | 35.0 | 0.192 | 80.0% | 0.0000 | 9 |
| High School | 0.858 | -0.010, 1.730 | 11.1 | 0.480 | 81.9% | 0.0399 | 3 |
| Tertiary | 1.340 | 0.791, 1.910 | 1.88 | 0.075 | 46.8% | 0.1700 | 2 |
| Gender | n/a | no observations | | | | | |
| Prior level of education | n/a | no observations | | | | | |
| Intervention category: | | | | | | | |
| MALL | 0.320 | 0.169, 0.417 | 16.0 | 0.013 | 62.5% | 0.0137 | 7 |
| MAML | 0.214 | 0.156, 0.272 | 36.0 | 0.599 | 94.4% | 0.0000 | 4 |
| Perceptions & Adoption | n/a | no observations | | | | | |
| Teacher development | n/a | no observations | | | | | |
| Experiential / inquiry-based | 0.661 | 0.316, 1.010 | 15.3 | 0.130 | 69.7% | 0.0182 | 8 |
| Teacher-centred | 0.322 | 0.184, 0.481 | 27.5 | 0.020 | 89.1% | 0.0000 | 3 |
| Learner centred | 0.663 | 0.411, 0.915 | 84.1 | 0.138 | 85.7% | 0.0000 | 14 |
| Pedagogy: | | | | | | | |
| Game-based | 1.190 | 0.694, 1.700 | 6.12 | 0.130 | 51.0% | 0.1060 | 4 |
| Context-aware | 0.490 | 0.257, 0.724 | 1.18 | 0.000 | 0% | 0.9470 | 7 |
| Collaborative | 0.666 | 0.352, 0.980 | 0.58 | 0.000 | 0% | 0.9010 | 4 |

Table 5.4 Moderator analyses

⁵⁹ Note that Papua New Guinea has been re-classified as a LMIC. However, during the conduct of the primary study the country was classified as a LIC.

Socio-economic context as a moderating variable: The first set of possible moderating variables related to the different socio-economic contexts in which the mobile learning programmes were implemented. As indicated above, a majority of ML4D programmes were conducted in urban areas and these programmes tended to apply more complex educational approaches. The moderator analyses provide statistical evidence that this structural inequality translated into a difference in programme impact.

Urban mobile learning interventions had a larger effect size of 0.80 (0.39, 1.20) SMD as compared to programmes in a rural context (g=0.20; 0.07, 0.33). This difference in effect sizes was statistically significant (Q: 13.1, p<0.05). The more complex intervention approaches, e.g. multiple educational components and richer pedagogical models, since seemed to have influenced the effectiveness of mobile learning programmes in urban areas. Rural interventions mainly used an SMS-based approach to deliver short snapshots of information to participants (e.g. Cole 2012). While this was found to be effective to increase learning, the gained improvements are significantly smaller compared to more complex interventions in urban settings.

Next, I assessed whether the impact of mobile learning interventions was dependent on the economic development of the countries in which they were conducted. I applied the World Bank's classification of economies and allocated studies to low income, lower-middle income, and upper-middle income status. On observation, interventions in low income countries have a smaller combined effect size (q=0.22) than both lower-middle (q=0.59) and upper-middle income countries (g=0.69). Yet, this analysis was based on small sample sizes and upon statistical investigation I could not rule out that the differences in effect sizes between these three groups came as a result of chance (Q=0.86, p=0.35). Mobile learning interventions in upper-middle income countries shared the common characteristic of being evaluated under closely controlled conditions. I have already established that this level of control explains a large amount of heterogeneity. Three out of the four interventions in low income countries, on the other hand, were informal mobile learning programmes with large samples sizes and a longer period of follow-up. In the absence of statistical significance for between country classifications, I refer to these experimental design conditions as an explanation for the differences in effect sizes. The same logic extends to the difference in effect sizes between programmes in formal and informal educational settings. The limited number of studies in these samples negated a meaningful statistical analysis.

Lastly, I investigated the type of school in which the mobile learning programmes was implemented as a potential moderator. While the meta-analysis generated a small nominal

difference between schooling types, this finding was not statistically significant (Q=0.10, p= 0.75). I since ruled out the hypothesis that mobile learning programmes in LMICs are particularly effective in certain types of schools. Rather, their impact on learning outcomes held true in each educational setting in which they have been implemented, that is informal education programmes (g=0.16), primary schools (g=0.71), high schools (g=0.86), and tertiary education (g=1.34). I could not assess gender or previous level of education as a possible moderator. None of the included studies did apply a gender specific focus and neither did studies report sufficient information on participants' level of education.

Applied technology as a moderating variable: I assumed that the mobile technologies applied in the mobile learning programmes might have an impact on the resulting programme effects. I therefore analysed the relative effectiveness of programmes using basic/feature phones, programmes using smart phone or PDAs, and programmes using tablet devices. On a glance, the meta-analysis results suggested that mobile learning programmes using more sophisticated mobile technologies have a larger effect size (g=0.73for smartphones and g=0.56 for tablets vs. g=0.18 for basic/feature phones). This observation was validated by the statistical sub-group analysis, which showed the differences in effect sizes to be statistically significant (Q: 13.1, p<0.05 for smartphones vs. basic phones; Q: 11.8, p<0.05 for tablets vs. basic phones). More sophisticated mobile devices, for example in terms of computing power and visualisation, generated systematically larger effect sizes. I was unable to assess the relative effectiveness of mobile learning programmes using a 1:1 model and shared models of device ownership due to the absence of quantitative evidence on the latter. I neither was able to use previous exposure to or perception of the applied technologies as a moderator. This information was not sufficiently reported in the included sample of studies.

Mobile learning intervention category as a moderator: I also investigated whether specific types of mobile learning programmes might be more effective in improving learning outcomes in LMICs. In this, I relied on the above-outlined ML4D intervention categories. Alas, the review did not identify any quantitative evidence on teacher development programmes or reliable numerical measures on changes in perception and adoption of mobile learning. I found a nominal difference between mobile learning programmes aiming to improve language learning (g=0.32) and programmes targeting mathematical ability (g=0.21). Yet, upon statistical sub-group analysis this difference was not significant (Q=0.94, p=0.33). Mobile learning programmes that facilitated experiential / inquiry-based learning, for example examining a local ecosystem, showed a larger effect size than both MALL and MAML (g=0.66). This difference was statistically significant and mobile learning since seems

to have been particular effective when using technology to contextualise learning experiences and allowing learners to explore content independently (Q=3.92, p<0.05).

This last finding underlined mobile learning's rationale as an innovative educational approach, which allows for a more relevant and practical learning experience through combination of different learning contexts. If this potential is fully exploited, as in the case of experiential / inquiry-based learning programmes, which specifically designed the learning experience around the combination of learner contexts (e.g. factual information about an ecosystem on mobile devices that are made accessible at the same time when learners experience this ecosystem in practice), then mobile learning in LMICs was found to produce larger knowledge gains. Finally, I was unable to assess whether learner-centred mobile learning programmes were more effective than teacher-centred interventions. Only three interventions targeted teachers as the main recipients of the mobile learning programme (Aker 2015; Kaleebu 2013; Piper 2015). The limited number of studies in this sample negated a meaningful statistical comparison with the learner-centred group of studies that featured a sample of 14 studies.

Mobile learning pedagogy as a moderator: The last potential moderator I analysed referred to the specific pedagogies which underlined the application of mobile devices in educational settings in LMICs. Of the different underlying pedagogies described in section 5.1.3, only three types prevailed in the included studies for meta-analysis (game-based n=4; context-aware n=7; collaborative n=4). A key finding in this moderator analysis was that pedagogy served as an effective moderating variable that highly reduced heterogeneity across the studies combined in each sub-group. Each of the three pedagogy-related sub-groups presented a homogenous sample (see table 5.4). I therefore conducted a more detailed narrative analysis for each of the sub-groups which is provided in appendix 5.4.

Each of the individual mobile learning pedagogies had a higher combined effect size than the overall average effect of ML4D programmes. That is, there is observational evidence that following an explicit pedagogy is in its own right a variable that increases the impact of using mobile technologies in education in LMICs. The formal statistical sub-group analysis confirms this hypothesis (Q=13.7; p<0.05). However, it is important to caution that studies using explicit pedagogical approaches applied more controlled evaluation designs in which the researchers oversaw the implementation of the innovative teaching and learning strategies enabled by the use of mobile devices. These findings therefore can only be interpreted as an indication of the importance of applying formal pedagogies supported and enabled by mobile learning. In addition, all but one of these pedagogical-rich studies have been conducted in UMICs; and, likewise, all but a single mobile learning intervention using explicit game-based, collaborative, or context-aware pedagogies have been conducted in urban areas.

Lastly, the meta-analysis results also indicated nominal differences between the combined effect sizes of different mobile learning pedagogies themselves. Using formal sub-group analysis, I found that the relative effectiveness of game-based mobile learning as compared to collaborative and context-aware pedagogies was statistically significant (Q=6.13, p<0.05; Q=6.24, p<0.05). The identified difference between collaborative mobile learning approaches and context-aware mobile learning, however, was not significant (Q=0.77, p=0.38). Based on a limited sample of evidence, the meta-analysis thus presented game-based mobile learning as the most effective approach to learning and teaching with mobile technologies in LMICs.

5.2.3 Do mobile learning interventions improve teaching practice in LMICs?

This section reports the overall impact of the use of mobile technologies in LMICs to change teaching practices in LMICs. Only four studies included in the meta-analysis attempted to measure a change in teaching practice using quantitative instruments. In three studies, this referred to observation of teachers facilitating lessons using mobile technologies and recording instances of applied teaching strategies, pedagogies, and teacher-learner interaction (Kaleebu 2013; Rosas 2002; Zurita & Nussbaum 2004). Unfortunately, only Kaleebu and peers' outcome measure was included in the synthesis as the remaining two studies only assessed teaching practices in the intervention group and not in the control group. A single study set out to investigate whether teachers' efforts had increased following the provision of mobile technologies (Aker 2012). As a result, there were only two studies eligible to contribute effect size data on changes in teaching practice into the meta-analysis. Unfortunately, neither study reported sufficient statistical information to calculate effect sizes and I am therefore limited to reporting the studies' outcomes in narrative form.

Aker (2012) investigated whether the provision of SMS quizzes and lessons supplementary to an existing adult literacy programme in Niger can increase learning outcomes. Teachers as well as adult learners were provided with basic phones and an increase in teachers' efforts given the provision of the phone was hypothesised as a possible mechanism to explain increases in learning. While the study established the latter, it ruled out a change in teachers' efforts as a contributing factor. Following the provision of basic phones, teachers

did not conduct significantly more or longer lessons as compared to teachers who received no phones.

On the other hand, in their 2013 evaluation of the SMS story programme in Papua New Guinea, Kaleebu established a significant effect of the mobile learning programme on teaching practice. SMS story provided teachers over two academic terms with a total of 100 daily text-message stories and 100 related text-message lessons. Aiming to support the instruction of students' reading abilities, teachers received tailored stories to read together with students as well as lesson plans on how to integrate and assess these stories in their daily teaching. Random visits to intervention and control schools showed that teachers in the intervention group were actively engaging with the content sent to them as text-messages and that there was a large change in the reported use of teaching strategies promoted by SMS story lesson plans. Overall, teachers using SMS story were found to be more engaged and to facilitate lessons in a more interactive and learner-centred fashion. These results were based on a proxy indicator consisting of classroom activities such as writing stories on chalkboards, reading stories to children, encouraging children to read from chalkboards, and listening to children reading to teachers one-on-one.

I also investigated the prevalence and effects of intermediate outcomes indicating potential changes in teaching practice. For this analysis, I consulted the entire sample of 20 studies included in the meta-analysis. However, there was insufficient information on intermediate outcomes related to a potential subsequent change in teaching practice to conduct a meta-analysis or narrative synthesis. Intermediate teaching practice outcomes mentioned but not investigated as part of the experimental design of studies referred to changes in teacher motivation, teacher training, perception and ease of use technology, and teacher confidence in learners' future performance. These outcomes were assessed using qualitative research methods and are therefore included in the configurative review module.

In sum, the review identified insufficient evidence to assess the aggregative impact of mobile technologies on teaching practice in LMICs. As a result, it can neither confirm nor refute the hypothesis that the use of mobile devices might alter teaching strategies and approaches. I will present further evidence on this question in the configurative review module in chapter 6.

5.2.4 Do mobile learning interventions improve EMIS in LMICs?

This section reports the overall impact of the use of mobile technologies in LMICs to support the collection and monitoring of educational information and outcomes. Only two studies included in the meta-analysis investigated the effects of mobile technologies on EMIS (Aker 2015; Cilliers 2014). Both studies assessed the impact of using mobile phones to monitor and increase teacher attendance. Aker's (2015) RCT was judged at a low risk of bias, while Cilliers's (2014) RCT was subject to a high risk of bias rating. Unfortunately, I was not able to calculate effect sizes for either study due to an absence of variance statistics being reported in the studies in reference to teacher attendance data. I am thus restricted to provide a brief narrative of the studies' results.

Cilliers's (2014) RCT in Uganda assessed the impact of different local monitoring schemes on teacher attendance. Headmasters as well as parents of pupils were provided with mobile phones to use a standardised SMS system to monitor teacher attendance at their schools. In two arms of the trial, teachers received a bonus for regular attendance based on the monitoring data. The study found mixed evidence of the effectiveness of mobile monitoring. Out of three different intervention designs only the mobile monitoring conducted by head teachers coupled with bonus payments increased teachers' attendance by 11%. There was no significant effect in the other two intervention designs. What is more, the quality of reporting using mobile phones was less reliable than in-person monitoring. Both, parents and head teachers, underreported teacher absenteeism by 14 to 18% despite the anonymous and rapid mobile tool to report attendance.

Aker's (2015) RTC presented strong evidence that mobile monitoring can increase teacher attendance. The authors further found that the increase in attendance led to a subsequent significant increase in learning outcomes of 7.5% (g=0.15; 0.07, 0.23). Adult literacy teachers received weekly phone calls by programme officers to inquire about their record of conducted lessons in the past week. There was no bonus or penalty tied to teachers' attendance and number of lessons taught. Teacher information was verified by calling adult learners part of the literacy programme. The regular phone calls led to a marginal increase of attendance: on average, teachers attended 1.27 additional days over six months. The authors therefore concluded that teacher attendance alone cannot explain the increase in learning outcomes. Rather, improved educational performance was driven by an increase in teacher motivation following the correspondence with programme officers, which was perceived by teachers as a recognition of the importance of their work. The same survey

data also revealed that while motivation increased, the weekly calls did not influence teachers' self-reported pressure, and perceived competence or choice. The mobile monitoring thus assumed a different role than theorised, being greeted by teachers as a form of motivation rather than supervision. This finding, if replicated in other studies, would question they very assumptions about the root causes for teacher absenteeism in resourceconstrained settings and the role of mobile monitoring in decreasing such absenteeism.

In sum, based on the systematic review evidence, it was not possible to arrive at an aggregate answer to the guestion of mobile technologies' effects on EMIS in LMICs. There is currently not enough rigorous experimental research evidence to answer this question. Evidence from two RCTs indicated positive effects of mobile monitoring on teachers' attendance, but the two monitoring programmes used different intervention designs and the mechanisms at play varied greatly. I am aware of two additional trials of mobile monitoring interventions in LMICs. However, the full research report of a World Bank funded RCT that has been concluded in December 2015 in Haiti is not yet in the public domain (Adelmann et al 2015)⁶⁰. In addition, past and ongoing impact evaluations of the Bridge International Academies in Kenya and Liberia have collected regular data on teacher attendance via tablet devices. These data were used in combination with other factors to determine teacher pay. Unfortunately, the NGO has not made this data or the evaluations publicly available. Given the recent publication dates of the included studies as well as the indication of largescale ongoing studies, it seems reasonable to suspect that the evidence-base on mobile monitoring in LMICs will increase in size in the near future allowing for a more meaningful synthesis.

5.2.5 Do mobile learning interventions increase empowerment in LMICs?

This section reports the aggregate effects of learning and teaching with mobile devices on empowerment in LMICs. In total, only three studies included in the meta-analysis collected outcome data on empowerment (Aker 2012; 2015; Pitchford 2014). However, in two studies the applied outcome measure of empowerment was of critical risk of bias and could therefore not be used in the synthesis. Pitchford (2014) asked 9-11 year-old learners in Malawi a 10-item survey investigating among other how many children the respondents intend to have and what they assumed their own future income to be (Pitchford 2014: 16). This data was assumed to yield insights on learners' ambition and empowerment. In addition

⁶⁰ Preliminary analysis has been disseminated through a World Bank seminar and policy brief: <u>http://documents.worldbank.org/curated/en/2015/12/25781533/haiti-can-smartphones-make-schools-better</u>

to the unrealistic assumptions underlying this outcome indicator, the control questions in the survey revealed that over half of the learners had not understood the survey questions. Pitchford's (2014) reported findings on learner empowerment in Malawi were thus excluded from the synthesis.

The second empowerment outcome measure excluded from the synthesis was reported by Aker (2012). As part of the adult literacy intervention in Niger reported above (5.2.3), the researchers attempted to measure the level of interest of adult learners in education; with level of interest being treated as an indicator of empowerment. Level of interest in education in return was assessed counting adult learners' use of a toll call hotline to express "their support for the adult education program" (Aker 2012: 114). In addition to the lengthy logic model linking collected data to empowerment outcomes, the research team was unable to collect information of the control groups' level of interest in education negating any conclusions on the attribution of the observed effect to the applied mobile learning intervention.

However, in their 2015 follow-up study of the 2012 RCT, Aker (2015) applied a more trustworthy outcome measure of empowerment consistently across both experimental groups. This study is therefore the only piece of evidence contributing empowerment outcome data to the synthesis. Following weekly phone calls to adult learners to allow them to report their teacher's attendance, learners' self-reported changes in empowerment were assessed through Rosenberg Self-Esteem Scale and the General Self-Efficacy Scale. Only findings on self-efficacy were statistically significant indicating a positive effect of the weekly calls on learners' perceived ability to increase their knowledge. This overall effect however did hide important variances in the reported data. Self-efficacy only increased in intervention communities in which learning outcomes increased too. Study areas in which learning was not enhanced reported the opposite effect on efficacy: that is, adult learners reported to have less control over the ability to increase their knowledge. The increase in self-efficacy, framed as empowerment in the study, is thus more likely to be a result of the increase in learning outcome (literacy gains in this case) than a result of the weekly phone calls per se. This is an important observation as it would suggest that if mobile learning is able to increase learning outcomes in disadvantaged settings, these increases in knowledge might support perceived empowerment too. The meta-analysis on learning outcomes reported in section 5.2.2 did indeed suggest that mobile learning is able to increase learning outcomes. However, only a minority of the mobile learning interventions included in the meta-analysis were conducted in disadvantaged rural settings comparable to the context of Aker's (2015) assessment of empowerment outcomes in rural Niger. In sum, I can only conclude, based

on the systematic review evidence, that there is an insufficient amount of research to allow for an overall assessment of mobile learning's effects on empowerment in LMICs.

5.2.6 Are mobile learning interventions a cost-effective approach to support education in LMICs?

The meta-analysis has produced rigorous statistical evidence that learning and teaching with mobile technologies in LMICs improved learners' test scores by 23.2% (g=0.47). In the context of meta-analyses of other educational interventions in LMICs this is a large effect size. However, mobile technologies—as all education interventions—require an investment in inputs. I therefore also extracted information on the cost-effectiveness of ML4D programmes. Information on input costs and rates of returns were only provided in six studies. While reported information was too heterogeneous to conduct a statistical cost-effectiveness analysis, the individual studies' cost-benefit findings presented in table 5.5 below revealed two patterns.

| Study | Intervention | Cost-effectiveness | Finding |
|-----------|------------------------|----------------------------|----------------------------|
| | | calculations | |
| Aker 2012 | SMS-supported adult | Additional costs of adding | for an additional US\$6.50 |
| | literacy campaign | SMS module to | per student, 4% more |
| | | intervention / additional | students were able to |
| | | number of learners | reach Level 1'. |
| | | reaching level A1 | |
| | | proficiency | |
| Aker 2015 | Mobile teacher | US\$ saved from in person | 'per-village savings are |
| | monitoring of adult | monitoring per village / | \$6.5, as compared with |
| | literacy campaign | SD learning gains | average gains of .20 |
| | | | SD in learning'. |
| Cole 2012 | Mobile-phone based | Comparison of | 'the cost of the |
| | agricultural extension | aggregated input cost for | intervention is quite low: |
| | | mobile and human | monthly cost of |
| | | extension service per | approximately USD \$1.13 |
| | | farmer | per farmer; the "all-in" |
| | | | costs for physical |
| | | | extension were about |
| | | | \$8.50 per farmer'. |
| He 2012 | PDA-supported | Aggregation of input costs | 'measured by academic |

Table 5.5 Overview of cost-effectiveness calculations

| | foreign language | for PicTalk machine / | gains, PicTalk's cost per |
|------------|------------------------|---------------------------|----------------------------|
| | learning: | Standard deviation | child per tenth standard |
| | | change in learning per | deviation is \$7.87 |
| | Mobile device with | child | (external) and \$3.11 |
| | interactive vocab | | (teacher). While the |
| | features & quizzes | | unadjusted per pupil cost |
| | | | is high, the gains in |
| | | | average score are |
| | | | commensurate with costs'. |
| Kaleebu | SMS messages to | Listing of input costs | 'the estimated cost of |
| 2013 | teachers to inspire | | delivering the text |
| | interactive and | | messages and |
| | learner-centred | | subsequent reading |
| | teaching | | improvement is |
| | | | approximately \$0.63 per |
| | | | child.' |
| Piper 2015 | Tablet with | Effect for each treatment | |
| | pedagogical | group / per-pupil per- | |
| | materials provided | subject unit cost of each | |
| | to: | of the mobile tools | a) \$100 leads to 7.8 |
| | a) Teacher | | students reaching full |
| | assistants | | reading fluency |
| | | | b) \$100 leads to 4.3 more |
| | b) Teachers | | students reaching full |
| | | | reading fluency |
| | | | c) \$100 leads to 0.5 more |
| | c) e-readers with pre- | | students reaching full |
| | loaded books | | reading fluency |
| | provided to learners | | . . |

First, table 5.5 indicates that all four mobile phone-based interventions were highly costeffective. Either using SMS or voice calls as a method to facilitate the intervention, the programmes had low input costs while still reporting increases in learning outcomes. Despite this encouraging findings, I caution that the meta-analysis findings suggested that these type of interventions have a significantly smaller effect on learning outcomes than more complex and pedagogy-rich mobile learning programmes. The evidence map further suggested that SMS-based interventions are primary targeted at rural, resource-constrained settings. So, while the reported interventions seem to be cost-effective, their educational impact is significantly smaller than other mobile learning interventions—an important consideration when interpreting these cost-effectiveness findings.

Second, table 5.5 also highlights that in formal educational settings investments into technologies for learners (i.e. PDAs and e-readers) are less cost-effective than investments into technologies for teachers. The provision of phones to teachers in Papua New Guinea (Kaleebu 2013) and tablets in Kenya (Piper 2015) led to large increases in students' learning at a low input cost. While 1:1 technology provision to students, too, increased learning outcomes, the financial investments required to achieve a similar increase in learning are significantly higher. Piper's (2015) trials illustrate this finding in particular: US\$100 invested in tablets for teachers led to 4 to 8 students achieving full reading fluency depending on the programme design. The same \$100 invested into e-readers for students, however, only led to 0.5 learners achieving full reading fluency.

In sum, there was some indication in the evidence-base that mobile technologies present a cost-effective tool to support education in LMICs. Mobile phone-based interventions using SMS as a delivery tool as well as teacher-centred interventions reported effective costbenefit ratios. Major investments in 1:1 models of technology delivery to students, on the other hand, were not found to be cost-effective. One needs to keep in mind though that the reported pooled effect of mobile learning was higher than the average pooled effects of educational interventions in LMICs reported in the literature, which supports the assumption that mobile technology investments can be cost-effective. However, this assumption needs to be balanced against the observation that most mobile learning programmes were by and large add-ons to existing educational efforts. That is, the technology provision built on the existing educational infrastructure (e.g. school buildings, trained teachers) and mobile learning programmes were able to exploit this established educational environment. A simple technology input cost vs. learning gains calculation therefore overlooks the initial educational investments providing a space in which mobile learning interventions can be implemented.

Conclusion

This chapter has presented the findings of my systematic review research on the aggregate effect of mobile learning on educational outcomes in LMICs. I first mapped the empirical evidence-base of research studies investigating the use of mobile technologies in LMICs to support education. Thereafter I used statistical meta-analysis to synthesise the reported numerical effects of mobile learning's impact to answer the question of what works for mobile learning in LMICs.

Starting with the descriptive map of the evidence, a diverse picture of ML4D interventions emerged. I found that most ML4D programmes were implemented in a formal educational setting and did not present independent programmes. The mere provision of mobile technologies seemed not to present a feasible programme approach to support education in LMICs. Mobile technologies assumed a supplementary role to existing educational interventions. Mobile phones were by large the dominant device applied and most ML4D interventions actively exploited the specific technological affordances of mobile technologies. Yet, the potential of mobile technologies to facilitate teaching and learning across contexts was hindered in half of the programmes due to deliberate limitations to device use in informal contexts.

The reviewed mobile learning interventions in LMICs aimed to facilitate a more learnercentred educational approach and targeted this objective through programme designs catering for and centred on learners' needs and preferences. The use of mobile technologies to support or target teachers was secondary. There was a built-in bias in the reviewed evidence to bypass teachers and other educational stakeholders in favour of targeting learners as the main users and beneficiaries of mobile learning. The most prominent ML4D approach applied MALL as a tool to support literacy or foreign language acquisition. Gamebased, context-aware, and collaborative learning strategies expressed the most common pedagogical underpinnings of ML4D interventions. Underlying pedagogies were, however, dependent on the applied technologies. Less sophisticated mobile devices, such as basic/feature phones, used less advanced teaching and learning strategies, whereas PDAs, smartphones, and tablets featured richer pedagogical models.

In the context of international development, it appeared that the reviewed mobile learning interventions overlooked the most disadvantaged groups in society. The interventions further were shaped by—rather than aimed to counter—existing structural inequalities within

education systems in LMICs. In particular, there was evidence of inequalities between rural and urban educational settings being reproduced in the educational design of ML4D programmes. Mobile learning interventions in rural areas subscribed to a less complex educational structure. Social transformation and empowerment was assessed by a minor number of programmes. Development indicators, for example official poverty lines, were not included in outcome measures of ML4D programmes.

The reported meta-analysis provided rigorous evidence of mobile learning's aggregate impact on education outcomes in LMICs. The meta-analysis established that the use of mobile technologies to support learning and teaching had a significant positive effect on student attainment. Synthesising the effects of 17 different mobile learning interventions comprising data from 17,909 participants, the meta-analysis found that there was an overall improvement of 28% in test scores between learners supported by mobile technologies and learners without access to technology. This positive and significant pooled effect of mobile learning interventions held true across contexts and various sensitivity and moderator analyses. Table 5.6 provides an overview of the key meta-analysis findings in the context of this thesis.

| Outcome | Standardised mean effect (g) | Percentage improvement in | |
|------------------------|--|---------------------------|--|
| | N of effect sizes in parentheses | treatment over control | |
| Main: | | | |
| Learning outcomes | 0.47 (17) | 23% | |
| Teaching practice | Insufficient quantitative evidence | n/a | |
| Empowerment | Insufficient quantitative evidence | n/a | |
| EMIS | Insufficient quantitative evidence | n/a | |
| Supplementary: | | | |
| Language | 0.32 (7) | 16% | |
| Math | 0.21 (4) | 10% | |
| Knowledge gains | 0.66 (8) | 32% | |
| Significant moderators | Difference in standardised mea | n Number of studies | |
| | effect | | |
| Urban context | +0.59 | 9 | |
| Rural context | larger <i>g</i> in urban areas | 5 | |
| Basic/feature phones | +0.54 | 4 | |
| Smart phones/PDAs | larger <i>g</i> for smart phones & PDA | As 11 | |
| Experiential learning | +0.34 | 8 | |
| MALL & MAML | larger g for experiential learning | g 10 | |
| Game-based | +0.53 | 4 | |
| Collaborative & | larger <i>g</i> for game-based learnin | g 11 | |
| Context-aware | | | |

Table 5.6 Overview of meta-analysis results

In addition to presenting evidence of the positive aggregative effect of mobile learning on student attainment, the meta-analysis also identified a number of variables that were associated with a larger impact of mobile learning on learning outcomes in LMICs. I found that mobile learning interventions were significantly more effective in urban than in rural areas and explained this effect due to the more complex educational and technological intervention design of urban programmes. I also established that programmes using more sophisticated technologies, which had the ability to generate richer mobile learning experiences, had a significantly larger effect size. Drawing conclusion from these two points, the meta-analysis findings suggest that pedagogies matter in the design of mobile learning interventions in LMICs. The mere delivery of content, e.g. through daily text messages (Aker 2012; Cole 2010; Kaleebu 2013) on mobile devices was significantly less effective in improving learning in LMICs. Interventions using mobile technologies to design context-
aware, game-based, and collaborative learning experiences, on the other hand, were found to be more beneficial in improving learning. I did not identify conclusive evidence on the other outcomes of interest in this systematic review. The evidence-base on mobile learning's aggregate impact on teaching practice, EMIS, and empowerment was too thin at the time of conducting this review. As a result, I can neither support nor refute the hypothesis that mobile learning is an effective tool to improve these three outcomes.

This chapter has provided evidence of mobile learning's educational potential in LMICs. The chapter was concerned with the aggregate effect of mobile learning programmes and has established that these programmes are effective to increase learning outcomes. During this investigation of ML4D's overall impact, however, a small number of important mechanisms and contexts that mitigate mobile technologies' educational and developmental potential were discovered. For example, the applied pedagogies presented a key mechanism to support mobile learning's impact and the context of reproducing structural inequalities in the design of mobile learning interventions questioned its developmental agenda.

These findings suggest that investigating mobile learning's aggregate effect alone provides an incomplete answer to the assessment of mobile learning's relevance and usefulness in LMICs. In addition to what works in ML4D, an investigation into how and why teaching and learning with mobile devices led (or led not) to education and development outcomes is required. Chapter 6 will therefore next investigate a configurative synthesis of mechanisms and contexts at play when using mobiles as an educational tool in LMICs. This synthesis will also examine whether and how the educational gains reported in the aggregative synthesis of mobile learning's effects can be linked to development outcomes. The question of whether mobile learning can be positioned to support poverty reduction, reduce inequality, and support a transformation in livelihoods cannot be answered based on the conducted aggregative review module. In this remit, the findings reported in this chapter will serve as the basis of a theory of change for ML4D, which will be fully developed using the findings of the thematic synthesis on contexts and mechanisms to zoom in on mobile learning's link to questions of development.

Chapter 6. Configuring a theory of change for the use of mobile technologies to support education in LMICs

Introduction

This chapter presents the second set of findings from my mixed-methods systematic review. It further combines both sets of review findings to develop a detailed theory of change for how the use of mobile technologies is assumed to support education in Low- and Middle-Income Countries (LMICs). This theory of change is thereafter juxtaposed with the prevailing conceptualisation and positioning of using mobile technologies as an educational tool in LMICs by international development and mobile vendor organisations. Through this juxtaposition, the chapter highlights gaps and contradictions between the understanding and positioning of mobile learning for development (ML4D) and the empirical evidence-base investigating the effects of teaching and learning with mobiles on educational actors.

The second set of systematic review findings refer to the thematic synthesis conducted as part of the configurative review module. Based on themes derived from qualitative research evidence, the results of the thematic synthesis establish contexts and mechanisms that unpack the aggregate effects of mobile technologies reported in the meta-analysis. For example, these contexts and mechanisms illustrate how technologies are perceived, how learning and teaching practices might (or might not) change, and what components of mobile learning interventions support effective educational processes.

Following the presentation of the thematic synthesis findings, I merge these with the findings of the meta-analysis to construct a theory of change for ML4D interventions. This theory of change outlines how mobile technologies are assumed to support education, and subsequently development outcomes, in LMICs according to the empirical research studies included in the systematic review. The theory of change illustrates a range of such pathways linking the educational use of mobile technologies to development outcomes⁶¹. The systematic review findings of the effects of ML4D interventions and the contexts and mechanisms configuring these effects are then plotted against this theory of change.

⁶¹ Development outcomes here refer to the definitions of development reported in the primary research studies.

In a last step, I juxtapose this theory of change for ML4D and the presented systematic review findings with the prevailing understanding and positioning of ML4D. This comparison aims to illustrate what assumptions about the effects of using mobile technologies as an educational tool in LMICs are supported by the empirical evidence and where there are gaps and contradictions between the evidence-base of ML4D's effects and technologies' perceived potential. These gaps and contradictions are then used to underline the case for applying the Capability Approach (CA) as a conceptual lens to explore the educational and developmental potential of ML4D interventions. The chapter concludes with a brief summary of the reported findings.

6.1 Thematic synthesis: contexts and mechanisms configuring the impact of ML4D interventions

The thematic synthesis explored the contexts and mechanisms that mitigate or reinforce the impact of using mobile technologies as an educational tool in LMICs. I therefore synthesised qualitative research studies that investigated the application of ML4D programmes in order to be able to unpack how and why teaching and learning with mobile devices led (or did not lead) to education and development outcomes. In this, I extracted the research findings of the qualitative primary studies, coded these findings for common descriptive themes, and then configured the descriptive themes into higher-level analytical themes. The coding framework targeted the identification of themes around contexts and mechanisms. As outlined in chapter 4, I define context as 'variables exogenous to the ML4D intervention that influence its impact'. The identification of themes followed largely an inductive coding process with a small number of a priori defined deductive codes (Online appendix 1). To recall, in order to be included in the systematic review, qualitative research studies had to meet minimum criteria related to the transparent reporting of research methodology, data collection, and data analysis.

6.1.1 An overview of the included qualitative evidence

A total of 52 studies were included in the configurative review module. The majority of studies (n=27) applied a case study design, followed by ethnographies (n=4). Participatory rural appraisal and action research were each used in only a single study. In 19 studies, the research design was described in terms of the methods of data collection such as 'in-depth interviews' or 'qualitative survey'. In general, individual interviews were the most common method of data collection (n=31) followed by observations (n=22) and survey methods (n=13) (e.g. of perceptions towards the use of mobile technologies). Ten studies used focus groups to collect qualitative data, while six studies used data collected by the applied mobile devices themselves (e.g. chat protocols, call logs). Regarding methods of data analysis, where explicitly stated, thematic analysis was the most common form of analysis (n=19) followed by content analysis (n=7), grounded theory analysis (n=4), and discourse analysis (n=2). Where interviews were conducted, the average reported number of interviews was eight (range: 3–32). Data collection predominately took place while teachers and learners were still engaged in the use of mobile technologies (n=24).

Similar to the aggregative module, studies focused on learners' experiences with mobiles

(n=36) rather than teachers' use of mobiles (n=16). This pattern further reflects in the type of ML4D interventions investigated by the qualitative studies: only eight ML4D programmes focused on teacher development. The majority of qualitative studies explored the perceptions of educational actors regarding the use of mobile technologies and their subsequent adoption (or non-adoption) (n=19). MALL and MAML were the subject of investigation in respectively 16 and 10 studies. Again, the most common device used in mobile learning programmes is the mobile phone (n=44) followed by some margin by tablets (n=8) and PDAs (n=6).

In terms of outcomes, the included studies mainly focused on qualitative changes in learning outcomes (n=38). These outcomes differed from the quantitative studies by having a stronger focus on the type of learning experiences and changes in learning practice. A total of 10 studies investigated qualitative changes in teaching outcomes. These followed a similar pattern with an emphasis in changes in teaching experiences and teaching practice. Only four studies explicitly tried to link the event of mobile learning to ideas of empowerment. These focused on adult literacy learning drawing on Freire's Critical Pedagogy and Sen's Capability Approach as well as gender empowerment outcomes (Andersson 2010; Balasubramanian 2010; Velghe 2014; Kumar 2010).

The studies included in configurative module had a stronger emphasis on intermediate outcomes than includes in the aggregative module. The majority of the qualitative studies provided rich information on the perceptions of teachers and learners on using mobiles as an educational tool. In terms of intermediate outcomes this targeted the following in particular: mobiles' pedagogical contribution (n=22); ease of use (n=20); perceived near/long-term usefulness (n=16); and technology acceptance (n=12). Reoccurring outcome measures in terms of what factors support integrating devices into educational practice referred to the importance of local content (n=7); device ownership (n=6); and teacher training (n=5).

6.1.2 How trustworthy is the qualitative evidence (critical appraisal of studies)?

As in the aggregative review module, I conducted a critical appraisal of the included research studies to ensure that primary findings used in the synthesis were trustworthy. This appraisal, however, followed different criteria more suitable for qualitative research (see Chapter 4, section 4.2.4) formalised in a detailed critical appraisal tool (Appendix 4.2). Figure 6.1 below illustrates the results of my critical appraisal of the qualitative research studies. Of 52 studies, 16 studies were subject to critical flaws undermining the

trustworthiness of the research findings, which were consequently not included in the thematic synthesis. As figure 6.1 indicates, the main reasons for exclusion from the synthesis following the critical appraisal were that the link between research findings and presented primary data was not evident (n=9) and that the research process was not reflecting sufficient quality and care (n=8)⁶². Challenges in terms of linking reported data to stated research findings and conclusions, for example referred to UNESCO's (2014) 'Reading in the mobile area' study. Despite not having collected data or conducted an analysis on the gender breakdown in the reading patterns of the Worldreader app, the study claims to identify such gendered patterns in the high downloads of romantic novels, citing an inherent preference of females for romantic fiction in Africa (UNESCO 2014: 55). The second main factor of exclusion referred to evident challenges in the technical conduct of the research itself. To illustrate, Worldreader's evaluation of its Project LEAP in Kenya was conducted by the NGO's own research manager. The report illustrates that the sample of respondents was changed during the study; that respondents were asked to estimate frequency of device use by individuals 12 months post the event; and that respondents were financially dependent on the occupation provided by Worldreader—the organisation leading the evaluation.

Following the exclusion of the 16 studies of a critical risk of bias, only 36 studies were eligible to contribute data to the thematic synthesis. That is, coding for descriptive and analytical themes was only conducted on the research findings of these 36 studies. No scaled critical appraisal was applied.

⁶² To recall, the critical appraisal questions related to 'research attends to context' and 'research is reflective' were not used to include or exclude primary studies. However, research appraised highly for being reflective of shortcomings in other appraisal areas could overrule the study's exclusion from the synthesis—though I did not identify any cases in this regard (Appendix 4.2).



Figure 6.1 Overview of critical appraisal: qualitative research

6.1.3 Thematic synthesis findings: What mechanisms and contexts mitigate and reinforce the effects of ML4D interventions?

This section reports the results of my thematic synthesis. The analysis and synthesis of the findings of each of the 36 included studies resulted in 16 analytical themes, which consist of 69 descriptive themes (Appendix 6.1). The analytical themes are separated into mechanisms for change and context themes. Taken together, these analytical themes unpack the identified effects on mobile technologies and investigate how and why these effects might (or might not) manifest. I first discuss the identified mechanisms for change before turning to the context themes.

Mechanisms for change: how and why do ML4D interventions work?

In the thematic synthesis, I identify ten mechanisms for change that present insights into what types of changes and processes are triggered through the application of mobile technologies as an educational tool in LMICs. These mechanisms to some extent unpack how the causal effects identified in the meta-analysis might be generated, but the thematic synthesis further highlights changes in teachers' and learners' behaviours that were not

identified in the quantitative impact evaluations on which the meta-analysis is based. So, some of the mechanisms for change below directly link to the findings of the meta-analysis, such as the mechanisms around pedagogies, which was identified in the meta-analysis too. Others, however, were not identified in the quantitative studies included in the meta-analysis, such as mechanism 4: teaching experience. These types of mechanisms for change illustrate the effects of using mobile technologies as an educational tool in LMICs that were not investigated and identified in the impact evaluation evidence.

The process of identifying these mechanisms for change in the thematic synthesis is outlined in detail in chapter 4, section 4.2.4 and appendix 6.2. In brief, for each included qualitative study, I coded the study findings according to reoccurring descriptive themes. For example, studies reported that learners described studying with mobiles as more interactive or more personal. Such findings were then coded in a descriptive manner as 'interactive learning' and 'personalised learning'. Column 1, in table 6.1 illustrative these descriptive themes. Having identified descriptive themes, I then configured these themes into higher-level analytical themes by investigating synergies, contrasts, and complementaries across descriptive themes. These analytical themes thus go beyond the actual findings reported in the included primary research and present the unit of analysis on which the results of my thematic synthesis are based. For example, the mentioned descriptive themes were configured together with three other descriptive themes into the analytical theme around mechanism 1: pedagogy and practice (Table 6.1).

The analytical themes were formed around mechanisms for change and context themes, which provided the overall framework to organise the descriptive themes. Behind each reported mechanism for change and context theme, there is thus a set of descriptive themes linked to the primary research findings in the included studies. It is important to keep in mind that while some analytical themes were based on a larger number of descriptive themes and primary research findings, I do not weight the analytical themes in their contribution to the synthesis. That is, none of my analytical themes (i.e. none of my mechanisms for change or context themes) is regarded as 'richer' than the others in their contribution to the synthesis. Lastly, analytical themes are not mutually exclusive and are based on overlapping descriptive themes as much as on the overlapping results of the included primary studies.

Table 6.1 provides a summary of the ten identified mechanisms for change and how they were constructed⁶³. The identified mechanisms for change can be broadly divided in three

⁶³ The full list of analytical and descriptive themes is provided in Appendix 6.1.

categories. First, mechanisms around changes in the way teaching and learning is experienced and practiced following the introduction of mobile technologies. This category presents the majority of mechanisms including mechanisms 1 to 5: pedagogy & practice; learner experience; teaching practice; teaching experience; and relationships & interactions. Second, mechanisms giving rise to a different set of socio-economic changes not identified in the meta-analysis. This category comprises mechanisms 7 to 10: social capital; economic capital; self-efficacy; and empowerment. Third, a strong and reoccurring mechanism around of the centrality of teacher training, mechanism 6.

| Reported descriptive themes based on primary studies' findings | | Mechanisms for change at play when teaching and learning with mobiles |
|--|---|--|
| | | |
| • | interactive and collaborative learning | Mechanism 1: Pedagogy & practice |
| • | authentic educational approach personalised and targeted learning increased access to information variety in teaching and learning activities | Mobile learning in LMICs can change the way education is practiced, leading to a number of more learner-centred educational approaches. |
| • | education is engaging and enjoyable | Mechanism 2: Learner experience |
| • | personalised and independent | |
| • | enhance the relevance of educational | Mobile learning in LMICs can change the |
| | experience | way education is experienced by learners |
| • | more open educational experience | open educational experience. |
| ٠ | a more learner-centred approach | Mechanism 3: Teaching practice |
| • | increased subject knowledge | |
| • | opportunities for collaboration, peer- | Mobile technologies can support |
| | support, and social learning | pedagogical and non-pedagogical |
| • | reinforce or trigger existing practices | changes in teaching. |
| | rather than establishing new practices | |
| ٠ | teaching as more enjoyable and less | Mechanism 4: Teaching experience |
| | stressful | |
| ٠ | more efficient and better organised | Mobile devices can facilitate the act of |
| • | more connected and, in rural areas, | teaching leading to a better perception of, |

Table 6.1 Summary of mechanism themes

| | less isolated | and satisfaction with, the profession (in |
|---|--|---|
| ٠ | more recognised and valued | return increasing the motivation to teach). |
| ٠ | a role of facilitating knowledge | Mechanism 5: Relationships & |
| • | shift not intrinsically connected to the | interactions |
| | use of technology | |
| ٠ | control is reason for adoption | The educational use of mobile technology |
| ٠ | some students did not agree with a | can change the relationships and |
| | shift away from teacher power | interactions between students and |
| • | tech use as exercise of power | teachers. |
| ٠ | training to be in control of technology | Mechanism 6: Teacher training |
| | use | |
| ٠ | training to focus on pedagogy and | Teacher training is key to ensure |
| | subject expertise | technology adoption and its pedagogical |
| • | collaboration and peer-support | use during teaching. |
| • | large social spill-over effects | Mechanism 7: Social capital |
| • | ability to connect | |
| ٠ | collaboration and peer-to-peer support | The educational use of mobile technology |
| ٠ | social standing | can support the accumulation of social |
| • | experience pride | capital. |
| ٠ | direct economic benefits of IT skills | Mechanism 8: Economic capital |
| ٠ | access to online job postings | |
| ٠ | support rural community development | The educational use of mobile technology |
| ٠ | intrinsic economic motivation | can support the accumulation of economic |
| | | capital. |
| ٠ | improved self-perceptions (pride) | Mechanism 9: Self-efficacy |
| ٠ | self-efficacy and aspiration | Mastery of technology can be a large |
| ٠ | community & peer recognition/status | motivation for learning with technology as |
| • | technology use as exercise of power | source of pride, confidence, self-efficiency. |
| ٠ | critical pedagogy | Mechanism 10: Empowerment |
| ٠ | women empowerment | |
| ٠ | co-construction of knowledge | A theme of using mobile technologies to |
| | | support empowerment ran through a sub- |
| | | set of studies, but was not focused on the |
| | | particular role of mobile learning in this |
| | | process. |

Mechanisms for change: teaching and learning experiences and practices The thematic synthesis generated a large variety of themes around how learners and teachers perceived education to have changed following the introduction of mobile technologies. These themes centred around the perception that mobile technologies' affordances can support the application of more learner-centred pedagogies. Qualitative evidence highlighted the ability of teachers and learners to exploit technologies' affordances such as connectivity, gamification, information sharing and visualisation to engage in more interactive and collaborative educational approaches (e.g. Hassler et al 2011; Kim 2012; Masperi & Hollow 2008). In addition, these affordances paired with others such as the portability of devices, camera and GPS, and multimedia features were linked to more authentic and situated learning experiences (e.g. Ekanayake 2013; Liu et al 2009). For example, adult literacy programmes in South Africa and India tailored mobile content depending on the domestic context of women in which they were assumed to access the educational content (Velghe 2014; Balasubramanian 2010).

Changes in educational practice also were linked to an enhanced ability to target and personalise educational lessons to the learners. An emerging sub-theme here referred to the ability to use mobile technologies to match educational content to learners' ability and rates of progression. This tracking and matching approach has been found effective in a number of impact evaluations (e.g. Chen et al 2010; Duflo 2015), but was only explicitly mentioned as a perceived contribution of mobile technologies in three studies included in the synthesis. An increased access to information facilitated by mobile technologies presented a further major change to educational practice. This was the most reoccurring descriptive theme identified in 25 studies. Teachers and learners both reported the value of having constant access to a wider range of information allowing them to extend learning beyond provided learning resources (e.g. Cole & Fernando 2012; Witt et al 2016). Taken together, these reported changes to educational practices were perceived to give rise to more interaction between teachers and learners, and between learners themselves; allow for more independent learner inquiries; and more contextualised and authentic learning experiences—each of which was assumed to support a more learned-centred educational approach.

A range of descriptive themes highlighted the perceived benefits students reported following the use of mobile technologies as an educational tool. Learners indicated a range of beneficial changes to their educational experiences including more engaging and enjoyable tasks and lessons in which, for example, 'I don't want to stop, I forget that I am learning' (Turtianined 2010). Increased personalisation and independence in the learning process

155

again were identified, as were an enhanced relevance and openness of the educational experience supporting more participation and interaction in lessons. These themes do overlap with the mechanism theme of learned-centred pedagogies⁶⁴.

Teachers, too, reported a more positive educational experiences as a mechanism through which mobile technologies support educational processes (e.g. Leach et al 2005; Onguko et al 2013). This includes a change to a more learner-centred pedagogy, but the thematic synthesis also identified a range of more pragmatic reasons. For example, teachers often reported that the use of technology made it easier to control the classroom as there was less need to lecture students because the devices kept them occupied and motivated (e.g. Sahni et al 2008). Similar pragmatic advantages include for instance: less paper work, easier marking of student papers, and ready-to-go educational content. A reoccurring benefit of using mobile technologies for teachers further referred to their ability to enhance their subject knowledge. Teachers often reported teaching subjects they had not studied for and perceived the mobile devices as a helpful tool to increase their subject expertise (e.g. Sohel & Kirkwood 2012; Wennersten & Qureshy 2012).

The reviewed primary studies reported that teachers valued the mobile devices in particular for the opportunity to collaborate, connect with peers, and engage in social learning. The ability to be connected to a wider community of educators and to be able to network and communicate with one another was a reoccurring theme (e.g. Onguko et al 2013; Sahni 2008). Teachers reported to feel less isolated, in particular in rural schools, and to feel more up-to-date (e.g. Ekanayake 2013). Collaboration with teachers in comparable contexts also served as a source for new teaching content and approaches. In sum, the thematic synthesis generated a range of themes linked to teachers' practices and experiences, not all of which were linked to pedagogical changes. An equally important mechanism associated with the use of mobile devices was that the devices can facilitate the act of teaching, which can lead to a better perception of, and satisfaction with, the profession by the teacher themselves.

Investigating the linkages between the mechanism themes around teachers' and learners' practices, there is a repeated emphasis on how the use of mobile technologies in LMICs can change the relationships and interactions between students and teachers. A range of studies provide rich evidence that the more learner-centred pedagogies can shift the educator's positon to assume the role of facilitating knowledge rather than being the holder and

⁶⁴ I provide a more critical interrogation of these findings in the discussion of the combined systematic review findings in section 6.2.2.

transmitter of knowledge (e.g. Anderson 2010; EIA 2011; Sohel & Kirkwood 2012; Sahni 2008; Hassler et al 2011). This shift is linked to teaching strategies that allow for more participation, learner-led activities, and more interaction and communication, which comes at the expense of less frontal teaching and content delivery. However, this theme was not unanimous. First, as eluded to above, teachers did not necessarily adopt more learnercentred teaching strategies for the pedagogical benefit, but for more pragmatic reasons (e.g. Kim et al 2011; Sahni 2008). Second, in a sub-set of studies, communities and students in different LMICs objected to this shift demanding the teacher to remain firmly in control of the educational process (e.g. Anderson 2010; Sohel & Kirkwood 2012). Third, there is also evidence that both teachers and learners employ mobile technologies as a tool to exercise power rather than to diffuse power relationships. Examples of this are teachers threatening to withhold devices to learners if they do not obey the teacher, or learners using the technology to dominate group work (e.g. Ale & Chib 2011; Kumar et al 2010). Fourth, in the above cited studies providing findings that support the mechanism 'Relationships and interactions', the reported shift is not necessarily linked to the application of the mobile devices. That is, the applied pedagogies do not depend on the use of mobile technologies and I would caution to draw a direct causal link between the observed changes and the provision of mobile technologies.

In summary, these five mechanisms unpack some of the positive learning effects identified in the meta-analysis. The application of mobile technologies as an educational tool in LMICs seems to be able to support a more learner-centred educational approach leading to more engaging and interactive learning experiences. These changes to educational practices, however, can also be non-pedagogical and teachers in particular applied the mobile devices to support a range of pragmatic changes to ease their teaching practices and experiences. More positive and engaging educational experiences as well as more learner-centred practices then might be able to explain the observed gains in educational outcomes. However, to be clear, these findings do not claim that there is a direct causal link between the provision of mobile technology and the observed pedagogical changes. The thematic synthesis is based on qualitative evidence and does not test hypotheses. Rather, the synthesis provides a structure on the diverse and rich *educational experiences and practices reported* in the primary evidence. *The thematic synthesis does not cover the question of attribution.*

The thematic synthesis findings I present are based on the body of evidence and need to be contextualised as such. For example, it would be presumptuous to claim that the identified themes around educational changes can be positioned to constitute mobile learning,

following the pedagogical definition by Kearney and peers' framework (Kearney et al 2010) presented in chapter 2, section 2.1.3. While across the themes, all of Kearney and peers' attributes of mobile learning are covered—that is, authenticity, collaboration, and personalisation—only a small minority of primary studies in fact apply all three attributes (Kim et al 2011; Pimmer et al 2014; Zurita & Nussbaum 2004). The assumptions and challenges this raises are discussed in detail in chapter 9, section 9.3.2.

Mechanisms for change: teacher training

The importance of teacher training in the adoption and application of mobile technologies emerged as a central mechanism in the thematic synthesis. A large body of quality evidence (n=12) highlighted different dimensions in which teacher training influences the educational changes associated with technology use (e.g. Kafyulilo 2012; Voigt & Matthee 2010). A first dimension referred to the need for teachers to be in control of the technology adoption. That is, teachers expressed a need to be proficient in the use of the mobile devices before students gained access to them. In a number of case studies, teachers actively blocked the application of mobile devices in the classroom if they did not feel in control of them (Ale 2011; Kafyulilo 2012). This findings positions teacher training as a main mechanism to support the effective use of mobile technologies. In the body of qualitative evidence that investigated such teacher training, reported findings emphasised for the training to focus on practical teaching strategies in which the technology can be embedded rather than to focus on training for ICT skills (e.g. Ekanayake 2013; Leach 2005). Within different training approaches, collaboration and peer-to-peer learning were mentioned as useful training strategies. The Digital Education Enhancement Project and the Digital StudyHall Project programmes, for example, used videos, secondment visits, and train-the-trainer strategies to let teachers observe and learn from other teachers in similar educational contexts on how to integrate mobile technologies into their teaching strategies (Leach 2005; Sahni 2008).

Mechanisms for change: socio-economic changes

My thematic synthesis identifies four mechanisms for change that describe how the use of mobiles in an educational setting is perceived to support socio-economic changes in the lives of teachers and learners. These mechanisms are based on qualitative evidence and therefore do not aim to indicate causal effects. Two sets of mechanism for change explored how the application of mobile devices might be linked to an accumulation of social and economic capital.

In terms of social capital, a consistent theme referred to large social spill-over effects of the introduction of mobile technologies. Both, teachers and learners, were reported to share the

devices and their educational uses in cases where device use was not limited to the educational setting (e.g. Voigt & Matthe 2008; Hassler 2011). Social capital could also be enhanced through using the mobile devices to connect with friends, colleagues, and relatives, which was highly valued for social reasons in particular among teachers. This also related to the ability to collaborate with and learn from other teachers, a theme discussed above already. Lastly, social capital was reportedly affected by a shift in perceptions around teachers and the educational system. The possession and mastery of mobile technologies was associated with an increased social standing of teachers. Likewise, being part of a ML4D programme including the provision of mobile devices was a matter of pride for parents and community members enhancing educational perceptions and motivation. This theme is discussed in more detail under contexts.

In terms of economic capital, four themes described how mobile devices could support economic outcomes for learners and teachers. Each of these four themes describes mechanisms bypassing the educational system. First, teachers and learners perceived the possibility to derive a direct economic benefit from their acquisition of IT skills. This related to assumptions of better employment opportunities and the ability to take part in the information society (e.g. Masperi & Hollow 2008; Ekanayake 2013). Searching for employment using mobile devices was repeatedly reported as a benefit of being able to use mobile technologies. Qualitative evidence of ML4D programmes in rural areas further identified beliefs across teachers, parents, and community members that the provision of mobile technologies could support the development of the local community and slow the urban migration of youth (e.g. Leach 2005; Sahni 2008). Lastly, a small but consistent body of evidence on adult literacy programmes unpacked an intrinsic economic rationale to be able to use mobile devices. Adult learners in Niger, South Africa, and India reported that being able to write SMS rather than having to make phone calls would increase their business activities and profit (e.g. Aker 2012; Balasubramanian 2010; Velghe 2014). Similarly, they would be able to use their phone to access business related information.

The final two mechanisms for change centre around the application of mobile devices to contribute to what could broadly be understood as processes of empowerment. There was a contrast in the identified qualitative studies between the richness of evidence underlying more granular personal changes in self-efficacy and locus of control and more formal assessments of empowerment on a more collective scale. For the latter, only a small number of qualitative studies reported linkages between the use of mobile technologies and a formal or collective process of empowerment (n=4). For example, despite the general theme that access to information can empower communities and people, my thematic

159

synthesis identified marginal evidence that pedagogies were built around this access, for example in a Freirean sense of Critical Pedagogy. Only Andersson (2010) and Balasubramanian (2010) aimed to empower communities and disadvantaged groups through the use of explicit pedagogies involving mobile devices. The same finding emerged in relation to gender empowerment. There were only three studies included in the thematic synthesis which set out the collect empirical data on how gender empowerment could be affected by learning and teaching on mobile devices. Likewise, while there were repeated suggestions in the literature that mobile devices can support a co-construction of knowledge, there were few empirical examples of this mechanism to be found in the included qualitative evidence. Instances of learners or teachers using mobiles to generate their own content were only reported in two studies (Pimmer et al 2012; 2014).

In contrast, I identify a range of themes related to changes in learners' and teachers' selfefficacy. Mastering technology usage and applying mobiles as an educational tool emerged as a source of pride, confidence, and improved self-perceptions. Related themes were identified in 13 studies. Teachers in particular considered themselves to be more 'advanced' and 'ready for the 21st century' leading to an increased professional standing (Leach 2005; Ekanayake 2013). These feeling were reinforced by community and learner perceptions of teachers, who in return reported to feel more valued as professionals and community leaders. Being able to master and teach with technologies thus became a source of pride and recognition for educators—a more granular shift in self-efficacy and agency. The same mechanism was observed for students, but based on a smaller body of evidence (n=5). All in all, however, claims to mobile learning's empowering attributes—in particular linked to empowerment through access to information-need to be cautioned. I do not find a large supportive body of evidence in the thematic synthesis or in the meta-analysis to fuel this narrative. Rather, the evidence-base seems to suggest that more individual and granular changes in teachers and learners' self-efficacy and social capital might be a mechanism through which mobile learning can link to ideas of empowerment and increased agency.

Context themes: how and why do ML4D interventions work?

The thematic synthesis further generated six context themes that unpack conditions, attributes, characteristics and other variables exogenous to the mobile learning programmes that were associated with the observed effects of technology usage. These context themes identify conditions that can facilitate or block the perceived educational and developmental impact of mobile learning programmes. Table 6.2 presents a summary of the six identified

context themes as well as the descriptive themes that informed them 65 .

| Re | eported descriptive themes based on | Contexts influences the teaching and |
|----|---|---|
| | primary studies' findings | learning with mobiles |
| | | |
| | | |
| ٠ | ubiquity & familiarity with mobiles | Context 1: Positive perceptions |
| • | convenience & blend in daily routines | |
| • | produce new and support existing | Mobile devices' particular affordances can |
| | forms of communication | nurture a general positive perception of |
| ٠ | affordability of devices | mobile technologies among educational |
| | | stakeholders in LMICs. |
| ٠ | ease of use | Context 2: Feasibility |
| • | interface design and space for | |
| | personalisation of devices | Mobile devices are technically able to |
| ٠ | technical queries relate to issues with | serve as educational tools in LMICs. |
| | hardware | |
| ٠ | respect and recognition from peers | Context 3: Technology as a social and |
| | and function as peer educators | professional status symbol |
| ٠ | pride of participation and higher | |
| | perceptions of teachers and schools | Mobile devices are intrinsically associated |
| • | teachers feel more professional and | with status and professionalism leading to |
| | valued | higher perceptions of schools / teachers |
| • | teachers express higher opinions of | who possess the devices. |
| | students | |
| ٠ | user training and teacher training | Context 4: Factors of adoption |
| • | integrated in the existing curriculum | |
| • | power issues on domination | While mobile technology as an educational |
| • | intrinsic economic motivation | tool is acceptable to teachers, learners, |
| | | and parents in LMICs, supporting factors |
| | | have to be in place to nurture its adoption |
| | | as an educational tool. |

Table 6.2 Summary of context themes

⁶⁵ The full list of analytical and descriptive themes is provided in Appendix 6.1. Context themes were identified following the same process as for the mechanism themes.

| use of limited by ial |
|------------------------------|
| use of limited by :ial |
| use of limited by :ial |
| limited by al |
| ial |
| |
| |
| |
| es |
| |
| atively |
| /IL4D |
| |
| es atively ЛL4D |

The six context themes include four themes linked to intrinsic attributes of mobile technologies and how they are perceived by educational actors in LMICs. These refer to contexts 1 to 4: positive perception; feasibility; technology as a social and professional status symbol; and factors of adoption. In essence, these four contexts explain how and why mobile technologies can experience rapid adoption as an educational tool. The themes also present the background to the mechanisms related to enhanced self-efficacy and social capital. The remaining two context themes refer to two key barriers to mobile technologies' impact on education in LMICs: informal use limited for learners and systemic challenges, contexts 5 and 6. To keep in mind, a range of contexts have already been included in the meta-analysis as moderator variables, for example socio-economic settings, and are therefore not discussed again in the thematic synthesis.

Contexts: adoption of mobile technologies as an educational tool

A key question in my thematic synthesis was how and why mobile technologies find adoption as educational tools in LMICs. Often reports about the use of mobiles in schools reference strong resistance among teachers and parents, ranging to outright bans of mobile devices in some instances (Trucano 2015; DBE 2015; Beland & Murphy 2015). I therefore explicitly coded the qualitative studies included in the thematic synthesis for any factors explaining uptake or rejection of mobile technologies in educational contexts. This led to four analytical themes explained below, but also established a strong and consistent finding that, on aggregate, the qualitative evidence reported barely any instances of non-adoption or rejection of mobiles. Of 30 studies providing qualitative data on the adoption of mobile technologies, only two studies (Ismael 2012; Kafyulilo 2012) reported a rejection of mobile devices by educational actors in LMICs, teachers in both cases. The empirical evidence-base thus suggest little support for popular narratives of the rejection of mobiles in education in LMICs.

Three main contextual themes might explain the adoption of mobiles. A first contextual theme focuses around pre-existing positive perceptions towards mobile technologies in general by teachers, learners, and parents in LMICs. A range of mobile devices' particular affordances seem to nurture these perceptions. For example, parents and teachers repeatedly referred to the ubiquity of and familiarity with mobiles (e.g. Liu et al 2010; Ale & Chib 2011). The technologies were not perceived as a foreign educational input but rather as an artefact that parents and teachers felt familiar with and used in their daily interactions themselves. The latter was emphasised regarding the ability to use mobiles to stay in contact and communicate with students and teachers (e.g. Jantjies & Joy 2014).

A second linked contextual themes highlighted that there was little reservation across teachers, learners, and parents regarding the technical feasibility of using mobile technologies for educational purposes. A reoccurring theme identified in 27 studies referred to the ease of use of mobiles. Learners and teachers were already familiar with mobiles and how to operate them, leaving few reservations among these groups that this use can be adapted in an educational context. Design aspects supporting such adaptation referred to the interface design of the educational application as well as the ability to personalise devices, for example through covers and individual log-ins (e.g. Leach 2005; Hassler 2011). Reported barriers to the use of mobiles mainly were caused by problems with the hardware of devices, such as battery life, screen size, memory storage, and breakage due to wear and tear. Overall, however, there was little qualitative evidence refuting the technical feasibility of using mobiles to support education in LMICs.

A third contextual theme highlighted the intrinsic association of mobile technologies with professionalism and success in LMICs. That is, ownership of mobile devices was seen as a social and professional status symbol. This theme was consistently found across the included qualitative studies (n=21) independent of the setting in which the mobile technologies were applied. Given mobile technologies role as status symbols, ownership and mastery of the devices therefore generated a range of social and professional benefits. Teachers in particular reported to feel more valued; both, in terms of being provided with devices akin to other government employees or 'office workers' (Leach 2005; Masperi 2008), and in terms of how local communities and learners perceived them. A common theme

referred to 'they [learners and community members] see us as professionals now'. Teachers thus reported to feel more respected and recognised as professionals, which was linked to enhanced self-perceptions and job satisfaction.

Further, the social and professional benefits of using mobile devices were not limited to teachers only. They likewise extended to learners and schools. For example, both teachers and parents expected students to perform better in school if mobile devices were used as teaching and learning aids. Parents in particular reported a preference to send their children to 'technology schools' (e.g. Masperi 2008; Voigt & Matthe 2012). As a result of the intrinsic high professional associations of mobiles, a virtuous circle of positive perceptions can be in place. This reinforced the adoption of technologies and, too, supported important social spill-over effects of device ownership.

Finally, despite these three context themes supportive of mobile technologies adoption, the thematic synthesis also identified a range of other supporting factors that influence the adoption of mobiles as an educational tool. Teacher training as already mentioned presented a key theme, but so did user training for learners. The latter was not so much focused on device literacy but rather on the responsible use of mobiles in an educational setting (e.g. Turtiainen et al 2010). The integration of the mobile learning programmes into the existing curriculum presented a second major factor of adoption (e.g. Kim et al 2011; Wennerstein 2012). Mobile learning content that did not follow the current curriculum and/or did not fit local contexts, such as local English accents, was often rejected (e.g. Masperi 2008; Wennerstein 2012). A linked theme again referred to teachers' need to be in control of the educational use of the technology. For instance, learner-centred and interactive teaching strategies in which the teacher is still in charge of the learning process (e.g. group work and presentations aided by mobile devices) found stronger adoption than strategies in which the teacher can be challenged or does not set the learning objective. In sum, the thematic synthesis therefore identified three contextual themes that support an intrinsic acceptance and adoption of the technology, but these are mitigated by specific educational contexts and needs.

Contexts: barriers to mobile technologies' impact on education in LMICs

In the thematic synthesis, I identify two context themes that provide rich evidence on factors mitigating the effects of teaching and learning with mobiles in LMICs. These two barriers are present across different settings and are likely to be faced by most ML4D interventions. First, in the large majority (n=20) of the mobile learning programmes included in the thematic

synthesis⁶⁶, the mobile and contextualised use of technologies is determined and limited by formal educational actors and social structures. That is, teachers and parents set strict boundaries around learners' use of mobile devices if they have the power to do so. This can lead to large restrictions on the learners' abilities to exploit the mobility of the devices and potential mobility of the learning experience. While teachers themselves report to value the ability to access learning materials 'anytime, anywhere' (e.g. Ekanayake 2013; Shohel & Kirkwood 2012), they rarely permit or encourage their students to use mobile technologies unsupervised during formal educational activities. Where such unsupervised mobile learning events occur, they are explicitly built into the mobile learning intervention design (e.g. Kumar 2010; Ekanayake 2013).

Teachers cite fear of theft and loss of devices as well as concerns that learners would not focus on the educational task as reasons for limiting device usage to formal classroom settings (e.g. Voigt & Matthe 2008; Kim 2011). For parents, fears of a financial loss of having to replace the devices in case of damages or thefts as well as fears over children accessing illicit online materials were the main reasons for rejecting an unsupervised use of the devices. Given the results of the meta-analysis and the thematic synthesis on mechanisms, in which rich mobile learning pedagogies that facilitated more learner-centred and contextualised learning approaches led to larger educational outcomes, this limitation to a mobile use of the devices across contexts presents a major barrier. Many attributes and benefits of mobile learning remain unlocked if the devices can only be used in a formal classroom setting.

A second contextual key barrier identified in the thematic synthesis refers to systemic challenges within the wider education system. Almost half of the studies included in the thematic synthesis (n=16) mentioned systemic obstacles that affected the ML4D intervention. The most frequent cited obstacles to effective programme implementation were teacher transfers (n=9), teacher strikes (n=8), and teacher absenteeism (n=8)—underlying the centrality of teachers to the use of mobile technologies (as well as the education system in general). None of the mobile learning programmes subject to these systemic challenges was able to use the technologies to circumvent these challenges. This finding highlights the extent to which programmes were embedded into the existing education systems and therefore subject to the system's functioning. There was an absence of empirical research evidence on ML4D interventions that attempted to influence systemic issues within

⁶⁶ But, interestingly, of the 20 studies included in the meta-analysis only five apply such restrictions. This is linked to the larger degree of control that researchers had over the intervention design in most of the impact evaluations included in the aggregative review module.

education systems in LMICs. In the light of this absence, the context of the existing education system is a major determinant of the ability to engage mobile technologies as an educational tool in LMICs.

Non-themes: deductive themes for which no evidence was identified

In the thematic synthesis, I also set out to explore a number of deductive descriptive themes that were specified a priori based on common themes in the literature on ML4D. Here, I briefly report on some of the deductive themes for which I did not identify any systematic empirical evidence in the included qualitative evidence. I refer to such themes as 'non-themes' and an exhaustive account of them is provided in Appendix 6.1. To be clear, I do not claim that the absence of evidence on these themes implies an evidence of absence. For various reasons the available research evidence might not have focused or reported empirical data on these themes.

In total, there were 12 themes for which I found no empirical data in the included studies. Most notably among the non-themes linked to contexts was that I did not identify any systematic empirical evidence that different groups or actors are able to exploit or access mobile devices differently in formal education. I explicitly attempted to code the included studies for gender-specific usage of technologies; age-related difference in use; and socioeconomic and ethnic-related differences. In mobile learning programmes in formal education structures neither of the above themes were reported. This differs from mobile learning programmes in informal education programmes. Here, I do find that gender emerges as a determinant of access to devices. However, this is based on a small number of studies (n=3) and is linked to mobile learning programmes that have an explicit focus on gender empowerment in general. Across the body of evidence available, therefore, little data reported differences in access and usage of mobile technologies.

There was also no systematic empirical evidence to indicate that mobile technology presents a mechanism to access otherwise inaccessible education services, that is the use of mobile devices as a substitute for formal educational structures. Only two studies linked to adult learning in informal contexts (Balasubramanian et al 2010; Velghe 2014) used mobiles as a formal replacement for education services. This links to the non-theme of systemic change mentioned above already. There was no systematic empirical evidence on mobile learning's effects on systemic change in education in LMICs. This finding might be partly related to the scale of most mobile learning programmes assessed in the thematic synthesis, which usually were focused on single schools. It further reflects, however, a lack of engagement with structural and systemic issues in the design of most ML4D programmes included in the synthesis. For example, I did not identify any studies on ML4D programmes focused on leadership in or management of education systems in LMICs. Systemic change, it seems, was not within the scope and objective of the evaluated programmes.

6.2 A theory of change for ML4D drawing on my mixed-methods synthesis

Following the meta-analysis as a method of synthesis in the aggregative review module and the thematic synthesis as a method of synthesis in the configurative module, I next combine both syntheses findings in a mixed-methods synthesis. In this, I merge both sets of synthesis findings to construct of theory of change of how the educational use of mobile technologies in LMICs can be linked to education as well as development outcomes⁶⁷. This theory of change thus reflects all the synthesised quantitative and qualitative research evidence on the empirical effects of using mobile technologies as an educational tool in LMICs. It thereby provides my answer to the question whether the empirical and conceptual claims to mobile learning contribution to development are warranted (c.f. chapter 2, sections 2.2–2.3)?

6.2.1 Developing a theory of change for ML4D interventions

The constructed theory of change is provided in Figure 6.2. From left to right, it follows the changes in educational settings in LMICs post the provision of mobile technologies as educational tools. It therefore tracks the immediate changes, intermediate outcomes, final outcomes, and impacts of the mobile learning programmes. In this, the theory of change aims to assess to what extent the empirical systematic review evidence supports a link between the application of the mobile learning programmes (far left) and the impacts on development (far right). In other words, it aims to assess the empirical and conceptual validity of the 'for' in Mobile Learning 'for' Development. Development here refers to the concept of development reported in the synthesised primary literature on which the theory of change is based.

The theory of change tracks three conceptual pathways starting from the provision of mobile devices: devices targeted at (1) learners, (2) teachers, and (3) administrators. These three conceptual pathways directly reflect the ML4D intervention designs reported in the 70 interventions included in my systematic review. Intervention pathways are not mutually exclusive and overlap multiple times within the theory of change. Blue arrows on the theory of change indicate direction of effect or influence. For example, a change in how learning takes place (intermediate outcome) influences increases in learning outcomes (final outcome). Green tick boxes indicate that the empirical results of the meta-analysis confirm the presence of an outcome. The mechanisms for change identified in the thematic

⁶⁷ It is important to keep in mind that the theory of change is based on the included primary research studies and its pathways and links to development outcomes thus reflect the assumptions of the primary literature and how these studies have conceptualised development.

synthesis are plotted around the theory of change in green boxes, while the context themes are represented by yellow boxes. Taken together, the theory of change thus combines all my empirical systematic review results and visualises the interplay and connection between different sets of findings and their relation to the applied ML4D interventions.



To provide an illustration of how the theory of change diagram can be interpreted: starting at the provision of mobile devices to, say, learners, this can result in an immediate new access to technological resources as well as new access to educational resources. Such access could result into more learning opportunities, or changes to how learning is taking place or is experienced. This, though, depends on changes in teaching practices, which in return are dependent on a range of mechanisms being in place. At this point, the meta-analysis only provides causal evidence for changes in learning to take place dependent on the applied pedagogy. The intermediate outcomes for the learners then might translate into actual increases in learning outcomes, again dependent on a range of mechanisms. The meta-analysis confirms that such increases in learning outcomes are observed in most mobile learning programmes. Alas, while targeted in the mobile learning programme design, there is currently no empirically verified link between changes in learning outcomes and impacts on human capital, employment, and economic growth—the impacts mentioned in the ML4D literature. Finally, each of these linkages is influenced by the context themes at the bottom of the theory of change.

As introduced, the main question is then whether the theory of change supports the assumption that mobile learning programmes can be linked to development outcomes as stated in the literature on ML4D? In the next section I am applying the developed theory of change to answer this question. I will discuss the systematic review results that the theory of change reflects and the implications of this review of ML4D evidence-base for the conceptual and empirical claims on mobile learning's role and contribution to international development.

6.2.2 Putting the theory of change to work: assessing Mobile Learning 'for' Development?

The findings of my mixed-methods systematic reject a link between using mobile technologies as an educational tool in LMICs and development outcomes as assumed in the literature on ML4D. Having systematically reviewed and synthesised both quantitative and qualitative studies investigating the effects mobile learning programmes in LMICs, I provide four linked sets of findings to support my argument that, in its current application, *mobile learning programmes cannot claim to support development outcomes*. First, I will highlight the evidence gaps on my theory of change related to development outcomes. Second, I will illustrate that current observed positive educational effects are not sufficient to claim a meaningful change in educational outcomes. Third, I will highlight that the design of mobile learning interventions is subject to two structural biases, which undermine the positioning of

the intervention to target development outcomes. Fourth, I will unpack how the current conception of development outcomes itself hinders the linkage between mobile learning and development outcomes.

The missing link: an evidence gap on development outcomes

The developed theory of change highlights an evidence gap on development outcomes. Looking at the impact column on Figure 6.2, I find an absence of evidence supporting empirical links between the observed education and social outcomes and impacts on development. The included mobile learning interventions applied three pathways in order to influence development impact. The first assumed that increases in learners' educational attainment could result in an increase in human capital, which then would enhance their employment prospects. Income-based or livelihood-based poverty and development outcomes were not targeted or measured in the included interventions. This pathway was the most popular narrative with 22 studies explicitly referencing it. However, as indicated on the theory of change, I did not identify any quantitative or qualitative evidence in either synthesis supporting it.

The second pathway assumed that an increase in learners' educational outcomes might be linked to empowerment outcomes. For example, for girl learners enhanced education might have a more intrinsic value of increasing their ability to take control over their lives. Again, however, neither of my two syntheses identified systematic evidence supporting this pathway. In addition, only a small set of studies across both review modules (n=8) explicitly referenced this pathway to development. The same applies to the third assumed link between the provision of mobile technologies in educational settings and potential development outcomes: using mobile technologies to empower learners and parents to monitor and enforce educational processes and standards. This pathway was only referenced in three studies, with none of the studies reporting supporting empirical evidence. There was no pathway reported between providing teachers with mobile devices and assumed development outcomes.

In sum, a simple record of the available empirical evidence clearly indicates an absence of evidence between providing mobile technologies as an educational tool in LMICs and impacts on socio-economic development as assumed in the literature. Again, I caution that an absence of evidence does not refute the assumption that such an impact could potentially take place. The current empirical evidence included in my systematic review, however, suggests that claims to mobile learning's impacts on development are currently not based on the available research evidence.

The size and nature of mobile learning's educational impacts

My theory of change for ML4D plots the positive results of the meta-analysis along a range of educational outcomes. Given the positive overall effect size of 0.47 SMD (23.2% change over the control group) on learning outcomes, one might argue that this impact translates into meaningful changes into learners' abilities and subsequent development outcomes⁶⁸. It is therefore crucial to contextualise this identified effect of a 23.2% change in educational attainment in order to be able to interpret whether it indeed reflects a transformational potential of teaching and learning with mobiles. A key question here is the baseline of the effect as well as the functionality of the gained educational outcomes. To assess both questions, I will focus on the seven studies of the lowest risk of bias, which too present the largest impact evaluations in the meta-analysis and drive 97% of the pooled effect (Aker 2012; 2015; Cole & Fernando 2012; He 2008; Kaleebu 2013; Piper 2015; Pitchford 2014). A narrative summary of all seven studies is presented in Appendix 6.3 for transparency.

Five out of the seven studies focus on mobile learning programmes to increase literacy outcomes (Aker 2012; 2015; He 2008; Kaleebu 2013; Piper 2015)⁶⁹. In three studies the programme design included SMS-based lessons and exercises (Aker 2012; 2015; Kaleebu 2013), while Piper applied a range of tablet-supported mobile learning strategies, and He used the PicTalk machine learning device. All mobile learning programs lasted between 6 and 24 months and all five programmes report positive effects on literacy outcomes. However, upon closer examination these positive literacy effects cannot be regarded as transformational learning outcomes that enable learners to enhance their human development. In fact, in none of the five experiments does a majority of learners experience meaningful shifts towards functional literacy.

In Kaleebu's (2012) impact evaluation, 20% of the children part of a 7-months mobile learning intervention still score an outright zero on endline literacy tests and the mean score on a scale of 0–10 has increased from 3.7–5.6. Likewise, in Aker's (2012) study of a mobile learning programme in Niger roughly 40% of participants improve their literacy levels from A0 to A1 after a two-year programme. In her 2015 experiment, a similar finding emerged with participants experiencing roughly a 20% change (0.15 SD) in reading abilities with a

⁶⁸ This would further assume that education outcomes in LMICs equate development outcomes. However, I would argue that education cannot be equated with development per se, and rather presents an integral component of and driver for development. For example, education outcomes are a *subset* of the HDI and other multi-dimensional poverty indexes. Therefore, education, and learning outcomes in particular, should not be equated with development outcomes.

⁶⁹ He et al (2008) and Piper et al (2015) in fact focus on literacy and maths outcomes, but I report their results on literacy outcomes here only.

baseline mean value of 3.4 on a 12-point test score. Piper's (2015) 10-months tablet programme in Kenya led to a slightly more positive but still marginal effect. Post intervention, 13.5% of pupils were able to attain the 65 correct words-per-minute benchmark threshold established by the Kenyan Ministry of Education Science and Technology. Broken down per \$100 invested, this translates into an average number of 0.5 students reaching full reading fluency post intervention⁷⁰. Lastly, in He's (2008) two-year implementation of the PicTalk programme, baseline data collection on English literacy scores using a standardised assessment had to be discontinued due to an insufficient number of students understanding the assessment. The authors then resort to using students' recognition of English words as an indicator of literacy. Unfortunately, baseline mean results for this outcome measure are not reported, but it seems reasonable to assume that the ~30% change (0.25 SD) commenced from low word recognition rates.

The observed positive educational impact of mobile learning programme in LMICs therefore can barely be regarded as to constitute transformational changes and to support learners' educational functionings significantly. To be sure, I do not refute the results of the metaanalysis which provided rigorous of evidence of a strong and consistent effect on learning outcomes. Mobile learning does indeed increase learning outcomes in LMICs. However, contextualising the observed impacts highlights that the positioning and interpretation of these positive effects need to be cautioned. The empirical evidence does not suggest that the generated learning gains support structural or transformational changes in learners' abilities. Rather, the observed effects—as the applied mobile learning interventions themselves—are firmly embedded in the overall structure of education systems in LMICs. In contexts where these systems are unequal or underperforming, the introduction of mobile technologies as educational tools cannot claim to positively influence the properties and outcomes of these systems.

Mobile learning for whom?

For mobile learning in LMICs to claim potential to support development outcomes, it is important to consider who has access to and is partaking in the mobile learning programmes. Based on the combined systematic review findings, two structural biases in who is included in mobile learning programmes undermine the positioning of the mobile learning interventions to target development outcomes. First, the reviewed ML4D interventions overlooked the most disadvantaged groups in society; and, further, the

⁷⁰ I am unable to provide similar cost breakdown for the remaining studies, due to an absence of reliable empirical data on cost and/or cost-effectiveness.

interventions were shaped by—rather than aimed to counter—existing structural inequalities within education systems in LMICs. Mobile learning programmes in LMICs clustered largely around urban areas and socio-economic indicators suggest that more affluent schools are targeted as hosts for mobile learning programmes in formal education. For example, flagship ML4D programmes such as Worldreader or Mobile Math each focus on learners in more affluent schools with a pre-existing track record of providing high-quality education (Worldreader 2012; Roberts & Vänskä 2011).

Of the small amount of mobile learning programmes that targeted education in rural areas, the vast majority applies less sophisticated mobile technologies (largely basic and feature phones) and less complex programme designs (largely mere information dissemination). As a result, mobile learning interventions that included more disadvantaged populations are not as pedagogical rich as their counterparts in more affluent and developed contexts. This directly translates into observed programme effects, which are smaller in rural areas, due to the moderating effect of the applied pedagogies on learning outcomes. This in-built inequality in programme design also affected the few mobile learning programmes that explicitly targeted social transformation and empowerment of disadvantaged populations (e.g. Andersson & Hatakka 2010; Balasubramanian 2010). These programs exclusively made use of basic and feature phones limiting the mobile learning design to applying SMS-based learning and information sharing exercises. In short, mobile learning programmes in more disadvantaged contexts generally applied the least advanced technologies and intervention designs. This finding challenges the positioning of mobile learning as a tool to support inclusive development for all groups in society.

The second structural inequality in mobile learning intervention design is the exclusion of teachers and communities. As indicated in the systemic review findings and reflected in the theory of change, the large majority of mobile learning programmes assumed learners as the population of interests. Supporting teachers or communities through the use of mobiles is a marginal programme component. In fact, the theory of change does not even plot a link between provision of technologies to teachers and development outcomes. And, as explained above, the link between providing mobiles as a monitoring tool to community members to empower them to enforce educational standards and processes was based on only three studies.

It is unclear why these key educational stakeholders are marginalised within the current mobile learning programmes in LMICs⁷¹. Though, the implications of this marginalisation are evident. It results in a strong human capital narrative to development. That is, if mobile technologies can support learners to obtain higher educational outcomes, this then translates into increases in human capital and subsequent socio-economic development. It is this narrative that strongly dominates the literature on which the theory of change was developed. However, as illustrated above, there is little empirical evidence supporting this narrative. In the context of this absence of evidence, the marginalisation of teachers and communities in mobile learning intervention design requires attention. It seems current mobile learning programme overlook a key link between technologies' affordances, educational changes, and potential improvements in livelihoods.

What counts as development in ML4D?

The available evidence-base on mobile learning and development seems to apply a narrow interpretation of what constitutes development⁷². The main understanding of development is linked to ideas around human capital and subsequent increased employment prospects and productivity. This narrative of human capital and economic growth as development is prevalent across much of the wider ICT4D literature of which the mobile learning literature is a subset—so this finding does not particular surprise (Castells & Himanen 2014; Unwin 2009). In ICT4D, however, there have been strong critiques of this human capital narrative, which I would argue apply similarly to ML4D (Kleine et al 2009; Zheng 2009; Oesterlaaken 2012; Gigler 2015).

The strong focus on human capital as a driver of development has led mobile learning programmes to overlook alternative pathways to development. It further has shaped a linear intervention design, focused on increasing learning outcomes in formal education. It is not so much that the human capital pathway per se is problematic; rather it is its dominance within the applied programme designs which crowds out alternative pathways to, and understandings of, development. I already have illustrated this in reference to the marginalisation of teachers and communities within the reviewed interventions. But, it is apparent within the positioning of empowerment processes and outcomes in the included studies too.

⁷¹ Though I provide potential reasons for this in Chapter 2, section 2.3.3.

⁷² Here, I directly critique the existing definitions and understandings of the concept of 'development' within the reviewed ML4D literature. This highlights the discrepancy between my personal definition of development following Sen and the definitions of development that my systematic review of ML4D literature empirically assessed.

The small amount of evidence on the link between the educational use of mobile technologies and empowerment outcomes applies a linear link between increases in learning outcomes and processes of empowerment that mirrors the positioning of learning outcomes and increases in human capital. Within the body of evidence on empowerment, studies assume and investigate a direct relationship between increases in literacy test scores and empowerment measures such control over one's life and changes in gender norms. This appears to reflect a naïve view of empowerment processes and outcomes. Empowerment is exclusively positioned as the direct consequence of enhanced literacy abilities. This shapes directly how these mobile learning programmes are designed with no provision made for the political and cultural processes reported as crucial in the literature on empowerment in development (e.g. Aslop & Heinsohn 2005; Kabeer 2010).

It appears then that current conception of empowerment as development in ML4D is too narrow and still subject to a linear focus on increases in learning outcomes as a key mechanism. The systematic review presents a range of different mechanisms linked to personal empowerment that could enrich the link between the provision of mobile technologies and concepts of empowerment; but these mechanisms are currently not integrated into mobile learning programme designs and largely emerge from qualitative studies of teachers and learners' perceptions on using devices. These refer in particular to the mechanism themes of self-efficacy, social and economic capital, and the context theme of technology as a status symbol. All of these systematic review findings point towards more granular and personalised changes and benefits that users of mobiles can derive from the devices' integrations in their social and professional lives. These changes are mainly linked to non-education outcomes such as feeling valued and respected or being able to enhance one's standing in the community. These mechanisms and outcomes seem to be closer connected to the affordances of mobile technologies that led to mobiles' adoption in LMICs in the first place, for example, connectivity, communication, and mobility. It therefore seems that an enhanced conception of what constitutes 'development' in ML4D might allow for a range of additional pathways to emerge that can link the provision of mobiles in educational settings to socio-economic outcomes. In other words, the definitions of development within the reviewed primary research are too narrow to capture the diversity of pathways and mechanisms between the use of mobile technologies, education, and human development and well-being.

In sum, my mixed-methods synthesis and theory of change finds no evidence that supports the positioning of mobile learning as a tool for socio-economic development in LMICs. *Development outcomes claimed in the ML4D literature such as poverty reduction, reduction*

in inequality, and transformations in livelihoods cannot be empirically linked to the applied mobile learning programmes. Further, observed positive effects in learning outcomes, on their own, do not translate in meaningful changes in educational functionings. Structural inequalities within mobile learning intervention design and a narrow conception of development as an increase in human capital also hinder mobile learning's potential to support development outcomes.

Empirically, it is therefore challenging to position mobile learning as a development intervention—let alone to claim a mobile revolution to be taking place in education in LMICs (UNESCO 2014). Some promising areas of where links between mobile learning and development could occur remain under explored. This refers to teachers' use of mobile technologies; changes in self-efficacy and personal and professional development; changes in social capital; and community-based monitoring.

6.3 What have we learnt from the evidence-base? Discussion of systematic review findings

My mixed-methods systematic review presents empirical evidence on the effects of using mobile technologies as an educational tool in LMICs. I next integrate the findings of the review into ongoing debates and discourses on ML4D. This aims to highlight gaps and overlaps between the systematic review findings and the wider literature. It also aims to flag potential implications for current and alternative conceptions of mobile learning as a development intervention.

6.3.1 A mobile revolution? Claims to ML4D's impact

As referenced in Chapter 2, section 2.2 the discourse on the impact of using mobile technologies to support education in LMICs features a strong assumption that mobiles can fundamentally alter education outcomes and achievements. This discourse is shaped in particular by international development organisations and mobile vendors and is perhaps best captured by UNESCO's proclamation of a mobile reading revolution:

"There are strong indications that the benefits of mobile reading are long-lasting and farreaching, with the potential to improve literacy, increase education opportunities and change people's lives for the better. A *revolution in reading* is upon us thanks to the massive proliferation of mobile technology [in LMICs] (...)" (UNESCO 2014: 83, emphasis added).

However, the UNESCO is not the sole organisation using research findings attributing large transformational impacts to mobile learning in LMICs in order to advocate for an increased investment in educational mobile technologies. The GSMA, for example, too claims that "a revolution is coming" based on the power of mobile to change education in LMICs (GSMA 2012a: 11). In support, the organisation references research from India in which mobile phone-based educational games allegedly raised learning outcomes by 60%. Likewise, the World Economic Forum states in the aptly titled 'accelerating the adoption of mLearning' report that "(m)obile learning represents an opportunity to systemically redefine the way that individuals and communities can contribute to society" (WEF 2012: 3). This report is based on qualitative research conducted by the GSMA in which youth in LMICs were interviewed regarding their perceptions on mobile learning. This qualitative study claims that "mLearning has a crucial role to play in improving the education and life prospects of young people in emerging markets [LMICs]" to then conclude with a call for "mobile industry, international development community and governments to collaborate and create services that will have a

profound, lifelong impact on the lives of young people" (GSMA 2012b: 4).

My systematic review findings refute these claims and provide reliable evidence that such wide-ranging claims to mobile learning's impact on education and development in LMICs are not warranted. The UNESCO (2014) and GSMA (2012b) studies were in fact excluded from the systematic review due to being of a critical risk of bias. For example, the underlying empirical data in the UNESCO study announcing a reading revolution finds that smart phone users in a range of African countries were logged into the Worldreader application on average for one minute (male sample) to seven minutes (female sample) per day. Further, the cited research findings on mobile learning from India in the GSMA (2012a) report are misrepresented. The study, He and peers (2008), is included in my systematic review but suggests a change in outcomes of between 0.25-0.35 standard deviations, which is based on a very low baseline level of literacy. To arrive at the 60% figure, the report authors of the GSMA report seem to have used the upper-bound confidence interval of a sub-group outcome in year two of the study.

In contrast, the findings of my systematic review seem to echo a range of existing reviews and academic studies. Wagner and peers' (2014) Mobiles for Reading landscape review finds a "lack of solid evidence of [mobiles] effectiveness" (2014: 9), which overlaps with the findings of my own review on development outcomes. The 2016 Word Development Report equally bemoans a lack of reliable evidence on the question of mobile technologies' impact on education and development in LMICs (World Bank 2016: 146). The same conclusion is reached by Tamin and colleagues (2015) in a review of large-scale, government-supported educational tablet initiatives in which the authors conclude that "the majority of these initiatives have been driven by the tablet hype rather than by educational frameworks or research-based evidence" (Tamin et al 2015: 2).

Tamin and peers (2015) findings are particular interesting as they explicitly reference the discrepancy between the excitement around mobile technologies as educational and developmental tools and the empirical track record of such initiatives. This conflict is perhaps most evident in the ITU's (2015) m-Powering Development Initiative report. In it, the ITU's working group on mobile learning commences with the common claim among mobile vendors to mobile learning's immense potential to change education: "Mobile devices have the potential to contribute significantly to education and learning across the world", only for the report to then immediately concede that, "(h)owever, uncertainty persists as to their precise impact, and how best they can be used to contribute positively to education and learning" (ITU 2015: 12). Further in the paragraph it is then acknowledged that there is "far
too little rigorous evaluation and M&E of mobile learning programmes", but this does not stop the authors from highlighting the "widespread recognition of the potential of m-Learning to improve the delivery of education and skills acquisition" (ibid).

The ITU's position might be best described as a 'faith-based' view of educational technologies' impact (Trucano 2015) in which mobile learning interventions 'ought to work' due to the investments made (c.f. Traxler 2013a). It seems not unreasonable for mobile vendors to assume such a position given their vested interest. However, their use of research evidence to claim support to the vast potential of mobile learning on education and development in LMICs is not warranted. My systematic review, as other related reviews, provides consistent and reliable evidence that such an interpretation of the available evidence-base is not accurate.

6.3.2 The limitations of access to technologies

The findings of my systematic review also have implications on ongoing debates on how to position access to technologies in relation to what users of technologies can actually do with them. These debates reflect similar concerns in the field of ICT4D, which outline how the translation of access to technologies into meaningful and effective use towards development outcomes is often challenged in LMICs (e.g. Gigler 2015; Kleine 2011; Zheng 2009). My systematic review findings challenge two perspectives of access to mobile technologies and their link to development outcomes: first, the techno-centric assumption that a mere access to mobile technologies can establish significant education and development outcomes; and second, the assumption that a mere access to content and information on the devices presents a meaningful education or empowerment intervention.

The techno-centric position assumes a close link between the provision of technologies and subsequent education and development outcomes (Unwin 2009; Traxler 2015). There is little need for further intervention aside from providing the technological devices to users. Prominent examples of this approach in relation to technology and education in LMICs are Sugata Mitra's Hole in the Wall project and subsequent development of minimally invasive education; the OLPC project; and the Solar Classroom in a Box project (Mitra & Vana 2001; Mitra 2003; Annay & Winters 2007; Solar Classroom 2017). In these educational technology interventions, the technology is commonly the only input provided and becomes essentially synonymous with the educational input of the intervention. For example, children are supposed to use the Hole in the Wall computers to teach themselves, an educational approach similar to the OLPC project through later iterations of the project do use the laptop

computers in a formal education context too. In my systematic review I found no empirical evidence supporting the effectiveness of a techno-centric approach to ML4D. There was not a single ML4D programme that supplied mobile devices to educational actors as the only intervention component.

This finding suggest that the mere provision of mobile devices is simply not a feasible intervention approach for mobile learning programmes in LMICs. Partly this can be explained by the fact that most educational actors do already have access to mobile technologies and that therefore interventions rather do design for the educational use and application of the technologies. Only a little bit more than half (55%) of all mobile learning programmes included in the systematic review actually had to provide the mobile devices as part of the intervention design. Another explanation might be linked to a general shift away from a focus on inputs and outputs in the education sector in LMICs in general. Following the large investment into access to education as part of the MDGs, recent efforts along the EFA and SDGs place a much stronger emphasis on the outcome and quality of education (Unterhalter et al 2015; WEF 2016). This is also reflected in recent disinvestments of large-spaced mobile learning programmes in LMICs. Since 2015, national-scale tablet programmes in Turkey, Kenya, and Thailand, which focus largely on the supply of technologies to teachers and learners, have been cancelled by governments.

Another manner in which access to technologies is commonly linked to education and development outcomes is to assume that access to mobile devices equates with access to educational content; and/or that access to information on mobiles equates with empowerment outcomes. Such perceptions are often expressed by evoking the metaphor of mobile technologies as "education you can hold in your hand" (WEF 2012: 4). Or, in Melinda Gates' words: "If every woman has a smartphone, imagine all the empowered people" (Gates 2016). Again, my systematic review refutes this positioning of technology access. Both, the meta-analysis and the thematic synthesis illustrate clearly the importance of the underlying pedagogies in order to exploit the technological affordances of mobiles for educational purposes. A range of sub-group analyses and mechanisms focus on the particular types of pedagogical approaches and how they can change teaching and learning practices to support educational outcomes in LMICs. In short, the review findings point to the centrality of what the different educational strategies supported by mobiles allow teachers and learners to do; the findings do not point towards a mere provision of educational content on mobiles which learners can then access independently or, in the case of teachers, to preexisting educational content being stored on devices, which teachers can then use as readyto-go materials for their lessons. The same applies to empowerment outcomes, which

cannot be reduced to a mere access to information or learning opportunities (see section 6.1.3 and 6.2.2).

A conception that assumes the access to educational content or information on mobile devices to present mobile learning links back to very early definitions of mobile learning as learning that takes place 'anytime, anywhere' (e.g. Quinn 2000; Kukulska-Hulme et al 2005; Traxler 2005). Current pedagogical definitions of mobile learning do not focus technological access or utilisation, but on the types of learning practices that can emerge from the interplay between technology usage and other educational inputs; and how these learning practices benefit from different types of mobile affordances (Wali et al 2008; Sharples et al 2007; Kearney et al 2010). This is informed by a socio-cultural view of learning and education and positions mobile learning firmly within this educational school of thought (Winters et al 2017; Pachler 2009).

From an educational perspective then, there is consensus that access to educational content on mobiles does not constitute mobile learning as it lacks an underlying pedagogy or theory of learning. The continued positioning of such access to content to constitute an effective educational and developmental intervention in LMICs is thus in contradiction to both educational theory and the empirical evidence reported in my systematic review. In the words of Winters (2015: 10) "the provision of content on mobile phone is neither pedagogically nor technically innovative (...) and positions mobile learning in a place where research was more than 10 years ago". The findings of the systematic review underline that mobile technology interventions need to draw on the full pedagogical value of mobile learning in order to unlock their educational and developmental potential.

6.3.3 Appropriate technologies?

My systematic review findings also pertain to ongoing debates as to what types of mobile technologies are most relevant and useful to support education in LMICs. A common narrative here advocates for what could be loosely termed 'appropriate' technologies to be used. This refers to the application of simple and affordable mobile technologies that are preferable already in use by people and communities (Trucano 2015; Trace 2016; Smith et al 2011; Balasubramanian 2010). It links to a wider academic school of thought based on Schumacher's (1973) intermediate technology concept, which is not discussed here. A prominent and much cited example for the use of simple and affordable mobile devices refers to Jensen's (2007) paper on how fishermen in India used their cellphones to access

market information in real time in order to determine at which harbours to sell their daily catch.

ML4D programmes have often subscribed to this narrative, which has resulted in programme designs that make predominately use of simple technologies such as radio devices and basic and features phones. For example, of the ML4D programmes included in the systematic review, all but two programmes implemented in conjunction with an NGO or other types of development organisations made use of basic and features phone. The idea that mobile technologies already in use by people and communities in LMICs are most suited as a tool for mobile learning programmes thus seems to enjoy much currency.

However, in the wider literature on ML4D this view is not unchallenged. Unwin (2015: 15) for example laments that "second-hand' technologies still continue to be passed-down to people living in poorer contexts". He assumes such technologies to hold back the potential of mobile learning programmes in LMICs as their technological features limit the interaction between people and technologies, and further do not reflect people's own choices in what devices to pursue. Winters (2016; 2017) adds to this critique highlighting that the use of less sophisticated devices necessarily limits the types of affordances that can be used to support learning and teaching practices. As a result, the range of pedagogies supported by mobile devices becomes limited. This observation is shared by Buckner and Kim (2013), who, following ten years of experimentation with different mobile learning intervention designs in LMICs, conclude that in order to integrating technology and pedagogy, sophisticated mobile devices with sufficient technological affordances are required.

The findings of my systematic review challenge the narrative of appropriate technologies to some extent. While not formally testing the impacts of different ML4D interventions against one another, there is observational evidence to the limitations of using simple and affordable technologies. Pedagogically-rich mobile learning programmes are found to yield larger effects, and too are associated with a range of mechanisms that enhance their relevance and usefulness to programme participants. However, these pedagogically rich mobile learning programmes are linked to the use of more sophisticated mobile devices, smart phones and tablets in particular. Mobile learning programmes using less complex programme designs were found to have a smaller effect. What is more, these less complex programmes—both technologically and pedagogically—were predominantly applied in rural areas and in more disadvantaged educational settings. This correlates with the link between NGOs and development agencies as ML4D programme implementers and the of use simpler technologies.

Therefore, there seems to be a paradox for advocates of appropriate technologies in ML4D programmes. Simple technologies such as basic/feature phones are positioned as technologies of choice by disadvantaged communities and on these grounds argued as the most relevant technologies to support mobile learning programmes targeting social transformation and empowerment (e.g. Balasubramania et al 2010; Andersson et al 2010; Trace 2016; Smith et al 2011). But, it is the very simplicity of these devices that constraints their educational and transformational potential as they allow for the mere dissemination of information or educational content. This model of content and information provision has already been shown to present an insufficient approach to ML4D. In sum, there is a contradiction between the necessarily limited pedagogies that basic/feature phones can transmit and their assumed transformational impact by development practitioners. From an educational perspective Rochelle (2003) urged mobile learning practitioners to "identify those simple things that technology does extremely and uniquely well, and to understand the social practices by which those new affordances become powerful educational interventions" (Rochelle 2003: 9). It seems that ML4D practitioners could pay closer attention to this advice. Applying mobile devices that have a wider range of affordances are likely to do a larger variety of things extremely and uniquely well, and therefore seem to present a more powerful education and development intervention.

6.3.4 Education as a broken service?

The last conversation that my systematic review finding relates to is whether the provision of mobile technologies and design of mobile learning programmes can be regarded as an endogenous or exogenous part to the formal education system. As discussed in chapter 2, section 2.3.3, there is a trend that technology provision deliberately bypasses the formal education systems and sets up parallel structures to it. Such a strategy is justified on the grounds that the current education system is regarded as weak and that its actors and processes are more likely to constrain the perceived potential of mobile technologies than to support it. Mobile learning programmes such as Solar Classroom in a box or the tablet iteration of the OLPC in Ethiopia (Talbot 2012) are examples of the setting up of such parallel structures in practice. Support for this narrative also comes from mobile vendors with the GSMA for example underling that "(m)obile technology's power to transform education is difficult to overstate, given the importance and impact of learning that takes place outside a traditional classroom environment." (GSMA 2012a: 4). This positioning might not surprise as a device- and content-focused understanding of ML4D provides a market opportunity for mobile vendors.

The systematic review findings strongly suggest that mobile learning programmes in LMICs operate within, and not outside the formal education system. What is more, the large majority of the reviewed ML4D interventions are in fact supplementary to existing educational structures and interventions. The mobile learning interventions in LMICs included in the systematic review are embedded within existing educational structures—either structures in the formal education system or existing informal education structures such as NGO literacy programmes.

The review suggests a number of reasons and mechanisms as to why this embeddedness in formal educational structures might be observed. It appears that most of the factors driving the adoption and effects of ML4D programmes are depended on inputs that only formal educational structures can provide. First and foremost, this refers to the centrality of teachers and pedagogies in the effective use of mobile technologies to support education in LMICs. My systematic review has provided ample evidence for the importance of both factors. Currently only formal educational systems in LMICs can provide both at scale. Likewise, more effective mobile learning programmes relied on complex technological and educational programme designs. These designs require supporting infrastructure to be in place such as internet connectivity, reliable electricity, provision and maintenance of devices at scale, etc. The same can be said about the established factors of adoption, for example integration of technology into existing curricula, user training, and peer-to-peer support, all of which are provided by formal education structures at scale. In sum, my systematic review findings indicate that mobile learning interventions are firmly embedded in the overall structure of education systems in LMICs. The framing of mobile learning as a parallel structure to the formal education system does not seem to be based on empirical research evidence.

However, mobile learning's embeddedness into formal education systems also infers that programmes will be subject to the negatives properties of the systems. In contexts where these systems are unequal or underperforming, ML4D is likely to reproduce rather than challenge these properties. This provides a challenge to my critique of the techno-centric positioning of mobile learning as exogenous to the formal education systems. A strong rationale for a techno-centric ML4D approach is that existing education systems suffer from a range of structural inequalities that disadvantage certain groups of learners. For example, girl learners in patriarchal societies might be systematically excluded from formal education or the language of teaching might exclude certain ethnic groups. By being positioned within existing education structures, ML4D programmes are unlikely to address existing inequalities and might actually sustain marginalisation. The systematic review findings support the assumption that this is indeed the case with current ML4D programmes.

The argument as to whether ML4D can or should be positioned as working within or outside formal education structures will be developed in full in chapter 9, section 9.3.1. To conclude the caveats around positioning mobile learning closely within existing education structures derived from my systematic review findings, two remarks need to be highlighted. First, as much as the majority of ML4D programmes were embedded within existing education systems, there was barely any evidence that these programmes aimed to strengthen the functioning of these systems. For example, mobile learning programmes rarely targeted structural support to sustain education systems such as supporting teacher training and professional development; EMIS; stakeholder engagement; and education management and leadership. Second, a very low number of ML4D programmes was driven by LMICs governments and education departments (n=4). The vast majority of programmes was driven by research interest, NGOs, and development agencies. It is challenging to claim a strong positioning of ML4D within formal education structures if decision-makers of these structures do not drive mobile leaning programmes.

Conclusion

This chapter has presented the second set of systematic review findings based on the thematic synthesis and provided a theory of change for ML4D combining both sets of systematic review findings. In the thematic synthesis, I identify a range of mechanisms and contexts that unpack how the provisioning of mobile technologies might influence education and development outcomes in LMICs. The identified mechanisms provide rich evidence on the changes in teaching and learning practices that can explain the observed positive educational effects of mobile learning programmes in LMICs. They too point to a wider range of socio-economic changes that educational actors might derive from the provisioning of mobile technologies. In terms of contexts, the thematic synthesis indicates that the acceptability and uptake of mobile technologies. However, factors within the education system also play an important role in the uptake and implementation of mobile learning in ML4D interventions.

I then combine the thematic synthesis and meta-analysis findings in a mixed-methods synthesis to construct a theory of change for ML4D. This theory of change aims to assess to what extent the provision of mobile technologies in LMICs can be linked to education and development outcomes. Based on the empirical evidence in both review modules, I show that there is currently no supporting evidence on the impact of mobile learning programmes on development outcomes in LMICs. In addition to this evidence gap, observed positive educational outcomes cannot be attributed transformational impact on education in LMICs. Structural inequalities within mobile learning intervention design and a narrow conception of development as an increase in human capital also hinder mobile learning's potential to support development outcomes. I therefore conclude that mobile learning cannot be positioned as development intervention based on current evidence and show how this finding runs counter to popular narratives on ML4D.

The findings of my mixed-methods systematic review suggests that a rethink of the positioning and conceptualisation of ML4D is required. While the absence of evidence on development outcomes cannot be equated with an evidence of absence, it seems justified to probe deeper into how else one might conceptualise the provision of mobile technologies in LMICs in relation to education and development. A more in-depth investigation into the perceived transformational potential of mobile learning on education in LMICs and how and why this change might lead to subsequent socio-economic changes is called for. In this, I

propose the Capability Approach (CA) as a conceptual lens through which to redefine how learning and teaching with mobiles might be linked to education and development in LMICs.

The CA has guided alternative conceptions of development outcomes in related research fields; and in particular unpacked similar techno-centric assumptions about access and use of technologies in relation to development. However, despite its wide application in development studies and ICT4D, empirical work on the CA in relation to mobile leaning in LMICs is scant. My review only identified three studies that explicitly referenced the CA, neither of which used it as a conceptual or analytical framework for data collection and analysis. Yet, a number of my systematic review findings indicate sufficient conceptual scope for the CA to guide a rethink of ML4D. For example, my thematic synthesis identified a range of themes around valued opportunities that teachers explored independently following the provision of mobile devices. In current conceptions of ML4D as laid out in my theory of change, there is however little scope for such pathways to development, which are dominated by a human capital approach to understand and assess education and development outcomes. From a capabilities perspective an investigation and conceptualisation of the use mobile technologies in educational settings in LMICs could pay more attention to what opportunities mobiles allow users to explore and how these are linked to individual changes in well-being, agency and human development. I therefore next present the findings of my qualitative case study of a ML4D programme in South Africa in which I applied the CA as a conceptual lens to investigate the transformational potential of mobile learning on education and development.

PART II:

DEVELOPING A CAPABILITIES CONCEPTUALISATION OF MOBILE LEARNING'S CONTRIBUTION TO DEVELOPMENT

Chapter 7. Using mobile technologies to expand teachers' capabilities: A qualitative case study of a mobile learning programme in rural South Africa

Introduction

This chapter presents the results of my qualitative case study investigating a South African mobile learning programme from the perspective of the Capability Approach (CA). The case study set out to explore how teachers in the Information Communication Technology for Rural Education Development (ICT4RED) programme in rural South Africa accessed and used mobile technologies in their social and professional contexts. I investigated whether the provision of personal tablet devices to teachers as part of the ICT4RED programme altered their valued capabilities and functionings. That is, I am interested in teachers' own views on whether and how their educational use of mobile technologies allowed them to explore valued beings and doings. And, if so, how these valued functionings support teachers' opportunities to live a life they had reasoned to value. To guide this investigation, I collected a range of qualitative data including in-depth interviews, focus groups, and classroom observations on teachers' perceptions and use of tablet devices in a rural education context.

The objective of this qualitative case study is to apply the CA as a conceptual lens through which to analyse the usage of mobile technologies in educational contexts in Low- and Middle-Income Countries (LMICs). Through the case study, I aim to operationalise the CA as tool to frame the outcomes of mobile learning for development (ML4D) programmes. I am therefore interested in whether the CA with its conception of development as freedom (Sen 1999) and its focus on the effective opportunities or freedoms of individuals to transform access to resources—mobile technologies in this case—into valued beings and doings provides a useful framework to unpack the role and contribution of mobile learning to development⁷³. This chapter zooms into my empirical investigation on whether the CA can indeed be applied as a conceptual tool to analyse the effects of ML4D interventions and what types of outcomes emerge from such an application. In further chapters, I will discuss

⁷³ This implies applying my preferred definition of development based on Sen (1999) from this point onwards in the thesis.

in more detail an analytical framework of ML4D interventions based on the CA that I derived of the case study findings (chapter 8) and what the implications of applying a capabilities perspective to the educational use of mobiles in LMICs has on the overall positioning of ML4D (chapter 9).

This chapter is structured as follows. I first provide a brief justification for the application of the CA based on the systematic review findings. This enhances the more substantial justification provided in chapter 3. I next briefly contextualise my qualitative research and applied research methodology before presenting the results of my case study research. These results are categorised according to the four overall capability dimensions emerging from my data. The chapter concludes with a brief summary of the reported findings.

7.1 Overlap between the systematic review findings and the CA

Chapter 3 provided a detailed conceptual justification for the application of the CA as a conceptual lens to analyse the application of mobile technologies as an educational tool in LMICs. This justification pointed to a range of conceptual overlaps with research bodies related to ML4D in which the CA has already been applied. It also pointed to a range of gaps between existing CA frameworks and the nature of mobile learning interventions in LMICs, which negated an off-the-shelf use of existing CA frameworks. In addition to this justification based on the existing literature, the findings of my mixed-methods systematic review present further support to the feasibility and usefulness of investigating ML4D from a capabilities perspective. I touch on these review findings briefly below.

My systematic review identified a range of gaps and contradictions between the evidencebase on ML4D and the perceived potential and subsequent positioning of mobile learning as a development intervention (chapter 6, section 6.2–6.3). It seems that some of the gaps between mobile learning's assumed transformational potential and its empirical track-record can be explored and framed using the CA as a conceptual lens. This relates in particular to the evidence gap of ML4D programmes on development outcomes.

The systematic review identified a lack of evidence of ML4D on development outcomes such as poverty indicators. It also highlighted a human capital perspective on both education and development as a dominant thought throughout the theory of change of ML4D interventions. At the same time, it illustrated a range of mechanisms, which emphasised the importance of using mobiles to support more granular changes linked to self-growth and attainment (e.g. self-efficacy, job satisfaction, social status). From a capabilities perspective, the emphasis on the theory of change for human capital approaches to education and development would be challenged. If we investigate the effects of teaching and learning with mobiles in terms of the valued opportunities they create for individuals, an instrumental view of education linked to skills is too narrow. Likewise, an enhancement in the levels of skills and productivity of the population would not be equated with development.

Consequently, contextualising my review findings from a capabilities perspective suggests that the findings might not indicate a lack of development outcomes per se; but rather that education and development outcomes are too narrowly defined. Applying a capabilities lens with its focus on individual well-being and agency would place a greater emphasis on individual's own appropriation of the mobile devices and the functionings she obtained from this appropriation. This would open-up the theory of change to require a more substantial

engagement with the mechanisms for change around self-efficacy and social status. It would further challenge the lack of engagement with the intrinsic value of education and some of the technological affordances of mobiles linked to non-educational technology usage. For example, while my review cannot link the observed learning effects to development outcomes as they do not present a significant change in educational abilities, a capability analysis would question whether this does not overlook other opportunities and functionings that learners might have derived from the programme. That is, a capability analysis would require a more substantial engagement of ML4D interventions with the interlinkages between educational, social, and economic outcomes. The systematic review findings show clearly that there is scope for such an analysis given the limited causal pathways between the provision of mobile technologies, their educational use, and development outcomes currently reported in practice in the ML4D literature.

Another major barrier identified in the systematic review affecting mobile learning's transformational potential in LMICs referred to the reproduction of structural inequalities. Mobile learning programme designs were found to discriminate against more disadvantaged groups in society. This programme design flaw could be alleviated by the application of the CA to conceptualise ML4D. In a capabilities conception, the emphasis in intervention design would shift from the provision of resources (i.e. mobile technologies here) to the conversion factors of individuals to apply these resources to an effective use to further their own valued being and doings. A capabilities lens would thus have highlighted the discrepancy between the conversation factors of more disadvantaged groups, the technology and programme design provided, and the assumed transformation impacts. If the equality of opportunities becomes the lens of analysis and programme objective in ML4D interventions, different design features would have been warranted to counter-balance the different abilities of users to apply the technologies to further their valued beings and doings. Arguably, designing ML4D for the equality of opportunity could have enhanced the development potential of interventions and addressed—rather than reproduced—these observed structural inequalities.

This argument then links back to the discussion on the limitations of access to technologies and the limitations of appropriate technologies. The former has already been sufficiently covered in chapter 2 and 6 and the systematic review findings add empirical support to the CA's critiques of techno-centric approaches to ML4D. In reference to the latter, applying a capabilities perspective to the use of technologies would question the inherent assumption that technologies already widely in use by the target population of ML4D programmes present the most relevant devices. If technologies are being assessed in their ability to

194

support individuals to enhance their set of opportunities to live a life they have reason to value, limiting ML4D interventions to technologies already in use consequently limits the sets of opportunities that can be achieved. A CA investigation of ML4D programmes requires to move towards equality of opportunity and choice, and to ask what mobile devices allow one to do. This seems to shift the emphasis away from the already existent access and use of technologies to what types of other usages and roles technologies could support if applied to enhance effective opportunities and capabilities⁷⁴.

In sum, the systematic review findings therefore provide additional support that the CA presents an effective tool to guide an in-depth investigation into the perceived transformational potential of mobile learning in LMICs.

⁷⁴ I discuss the potential risks of the CA's strong focus on individuals in chapter 4, section 4.3.5 and chapter 9, section 9.3.1.

7.2 Contextualising the case study research

Before presenting the result of my qualitative case study of a ML4D programme in rural South Africa, I comment briefly on a few aspects related to the research design and process not covered in chapter 4. This aims to provide additional context on how I arrived at the presented results and how they might be understood.

The vast majority of qualitative data used in the analysis presented here was collected between January and March 2015. During this period, I accompanied 25 teachers involved in the ICT4RED mobile learning programme during their daily teaching activities and conducted in-depth interviews with each teacher. I also attended a series of professional development workshops part of the ICT4RED intervention which involved the same teachers; graduation ceremonies of different phases of the programme; as well as policy meetings and policy roundtables related to the programme. These events extended from June 2014 to October 2016. Most of the teachers interviewed and shadowed in the 2015 data collection became 'ICT champions' following the completion of phase 2 of the ICT4RED programme and subsequently served as trainers of trainers for the programme in phase 3 in 2016. I therefore continued to interact with the majority of research participants post the initial data collection. However, the results and analysis presented here are exclusively based on the data collected until March 2015. In my later engagement with the ICT4RED programme and the teachers involved, I did not assume the role of a researcher. Through a separate research project⁷⁵, I was subsequently advising programme staff and the government departments funding the programme on how to use the overall programme's research results to inform the revision of a white paper on e-education in South Africa. My role in and interaction with the ICT4RED programme had thus changed to an extent where it did not seem appropriate to regard myself as an external researcher.

In my case study research of the ICT4RED programme I exclusively collect data from the teachers involved in the programme. While the entry point of ICT4RED indeed is teachers as mobile technology users, the programme does extend to learners as well. As part of the mobile learning design, a 1:1 model of tablet provision is applied; though this is phased in gently with teachers receiving tablet devices first before rolling the programme out to selected classes of learners, and eventually to the entire school involved. The reasons for focusing on teachers in my case study relate to access and ethics.

⁷⁵ The UJ-BCURE programme: <u>https://africacentreforevidence.org/uj-bcure_description-4/</u>

In terms of access, a language barrier between myself as an English speaking researcher and the learners as isiXhosa speakers prevented an extension of the research to students. In rural Eastern Cape, a translator would have been required to involve a diverse group of learners within the case study, which I did not have access to. In addition, to conduct research on students in the Eastern Cape informed consent of the guardian of each learner was required in 2015⁷⁶. I did not have sufficient resources to track each guardian and seek their permission to involve their children in the research. This points to a range of additional ethical concerns when conducting research with minors (e.g. Alderson & Morrow 2011). In addition to the pragmatics of seeking informed consent, I also felt unable to provide a research environment in which I could have sufficiently balanced the multiple inherent power relations between me as a researcher and the involved minors. This would have required a larger pool of research resources, for example translation services, time spent in the local communities, and research skills. It is important to note that the decision to limit the research to teachers only was taken before the findings of my mixed-methods systematic review were available.

It is also important to keep in mind the nature of the collected data and its relation to the analytical frame of the CA. As outlined in chapter 4, section 4.3.3, while the nature of data collected followed an inductive approach with interview and observation schedules being semi-structured, the introduction of the CA as an analytical frame was deductive. That is, I explicitly coded the qualitative data according to the analytical devices of the CA: functionings and capabilities. In this, the guiding frame to code my data referred to indications of whether teaching with mobile technologies foster valued being and doings and enhanced the perceived opportunities of teacher to live a life they have reason to value. To operationalise the conceptual coding framework⁷⁷, I relied on a range of guiding questions such as whether there is an indication that teachers can transform access to mobiles in order to become the kind of teacher they aspire to be. It is therefore important to keep in mind that the research results expressed in terms of functionings and capabilities is but one way to read the available data. A different conceptual framework would have generated different types of themes.

Within these analytical frame and the applied conceptual devices of coding for functionings and capabilities, individual content themes were then derived inductively following standard

⁷⁶ The Eastern Cape Department of Education has undergone extensive changes in senior management since 2015. ICT4RED has been informed that this access policy is in the process of being revised and there is currently uncertainty on the Department's policy in this regard.

⁷⁷ For related examples of operationalising the CA as a conceptual framework in ICT4D, see Kleine (2013) and Gigler (2015).

processes of thematic analysis of qualitative data. There is some overlap between the content themes derived in my case study research and in my thematic synthesis in the systematic review. In terms of research process, the data collection and analysis of the case study was conducted after the meta-analysis but before the thematic synthesis. Given the methodological overlap between the thematic analysis in the case study and the thematic synthesis in the systematic review, I cannot rule out the possibility that the themes derived in the case study influenced my coding of the data in the thematic synthesis. However, the thematic synthesis did not adopt the CA as an analytical device and instead focused on a mechanism-context framework. All developed codes in the thematic synthesis are linked on EPPI-Reviewer to text on research findings reported in included primary studies allowing for an interrogation of the link between the underlying data and my interpretation of it in the thematic synthesis; and whether this interpretation might have been influenced by my prior case study research.

Lastly, as flagged in chapter 4, there is also a question to what extent my primary research might have been subject to desirability bias, which I discuss in detail in appendix 7.1

7.3 Investigating teachers' use of mobile technologies from a capabilities perspective

This section present the results of my qualitative case study research investigating teachers' use of mobile technologies in rural South Africa from a capabilities perspective. The section is structured according to the four dimensions of capabilities emerging from the data: informational, educational, economic, and societal capabilities. Each dimension of capabilities indicates a set of valued opportunities that teachers reported to being able to explore following the provision of mobile technologies. I first developed an initial list of all capabilities indicated by teachers before thematically grouping these into the four dimensions of capabilities. For example, a number of capabilities were related to valued educational opportunities, while other pertained to valued opportunities in teachers' social rather than professional lives. This thematic grouping of capabilities resulted in the four dimensions of capabilities.

In total, the four dimensions of capabilities comprise 21 valued opportunities linked to teachers' use of mobile technologies. As explained in chapter 4, each of these capabilities is based on teachers' functionings identified in the collected qualitative data. Therefore, behind each of the 21 capabilities is a set of functionings. Appendix 7.2 provides an exhaustive list of all 91 functionings and how they relate to the capabilities reported here.

7.3.1 Informational capabilities

The first dimension of capabilities identified in the case study of the ICT4RED programme refers to how teachers used their tablets for a wide variety of functionings associated with the expansion of their informational capabilities. I define informational capabilities as the teachers' capabilities to transform access to the tablets into valued functionings related to the use of the specific affordances of the mobile technologies. Informational capabilities to transform access to the tablets state teachers reported in reference to their use of the technological affordances of the tablet devices such as their mobility and constant connectivity. In total, the dimension of informational capabilities consists of five capabilities, which in return are based on 26 reported functionings (Appendix 7.2). The five informational capabilities are listed below in no particular order and discussed in turn in this section.

- ICT usage—to be able to use ICTs effectively
- Information literacy—to be able to find, evaluate, use, and process information
- Connectivity-to be able to connect and communicate

- Mobility—to be able to engage in one's professional identity and practice anywhere, anytime
- Merging contexts—to be able to extend professional beings/identities across contexts

ICT usage—to be able to use ICTs effectively

Teachers expressed satisfaction and pride from achieving what some termed 'ICT literacy' or 'being a tech champion'. This referred to the independent operation of the tablets and their affordances as well as, importantly, the process of obtaining this proficiency. For many teachers being able to operate tablet devices themselves was a major motivation to participate in the ICT4RED programme. The acquisition of ICT skills and associated functionings such as being able to use emails, social networks, and multi-media features of the tablets was highly valued. Teachers repeatedly compared the tablets to PCs highlighting how the applications they found on the tablets made their integration into classroom activities easier. The value of ICT skills also emerged in the context of little pre-existing exposure to educational technology. As one teacher observed: "I spent 22 years in a teaching system but I didn't know anything about technology. I knew nothing about tablets or computers. So I was so interested in this project to see if I can also use these tablets" (Primary teacher; day 3.)

The process of acquiring ICT skills itself was a strong source of well-being as teachers reported high levels of pride for mastering technology usage. In some contexts, teachers reported that their initial inability of using mobile technologies made them feel ashamed given how proficient their students were with the devices. Overcoming such fears and becoming 'confident ICT users' motivated teachers to 'keep trying new things' and teachers valued the process of experimenting with the devices to discover new affordances that might be of support to their teaching.

Teachers also referred to the intrinsic value of ICT skills and their choices in engaging with technologies on their own terms. For example, while few teachers used the blogging facility of the ICT4RED website, the blogging itself was repeatedly mentioned in the data as a valued opportunity. This variation between the opportunity sets identified by teachers and their actual pursuit of such opportunities was also observed between schools. While all schools were provided with the same ICT infrastructure throughout the programme, the conversion of these resources highly varied. In the school with the most advanced use of tablets, teachers completely facilitated lessons using white boards linked to their tablets, whereas in the least advanced contexts, tablets were mainly used as a substitute for a lack

of text-books and writing pads.

Information literacy-to be able to find, evaluate, use, and process information

Teachers perceived the constant and up-to-date access to information, which the tablets facilitated, as a major contribution to their lives. As one teacher expressed it, "these tablets are filled with knowledge" (Primary teacher; day 15). Accessing this knowledge independently and tailored to personal preferences enhanced teachers' sense of choice as the below quote indicates:

"They [tablets] are an educational tool and the most important educational tool as we are getting a lot of information from them—it is not just for fun. It upgrades your knowledge. Internet is the most important because you're able to get anything you want from it. And to connect with friends and teachers I use the internet and email. You get so excited checking your inbox and I am able to look for anything I want. Information is quick and you don't have to go somewhere to research something. Even if you are travelling, once you are at home you can access information." (Primary teacher; day 9).

The above quote highlights the value that teachers placed on having a choice in how to apply the tablets in line with their own needs. Being able to 'look for anything you want' and to merge the translation of access to information into professional and social uses was highly valued. Teacher could use the tablets to access information relevant to their teaching profession, for example using google for teaching materials or asking colleagues for advice; but they could also use the tablets to access information relevant to their personal and social needs. Examples of this use of social information was the often mentioned 'bible app' that teachers used to participate in church services and to teach their own children.

Teachers valued this choice and opportunity of being able to find and use information and to feed this information into both their professional and social realities. More so, being able to engage in this opportunity on their personal devices was described as 'being independent'. Teachers explained that they no longer had to wait for information and updates from the education department or the unions to reach them, but that they now could proactively access them on their own terms. The convenience and independence of accessing information on mobile devices was thus another valued functioning.

Connectivity—to be able to connect and communicate

The capability to be able to connect and to communicate also emerged as a key informational capability. Connectivity and communication allowed teachers to maintain and build their social networks as the tablets provided a channel for instant and personal

communication. Teachers made deliberate choices in what connectivity features of the tablets to pursue and to what ends. In this, teachers again valued this capability for both personal and professional reasons. The opportunity to exchange ideas with other teachers and staying in touch with graduate learners was valued as much as the opportunity of connecting with former colleagues, family, and friends.

The capability to build and maintain professional networks was valued particular highly by teachers who had conducted their teacher training in urban institutions and were now assigned to rural schools. Two of those teachers for example referred to their most memorable moment of using the tablets when they received an email from a former colleague at graduate school whom they had lost touch with after completing their courses. Similarly, the capability to remain in contact with former learners who had received scholarships to go to South Africa's most prestigious universities in Cape Town and Johannesburg was cited as a major source of professional well-being and motivation.

In the context of the rural location of the schools, this capability to connect and communicate seemed to present an important source of agency decreasing teachers' sense of isolation:

"I have joined LinkedIn and though I'm not at hundred percent, I know what is going on. It makes you to connect with other people that you have never met before. You look at other profiles and you realise people are working and it motivates you. It motivates you to post what you doing in your own corner and people must know that is your school, and we're doing our work in our little corner (...) It makes you so happy when you connect." (Primary teacher; day 6).

This opportunity to overcome a sense of isolation was referred to throughout the collected data and teachers cited examples of using the tablets to connect to friends and colleagues from countries as diverse as Sweden, USA, Russia, Kenya, and Australia. Not all of these mentioned interactions and connections had evident professional benefits, but were largely valued by teachers for 'making us feel to be closer'.

Mobility—to be able to engage in one's professional identity and practice anywhere, anytime

Teachers highly valued the capability to use the tablets anywhere, anytime; and by extension, to engage in their professional practice at time and location of their choice. 'The tablet goes where I am going' and 'it's all there in the tablets' were commonly used phrases to describe the virtue of being mobile. Teachers described a range of benefits from being able to use the devices anywhere, anytime. Most teachers travelled more than 30 minutes to

work every day and attended at least three departmental or union meetings per month in Cofimvaba. Being able to have all their administrative material and teaching resources in one place was therefore highly valued in this context. Teachers also referred to the convenience of not having to carry text-books and other teaching materials. There was a general perception that carrying tablets appeared more professional and allowed one to behave like a 'dynamic professional'.

Teacher reported many instances of educational uses of the tablets facilitated by the devices' mobility. This included the creation of teaching content on the go, most commonly by taking pictures and audio/video recordings of content relevant to the planned lessons. Teachers also reported using the devices at home to prepare for next day's lessons, which was preferred over having to prepare the lessons at the school's facilities. Teachers also hypothesised that it would be possible for them to teach pupils remotely via their tablets, for example when they had to attend departmental meetings. However, more common instances of this in practice referred to situations in which a teacher had to teach and monitor two classrooms at the same time. In an observed lesson, the teacher switched between using instructional videos on her tablet connected to a white-board in one classroom with her regular lesson plan in another classroom. Though, this type of mobility was not valued by educators for a loss of teaching time and quality in both classrooms. In summary, the mobility and constant availability of the tablet devices allowed teachers to convert the devices into centralised teaching tools, which more conveniently stored all their educational content and allowed them access to it on their own terms. In reference to this, a High school teacher (Day 12) remarked: "It [the tablet] is my office". Teachers valued having such a tool for personal reasons too as they perceived it to be more professional and allowed them more opportunities to engage in their professional identity and practice.

Merging contexts—to be able to extend professional beings/identities across contexts A cross-cutting informational capability was teachers' opportunity to move in and out of their professional practice and identity across contexts. Using the affordances of the tablets, teachers explored different ways in which they could act and be perceived as professionals outside the classroom. A range of these have been discussed as part of other informational capabilities already and others will be discussed in the remaining dimensions of capabilities. This indicates that the opportunity to merge contexts was valued by teachers and, given its nature, influenced a range of educational and societal capabilities too.

Without pre-empting or repeating this cross-cutting capabilities, in summary teachers valued the use of tablets to cross and merge three main contexts. First, they used the tablets to

cross geographical contexts, for example collecting educational content on the go or connecting virtually with teachers from other schools. Second, teachers crossed temporal contexts using the tablets to connect with past colleagues, to stay in touch with learners leaving the schools, and to relive and access educational opportunities denied to themselves during the apartheid regime. Third, teachers bridged social and professional contexts using the tablets to extend their professional identities into their social lives, for example recording their lessons for children and community members not being able to attend and access lessons in school; or, vice-versa, to extend their social identities into their professional contexts, for example recording speeches of themselves at church and using these as content and inspiration for learners in class.

7.3.2 Educational capabilities

The second dimension of capabilities related to teachers' use of the tablets to create and explore a range of educational opportunities. Educational capabilities are defined here as teachers' capabilities to transform access to the tablets into valued educational beings and doings. Educational capabilities thus comprise the opportunities related to teachers' use of the tablets to expand their functionings in a professional context, for example achieving their professional goals as a teacher. I identified six valued educational opportunities associated with the dimension of educational capabilities. These six capabilities consist of 28 functionings identified in the qualitative data (Appendix 7.2). The six capabilities are listed below and will be discussed in turn in this section:

- Student success—to be able to improve learning achievements
- Pedagogies—to be able to change and explore teaching methods
- Role of the teacher-to be able to become a facilitator of knowledge
- Occupational satisfaction—to be able to enjoy teaching more
- Innovation—to be able to innovate
- Collaboration—to be able to work in collaboration

Student success—to be able to improve learning achievements

Central to teachers' educational use of the tablets was the ambition to integrate the tablets into their overall efforts to provide their learners with quality education. Teachers valued the tablets in this remit in a practical sense as a tool to support them in supporting their learners to achieve academically. Being able to improve their learners' academic achievements, for example through increasing engagement and providing a wider variety of educational content, was one of the most frequent cited valued doings of the teachers involved in my research.

Teachers reported that using the tablets as an educational tool made it easier to motivate learners for education. Learners were for example described as 'excited', 'having more fun', and 'paying attention'. Teachers linked this change in leaners' enjoyment both to the novelty effects of the tablets and the teaching strategies that the tablets allowed teachers to facilitate. Being able to motivate learners for education and being able to provide them with a more enjoyable learning experiences then supported teachers' key teaching goals of supporting learners to do well academically. This capability set was captured in one of the teachers' own words as: "I like my learners and I want to move them to have better chances. I want them to fly high on the sky. And the tablets will help them flying high" (Primary teacher; day 4). Supporting learners' success was a key driver of teachers' own motivation and identity as professionals. And teachers believed the tablets to enhance their capabilities to achieve this goal. The tablets were therefore regarded by teachers as an additional educational input that they could apply in their pursuit of improving the learning achievements of their own students.

Pedagogies—to be able to change and explore teaching methods

In line with ICT4RED's main objective, the teachers explored a variety of different teaching opportunities based on the educational affordances of the tablets. In the ICT4RED programme, technology adoption and usage was embedded into efforts of promoting more learner-centred teaching methods in rural schools. Teachers valued these more learner-centred teaching strategies (e.g. role plays; jigsaw; group work) as they believed them to enhance a range of their valued educational abilities. Being able to increase interaction and participation in their classroom lessons, for example, presented a valued educational capability that teachers perceived to be able to explore with the help of the tablets.

However, often it was not so much a particular teaching strategy itself that teachers valued, but their own ability to adapt their teaching to what teachers' perceived as a rapidly changing educational environment. Teachers expressed for example that the 'traditional way' of teaching was not suitable for '21st century learners' and that they wanted to be able to 'keep up'. Teachers were convinced that frontal teaching methods with little learner involvement did not stimulate learners sufficiently, which they then contrasted to learners' reaction when being taught with the integration of tablets in lessons. As a result, teachers valued the opportunity to adapt their teaching strategies; but this was not limited to any particular strategy or underlying pedagogy per se. I will comment more on the implications of this capability set on the applicability and usefulness of formal mobile learning pedagogies in chapter 9, section 9.3.2.

Teachers further valued to their ability to use the access to additional information and educational content on tablets to integrate different and more appropriate content in their lesson plans, for example supplementing the lessons with localised multi-media inputs such as YouTube videos of local events. Teachers valued this access to more information and content in relation to their own subject knowledge too.

This general perceived increase in educational capabilities provide teachers with more choices on how to teach allowing them to be more creative and diverse in their chosen teaching strategies—an educational opportunity that had intrinsic value to teachers. Also, being able make their educational activities more interactive and diverse—characteristics that were attributed with increased learner motivation and performance—linked back to teachers' valued opportunity of providing their students with high quality education.

Role of the teacher-to be able to become a facilitator of knowledge

Teachers in particular appreciated the opportunity to alter their teaching methods to allow learners to 'become the problem solvers' and to co-produce knowledge. This educational capability was a sub-set of the capabilities related to pedagogies presented prior to this. However, this capability to be able to become a facilitator of knowledge was mentioned so consistently and in great detail by teachers that it merits to be mentioned in its own right. Using the tablets allowed teachers to redefine their educational role and to become "the facilitators of knowledge, not the source of knowledge" (High school teacher; day 13). This shift in teaching approach was valued as teachers reported it to lead to more engaged learners and interactive lessons, which in return were assumed to lead to increased learning outcomes. Hence, being able to become a facilitator of knowledge allowed teachers to pursue their valued educational functioning of supporting learners to succeed. The educational value of using the tablets to become facilitators of knowledge in the eyes of the teachers is captured in the below excerpt:

'It [the tablet] has improved my teaching and learning styles and the learning outcomes have improved. (...) My teaching style is no longer teacher-centred. When I'm in class I'm no longer standing in front, I let the learners do the work so I'm not the person who keeps the knowledge and I am just facilitating it. The teaching strategies we learnt helped a lot. I did the storytelling for example and the children told really exciting stories (Primary teacher; day 10).

I observed a number of examples of this teaching strategy in practice. Teachers in particular liked the teaching activity of story-telling in which learners were given tablets to create a multi-media story about the customs and life histories of elders in the community. Teachers,

in particular if they were assigned to the rural schools from urban teacher training institutions, explained how this activity had led to a two-way learning experience and that they had invited community elders in some instances to co-present lessons on local histories. I note that these teaching strategies are close to social-constructivist ideas of education but were not framed by teachers in this manner.

Occupational satisfaction—to be able to enjoy teaching more

Teachers also reported a range of cross-cutting functionings derived from the use of the tablets that allowed them to being able to enjoy teaching more. These functionings merged educational and informational affordances of the tablets. As already reported, teachers believed the tablets to support them in controlling the classroom as learners were better behaved and more attentive. Tablets were seen as a tool to reduce teachers' workload and to facilitate teachers' daily tasks as teachers for example no longer had to carry heavy textbooks on their daily commutes. This was often referred to as to 'boost our morale' and teachers indicated the pleasure derived from the ease of use of administering educational tasks on the tablets. 'You just tab and it is all there' was a common used expression and teachers applied the tablets to store learners' assignments, attendance records, and reports on their tablets, among other. This was contrasted to keeping paper records in the school's facilities which the teachers regarded as an inferior method of EMIS. In sum, these practical and educational functionalities of the tablets were generally regarded as to make being a teacher 'more enjoyable'.

Innovation—to be able to innovate

Another valued educational capability related to teachers' opportunity to innovate their educational practices. This capability encompasses the pedagogical innovation mentioned above already but extended beyond it emphasising the intrinsic value of being able to engage in a process of innovation itself. Many teachers reported that they were initially scared of the technologies as they were worried that they would make mistakes when integrating them into their teaching. Teachers do not have many opportunities for professional development in the rural Eastern Cape and therefore were worried that a change to their teaching undermining their ability to support learners' educational successes. The ICT4RED programme actively designed for this contextual factor ensuring that the professional development training provided a 'fail-safe' environment in which mistakes were framed as learning opportunities (Herselman & Botha 2015).

Teachers valued this ability to try out something new without fearing the professional

repercussions of making mistakes. For example, one teacher observed that, "the main advantage [of using tablets] is that you realise not everything in life is difficult, even if you don't know something you can try to do it." (Primary teacher; day 16). The value of this sense of discovery accompanying teachers' joint exploration of the tablets as educational tools was expressed in multiple themes such as 'learning by doing', 'make mistakes together' and 'you just try again'. This experience of reframing failure as an opportunity for innovation then reflected in an enhanced tolerance of learners' mistakes too, which teachers began to refer to as 'mistakes are part of the solution'.

Teachers were impressed with their own ability to master the tablets rapidly as educational tools and to innovate their educational practice. This opportunity to innovate was also valued in the contexts of 'keeping up' with learners. As indicated above, teachers believed educational contexts and practices to be in flux and did not want to be left behind. Being able to teach with tablets was seen as an opportunity to become 'modern' or 21st century teachers—a deliberate expression to match their perception of dealing with 21st century learners who are advanced in their technology usage already.

Collaboration—to be able to work in collaboration

The final capability mentioned by teachers referred to their value of the opportunity to work in collaboration. Aspects of this capability have already been covered under the informational capability 'connectivity'. From an educational perspective, teachers in particular valued the tablets' connection and communication affordances to collaborate with other teachers. Observed examples of this included the exchange of teaching content, asking for and providing advice, joint development of lessons plans on Google Docs, and the collaborate planning of school districts events using the full suite of Google's collaborative tools. Collaboration was also a key facilitator of technology adoption and peer-to-peer learning and joint technology assignment presented valued mechanisms through which to integrate the tablets into teaching practices.

7.3.3 Economic capabilities

The third dimension of capabilities referred to the economic opportunities that teachers associated with their ability to use mobile technologies in an educational setting. Economic capabilities can since be defined as teachers' capabilities to convert the technological input into valued economic functionings. The dimension of economic capabilities therefore groups together valued economic opportunities, for example looking for a job or supplementing one's professional credentials, that teachers explored as part of the ICT4RED programme.

In the rural context of the ICT4RED schools, teachers sensed five economic capabilities based on a total of 18 observed functionings (Appendix 7.2). The five capabilities are listed below and discussed in return:

- Employment—to be able to be employed
- Economic capital—to be able to build economic capital
- Online education-to be able to extend educational opportunities
- Marketing-to be able to 'build the brand'
- Digital services—to be able to access digital services

Employment-to be able to be employed

Above all, teachers assumed that their own—as well as their learners'—capability to be employed had increased due to the involvement in the ICT4RED programme. The opportunity to translate access to mobile devices into enhanced employment prospects was a strong theme in the collected data and teachers cited it as a central capability derived from their use of tablets. For example, being able to teach with mobile technologies was seen as a major advantage in order to apply for teaching posts and to stay employable in the 'modern' times. There was a notion among teachers that 'ICT is in demand these days' and that being proficient in the use of mobile technologies had made teachers an 'asset' for current and future employers. This notion extended to their learners too, and teachers assumed them to have higher chances of finding employment due to their gained IT skills as well.

As a result of their enhanced and in-demand ICT skills, teachers mentioned a range of different employment opportunities that they now could explore as 'ICT champions'. This related to posts in schools in more affluent areas that were already using educational technologies and to posts in the provincial education department. Teachers also expected the ICT4RED programme to be expanded to additional schools in the province and assumed that they would be able to act as trainers of the trainers in this process. Within this theme, there was no observed functioning of teachers actually transferring and it seemed to be the opportunity of having professional choices itself that teachers valued. For example, one teachers remarked:

"It [the tablet] also gives me more opportunities. If I want to transfer from my school I will have a great advantage to find a new job because I know how to use tablets. Being a mobile teacher gives me more choices where to work and how to work." (Primary teacher; day 9).

Teachers positioned their opportunity be employed in the context of a changing education profession. There was a strong perception that the future of the education profession lies in embracing technology. 'In future, only technology will be used' or 'technology is everywhere' were common expression to describe this perception. Teachers' valuing of the gained ICT skills and subsequent perceived employment opportunities, then, not only related to exploring new opportunities, but as much to being able to stay employable in a changing time.

Economic capital—to be able to build economic capital

Teachers also applied the tablets to explore different aspects of their economic capital. This exploration drew in particular on the connectivity and access to information affordances of the tablets. Teacher highly valued the ability to access and upload employment related information. This opportunity was two-fold as one teacher explained:

"You have more chances. Even if you are looking for a certain post, it is easier to get it. First, you can know about it online first before everyone else. Second, you can advertise your own work online. People can find you." (High school teacher; day 13).

This quote illustrates the parallel aspects of using the tablets to enhance economic capital. Teacher employed the tablets widely to gain access to more up-to-date employment information, for example posts advertised by the Department of Education. The tablets allowed them to become aware of opportunities ahead of others as the usual advertisement channel was a paper-based monthly bulletin distributed to all schools. As a result, teachers reported to be able to have advantages in their job search and that they had an increased choice and access to employment-relevant information. But, teachers also valued the capability of being able to proactively advertise themselves online. Teacher referred for example to their 'profile' online, which usually related to accounts they maintained on LinkedIn and SchoolsNetSA, and that people were now able to find them.

A further aspects of the economic capital capability relates to the ability to share this employment-related information with learners in order to guide their career paths. This capability overlapped with teachers' professional goal of supporting learners' to succeed and often was valued particular highly in references to information and careers paths that were not accessible to teachers themselves when they were in school. Below one teacher describes how a lack of relevant information and career advice in her student days almost led to her becoming a nurse rather than a teacher:

"Teaching was my second choice and it was nursing that was my first choice. I was so stupid, I only liked it because of their [nurses] uniforms. That is a rural environment for you. I didn't know

better and no one gave me any other option. And what was it with the uniform? Why would you want to be a nurse just because of the uniform? But back in the day all I knew about was a policeman or a nurse—you don't know about other professions. Today this would be much better and learners can find out about the world. Of all, learners have more information about this also because we are more informed as teachers. Now I can tell them to google all the jobs they can do with a matric and explain to them what is what." (Primary teacher; day 6).

Teachers valued this opportunity to guide their learners to make more informed professional choices. Again, the main emphasis was on the intrinsic value of having a choice and knowing one's options, rather than prescribing a particular career path or choice.

Online education—to be able to extend educational opportunities

Teachers also applied the tablets to enhance educational opportunities for themselves and others. These opportunities were framed in narrow instrumental terms as either increasing teachers' own qualifications or as supporting their own and other relatives' children's academic achievements. In terms of their own qualifications, a sizable number of teachers were enrolled with distance learning institutions to further their formal qualifications. The University of South Africa's BeD Education was the most popular programme and teachers highlighted how much easier the administration and pursuit of their distance learning studies had become due to the tablets. For example, before obtaining the tablets, teachers had to use the internet cafe in the districts capital to submit their assignments, which presented a challenge for many given the travel distance and operating hours of the cafe. Content-wise, teachers involved in distance learning courses also valued the ability to engage in research for their course assignment on their tablets.

But, equally so, teachers valued the extension of educational opportunities through the tablets in terms of providing education to others after school hours. This usually referred to teachers' own children and/or children of teachers' relatives. The use of tablets for commercial after-school tutoring was not reported. Interestingly, the opportunity to make use of online education was valued so highly that teachers reportedly installed WIFI connections in their homes too in order to allow their spouses and children to supplement their educational efforts without depleting the teachers' mobile data allocation.

Marketing-to be able to 'build the brand'

Teachers also reported a valued economic opportunity related closely to the ICT4RED programme itself. The capability to use the ICT4RED brand to support their own schools and reputation. Teachers reported that the news of the schools' approach to teaching with

technologies had spread rapidly throughout the community leading to an increase in enrolment of learners at ICT4RED schools. Teachers valued this trend and perceived it as a recognition of their schools' ability to provide high-quality education by local communities. Teachers also derived personal benefits of being associated with the ICT4RED programme and reported that community members dubbed them 'technology teachers' —a reputation teachers received positively. This combined valued functionings obtained from being able to market oneself as part of the ICT4RED was captured in the below quote.

"They recognise us, in fact not only the community the entire district in the entire Eastern Cape. They all knows us. When we go to the workshops we used to take pictures with our tablets and everybody was so impressed. Other people are asking when this project is coming to them and they want to know if it is working. And believe me it is working so much for the children. We even have more learners coming to us now. I have over 50 [learners] now and I had 30 last year." (Primary teacher; day 7).

Digital services—to be able to access digital services

A final economic capability referred to the use of tablet devices to explore a range of practical applications related to digital services. Teachers mentioned using the tablets to make online appointments, use online banking, access online call rates, and order catering service, among other. Access to these digital services was valued highly for teachers residing in remote locations as they otherwise would have to travel to the nearest urban centre to conduct their business. Digital services had economic value as it saved teachers time and money. The opportunity to have access to digital services further was valued as teachers assumed it to be the 'advanced' way of doing things and did not want to be excluded from it as the below quote shows:

"Now we are living in an advanced technological world whereby you don't even have to queue to go to the bank in order to send money to your kids. You can use your e-wallet. And it makes life easier. You can even make an appointment through your phone. So teachers need to change too." (Primary teacher; day 9).

7.3.4 Societal capabilities

The mastery and application of the tablets as an educational tool also did affect teachers' societal capabilities, which is the fourth and last dimension of capabilities identified in my case study. Societal capabilities refer to teachers' capabilities to explore and pursue valued being and doings in society as a result of gaining access to and proficiency of the tablets. This dimension of capabilities thus encompasses valued opportunities in society, for example being respected as a professional, that teachers were aiming to achieve through the educational use of tablets. The teachers involved in the ICT4RED programme

appropriated the technological input to support five societal capabilities, which consist of 20 reported functionings (Appendix 7.2). The five societal capabilities are listed below and discussed in return:

- Professional identity—to be able to create a professional teaching identity
- Respectability-to be able to be impress
- Recognition—to be able to be 'on the same page'
- Self-actualisation—to be able to grow as a person
- Community development—to be able to support the local community

Professional identity-to be able to create a professional teaching identity

A major source of valued being and doings reported by teachers was the capability to be regarded and to present themselves as professionals. Teachers applied the tablets in a variety of ways in order to enhance their standings as professionals and to display their professional identity. The ability to apply the tablets in this pursuit of constructing a professional identity was aided heavily by the intrinsic association of the possession of tablet devices with being a professional. As one teachers remarked: "People just assume that you are a more professional being. They are impressed automatically by just seeing you with the tablet." (Primary teacher; day 1). In the rural context of the Nciba district, the mere possession of tablets was regarded as a social and professional status symbol. Teachers were able to exploit this perception of the devices and applied it in order to support their standing as professionals.

Teachers enjoyed their possession and use of tablets and often referred to how they 'looked like someone who is professional and organised'. In a telling quote one teachers observed that she appeared as "someone who knows what you are doing" (Primary teacher; day 5). A large part of teachers' opportunity set to use the tablets to construct a professional identify was thus driven by outside perceptions rather than observed changes in teachers' actual behaviour. This positive outside perception, however, set in motion a virtuous circle of encouragement for teachers who observed this shift in outside perceptions carefully and used it to carve a niche for themselves as professionals. This sentiment is captured well in the below two quotes:

"Yes It makes us look more professional than before. Before we normally took ourselves as layman, as traditional teachers; now we can see that we are modern teachers; we are teachers of time." (Focus group; day 17).

"It has given us confidence to feel that we are teachers and that we are unique from other

teachers." (Focus group; day 17).

Here it becomes evident that teachers not just adopted the narrative of being professionals. They further defined themselves as 'modern' and professionals 'of time'. The teachers involved in the ICT4RED therefore used their functionings as teachers who use technologies to define their professional identity as technology users. In line with this constructed professional identity, teachers rebranded their school's names during the public ICT4RED graduations in the communities as 'Technology University of Mdanstane' or 'Thembalethu Institute of Technology'—emphasising their ability to use technology in an educational setting.

Teachers valued to be regarded as successful and qualified in their profession and pointed to a range of examples how these experiences supported their professional identities. In a context of low pre-existing perceptions of the teaching profession, teachers were in particular pleased to be able to attract learners into the career path of being a teacher. This attraction was directly supported by the ability to use tablets in an education context as expressed in the below quote.

"We have some learners now, when they talk about careers they tell us now that as you [teachers] are advanced we also want to become teachers. Our country is providing you with technology and I can follow the same teaching profession as our government is updating you." (High school teacher; day 11).

In sum, teachers felt more valued in their profession and were able to carve out a professional niche identifying as technology teachers. Such a positive professional identity was often lacking as explained in the next section, and using the tablets to support the construction of such an identity was a highly valued opportunity—the opportunity to become 'a full teacher'.

Respectability-to be able to impress

Linked to their capability to construct a valued professional identity was the capability of being able to impress. This referred to teachers' valued functioning of being respected by learners, other teachers, and the wider community. This active desire for respect and the subsequent application of tablets to pursue it comes in the context of South African teachers having been subject to extensive public (including government) criticism due to their perceived lack of performance (e.g. Motala et al 2012). How important the functioning of being respected is to teachers shows the below quotation:

"It [teaching with tablets] makes us being more respected. They think we are BEE, because we

have those fancy tablets. They think we are the most important people. Now we have dignity as teachers." (Focus group; day 17).

The teachers here refer to the dignity of teachers and in a previous quotation teachers referred to themselves as 'laymen'. This language seems to give a sense of the low professional perceptions teachers had of their own occupation and how it affected their societal capabilities. In the above quote, BEE stands for the South African government's 'Black Economic Empowerment' scheme, an affirmative action programme aimed at previously disadvantaged groups to create a wealthy African middle-class. The reference shows how, in this rural setting, the possession and use of mobile technologies is intrinsically associated with being successful and qualified. It is a stark shift from the perception of teachers without dignity to teachers being on par with BEE professionals. The opportunity to elicit respect and increase the standing of the teaching profession presented an important societal capability for teachers.

Recognition—To be able to be 'on the same page'

A second capability linked to teachers' construction of their professional identity and quest to be respected was teachers' capability to be recognised. This capability was more specific than the overall capability of being respected. Teachers' reference of the value of being recognised was linked to specific grievances that they had experienced as part of their teaching positions in the rural Eastern Cape. At the most micro level, teachers felt the tablets allowed them to be on the same page as Model C schools in the Eastern Cape, such as Selbourne. Model C schools refer to public schools in the highest income quintile of South Africa's schooling districts, which too have the highest educational achievements across South Africa's public schools. Teachers repeatedly expressed statement such as 'now we are on the same level as Selbourne' or 'I can go teach at Selbourne now'.

The Eastern Cape is South Africa's worst performing region in terms of educational outcomes. At the time of submission, the provincial education department is under national administration due to its inability to provide educational services to all learners in the province. This dimension of deprivation was also repeatedly mentioned and teachers remembered how their friends and colleagues from other provinces had 'looked down on us'. The opportunity to become technology teachers, however, changed these relations and teachers felt that they were able to level the playing field. In a personal anecdote one teachers recalled:

"For example last year I was at the funeral and so then my friend from the Western Cape attended as well. And I wanted to boost in front of her. So I took out the tablet and recorded the funeral because they always say education is better in the Western Cape. So I wanted to show what we have in the Eastern Cape. She was so impressed." (Primary teacher; day 15).

The last reference of being able to be on the same page referred to teachers' perceptions of spatial inequalities in the education system. Teachers expressed hopes that the use of tablets at their schools might allow them to balance the flight of learners from rural to urban schools. In this, teachers assumed that learners and their parents could realise that they can receive 'modern', high quality education at their rural schools too; and that therefore there would be no need to send their children to more expensive urban schools. The educational use of tablets and their intrinsic positive associations with educational quality was thus valued by teachers as a means to enhance the status of their own rural educational contexts.

Self-actualisation—to be able to grow as a person

Teachers' societal capabilities of being able to construct their professional identity, gain respect, and recognition as professionals then directly translated into their perceived capability to grow as a person. This process of self-actualisation was a highly valued societal capability as it directly affected teachers' social well-being. Teachers described themselves to move from a state of 'being blank' or 'uneducated' to a state of 'being enlightened' or 'uplifted'. Being able to master technology, assume a more positive professional identity, and to support their learner academically left the teachers 'proud', 'confident', or 'accomplished'. Using the tablets in support of their professional societal capabilities therefore spilled over into teachers' personal development outcomes and their general self-perceptions. Teachers' ability to 'grow as a person' through exploring the use of tablets at the intersection of their social and professional contexts is captured well in the below quote:

"The tablets increased my well-being. I have become a trainer by training myself. It equipped myself with a lot of skills. I never knew I could divide my attention like this. Now I can even present for an older audience. I can facilitate anything now. And before I didn't have that skill but now I do. I was afraid of presenting to them, but now I'm the expert of ICT." (High school teacher; day 14).

Teachers further framed this capability of self-actualisation in terms of their professional contribution to society. As explained above, a central valued educational capability was the opportunity to support the students' academic success. A core part of teachers' professional and social self-actualisation therefore was linked to enabling this educational success of students. Teachers referred to this valued functioning of seeing their learners achieve as their 'patriotic duty' and 'contribution to society'. Another metaphor used to evoke this sense
of social and professional self-actualisation referred to the 'production of professions' by teachers. As one teacher explained:

"I value being a teacher because you can produce more professions from you. I produce doctors, politicians, and businessmen. Even you [points at me] are a product of a teacher. I'm very proud of that. And the tablet has helped me to produce more. (High school teacher; day 11)

The last sentence of the quotes indicates how the technological input fits into this teacher's construction of their societal capability of being able to grow as a person. The use of the tablets is appropriated and embedded into an interconnected set of different dimensions of educational, informational, and societal capabilities, whose interplay allows the teacher to pursue their personal well-being and development objectives.

Community development—to be able to support the local community

The final societal capability emerging from the data referred to teachers' opportunity to support the local community. This capability has been touched on already with related functionings being reported as part of the capability of building 'economic capital' and 'building the brand'. At its core, the community development capability referred to the spill-overs effects of teachers being able to provide higher quality education and the status associated with schools teaching their pupils with tablets. In addition to the increase in student enrolment, teachers also reported community members visiting schools to experience the tablets for themselves. Teachers described extensively how excited local communities and partners were about the provision of tablets to their local schools and how it became a source of local pride. Parents, as much as teachers, believed the tablets to allow their learners to obtain higher grades and that being skilled in ICT usage would increase their employment prospects. As one teachers observed, "they [parents] see the tablets as an integral part of their children's future". (Primary teacher; day 10).

Teachers valued the opportunity to contribute to the local community development through the tablet programme. For one, it increased their own status in the local community linked to their professional identity. More so, teachers also reported a range of practical benefits of being able to support the local community. At one school, regular break-ins reportedly ceased after the start of the ICT4RED programme, which teachers interpreted as reflecting the community's recognition that the teachers now provided a quality education to the learners and that stealing the tablets would compromise the learners' opportunity of gaining access to tertiary education. At another school, the local community reportedly contributed materials to build a storage for the tablets. In all schools more positive relationships and interactions between the school and local communities were reported by the school's principal.

Conclusion

This chapter has presented the results of my qualitative case study of a ML4D programme in rural South Africa. I provided qualitative data on how teachers involved in the ICT4RED programme used the mobile technologies to expand their capabilities. That is, I collected data on how teachers appropriate the technologies to aid the pursuit of what they had identified as valued beings and doings. Through this, I derive a list of valued opportunities that the teachers themselves attempted to explore by using the mobile devices. I find that these valued opportunities can best be understood as four thematic dimensions of capabilities: informational, educational, economic, and societal capabilities.

In this chapter, I have described the four identified dimensions of capabilities and to provide evidence of the underlying data for each. I have made minimal references to the detailed interplay between the dimensions of capabilities and how they relate to teachers' well-being and human development. Throughout the presentation of the four dimensions of capabilities, however, these intersections of capabilities have been hinted at. I therefore next transform my case study findings into a more formal analytical framework of teachers' use of mobile technologies from a capabilities perspective, which is outlined in chapter 8.

Chapter 8. Conceptualising mobile learning for development from a capabilities perspective

Introduction

This chapter uses the findings from my qualitative case study to develop a conceptualisation of mobile learning for development (ML4D) based on the Capability Approach (CA)⁷⁸. To be more precise, it first presents a descriptive framework to synthesise and visualise the main case study findings on teachers' use of mobile technologies in a resource-constrained setting from a capabilities perspective. The framework is derived from the valued functionings and capabilities that teachers reported to explore as part of the Information Communication and Technology for Rural Education Development (ICT4RED) programme. I synthesise these primary research findings into a descriptive framework to outline the interplay and relations between the four identified dimensions of teachers' capabilities. To recall, in the case study I identify four dimensions of valued beings and doings and corresponding capabilities that teachers reported: informational, educational, economic, and societal capabilities. In this chapter, I explore how these four dimensions of capabilities relate to teachers' well-being and human development and how these relations can be used to guide a conceptualisation of ML4D based on the CA.

The objective of formalising my case study findings into a descriptive framework is to generate a visual tool to represent the case study research results on assessing ML4D from a capabilities perspective. As explained in chapter 4 and 7, my qualitative case study of the ICT4RED programme served an instrumental purpose of operationalising the CA as a conceptual device to assess ML4D. This operationalisation is of interest to my wider attempt of providing an alternative conception of ML4D based on the CA. Therefore, I require a descriptive overview of my case study findings in order to translate these findings into a foundation on which to establish my alternative positioning of ML4D, which the framework provides.

This chapter is structured as follows. I first present and describe the framework. I then

⁷⁸ In chapter 9, I will lastly bring together the developed capabilities conception of ML4D with the systematic review findings.

expand on the case study's findings in understanding the interplay between the four dimensions of capabilities and how these related to teachers' well-being and human development. To illustrate this interplay in more detail, I then provide three vignettes of the intersectional nature of the dimensions of capabilities and how teachers' explored their conception of a good life through the pursuit of overlapping capabilities. I then use these insights to propose a conception of ML4D from a capabilities perspective. The chapter ends with an overview of the key findings presented and a brief conclusion.

8.1 A framework of teachers' use of mobile technologies as an educational tool from a capabilities perspective

This section introduces and outlines my descriptive framework for teachers' use of mobile technologies as an educational tool based on my case study findings of the ICT4RED programme. I first outline the key components of the framework before discussing how the framework should be interpreted and the limitations to its application.

8.1.1 Introducing the capabilities framework

My framework of teachers' use of mobile technologies as an educational tool from a capabilities perspective is presented in figure 8.1 and explained here in detail. As outlined in chapter 7, my qualitative case study of the ICT4RED programme found that teachers perceived their educational use of tablet devices to contribute to four dimensions of capabilities: informational, educational, economic, and societal capabilities. This distinction between the four dimensions of capabilities became apparent after reviewing the initial list of capabilities developed in the thematic analysis of the case study data. This list consisted of 21 valued opportunities that teachers explored using the mobile technologies. It reflects the valued beings and doings that teachers pursued through the conversion of access to mobile technologies. Reviewing this list, I identified the four dimensions under which the 21 capabilities could be grouped. For example, a number of capabilities were directly related to valued educational opportunities, such as to be able to improve learners' educational attainment and to be able to act as facilitator of knowledge. Other capabilities, however, did not fit the label of valued educational opportunities and were linked closer to teachers' economic opportunities or valued beings and doings related to the use of mobiles' technological affordances. As a result of this thematic grouping, four dimensions of capabilities were identified, which do have some overlap between individual capabilities, but in themselves present distinct dimensions of teachers' valued capabilities.

These four dimensions of capabilities—alternatively also referred to as dimensions of opportunities (Sen 1999; Deneulin 2009; Gasper 2007)—present the building blocks of the framework and are presented in four boxes in figure 8.1. In each box, the relevant capabilities are listed with an abbreviation in brackets to ease overview. It is important to keep in mind that these lists of capabilities are not normative and only attempt to reflect the range of capabilities valued and explored by the teachers involved in the ICT4RED. As described above, each capability listed in the boxes was derived inductively from teachers'

reported and observed valued functionings.

Figure 8.1 A framework of teachers' use of mobile technologies as an educational tool from a capabilities perspective



The first proposition of the framework is then that the use of mobile technologies to enhance teachers' capabilities to live a life they have reason to value can be understood to comprise four dimensions of capabilities. Each dimension represents a different set of valued opportunities that teachers perceived to result from the application of the tablets in an educational setting.

The second proposition of the framework is that teachers' human development comes as a result of the interplay between these four dimensions of capabilities, not through a pursuit of individual capabilities (see 8.2 for more detail). To illustrate this, human development is placed in the centre of the framework in a circle crosscutting all four dimensions of capabilities. This circle in the centre represents teachers' conception of a life they reasoned to value.

However, this is not to suggest that there exist a singular formula or configuration of capabilities that reflects and contributes to teachers' human development and well-being. Teachers will vary in their exploration of capabilities as they vary in their conception of the good life and teachers' configurations of capabilities will therefore vary between and within different dimensions of capabilities. To illustrate that this interplay of different dimensions of capabilities is not unidirectional and does not have to involve all four dimensions, the framework indicates four areas of interplay between different sets of two dimensions of capabilities that translate into teachers' well-being and human development. These are positioned between the four boxes and indicated with arrows. For instance, combining their educational and societal capabilities, teachers were able to explore alternative ways to conceptualise what being a teacher means, which translated in reported contributions to their human development and well-being. *The emphasis of the framework is therefore firmly on the interplay between capabilities and dimensions of capabilities*.

Lastly, the framework illustrates the centrality of the mobile technologies and the mobile learning programmes in teachers' exploration of the reported capabilities. Each dimension of capabilities is explored by the use of mobile technologies, which is articulated in each of the four boxes clearly. The framework presents a bottom-up view, based on teachers' own perspectives, about what value the educational use of the tablets contributed to their lives. The mobile technology and its educational use as part of the mobile learning programme is therefore informing every single aspect of the framework. That is, each capability presented on the framework, was explored by the teachers through the appropriation of the mobile devices as a tool to explore valued opportunities in line with the teachers' definition of a

good life⁷⁹. Mobile technologies and the applied mobile learning programme can therefore best be understood as a resource that contributes to the capabilities; but the use of the technology, or the technology itself, cannot be linked to a single dimension of capabilities. Teachers' use of mobile devices in an educational setting was framed and understood by themselves in multiple dimensions of opportunities and promoted their human development and well-being through a combination of different capabilities—which is illustrated in more detail in section 8.2.

8.1.2 How to interpret and apply the framework

There are a few aspects to consider when interpreting my capabilities framework and how it could be taken forward. First, it is important to reiterate that my identified list of 21 valued capabilities is not a normative framework for the use of mobiles as an educational tool in Low- and Middle-Income Countries (LMICs). That is, it should not be seen as an attempt akin to Nussbaum's version of the CA to develop a normative list of capabilities that all ML4D programmes should strive to attain in order to support human development. The 21 listed capabilities as well as their four corresponding dimensions are a synthesised representation of the rich set of opportunities valued and explored by the 25 teachers in my case study. These capabilities should be seen as descriptive and indicate that these 21 capabilities are important aspects that contribute to teachers' conception of a good life; but they do not claim to represent a conception of the good life itself. The framework can therefore not be used as a tick-box exercise for mobile learning programmes assuming that that the use mobile technologies in an educational setting can only be linked to human development and well-being if all 21 capabilities are enhanced.

Second, and following from this rejection of a normative reading of the list of capabilities, the framework is firmly meant as a descriptive overview of my case study results. The case study's key argument that the interplay between the four dimensions of capabilities allows mobile learning programmes to make a contribution to human development should not be seen a normative attempt to define ML4D either. That is, I am not putting the framework forward as my conceptualisation of ML4D comprised of the four dimensions of capabilities and their interplay. The framework is too bounded by the case study data on which it is based in order to make a universal claim to the applicability of the four dimensions of capabilities as a conceptualisation of ML4D at this stage. Rather, I make a normative case

⁷⁹ This is of course not to say that these capabilities can *only* be explored through the use of mobiles. In the context of this research and my findings, however, the capabilities were primarily explored through the use of mobiles, which the framework attempts to capture.

for using the CA as an analytical tool to conceptualise mobile learning's contribution to development, which is outlined in 8.3.

Third, the framework is not an empirical measurement of capabilities or well-being or agency achievement. It should be clear from my methodological set up in chapter 4 that the framework is derived from a qualitative case study that assessed teachers' reported and perceived opportunities following participation in a mobile learning programme. The framework is not a reflection of an impact evaluation of teachers' freedom to achieve well-being and agency or their actual achievement as provided by other capabilities scholars such as Buckler (2015) and Brandolini and D'Alesio (2009); neither does it attempt to rank or weight different capabilities or capability sets in their contribution to human development as for example suggested by Walker (2006).

How should the framework be interpreted and applied then? I would suggest to read the framework in line with Sen's version of the CA, which is much less prescriptive than Nussbaum's version. That is, the developed framework presents an important contribution to the literature on ML4D as it illustrates that the CA can be operationalised to assess the effects of mobile learning programmes in LMICs. Based on this empirical verification of the operationalisation of the CA as well as the conceptual argument for the CA outlined in chapter 2 and 3, I contend that there is a strong normative case for applying the CA to guide a conceptualisation of ML4D. This includes a strong case for the use of the CA as an analytical framework in primary research on ML4D as evidenced in my case study.

In sum, I would position the framework as an open-ended tool to guide empirical and conceptual research on ML4D. It does not present a conceptualisation of ML4D. It does, however, present a structure to unpack mobile learning's contribution to development in terms of four dimensions of capabilities and their interplay in supporting human development and well-being. This investigation is illustrated in detail next.

8.2 Development as freedom and ML4D

Sen's definition of development as freedom repositions development to be evaluated in terms of an expansion of capabilities. This is both the primary end and the principal means of development (Sen 1999). In my framework, the central circle presents enhanced capabilities as an end, whereas the capabilities in the boxes present enhanced capabilities as a means of development. This section is concerned about the interplay between capabilities as a means of development and how this interplay translates into the final outcome of enhanced capabilities as a primary end of development. For ease of discussion, I am referring to the enhancement of capabilities as an end of development as human development or well-being.

In the context of my case study findings, development thus refers to teachers' effective freedom to live a life they have reason to value⁸⁰. This positioning of an expansion of freedom—defined more technically as capabilities—inescapably focuses on individual's—teachers'—agency as a core component of the process of development. Freedom cannot meaningfully be supplied top-down or provided access to; freedom needs to be explored and generated by the teachers themselves to have meaningful value. Capabilities without the agency to explore them in a self-determined manner do not present effective freedoms. Teachers need to be able to conceive their own conception of well-being and human development and then to be able to explore this in a self-determined manner. The argument presented in my case study is that the ML4D programme supported teachers' self-determined exploration of their conception of a good life; if so, we can then speak of mobile learning for development as *freedom*.

To be more precise, this argument can be unpacked further to include my four dimensions of capabilities and the centrality of their interplay. This leads to an expanded conception of mobile learning for development as freedom as: that the ML4D programme supported teachers' self-determined exploration of *four dimensions of valued capabilities whose interplay enhances teachers' reported human development and well-being*. This expansion seems necessary to highlight the four dimensions of capabilities and their interplay which are a unique result of the particular nature of ML4D interventions. While I by no means claim that any of the identified capabilities can only be enhanced through ML4D interventions, their simultaneous exploration is a key contribution of ML4D programmes. It is precisely this simultaneous exploration which can bring about teachers' human development and well-

⁸⁰ Note that this definition differs from most definitions of development in the wider ML4D literature.

being and the facilitated interplay between different dimensions of capabilities therefore constitutes ML4D's specific contribution.

I next attempt to unpack this abstract discussion of the need to focus on the interplay of dimensions of capabilities with two practical examples from the ICT4RED case study.

My argument that to frame teachers' well-being and human development in terms of only a single capability dimension (or single capability) would be overly narrow and neglect the diverse ways in which teachers appropriate their usage of the technology to foster their agency and well-being is illustrated in the response of a primary teacher (Day 5):

'I like imparting knowledge to learners. Besides, I am a parent to them. I value it and it doesn't end in the classroom. Even after they go to high school and tertiary you wonder how they did in life? For example, I didn't do some subjects in high school I wanted to do; I couldn't do BCom as it was those days [Apartheid]. But when I taught BCom for the first time, I felt like I was fulfilling my past dream and the tablets gave me the access to the content so I can get more information. Now my first group of learners is at university level and they do BCom accounting and economics at the university in Cape Town. So, I am so proud of that and I always check their progress using my tablet through WhatsApp and Facebook.'

In this excerpt it is evident how the teacher constructed her professional and social identity in multiple and overlapping ways (being a teacher and being a parent) and how she navigated the newly gained capabilities from using the tablets to pursue what she defined as valued beings and doings. This teacher has discovered that the use of the tablet supports her with a valued opportunity that she was denied by institutional discrimination during Apartheid. In a literal sense, she did not have the freedom to study for a BCom, but during the ICT4RED programme explored whether the tablet could support her to start teaching her own pupils in the subjects that she was denied. This valued educational capability then is further enhanced by the informational capabilities of remaining connected with her pupils throughout their tertiary careers. The appropriation of the technology to pursue multiple dimensions of capabilities has provided this teacher with the agency and capability to transform her educational and social reality in line with her own definition of freedom.

This exploration of capabilities linked to past unfreedoms was a rich theme in collected data. Without leading too much on numerical frequencies, 11 out of 25 teachers made this reference. Teachers explained how they could not study for certain subjects or pursue certain professions because of growing up under the Apartheid regime. They appropriated the use of tablets to explore whether they could now address some of these past

229

unfreedoms by providing their learners with the opportunities denied to themselves. To recall, supporting their learners' academic and professional success was a core capability and part of teachers' professional identity. Being able to apply the tablets, for example through accessing content and lesson plans linked to subjects they had not been formally trained to teach, enhanced this valued capability and through merging it with the pursuit of an opportunity denied to themselves became a major source of well-being and human development. The value teachers attached to this pursuit is evident in their subsequent efforts to stay in touch with and to continue support their learners, a capability that again is supported by the use of tablets.

A second illustration of this concept of addressing unfreedoms or barriers to their human development, emerged in relation to rural education. Teachers illustrated how the rural areas in which they taught were systematically excluded during Apartheid and how rural education was still perceived as inferior to urban education. The provision of 'modern' mobile technologies in a rural education context was in stark contrast to general perceptions of rural education with one teacher expressing a common sentiment in a focus group of: "we were thinking we are not supposed to get this technology and do this [teaching with technology]. It is something we have been crying for." (Focus group; day 8)

Teachers, in particular younger ones, explained how they had a strong interest in technology in general, but that neither the social or professional context in which they operated provided them with an opportunity to explore their interest in technology. This narrative of not having had the freedom to explore technology was then transformed as part of the ICT4RED into a configuration of valued opportunities such as 'ICT usage', 'Innovation', 'Self-actualisation', and 'Marketing'. Taken together, this configuration of capabilities and the agency to explore them, in the context of past deprivation and structural discrimination, presented a major avenue to teachers' well-being and human development. This notion is summarised in a teacher's own words as:

"It [ICT4RED] has changed me, I have changed. I have always been interested in technology and now it has changed my life. Even at home when my children ask me something unpleasant, I just google it and now they say how come you've been so advanced? I have grown. It has made a difference in my life. Firstly, I'm computer literate now. Secondly, it supplements my teaching methods. It is also my office. It gives me the option to teach, I can either use that or not use it." (Primary teacher; day 9)

To conclude, to pursue their own conception of a live a life they valued teachers explored the use of tablets to support a configuration of overlapping dimensions of capabilities. In this process, they appropriated the technology to explore valued beings and doings not directly related to the intended ICT4RED programme design, and closer related to addressing past and current unfreedoms. It is this agency or ability to choose a life one has reason to value rather than following an exogenous or predefined notion of the good life that constituted a major source of well-being and development. Adapting Kleine's (2013) question of 'technologies of choice?', this then opens the question of whether tablets can be regarded in this ML4D programme as a 'technologies of freedom'? I will return to this question at the end of this chapter in section 8.3 when I have presented a more diverse set of configurations of dimensions of capabilities that teachers explored to further their well-being and human development. In this remit, I next present three brief vignettes capturing a diverse set of these capability configurations⁸¹.

⁸¹ Note that figure 8.1 presents four vignettes. The fourth vignette, 'alternate ways to teach: the teacher as a facilitator of knowledge' is discussed in chapter 9, section 9.3.2.

8.2.1 Vignette: Building the nation

"As a teacher you are like a leader, you are a role model. If you see the child doing well you get this patriotic feeling because you are building the nation. The teacher is a builder of the nation and the tablets have helped me a lot to build the nation." (High school teacher; day 13).





A key configuration of the dimensions of capabilities referred to teachers' appropriation of tablets to pursue their objective of 'building the nation' (Figure 8.2). Building the nation expresses a narrative generated by South Africa's ruling party after leading the country through a peaceful transition from apartheid dictatorship to democracy, and calls on teachers to educate the first generation of 'born-free' South Africans (Motala et al 2012; SADTU 1996). Teachers' professional mandate is since framed at delivering high-quality education to contribute to the nation's socio-economic development. It presents their professional contribution to society.

The teachers interviewed were highly supportive of their role in building the nation and flourishing in this role was reported as a major component of their professional identity as teachers and the life they valued. Teachers derived professional and social well-being from

being able to contribute to building the nation. As a result, teachers actively arranged the capabilities derived from using the tablets to shape their contribution towards building the nation. Figure 8.2 provides an example of a capabilities configuration to pursue teachers' goal of building the nation: The capabilities of being able to use mobile technologies and to gain access to more information (informational capabilities) contributed to teacher' educational opportunities. Being able to integrate mobile devices and the enhanced access to information into their teaching efforts gave teachers additional opportunities to support learners through more diverse and innovate teaching methods (educational capabilities). Being able to help learners achieve in return supported teachers' social capabilities as they felt they were fulfilling their destined role of building the nation and thus earned enhanced status and recognition in society. It lastly also supported their economic opportunities vested in the assumption that the school and the teachers would be rewarded for contributing so effectively towards national development.

This short vignette illustrates how teachers applied and appropriated the mobile technologies in a self-determined matter to pursue their personal objective of contributing towards the building of the nation. Building the nation was a core component of the life they had reasoned to value and the teachers configured overlapping dimensions of capabilities in order to be able to pursue this source of well-being and human development.

8.2.2 Vignette: Alternative ways to conceptualise what being a teacher means

"We are more respected. We knew all about laptops and tablets of other office people, so now we have them too. It [tablet] promotes the standard of teaching, even children now want to be teachers. They see that teachers are acknowledged and advanced." (High school teacher; day 12).





A second way in which teachers arranged different dimensions of capabilities to enhance their well-being and human development was to pursue the exploration of an alternative and more positive professional identity (Figure 8.3). The need for this professional identity arose in the context of low public perceptions of the teaching profession as well as government and media criticisms paired with challenging working conditions and modest salaries (Motala et al 2012; Booyse et al 2011). The construction of this professional identity encompassed a range of different capabilities, an example of which is provided in figure 8.3. To start with, teachers reported how their use of tablets as educational tools facilitated a more enjoyable teaching experience. For example, the use of tablets eased some administrative burdens and allowed teachers to apply strategies that made it easier to control the classroom. A second valued educational and informational capability supporting increased occupational well-being related to the use of tablets to connect and collaborate, leading to less isolating

teaching experiences.

This increased perception of a more enjoyable teaching experience then was complemented by a change in outside perceptions of the teaching profession too. Societal capabilities such as recognition and respectability explained how teachers' status within the local community and within schools themselves had increased due to the mastery of tablets as an educational tool. But, a key in the configuration of these capabilities to support teachers' construction of a professional identity to further their well-being was the capability to merge this identity across contexts. That is, the tablets' technological affordances allowed teachers to transfer and apply their professional identity in social contexts. For example, teachers used the tablets to showcase their technological and educational skills at social events in order enhance their standing within communities. Likewise, teachers transferred their social identities to the classroom by showcasing to their colleagues and pupils important social achievements, such as leading the community prayer. As a result, teachers could merge in and out of their preferred identities in changing contexts.

In sum, teacher appropriated the use of the tablets to explore a range of capabilities linked to their own construction of an alternative and more positive professional identity. These capabilities allowed them to feel more valued in their profession and to practice teaching in a proud and confident manner, which then was transferred across contexts to support both their social and professional well-being.

8.2.3 Vignette: We are moving with time—we are in demand

"A perfect teacher is one that is not left behind by time. If you don't know technology you know nothing currently. Because we are now living in an advanced technological world whereby you don't even have to go to the bank to queue. Being a mobile teacher gives me more choices where to work and how to work." (Primary teacher; day 6).





A third configuration of different dimensions of capabilities that allowed teachers to pursue a life they had reasoned to value referred to teachers' objective of being able to fully participate in the 'modern' society and economy (Figure 8.4). Teachers applied the tablets to explore a range of opportunities mitigating their perceived fear of being left behind by time. This fear was a major component of perceived unfreedoms and reinforced by the rural education context in which the teachers operated. As a result, the teachers directly applied the tablets to explore a range of functionings to stay up-to-date. This referred initially mainly to informational capabilities such information literacy and ICT usage, which were valued intrinsically as teachers assumed that they presented the required key capabilities in the 21st century.

However, teachers then integrated these informational capabilities into their educational and

economic capabilities to enhance their agency to mitigate the fear of being left behind by time. Being a teacher of time who masters and integrates mobile technologies in her teaching strategies became an integral part of teachers' professional objectives. This valued educational opportunity then further translated into economic capabilities since, as a teacher of time, it would be easier to secure new and current employment opportunities. Instead of being left behind, teachers perceived the opportunity to be able to move with time and benefit professionally from possessing the ICT skills that were in demand. Further configurations included using the informational capabilities to enhance one's own educational efforts through distance learning and thereby progression in the teaching career; or, alternatively, to use one's informational capabilities to share access to economic opportunities to thereby support societal capabilities such as community development.

This last vignette again illustrated the intersectional nature of the dimensions of capabilities. Teachers' human development and well-being did not result from the mere possession of informational capabilities or the ability to integrate technology into their teaching activities. It resulted from the opportunities that these configurations of capabilities created in their self-determined pursuit of a good life—in this case the agency to not be excluded from the 'modern' society and economy and to be able to fully participate in the professional chances and choices associated with both.

8.3 Towards a capabilities conception of ML4D

My case study of the ICT4RED programme outlines teachers' use of mobile technologies as an educational tool from a capabilities perspective. It attempted to test the feasibility of the CA to assess and unpack the effects of an ML4D intervention. In this section I will move beyond the insights gained grounded in the case study data and explore the implications of my case study findings for a wider conceptualisation of ML4D using the CA as relevant theoretical foundation. That is, I am aiming to use my case study results to develop a capabilities conception of ML4D.

My case study research provided an empirical grounding for the feasibility of the CA as a conceptual lens to unpack the effects of ML4D interventions. It illustrated how teachers' own experiences of using tablets as an educational tool as part of a mobile learning programme in rural South Africa could be framed and analysed using the CA. This analysis led to the identification of 21 valued capabilities that teachers explored through the use of tablets in a rural educational setting. The CA therefore served as an effective conceptual framework to investigate the links between the use of mobile technologies, education, and development in LMICs.

Following the successful application of the CA as an analytical tool, I then expanded on my case study data to unpack how the identified capabilities could be understood in relation to Sen's notion of development as freedom—the final outcome of development in a capabilities conception of ML4D. In this, I make two propositions: (1) that the use of mobile technologies to enhance teachers' capabilities to live a life they have reason to value can be understood to comprise four dimensions of capabilities; and (2) that teachers' well-being and human development comes as a result of the interplay between these four dimensions of capabilities, not through a pursuit of individual capabilities. In a last step, I merge these two propositions with the CA's emphasis on individuals as active agents to control and determine the conception of what constitutes relevant freedoms and a live they have reason to value. This investigation leads me to conclude that, in the context of my case study, ML4D can be conceptualised from the capabilities perspective as: the use of mobile technologies as an educational tool to support teachers' self-determined exploration of four dimensions of valued capabilities whose interplay enhances teachers' reported human development and well-being.

However, there are a few obstacles in this capabilities conception of ML4D that negate its application beyond the particular context of my case study. First, the developed list of valued

functionings, corresponding capabilities, and subsequent dimensions of capabilities is based on data collected in one particular context. The context of the nature of the ICT4RED programme as a ML4D intervention as well as the particular nature of the South African education system bound the transferability of my developed capabilities framework and the conception of ML4D derived of it. It is crucial to highlight that the developed framework does not claim to provide an exhaustive list of capabilities that constitute an effective application of mobile technologies as an educational tool in LIMCs. It since should not be confused with an attempt to develop a list of essential capabilities as proposed by Nussbaum (2011). I agree with Sen's (2009) position that while a normative focus on capabilities and functionings to conceptualise development processes and outcomes is justified, a normative prescription of specific important, arguably even universal, capabilities can only be developed on a case to case basis. As a result, I am reluctant to position my four dimensions of capabilities and their interplay as a necessary component of a general CA conception of ML4D.

Second, and linked to the first line of critique, the developed capabilities conception is linked to teachers' use of mobile technologies. Any broader conception of ML4D needs to apply a more inclusive focus, which is likely to change the composition of capabilities identified. I therefore require a conception that can accommodate all educational agents and processes, not just teachers and the integration of mobiles into teaching activities. Third, to serve as a more general conception of ML4D, I require a more open-ended formulation of the outcomes and processes associated with ML4D. The developed conception is fairly prescriptive in terms of dimensions of capabilities and the need to focus primarily on the result of their interplay. This limits the conception's application to different ML4D programmes that use technologies in different ways and for different purposes.

Taking into consideration the above caveats, I reformulate my conception of ML4D from a capabilities perspective as:

the use of mobile technologies to support an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of educational agents.

In this conception, I retain the focus on an expansion of valued beings and doings as the space in which the effects of ML4D programmes should be assessed, but exchange the prescriptive nature of my case study-bounded conception with a more open-ended understanding of 'enhancing the capabilities and agency of educational agents'. This

sequence likewise drops a focus on teachers in favour of a more inclusive focus on any educational agent (learners, teachers, parents, school administrators, etc.). In addition, my proposed definition introduces a focus on the use of mobiles to support 'an endogenous transformation of education in LMICs'. This sequence aims to introduce an indication of the nature and scale of the educational changes and contribution that one might expect of ML4D. The rationale for this addition links directly to the discussion started above on whether to position ML4D and the use of mobiles as: 'technologies of freedom'.

To recap, in section 8.2 I introduced the notion that if the use of mobile devices supported teachers to advance their effective freedoms including the removal of key unfreedoms, then tablets might be regarded as 'technologies of freedom'. This expression presents an adaptation of Dorothea Kleine's concept of 'technologies of choice' in the ICT4D sector. Adopting this expression would indicate a positioning of ML4D and its potential contribution to education and development in a grander sense than my current expression of being able to support an endogenous transformation of education in LMICs. Surely, if ML4D were to be positioned in line with an idea of 'technologies of freedom', the suggested nature and scale of the educational changes and ML4D contribution to it as an endogenous transformation seems modest.

The concept of 'technologies of freedom' was introduced by Sen (2010: 4) in reference to mobile phones, which he positioned as "generally freedom-enhancing". It assumes that technologies and their affordances support inherently freedom-enhancing usages due to the vast range of opportunity sets they allow users to explore. Notwithstanding, I would propose a more cautious and modest interpretation based on two shortcomings of this line of argument given the data collected in my case study. First, if I equate freedom with choice as Kleine suggest, tablets as part of ML4D intervention did enhance teachers' perceived choices in line with four dimensions of valued opportunities. However, neither of these four dimensions in themselves nor the configuration of them is dependent on the actual use of technologies. Take the educational and societal capabilities as an example: it seems challenging to argue that most of these capabilities require a technological input for teachers to explore them. Educational and societal capabilities were an integral part of teachers' freedom to live a life they have reason to value; technologies, on the other hand, were but one tool to support these capabilities. It therefore seems as an overreach to claim that the tablets can be positioned as inherently freedom-enhancing. Their main contribution seems to lie in the simultaneous support to multiple dimensions of opportunities.

Second, one could argue that by directly addressing past unfreedoms (e.g. providing what teachers have been denied under Apartheid), technologies become a tool of freedom. However, this overlooks the importance of the choice of teachers to use the tablets in this manner. Applying the tablets to explore capabilities linked to the rectification of past discrimination was one configuration of capabilities; and, albeit being a powerful configuration, remains one of many options of how teachers appropriated the use of the technologies. It does not seem warranted to equate tablets with constituting a technology of freedom only because of this being one of the identified appropriations. For some teachers, the main valued beings and doings derived from the use of tablets were the enhanced ability to control the classroom and that they had to carry less text-books on their daily commutes. There was rich evidence in my case study data that tablets can be positioned as freedom-enhancing in certain contexts, but to generally position them in this matter seems too 'grande' a positioning based on the evidence that I have presented and formalised in my case study.

While the above discussion seems to be able to refute a positioning of the nature and scale of the educational changes induced by ML4D in line with the conception of 'technologies of freedom', it does not explain my adoption of a positioning of the nature and scale of the educational changes induced by ML4D as to present an endogenous transformation. The term endogenous transformation attempts to capture the locus and space of the educational change (i.e. endogenous) and the nature and scale of the change (i.e. transformation).

In the CA, there is no formal conception of the term transformation and it does not present one of the CA's analytical devices as functionings or effective freedoms do. However, the CA does require an in-depth analysis of how access to and use of mobile technologies leads to changes in individual's valued beings and doings. This analytical technique of the CA necessary sensitises and forces one to focus on the transformative effect that using technologies as part of the ML4D programme has on individuals as the initial unit of analysis⁸². In an CA analysis of ML4D programme the starting point is thus very likely to be how individual technology users can appropriate the mobile devices in order to align technology's affordances with users' pursuit of the life that they had reasoned to value. There then should be little contestation for my adoption of the term 'endogenous' to represent that the locus of change is likely to be within and aligned to existing conceptions of the good life, most likely those of individuals.

⁸² For a full discussion on designing a systematic in-depth investigation of individual's technology usage applying the CA to focus on transformational effects, I refer to Kleine (2013) chapter 2 and 7.

However, the alignment of technology usage to pursue the life one has reasoned to value then can take multiple forms. In my case study research, this alignment let to a series of small steps, first developing individual capabilities to then leading to a configuration of different dimensions of capabilities, whose interplay finally contributed to teachers' human development and well-being. The observed nature of change in my research did not work through the exploration of one central or uber capability associated with an inherent affordance of the technologies, for example being ICT literate or being connected, which then led to an unlocking of teachers' effective freedoms and path to explore their conception of a good life. Such a linear link between a capability inherent to the technological device and expansions in technology users' real freedoms could rightfully be positioned as a notion of 'technologies of freedom'. Rather, I observed a careful configuration of capabilities, which, according to teachers' preferences and contexts, allowed them to step-wise develop agency to move closer to their conception of a good life. It is this process that I attempt to capture under the term 'transformation' and it represents a type of educational change that works largely through changing educational agents from within in a process of small steps informed by the objective of enhancing the capabilities and agency of agents at each step. It is important to note though that this distinction between the nature and scale of the educational change induced by the ML4D intervention and my preference to describe this change in the language of an endogenous transformation, at this point, is not normative but a reflection of my case study data.

Conclusion

To conclude, I retain my proposed capabilities conception of ML4D as: the use of mobile technologies to support an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of educational agents. This conception defines the space in which the effects of ML4D interventions should be assessed as the valued beings and doings and corresponding capabilities of educational agents. It defines ML4D's contribution to education and development as to foster an endogenous transformation located within educational agents and structures based on their self-determined exploration of beings and doings aligned to their conception of a live they reasoned to value.

In my case study, I illustrate one practical manifestation of this specific contribution of ML4D to education and development. Based on the ICT4RED programme as an example of a ML4D intervention in a rural education context, I showcase how the specific properties of a ML4D programme allowed for a simultaneous exploration of four dimensions of capabilities. This simultaneous exploration and the facilitated interplay between different dimensions of capabilities, which brought about teachers' human development and well-being, therefore constituted ML4D's specific contribution. However, for a range of reasons, I shun away from positioning this case study-bound conception of ML4D as a general framework for ML4D. Though, I would argue that the developed four notions of informational, educational, societal, and economic capabilities probably do have a wider application in investigating the transformational potential of mobile technologies as an educational tool in LMICs. But, at this point, and given the empirical data presented, a prescription of these four dimensions of capabilities and how their interplay supports human development and well-being seems premature.

In this chapter, I have developed a capabilities conception of ML4D. However, I have not discussed the implications of this conception for the positioning of ML4D in the wider discourse on mobile technologies, education, and development. If the potential of the educational application of mobile technologies in LMICs can be expressed in terms of a transformative expansion of human capabilities and agency of educational agents, as argued here, this claim needs to be juxtaposed and contextualised with existing discourses on ML4D. It is not sufficient to define the specific contribution of mobile learning to international development through the CA without comparing this perspective against existing definitions such as framing its contribution in terms of human capital expansions. These discussions are provided in the next chapter in which I attempt to investigate the

implications of my capabilities conception of ML4D in order to guide a rethink of the positioning of ML4D.

Chapter 9. Rethinking ML4D—positioning mobile learning in international development

Introduction

This chapter presents my alternative conception of using mobile technologies as an educational tool in Low- and Middle-Income Countries (LMICs). It combines my capabilities conception of Mobile Learning for Development (ML4D) with the findings of the mixed-methods systematic review in order to position mobile learning's role in international development. In this, it discusses my capabilities conception of ML4D in the light of the available evidence-base on the use of mobiles to support education and development in LMICs in order to investigate gaps and overlaps between both. I then juxtapose the capabilities conceptualisation of ML4D with existing conceptualisations of mobile learning in international development. This aims to show how the developed conception differs from existing understandings of ML4D and why I assume that a capabilities conception allows for stronger links between the educational use of mobile devices in LMICs and development outcomes. This justification directly answers the thesis's main research question of how the use of mobile technologies as an educational tool in LMICs can be conceptualised as a development intervention.

This chapter is structured as follows. I first provide a discussion of my capabilities conception of ML4D in the context of the mixed-methods review findings. This illustrates whether my proposed conception can be reconciled with the implications of the systematic review findings, in particular with the developed theory of change for ML4D. Having done so, I then proceed to compare my conception of ML4D against existing understandings of mobile learning in international development. This presents my rethink and positioning of ML4D. In the remainder of the chapter I then elaborate on the overlap and contrast of my capabilities conception of ML4D with a wider body of literature. This discussion aims to explore the periphery of the thesis's argument and to place boundaries around its contribution to the literature.

9.1 Comparing the capabilities conception of ML4D with my systematic review findings

In chapter 6, I argued that the main contribution of my systematic review findings is to empirically set up the need for an alternative conception of ML4D. By establishing that the synthesised evidence-base on the effects of using mobile technologies to support education and development in LMICs does reject the positioning of mobile learning as a development intervention, the systemic review provides my empirical justification for the exploration of an alternative conception and positioning of ML4D. However, it would seem hypocritical to not subject my own proposed capabilities conception of ML4D to the same scrutiny of the available evidence-base. That is, if I use my systematic review findings to challenge a techno-centric conception of ML4D and refute its claims to disruptive and large-scale impacts on education and development, my own conception needs to be contextualised in the light of my systematic review findings too. I therefore first elaborate on to what extent my capabilities conception is reflected in the systematic review findings.

To recall, in the previous chapter I positioned mobile learning's contribution to development as:

the support of an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of actors in the education system.

This definition was derived from my qualitative case study and the developed capabilities framework featuring four dimensions of capabilities that reflected teachers' valued opportunities, explored through the educational use of mobile technologies. These dimensions referred to informational, educational, economic, and societal capabilities and reflected 21 individual capabilities. In my systematic review, I first conducted a statistical meta-analysis of the effects of using mobiles to support education in LMICs, which found mobile learning interventions to have a significant effect only on learning outcomes. With regard to my meta-analysis findings, I cannot provide an empirical assessment of ML4D interventions that apply, say, a techno-centric design. There were no CA-inspired ML4D interventions identified in systematic search for evidence. I can therefore not comment on whether a capabilities conception of ML4D would lead to more effective mobile learning programmes. This is a question that requires further primary research first.

With regard to the thematic synthesis findings, there is a strong overlap between dimensions of capabilities as well as individual capabilities represented in the case study framework and the mechanisms and contexts identified in the systematic review. For example, the mechanisms of 'changing pedagogy and practice' and 'teaching practice' directly relate to valued educational capabilities reported by the teachers involved in the Information Communication and Technology for Rural Education Development (ICT4RED) programme. Contextual themes such as 'technology as a social and professional status symbol' and 'positive perceptions' are also reflected in valued informational and societal capabilities identified in the case study. There is a particularly strong overlap between the mechanism themes of 'social capital', 'economic capital', and 'self-efficacy' and valued societal and economic capabilities. However, this overlap cannot be assumed to be a validation of my capabilities conception of ML4D. Rather, it presents a validation of my primary research findings indicating that a range of themes discovered overlap with findings of teachers' use of mobile technologies reported in the wider literature.

Once I turn to my combined systematic review findings and the developed theory of change for the use of mobiles to support education in LMICs, the overlap between a capabilities conception of ML4D and the systematic review becomes stronger. In chapter 7, section 7.1, I outlined a range of potential overlaps, which I can now flesh out with empirical data from my case study research. First, I argued that the observed lack of effects on development outcomes might be explained by a too narrow framing of development outcomes and pathways to such outcomes in terms of human capital. The vast majority of ML4D interventions conceptualised pathways to development through an increase in skills and human capital. Based on my capabilities framework, I can now offer a concrete expansion to this narrow conception, positioning my four dimensions of capabilities as additional components of individual's well-being and agency. These four dimensions of capabilities could thus be integrated into the theory of change spanning the pathway between final outcomes and impacts. This would allow, for example, a greater emphasis on teachers' professional well-being as an outcome of development. It could also facilitate the introduction of agency and control over one's life as an outcome and driver of development, which is currently only partially referenced under the mechanism of 'self-efficacy'. In sum, the introduction of my four dimensions of capabilities into my theory of change could significantly alter its design highlighting a richer and more complex set of pathways to human development.

With regard to my second line of argument based on the review finding that mobile learning's positioning as a development intervention is held back by a reproduction of

247

structural inequalities, the developed capabilities conception of ML4D makes explicit provision for this. By arguing for an emphasis on anchoring ML4D programmes in the primary objective of enhancing the capabilities and agency of educational agents, the focus of the programme design and theory of change should shift towards the effective opportunities that learners and teachers can generate of the technological inputs. This would lead to the introduction of conversion factors as part of the theory of change, which could be expressed as either mechanisms or contexts. It would likewise shift the focus of the programme design towards more participatory and people-focused approaches as programme objectives would have to be aligned with people's reasoned choice of how to integrate the use of mobiles into their pursuit of valued beings and doings. Taken together, adopting the proposed capabilities conception of ML4D would force the theory of change to introduce conversion factors and agency and choice in addition to the current components. This would be enhanced with the above introduction of the four dimensions of capabilities leading towards a programmatic outline of ML4D interventions that seems to offer a more relevant representation of the links between mobile technologies, education, and development in LMICs.

All in all, I therefore conclude that the capabilities conception of ML4D can indeed be reconciled with the findings from the mixed-methods systematic review. Introducing a capabilities perspective to the review findings, the developed theory of change in particular, seems to be able to address some of the shortcomings in the existing ML4D intervention designs that were identified as barriers to mobile learning's contribution to development outcomes. Having reconciled my capabilities conception of ML4D with the systematic review findings, I next discuss the implications of this conceptualisation of ML4D for the existing positioning of the role of mobile learning in international development.

9.2 Finding a space for mobile learning in international development

In chapter 2, section 2.2–2.3, I outlined the existing conceptualisation of mobile learning's role and contribution in international development as well as three key critiques of this conceptualisation. I then conducted a mixed-methods systematic review in order to assess the empirical validity of the claimed contribution of mobile learning to education and development in LMICs. My review finds that claims of ML4D's effectiveness and potential to support development in LMICs are based on simple causalities rather than reliable systematic evidence. What is more, the magnitude of claims to mobile learning's large impact on education and development in LMICs seems to be subject to a displacement of scope (Wagner 1962), i.e. that the claims made about ML4D are out of perspective with what could possibly be achieved. This raises doubts whether proponents of ML4D at a development policy level have thoroughly conceptualised how the use of mobile devices is believed to support education and development in LMICs. I find that on closer examination techno-centric and access- or input-based conceptions of ML4D run the risk of justifying the creation of separate educational structures circumventing, rather than supporting, the existing educational structures. In sum, I conclude that existing conceptualisations of ML4D fail to make a convincing conceptual and empirical link between teaching and learning with mobiles in LMICs, increased education outcomes, and enhanced livelihoods and socioeconomic development.

Based on my rejection of existing positionings of mobile learning's role and space in international development, I then develop a capabilities conception of ML4D based on the qualitative case study of the ICT4RED programme. This capabilities conception provides an alternative perspective on how the use of mobile technologies as an educational tool in LMICs can be conceptualised as a development intervention. I argue that conceptualising ML4D through the CA supports a focus on an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of actors in the education system. In the absence of evidence supporting mobile learning's impact on development outcomes, a focus on the role of mobile technologies to expand teachers' and learners' valued functionings and capabilities presents an alternative conception of the links between mobile technologies, their use for educational purposes in LMICs, and development outcomes. This understanding and positioning of mobile learning in international development differs from existing conceptualisations by shifting the focus of mobile learning interventions away from access to technologies and content provision towards four different

foci of interest: (1) an opportunity focus; (2) a people focus; (3) an education focus; and (4) a transformation focus. I discuss each briefly below.

9.2.1 Opportunity-focused ML4D

In a capabilities conception of ML4D, the effective freedom or opportunity of actors in the education system to achieve agency and well-being becomes the central objective of using mobile technologies. This presents a shift from a techno-centric positioning of technology provision and access to technology and content as the main objective of ML4D interventions. Rather than assuming that a mobile phone presents a 'teacher in the pocket' or that all what learners in LMICs are lacking is a 'classroom in a box', my CA conception of ML4D focuses on the valued opportunities of teachers and learners in their educational context. That is, it neither focuses on access to technology nor on predefined instrumental uses of technologies, but aims to explore what effective freedoms teachers and learners would like to explore and how the use of technologies can be integrated in their exploration of these opportunities. Such an opportunity-focused objective of ML4D then requires an explicit design of mobile learning programmes to address this objective. An example of such an opportunity-focused design could relate to a closer integration of mobile learning with critical theories such as Freire's Pedagogy of the Oppressed (Freire 1972).

An opportunity-focused understanding of ML4D also shifts the focus of mobile learning programmes away from reductionist notions of the processes and outcomes of both education and development. In existing conceptions, the use of mobile technologies in education is overly framed in an instrumental fashion (Winters et al 2017) focusing on increases in skills and human capital. These are then linked to development outcomes, which are largely defined as finding employment and increasing productivity. A capabilities conception of ML4D enhances these objectives of using mobile technologies in education in LMICs, for example highlighting the intrinsic value of education in supporting learners' and teachers' agency to take control over their lives. Development is understood as an expansion of people's freedoms, which does include economic freedoms such as being able to find decent employment and to make a productive contribution to society, but goes beyond these economic capabilities to include a richer and more diverse understandings of valued beings and doings. These richer and more diverse freedoms, for example, could relate to the four dimensions of valued opportunities presented in my capabilities framework.

9.2.2 People-focused ML4D

A second shift in ML4D's positioning in international development supported by a capabilities conception is an increased focus on educational actors, teachers and learners in particular, as the key agents in mobile learning programmes. The CA stresses that people are at the centre of any development effort and cannot be regarded as passive recipients of "the fruits of cunning development programs" (Sen 1999: 53). People need to be in control of the pursuit of their valued beings and doings and need to have the ability to define their own conception of a live they have reason to value. Mobile learning programs then assume developmental potential by people's active and deliberate appropriation and integration of the educational use of technologies in their pursuit of valued being and doings. In short, it is people's own choices and aspirations that determine mobile learning's contribution to development. This understanding seems to be at odds with top-down conceptions of ML4D such as provided by UNESCO and the GSMA in which the contribution of mobile learning to development is already predefined (in relation to the former MDGs and EFA targets in UNESCO's case, and in relation to disrupting education in LMICs in GSMA's case).

From a programme design perspective, it is equally crucial to apply a people-focused lens to ML4D interventions. The CA highlights the need to take into consideration a range of conversion factors that individuals possess when translating access to inputs, such as mobile devices, into capabilities in order to achieve well-being. This is particular important in the context of learning and teaching with mobiles in LMICs as the conversion factors of educational actors and thus their ability to transform access to mobiles into effective opportunities will vary greatly. For example, a girl learner in a society with patriarchal norms will have a different ability to explore learning across contexts as a girl learner in an affluent urban environment with more egalitarian gender norms. Applying a participatory, peoplefocused approach to the design of ML4D programmes is therefore a key implication of a CA conception of mobile learning's role in international development. Such a people-focused approach could then be able to counter ML4D's existing risk of reproducing structural inequalities as it requires a focus on the diversity of people's abilities to translate access to technological inputs into effective opportunities. This also allows for a greater integration of teachers into mobile learning programmes, who often are bypassed and marginalised in current programme designs (c.f. Winters 2015).

9.2.3 Education-focused ML4D

An implication of the above two shifts in the positioning of mobile learning's role in

international development is further that the contribution of mobile learning has to be framed more strongly in educational terms. That is, the contribution that mobile technologies make requires a stronger pedagogical grounding. The systematic review as well as the case study both found that the richest applications of mobile technologies in education in LMICs are connected to the use of explicit mobile learning pedagogies. In the systematic review, pedagogically-rich mobile learning programs were found to have a larger effect, while in the case study the embedding of technology use in an explicitly pedagogical mobile learning programme led to a range of rich opportunities explored by teachers. This is in contrast to techno-centric conceptions of ML4D in which the educational underpinning of mobile learning is often an after-thought rather than an integral programme component.

In a CA conception of ML4D, focused on expanding people's effective opportunities, mobile learning in LMICs therefore needs to be primarily positioned as an education intervention. It is through the exploration of mobile technologies' particular affordances, which give rise to a range of different teaching and learning approaches, that the educational actors in LMICs can explore a diverse range of opportunities. Without a primary focus on these educational opportunities, mobile learning lacks a plausible theory of change to support a transformative change on education and development outcomes. Positioning mobile learning in international development from a CA perspective, therefore would require a shift back towards a learning activity-based, pedagogical definition of ML4D.

9.2.4 Transformation-focused ML4D

Lastly, conceptualising ML4D through the CA leads to a transformation-focused positioning of the educational use of mobile devices in LMICs. That is, the contribution of mobile learning to development is framed in terms of an endogenous transformation of education in LMICs, not in terms of a 'disruption' or 'revolution' of education in LMICs. In Chapter 2, section 2.2.3–2.3.3 the implications of an exogenous positioning of mobile learning as an adjunct rather than integral part of the education system are discussed, highlighting that this fuels a deficit model of education in LMICs with negative consequences for cost-intensive educational inputs and actors, in particular teachers. My systematic review then further underlined this critique illustrating that there is little evidence to support the assumption that mobile learning programmes can effectively be implemented outside of existing educational structures. Framing mobile learning's contribution to development in terms of disrupting and revolutionising education is thus conceptually and empirically problematic.
Applying a CA conception of ML4D supports this critique and showcases how mobile learning interventions can be positioned as endogenous to the existing structures. The identified configurations of capabilities in the case study and how they relate to teachers' well-being and human development pay testimony to this. For example, the narrative of building the nation illustrates clearly how teachers' appropriation of technologies in line with the pursuit of their valued beings and doings took place within educational structures. Only through their perceived professional mandate of 'building the nation' and subsequent professional identity did the application of mobile devices receive a space in which it was valued by teachers for professional beings and doings. While there certainly were a range of capabilities explored by teacher that lay outside the existing educational structures, the main application of the devices and exploration of valued opportunities lay inside the existing educational structures. In this, I observed a careful configuration of capabilities, which, according to teachers' preferences and contexts, allowed them to step-wise develop agency in order to move closer to their conception of a good life. This transformation represents a type of educational change that works largely through changing educational agents from within in a process of small steps informed by the objective of enhancing the capabilities and agency of agents at each step. As a result, a CA conception of ML4D proposes a type of educational change that can best described as an 'endogenous transformation' as a contrast to a disruptive, revolutionary educational change.

Summary: Rethinking ML4D

In summary, applying a capabilities perspective to conceptualise ML4D shifts mobile learning's role and contribution in international development in line with an opportunity-, people-, education-, and transformation-focused understanding of using technologies as an educational tool in LMICs. It is in this conception that I would position mobile learning as a development intervention. This shifts the conceptual focus away from a techno-centric, access-based view of mobile technologies towards a view that prioritises how educational actors can apply these technologies to enhance their capability and agency to live a life they have reason to value.

Rethinking ML4D, I thus position the educational application of mobile technologies in LMICs to carry the potential to contribute to an expansion of human capabilities and agency. However, in order to achieve this contribution of mobile learning to development, mobile learning programmes need to be carefully designed for this particular objective. It is not an inherent property of ML4D interventions to exercise this expansion of human capabilities and agency. A key consideration in this explicit design effort refers to acknowledging that the

253

particular educational and informational attributes of mobile learning can allow for a more multidimensional use of the technology, which supports individuals to simultaneously explore different dimensions of capabilities integrating their social and professional contexts and identities. Given these particular attributes of mobile leaning, it can therefore be positioned to support the expansion of a different set of capabilities, which thus far have been overlooked in previous research applying the CA in education and technology.

This positioning of ML4D and line of argument would call for more humility in the advocacy on ML4D than currently displayed by international development and mobile vendor organisations. It positions mobile learning's contribution as to foster a series of step-wise endogenous changes in educational practices and processes driven by educational actors' own definitions of valued beings and doings. It thereby targets an endogenous transformation of education in LMICs to which the educational use of mobile technologies can make but one contribution. By no means, however, does mobile learning present the potential to single-handedly revolutionise educational interventions in LMICs whose contribution needs to be considered against its costs and the respective contributions of other education interventions. To be clear, ML4D has a strong contribution to offer to educational change in LMICs, that is the ability to foster a simultaneous exploration of dimensions of capabilities comprising human development and well-being; but whether its contribution is the most relevant among the many potential educational contributions of other programmes needs to be considered carefully on a case to case basis.

9.3 Defending the argument for my conception of ML4D

I have now proposed my capabilities conception of how the use of mobile technologies as an educational tool in LMICs can be positioned as a development intervention. However, there are a few lines of critiques that could be put forward to challenge this conceptualisation and positioning. Below, I address what I perceive to be the five most pressing challenges to my capabilities conception of ML4D and subsequent positioning of mobile learning as a development intervention.

9.3.1 Transformation from within, undermining structural changes?

In section 8.3, I outlined my case to use the term endogenous transformation to characterise the nature of the educational change targeted in a capabilities conception of ML4D. This attempted to capture the locus and space of the educational change (i.e. endogenous) and the nature and scale of the change (i.e. transformation). I justified my use of this term based on the data collected and analysed rather than through a conceptual argument. An endogenous transformation represents a type of educational change that works largely through changing educational agents from within in a process of small steps informed by the objective of enhancing the capabilities and agency of actors at each step.

I then supplemented this descriptive argument with a conceptual argument outlining the risks of positioning technology as exogenous to the existing educational structures and to attribute it a disruptive, revolutionary potential for change. In short, positioning mobile learning as an exogenous educational input risks creating conflict between (i) existing educational structures and its actors vs the new technological innovation; (ii) between a focus on technology innovation vs a focus on education innovation; and (iii) between a disruption of education from the outside vs a transformation of education from the inside. Based on this conceptual exploration and the empirical findings of my systematic review, I concluded that a capabilities conception of ML4D needs to focus on an endogenous transformation of education in LMICs.

However, there is an important counter-argument to this conception that needs to be taken into consideration. By embedding ML4D firmly within existing educational structures, mobile learning programmes might not be able to affect structural inequalities within these structures. That is, if mobile learning aims to influence a step-wise transformation from within, it might lack the ability to change larger structural issues in the education system. For example, if the existing educational structures discriminate against girl learners, embedding mobile learning in these structures risks reproducing these inequalities. If, on the other hand, we position mobile learning as exogenous to these structures, the technology input might be able to create separate, more equal structures that replace the existing unequal structures. This rationale for example underlines the idea that providing girl learners with access to more information on mobile phones can lead to their empowerment.

As much as this is a conceptually coherent challenge, my data sets also underlines that there is a danger that a 'transformation from within' position of ML4D risks trapping mobile learning in existing systemic inequalities. In my systematic review, I establish that indeed mobile learning programmes in LMICs are designed in such a way that they do not run counter to existing inequalities such as prioritising urban schools over rural schools. However, the same holds true for my capabilities framework of teachers' use of mobile technologies. For example, in the teachers' narrative of building the nation, which presented a key source of their well-being and human development, one could argue that teachers in fact follow a predefined conception of a good life, and not their own. The narrative of building the nation was developed by the South African government and teachers have internalised it as a part of their professional identity. Teacher also felt thankful for, and valued by, the government for providing them with tablets.

Yet, at the same time teachers were employed at schools and school districts that exhibit large structural inequalities within the South African education system. There was no indication at all within my primary research that teachers intend to use the tablets to actively address these structural inequalities. While being conscious of them, it did not challenge their narrative of wanting to build the nation. To the contrary, teachers directly resisted any application of the tablets that could be seen as undermining their building of the nation. This referred in particular to using the tablets to access or generate teaching materials that were not part of the prescribed national curriculum. Teacher rejected this application of the tablets as it would not support their learners in passing the annual assessment, thereby undermining teachers' contribution to supporting socio-economic development in South Africa. The ICT4RED mobile learning programme is thus embedded to such an extent into the existing education system that it might lack the ability to affect large-scale structural change.

In sum, by focusing on transformation from within and the valued beings and doings of educational actors, mobile learning might be limited in its ability to address existing inequalities within the education system. If educational actors do not value structural changes and these are not part of their conception of a good life, it is difficult to see how the

mobile technologies would be applied to this end. In such a situation an exogenous intervention might be more relevant to create effective educational changes. But, before I proceed to formulate a reply to this challenge, it is important to note that this conceptual challenge is not unique to my case study. There is a rich debate on the CA's ability to deal with structural changes. Scholars, for example, have extensively debated the individual focus of capabilities and suggested that collective capabilities might be a more relevant level of analysis (Ibrahim 2009; 2006; Stewart 2005). Likewise, Zheng and Stahl (2011; 2012) and Popova and Roberts (2017) suggest to marry the CA with critical theory in order to allow a more structural type of investigation. Sen, too, has acknowledged this issue, for example in form of an emphasis on taking into account adaptive preferences of individuals (Sen 1992; 1999).

In order to defend my 'transformation within' conception of ML4D, I turn to the distinction of defining developmental changes between a 'liberation within' and a 'liberation from' developed by Kullmann and Lee (2013). Their suggestion combines Latour's concepts of 'actor-networks', 'collectives' and 'moments of translation' with Sen's capabilities, agency, and conversion factors—a combination which I found most adequate in the context of my conceptual challenge as it is not only focused on the nature of the capabilities itself, but further also on the described nature of the change observed. Kullman and Lee introduce Latour's (1993) conceptual devices into Sen's definition of freedom. This allows them to make a distinction between two distinct concepts of freedom:

"Where 'freedom' is often understood as liberation *from* one's social and material surroundings, we will use the Sen/Latour encounter to clarify a view of freedom as liberation *within* one's environment. Liberation does not involve rendering the environment irrelevant by breaking existing relations with it. Instead liberation is composed through gradual changes in everyday socio-technical relations of specific collectives." (Kullman & Lee 2013: 40, emphasis in the original)

In an understanding of 'liberation from', individuals are able to become independent and removed of their immediate environments; or, more formally, individuals can "transcend the surrounding social and material relations so that it becomes safe to ignore those relations" (Kullman & Lee 2013: 44). The authors illustrate this practice of 'liberation from' with a case study of the OLPC program showing how the OLPC attempted to allow children to remove themselves from the limitation of an educational context in LMICs by providing them with laptop computers that were assumed to facilitate learning without the need for a formal educational intervention. Children thereby were 'liberated from' absent teachers, a lack of relevant content, poor educational infrastructure, etc.

257

'Liberation from within', on the other hand, assumes individuals to be defined by their immediate environments and regards the surrounding social and material relations as crucial to individual's development as it is these relationships that foster individual or collective agency. Liberation from within thus asks individuals to explore their opportunities to change their immediate environments, which is described as a process of "careful and gradual 'conversion' or 'translation' of everyday relationships in order to make more space for situated forms of human 'development' and 'growth'" (Kullman & Lee 2013: 45). Applying this concept of 'liberation within' similarly to the OLPC, the authors show how collaborative learning activities that children explored with the provided laptops generated a stronger set of capabilities and functionings than using the laptops to "reach an abstracted state of 'liberation from'" (Kullman & Lee 2013: 53). Reflecting on their contrasting of both concepts of freedom, the authors conclude that:

"Taken together, Sen and Latour invite a systematic empirical analysis of developmental projects in terms of 'liberation within' rather than 'liberation from'. This means that the quality of the relations among persons, technologies and environments is more important than minimizing dependencies between people and their surroundings" (Kullman & Lee 2013: 54).

The distinction between 'liberation within' and 'liberation from' is conceptually more rigorous than my distinction between an endogenous transformation and a disruption or revolution of education in LMICs. I therefore adopt Kullman and Lee's supplementation of Sen's concept of freedom and position ML4D from a capabilities perspective as to support a 'liberation within' educational actors in LMICs. In this, I assume that mobile learning is better suited to enhance the quality of the relations among persons, technologies, and environments allowing for a gradual and situated process of development. This does not negate that mobile learning might have the potential to support structural changes and address existing unfreedoms and inequalities embedded within the education system. In fact, teachers suggested that the tablets supported them to rectify past unfreedoms to some extent. However, based on my review of the overall evidence-base, mobile technologies as an educational tool do not seem to provide a close fit to address structural changes in the education system. In my conception, I therefore position mobile learning to support a 'liberation within' as the small, well-defined functions that mobiles do uniquely well in an educational context can only translate into powerful educational interventions through the mediation of social practices (Roschelle 2003). In regard to fostering a 'liberation from', which I do see as important too, I believe there to be more potent educational interventions that can lay claim to institute such structural changes; for example, changes in teacher training, national curricula, budget mechanisms such as gender mainstreaming, and schoolbased decision-making, among other. Also, perhaps once ML4D has established itself as a

viable tool to support a 'liberation within', research and practice on mobile learning in LMICs might reconsider the objective of using mobiles to support a 'liberation from'. At this point, however, I am positioning ML4D firmly as to primarily relate to efforts of contributing to a 'liberation within' education in LMICs.

9.3.2 Mobile learning pedagogies and development as freedom

As part of my positioning of mobile learning as a development intervention, I claimed that the educational use of mobiles in LMICs needs to be anchored in a stronger pedagogical understanding. That is, that the specific pedagogies that the particular affordances of mobiles can support need to be more closely integrated with the conception of ML4D. However, throughout my thesis I have mainly provided evidence for the importance of a deliberate pedagogical approach to teaching and learning with mobiles per se, rather than evidence for the specific mobile learning pedagogies themselves. This is largely due to the positioning of the thesis in literature on international development rather than in the educational domain. Yet, throughout my thesis the importance of not just any, but particular mobile learning inspired pedagogical underpinnings of the educational application of mobile devices is touched upon. Below, I aim to illustrate these dispersed references to the more specific pedagogical approaches identified with mobile learning in a development context in a bit more detail.

To start with, in the meta-analysis of my systematic review I found that pedagogies mattered in the design of mobile learning interventions in LMICs. The mere delivery of content, for example through daily text messages (Aker et al 2012) on mobile devices, was significantly less effective in improving learning in LMICs. Interventions using mobile technologies to design context-aware, game-based, and collaborative learning experiences, on the other hand, were found to be more beneficial in improving learning. This finding about the importance of explicit pedagogical underpinnings was echoed in the thematic synthesis, which identified pedagogy to be a key mechanism supporting the effects of mobile learning on education outcomes. The thematic synthesis identified the following features of pedagogical approaches as most prominent in the qualitative evidence on teaching and learning with mobiles in LMICs: interaction, collaboration, authenticity, personalisation, targeting, access to information, and an enhanced variety of teaching strategies.

In my qualitative case study, I then investigated a mobile learning programme in rural South Africa that was explicitly designed to nurture pedagogical innovation in line with a more learner-centred teaching approach. I found that this mobile learning programme did indeed allow teachers to explore a range on valued educational opportunities, including a more learner-centred education approach in which the teacher becomes the 'facilitator of knowledge'. This could be positioned as to constitute strong evidence that the particular mobile learning pedagogies taught by the ICT4RED were the key contribution of the mobile learning programme to teachers' and learners' development.

But, this conclusion would be at add odds with my developed capabilities framework of teachers' use of mobiles, which positions teachers' well-being and human development to be a function of an interplay between different dimensions of capabilities. It also hides the active facilitation of these particular pedagogies by the ICT4RED. Teachers further varied in their exploration of educational opportunities, and while some valued integrating mobile learning into their daily teaching, others merely valued the tablets as a tool to facilitate their daily routines. It is also interesting to note that teachers often linked the changes in educational capabilities to the use of tablets when it is not clear why non-technological inputs could have not been applied to the same effect.

Development as freedom in the capabilities conception of ML4D should not be linked directly to any specific pedagogy at this stage. That is, it would be an overstatement to claim that a positioning of ML4D is only valid if an explicit mobile learning pedagogy is embedded into the programme design. In my current conception, I therefore deliberately highlight an 'education-focused' approach, but refrain from prescribing a specific 'mobile learning-focused approach' that would advocate for particular mobile learning pedagogies such as experiential or context-aware learning. This attempts to reflect that a pedagogical anchoring of ML4D is crucial for mobile learning to make a relevant contribution to development outcomes, but that the specific nature of this pedagogical anchoring is open-ended.

In conclusion, mobile learning pedagogies can and should be integrated with the conception of ML4D. And, there is a reliable evidence that points towards the beneficial impacts of this integration, for example by merging informational and educational capabilities. However, in order to support an endogenous appropriation of technologies and a people-driven exploration of their educational usages, specific pedagogies should not be singled out as the defining attributes of a conception of ML4D. To illustrate the range of potential pedagogical applications of mobile technologies in LMICs, I conclude this discussion with a last vignette of my case study research.

Vignette: Alternate ways to teach—the teacher as a facilitator of knowledge

"It (the tablet) has improved my teaching and learning styles. Also, it has improved learning outcomes and they (the students) learn quicker. My teaching style is no longer teachercentred. When I'm in class I'm no longer standing in front; I let the learners do the work so I'm not the person who keeps the knowledge. I am just a facilitator and the teaching strategies help us a lot in this. I did the storytelling exercise and the children told really exciting stories. Also learning with tablets in general is very helpful. In geography, to have evidence that they visited the place I need them to take pictures. We just recently did map work. I asked them to go outside and create a map of the area using the templates and Google maps and pictures. They also put up the constructed and natural features." (Primary teacher; day 7)

In my case study research, I identified a rich range of different teaching approaches that teachers explored with the support of mobile technologies. The above quote illustrates one prominent example. Here, the teacher used the particular affordances of mobile technologies to enhance her teaching of Geography. Learners used the technologies to learn across contexts and towards an authentic and relevant task, that is enriching the Google map features of their communities. I also observed other such examples in particular in natural science teaching and history and economic lessons such as the story telling exercise mentioned in chapter 7, section 7.3.2. In addition, one innovative example stands out: in order to teach the concept of arbitrage, a teacher split her class in six teams that were sent into the local community, while one team stayed behind in the classroom. The teams in the community surveyed local Spaza shops and recorded the prices for a basket of goods, which were sent in real time to the coordinating team in the classroom. The classroom team then compared the different prices of the community shops against each other, as well as against whole sale market prices of these goods. The learners then calculated profit margins per shop as well as for their own fictitious start-up business, which was supposed to act as a middle entity between wholesalers and Spaza shop owners.

My case study therefore illustrated that pedagogical-rich mobile learning did take place in a resource-poor rural setting. These pedagogical-rich practices largely resulted from a configuration of teachers' informational and educational capabilities in an effort to create more learner-centred educational processes. In order to enhance their valued educational opportunities, such as supporting their learners to succeed, teachers subscribed to a narrative of 'becoming the facilitator of knowledge', which expressed their desire to allow students more participation and interaction in the learning process. However, it needs to be

mentioned that many of the resulting teaching strategies, as well as the overall learnercentred pedagogical approach, were suggested in the ICT4RED teacher training manual and further practiced in the professional development classes. Teachers' adoption of these learner-centred teaching strategies and the desire to become a facilitator of knowledge therefore need to be understood in the context of this professional development programme. Teachers also varied in their adoption and implementation of these learner-centred teaching strategies and oftentimes the link between the conducted teaching strategy and the need for it to be facilitated through the tablet devices is not clear. For example, the popular teaching strategy of story-telling in which learners surveyed community elders to make a multi-media portfolio on the tablets could arguable also have taken place using traditional learning materials. The tablets certainly added richness to the task in terms of information presentational and storage, but were not inherently needed to facilitate the learning activity.

9.3.3 Meta capabilities: A quest for the magic bullet capability?

A third challenge against my conceptualisation and positioning of ML4D could arise from a different direction, questioning whether my positioning of mobile learning might not be too modest. It could be argued that, if I claim that mobile learning through its intersection between technology and education pertains to the expansion of multiple set of capabilities, I should position mobile learning as some form of meta- or uber-capability. Seeing that the educational use of mobile devices nurtures the subsequent generation of dimensions of capabilities, it could be seen as a higher-order capability. In fact, this line of argument, i.e. to identify meta-capabilities, has some currency in the capabilities literature. Sen (1999) suggests some form of basic capabilities such as education and health to be required to explore further capabilities. Nussbaum (2011), too, sees a range of core capabilities while Wolff and De-Shalit (2009) coin the concept of 'fertile capabilities' to explain a form of capability that is required to be present in order to generate additional capabilities.

Outside the theoretical CA literature, a range of capabilities have been positioned as of special importance too. Venkatapuran (2011) argues to regard the capability to be healthy as a meta-capability. However, this would contradict Kleine (2013) and Gigler (2015) who position technologies and informational capabilities respectively to be of central importance among different types of capabilities. It seems then that there might be a risk to follow a quest for a magic bullet capability by adding mobile learning to this list.

I have already discussed why I do not regard mobile technologies in education to constitute 'technologies of freedom' (see section 7.3). And, I would again caution here that while

mobile learning in LMICs can support a diverse set of capabilities, this is neither a necessary outcome of the educational use of mobiles, nor are these capabilities a unique property of mobile learning itself. Other interventions might yield similar capability configurations and I do not see the educational use of mobiles to present some form of higher-level capability construct. In this, I disagree with Gigler (2015), who positioned informational capabilities as a core central capability required for technology users to then explore their social capabilities. While I do agree that this linear progress might indeed be the case, I do not assume that informational capabilities are a necessary condition for social capabilities to be generated.

To position myself in this debate on meta-capabilities then, I would agree with Sen's position that there probably exist a range of basic capabilities; but as he does, I do not think a universal list of these can be created, and neither would I add a capability to use mobile technologies to this. Sen (1999) and Unterhalter (2005) both regard education as a basic capability. I agree with this conceptualisation and it could be argued that this sufficiently covers mobile learning's capability set too. Suggestion to add a basic capability around access and use to information and ICTs seem to be increasingly justified too, and I would again regard mobile learning to present a sub-set of this higher-level capability.

Sectoral thinking and multipurpose technologies

The above discussion, however, highlights the needs to elaborate a bit further on the implications of mobile learning's contribution to multidimensional capability sets. Mobile technologies do indeed present multipurpose technologies (Kleine 2013) allowing them to support the exploration of multiple dimensions of capabilities simultaneously. Alas, sectoral thinking within the Capabilities and Development literature seems to "not adequately grasp the full transformative and often highly personalised effect that access to the internet or even a mobile phone can have on people's lives" (Kleine 2013: 7). Current capability frameworks therefore run at risk to analyse human well-being and development in silos. It seems unintuitive, for example, that both the capability to be healthy and the capability to be educated are discussed as if one capability should take precedent over the other (c.f. Venkatapuran 2011; Walker 2006), when both seem to be of equal importance to any individual's well-being and development. This sectoral thinking is in particular irrelevant for cross-cutting interventions such as ICTs and mobile technologies, who by design allow individuals to choose what type of capability set and opportunities to explore.

An interesting suggestion in this regard in raised by Donner (2016). An early proponent of mobiles for development (M4D), in his 2016 book 'after access' he suggests that "it is time to

retire the term M4D" (Donner 2016: 190). He reaches this conclusion based on the observation that the boundaries between devices, functionalities, and application between mobiles and other ICTs have blurred to such an extent that is makes little sense to distinguish between ICT4D and M4D. While this is a welcome contribution to the wider discourse on technologies and development, I do not think this argument is applicable to ML4D too. The educational component of mobile learning is too central in my conception of ML4D as to allow a full merging of mobile learning in LMICs with the ICT4D domain.

9.3.4 Revisiting the gap in existing capabilities frameworks

In chapter 2, section 2.4.1–2.4.4 I discussed existing capabilities frameworks in ICT4D and education in LMICs, and held that none of them provided an off-the-shelf conception that could be transferred to ML4D. In order to fill this conceptual gap, I therefore generated a capabilities conception unique to ML4D. However, to safeguard against constructing a straw man argument, it seems relevant to briefly revisit my argument for rejecting existing frameworks and whether my developed capabilities conception is indeed substantially different from these.

My argument for rejecting existing frameworks was based on three pillars. First, there was no existing framework tailored to mobile technologies and the potential capabilities that users might develop from their particular affordances. Most work on ICT4D has focused on ICT policies and fixed infrastructure such as telecentres, desktop computers, and access to the internet in general (c.f. Kleine 2013; Gigler 2015). This overlooks important aspects of mobile technologies, which are far more widespread in terms of access, and personal in terms of use, than other ICTs. Frankly, mobile technologies might be relevant to a more diverse set of valued opportunities and functionings. Second, because conceptual work on the CA and ICTs has focused on fixed technologies, it is often overly sectoral and not geared towards the diverse applications of mobile technologies to support users' pursuit of a good life. Mobile technologies, due to their constant and personalised use in all areas of life, might be able to help users explore multiple types of capabilities simultaneously.

Third, from an educational perspective existing work on the CA has not investigated the use of technology, either fixed or mobile. While there is a rich body of literature on teachers' capabilities, this literature does not feature the use of technologies as an educational tool at all. There seems to be good reason to believe that learners and teachers can generate different types of capabilities and valued functionings from the conversion of mobile technologies as an educational resource. Existing work on the CA in education and LMICs

has not explored these opportunities and there thus remained a gap on investigating mobile learning in LMICs from a capabilities perspective.

Reviewing these arguments, it appears that my capabilities framework indeed fills a relevant gap. The four dimensions of capabilities illustrate the interplay of different opportunity sets that users explored through the educational use of mobiles. The four dimensions of capabilities and their interplay to support human development come about through the simultaneous exploration of diverse types of capabilities related to different affordances of mobiles. For example, the ability to use mobiles to communicate and access information across contexts, which fixed ICTs do not provide, can give rise to informational, educational, and societal capabilities at once. Existing capabilities frameworks of ICT4D and education in LMICs have not catered for this simultaneous exploration and overlap in capability sets. This is probably the strongest rationale for the development of a tailored capabilities conception for ML4D.

However, in addition, my descriptive framework also reflects a range of individual capabilities that are not covered in existing work. Seeing that there is no existing work on investigating the use of educational technologies in LMICs from a capabilities perspective, the developed educational capabilities and their interplay with the other three dimensions of capabilities present a contribution to the literature. Likewise, a range of induvial capabilities linked to the particular affordances of mobile devices have not been reported in the context of LMICs before. Though, I would position these individual capabilities as slightly less prominent as they are contextualised by the nature of the ICT4RED intervention and the socio-economic realities of teachers in rural South Africa. All in all, I would therefore reinforce my initial assessment that the development of a descriptive capabilities framework tailored to ML4D was warranted.

Comparing my capabilities framework for teachers' educational use of mobile technologies directly with Gigler's (2015) capabilities framework for 'Development as freedom in a digital age' illustrates a few additional differences. For Gigler (2015), informational capabilities consist of: (a) ICT capability; (b) information literacy; (c) communications capability; and (d) content capability. In my case study research, the technological focus from fixed ICT infrastructure to support indigenous development was changed to a mobile learning initiative to support rural education development. This change in context changed the composition of valued informational capabilities adding the notions of being able to be mobile and to merge social and professional contexts and realities. In the context of the ICT4RED case study, teachers also attached little value to the idea of using the tablets to generate independent

teaching content—akin to Gigler's content capability. Creating educational content was seen as the government's responsibility, whereas it was teachers' responsibility to ensure that learners successfully mastered the provided content in order to build the nation. To develop their own independent content then was perceived by teachers as to undermine their educational efforts. Again, this highlights the importance of context in the definition of valued capabilities and underlines my caution to position my 21 individual capabilities as a general framework to guide the assessment of ML4D interventions.

Furthermore, my capabilities conception of ML4D differs from Gigler conception in terms of the positioning of informational capabilities. As discussed in the prior section (9.3.3), I do not regard informational capabilities to present a catalyst for the development of subsequent capabilities. I regard them as one of four dimensions of capabilities whose interplay should be the focus of the analysis. This focus on the interplay then presents a second distinction to Gigler's capabilities framework, which assumes a linear progression from informational capabilities. Such a linear progression of capabilities is also inherent in Venkatapuram's conception of the capability to be healthy, which presents a starting point for the exploration of different capabilities once it is met. In addition to rejecting the idea of informational capability as a meta-capability, I would therefore also reject the idea of a linear progression between capabilities in favour of a multi-dimensional interplay between capabilities.

Comparing my framework to Kleine's Choice Framework, also highlights a range of differences. First, and crucially, her framework is normative while my framework is descriptive. That is, Kleine's framework was developed a priori and then applied to a case study in Chile to test it; in contrast, my framework was developed as a tool to describe teachers' educational use of mobile technologies from a capabilities perspective. I only positioned my capabilities conception of ML4D as normative.

Second, as a normative framework, Kleine offers a range of additional conceptual lenses that my framework does not provide. This refers in particular to her rich unpacking of conversion factors (structure), aspects of agency, and degrees of empowerment. It is in the former two—that is, conversion factors and aspects of agency—that my framework could benefits from a closer integration with Kleine. Third, Kleine's focus on empowerment as an operationalisation of the freedom to achieve development outcomes is too narrow in the context of ML4D. This layer of her framework is akin to my four dimensions of capabilities. The particular nature of mobile learning interventions requires a more diverse conception of capabilities, which is reflected in my four dimensions. Fourth, and last, her framework is not

catering for the educational use of technologies as it departs from a different starting point, that is ICT use by Chilean entrepreneurs. All in all, Kleine's framework provides helpful insights into the conversion factors and agency aspect when using technologies, which seem directly relevant to a potential redesign of my theory of change for ML4D as suggested above in 9.1

9.3.5 In defence of human capital and economic growth

A last consideration in the defence of my capabilities conception of ML4D refers to my positioning of human capital and growth-focused understandings of education and development in LMICs. Throughout the thesis, I have criticised both and bemoaned an overly narrow focus on these as a manner in which to conceptualise ML4D. It thus seems helpful to briefly clarify the context of these critiques and their applicability given the empirical data presented in this thesis.

First, in terms of the empirical data presented, both the systematic review and the case study highlight the intrinsic importance of increases in human capital and economic growth (i.e. employment opportunities). While the meta-analysis and theory of change do not establish any impacts on development outcomes as defined in terms of human capital and employment, this is based on a lack of evidence evaluating these outcomes. In the thematic synthesis, however, a range of mechanisms pointed to increases in economic capital. The same applies to my case study findings, in which economic capabilities presented one of four dimensions of valued opportunities explored by teachers through the use of tablets. What is more, one of the teachers' key configurations of capabilities in order to pursue their valued beings and doings—the configuration of building the nation—had a strong human capital and productivity focus. This thus suggests that my capabilities conception of ML4D is indeed compatible with a human capital / growth-focused conception of development.

Second, this finding is in line with the general positioning of Sen's CA, which is not assumed to be mutually exclusive with different approaches to development including growth-focused or income-based approaches. In the words of Sen (1999: 7): "It is hard to think that any process of substantial development can do without very extensive use of markets, but that does not preclude the role of social support, public regulation, or statecraft when they can enrich—rather than impoverish—human lives" and, further, "Growth of GNP or of individual incomes can, of course, be very important *means* to expanding the freedoms enjoyed by the members of society." (Sen 1999: 2, emphasis in the original). That human capital- and employment-focused aspects then feature strongly within my empirical findings including the

case study results should not surprise as much. The CA inherently is positioned to present a more encompassing conception of human well-being and development, which includes the important role that increases in human capital and economic growth have to play in this. However, the CA extends beyond these as does my conception of ML4D.

In sum, my position of ML4D should not be seen as in conflict with human capital and growth-focused approaches to development. They provide but one component of the many components of human well-being, agency, and development. My critique of these is in the context of these approaches to development dominating conceptions of ML4D at the expense of other voices on how the use of technologies can support education and development in LMICs. Chimamanda Adichie's parable of the 'Danger of a single story' perhaps best captures my positioning of human capital and growth-focused conceptions of development. As she explains: "The problem with single stories is not that they are untrue, but that they are incomplete. They make one story become the only story...." (Adichie 2014: 4).

Conclusion

This chapter has concluded my exploration of the positioning of mobile learning's role and contribution in international development. It conceptualises mobile learning's contribution to development as the use of mobile technologies to support an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of educational agents. In the absence of evidence supporting mobile learning's impact on development outcomes, I present a focus on the role of mobile technologies to expand teachers' and learners' valued functionings and capabilities as an alternative conception of the links between mobile technologies, their use for educational purposes in LMICs, and development outcomes. This understanding and positioning of mobile learning in international development differs from existing conceptualisations by shifting the focus of mobile learning interventions away from access to technologies and content provision towards four different foci of interest: (1) an opportunity focus; (2) a people focus; (3) an education focus; and (4) a transformation focus.

To support this line of argument, this chapter has juxtaposed my capabilities conception of mobile learning with the findings of my mixed-methods systematic review. It also has outlined five challenges to my conception and positioning of ML4D and suggested responses to each. I thereby illustrate that my developed conceptualisation of ML4D directly answers my thesis's research question of how the use of mobile technologies as an educational tool in LMICs can be conceptualised as a development intervention. Having done so, I next discuss limitations of my research and implications for future work in the final chapter of thesis, which will also provide a brief overview of the main findings and conclusions.

Chapter 10. Conclusion of the thesis

Introduction

This chapter concludes my thesis. It provides an overview and summary of the main findings of my empirical research and the argument developed. It also recounts the contributions of my thesis to the wider research and practice on mobile learning for development (ML4D) and outlines some implications of my research findings for policy and practice. The chapter commences with a summary of the thesis's findings and contributions followed by some of the implications thereof and suggestions for further research. The chapter concludes with a few last remarks on the future of ML4D.

10.1 Limitations of the research

As all research, the empirical research conducted as part of this thesis is subject to a range of limitations. This applies equally to the systematic review research and the case study research. I outlined the limitations for both research methods in chapter 4, section 4.2.6 and 4.3.5. These limitations should be kept in mind when considering the below summary of findings and implications of my research.

10.2 Summary of findings and contribution of the thesis

Having reviewed and cautioned the limitations of the empirical research conducted as part of this thesis, I next provide a summary of the main findings and the contribution of the thesis.

Chapter 10: Conclusion

10.2.1 Summary of findings and argument

My thesis commenced on the premise that the positioning of mobile learning's role and contribution in international development was neither grounded in an empirical evidencebase nor grounded in a thorough conceptual foundation. In my literature review, I outlined how mobile learning is assumed to contribute to education and development outcomes but then highlighted three key critiques that challenged how ML4D has been understood and positioned in development policy and practice. First, there is a disconnect between the claimed vast development potential of ML4D and the empirical evidence-base cited to support these claims. Second, there is a lack of an explicit theory of change for how the use of mobile technologies is assumed to support education in LMICs and subsequent development outcomes. Current ML4D programmes and policy positions assume simplistic causalities and under-define both education and development. Third, due to its technocentric positioning, ML4D interventions run a risk of circumventing and undermining the existing education system. This positioning presents ML4D as exogenous to existing education and is at odds with a people-focused perspective of education and development in Low- and Middle-Income Countries (LMICs).

I then conducted a mixed-methods systematic review in order to assess the empirical validity of the claimed contribution of mobile learning to education and development in LMICs. My review findings reject a link between using mobile technologies as an educational tool in LMICs and development outcomes. Having systematically reviewed and synthesised both quantitative and qualitative studies investigating the effects mobile learning programmes in LMICs, I provided four linked sets of findings to support my argument that, in its current application, *mobile learning programmes cannot claim to support development outcomes*.

First, my mixed-methods systematic review found an absence of evidence between the provision of mobile technologies as an educational tool in LMICs and impacts on socioeconomic development. Second, while my systematic review identified a statistically significant impact on learning outcomes, these effects were not sufficient to claim a meaningful change in learners' abilities and human development. Third, two structural biases in who is included in mobile learning programmes undermined the positioning of the mobile learning interventions to target development outcomes: mobile learning interventions overlooked the most disadvantaged groups in society and reflected existing structural inequalities within education systems in LMICs. Fourth, a narrow conception of development as an increase in human capital hindered mobile learning's potential to support development outcomes. I therefore concluded that mobile learning cannot be positioned as development

271

intervention based on current evidence and showed how this finding runs counter to popular narratives on ML4D.

The findings of my mixed-methods systematic review provided my empirical justification that a rethink of the positioning and conceptualisation of ML4D is required. Based on its application in related bodies of literature, I identified the Capability Approach (CA) as a relevant theoretical framework to guide a conceptualisation of mobile learning's role and contribution in international development. I then tested the CA's operationalisation for this purpose using a qualitative case study of a mobile learning programme in rural South Africa. In the case study, I investigated whether teachers' use of mobile technologies allowed them to explore valued functionings and capabilities. The case study found that teachers' use of mobile technologies can best be understood as an expansion in four dimensions of capabilities: informational, educational, societal, and economic capabilities, which taken together can enhance teachers' well-being and human development. This was based on the observation of 22 individual capabilities that teachers explored using mobile technologies, which were then synthesised into an analytical framework to illustrate how the careful configuration of these capabilities, according to teachers' preferences and contexts, allowed them to step-wise develop agency in order to move closer to their conception of a good life.

I then applied the case study results on teachers' use of mobiles technologies to derive a conceptualisation of ML4D based on the CA. I formulated my conception of ML4D from a capabilities perspective as: the use of mobile technologies to support an endogenous transformation of education in LMICs anchored in the primary objective of enhancing the capabilities and agency of educational agents. This conception defines the space in which the effects of ML4D interventions should be assessed as the valued beings and doings and corresponding capabilities of educational agents. It defines ML4D's contribution to education and development as to foster an endogenous transformation located within educational agents and structures based on their self-determined exploration of beings and doings aligned to their conception of a live they reasoned to value. In the absence of evidence supporting mobile learning's impact on development outcomes, a focus on the role of mobile technologies to expand teachers' and learners' valued functionings and capabilities presents an alternative conception of the links between mobile technologies, their use for educational purposes in LMICs, and development outcomes. This conceptualisation and positioning of mobile learning in international development differs from existing understandings by shifting the focus of mobile learning interventions away from access to technologies and content provision towards four different foci of interest: (1) an opportunity focus; (2) a people focus; (3) an education focus; and (4) a transformation focus. I concluded my argument with a

272

defence of my capabilities conception of ML4D and the subsequent positioning of mobile learning in international development against a selected range of potential critiques.

In sum, rethinking ML4D, I thus position the educational application of mobile technologies in LMICs to carry the potential to contribute to an expansion of human capabilities and agency. Applying a capabilities perspective to conceptualise ML4D shifts mobile learning's role and contribution in international development in line with an opportunity-, people-, education-, and transformation-focused understanding of using technologies as an educational tool in LMICs. It is in this conceptual focus away from a techno-centric, access-based view of mobile technologies towards a view that prioritises how educational actors can apply these technologies to enhance their capability and agency to live a life they have reason to value.

10.2.2 Review of thesis contributions

In chapter 1, section 1.3 I outlined the specific contributions of my thesis, which I briefly revisit here. On an empirical level, my thesis advances the evidence-base on the use of mobile technologies to support education and development in LMICs. It systematically brings together, appraises, and synthesises the published research evidence on mobile learning in LMICs. I used this rigorous synthesised evidence-base to contribute to existing debates on the perceived impact of mobile learning on education outcomes and development progress. This contribution aims to ground the narrative and claimed potential of ML4D in reliable research evidence. The systematic review also allowed me to construct a detailed theory of change for ML4D programmes. This theory of change contributes an explicit investigation of the presumed links between mobile technologies, education, and development allowing for a detailed investigation of their empirical and conceptual validity. Taken together, the systematic review supports a shift in the discourse on ML4D away from anecdotal evidence, good intentions, and plausible but simple theories of mobile learning's contribution to development.

The thesis also contributes the first case study of applying the CA as an analytical framework in a primary research project on ML4D. It presented a list of capabilities that teachers explored resulting from the use of mobile technologies in an educational setting and highlighted the interplay between the identified capabilities to contribute a novel empirical lens on the effects of mobile learning programmes in LMICs. The case study further confirms the assumption that the CA can be operationalised to investigate mobile

learning programmes adding to its existing application in related bodies of research such as ICT4D.

On a conceptual level, my thesis developed a detailed conceptualisation of ML4D, which I have outlined above. This conceptualisation defines and grounds the contribution of mobile learning to development firmly in the use of technologies to support educational agents in LMICs to live a life they have reason to value. I have outlined how this shifts the discourse on ML4D towards a more opportunity-, people-, education-, and transformation-focused understanding of using technologies as an educational tool in LMICs. This conceptualisation aims to complement existing understandings of ML4D, in particular human capital and growth-focused approaches. It contributes an alternative way to think about ML4D that hopefully can contribute towards more effective mobile learning policy and programme design in LMICs.

Lastly, the conducted mixed-methods systematic review makes a methodological contribution to a nascent body of work on using mixed-methods approaches to systematic reviews of development programmes and policies. My thesis also makes a minor methodological contribution to the operationalisation of the CA as an analytical device by illustrating the CA's versatility in a body of research in which it had not been applied before.

10.3 Implications of the research findings for policy and practice

The findings of my research carry a range of implications for both ML4D policy-making and practice. These implications can be separated into the implications of the empirical findings of my systematic review of the ML4D evidence-base and the implications of my developed capabilities conception of ML4D. Both are discussed in return.

10.3.1 Implications of my systematic review findings

The findings of my mixed-methods review indicate a clear mismatch between the claimed and the empirical impact of mobile learning in LMICs on development outcomes. Above all, the implications of these results are a strong call for a more evidence-informed approach to ML4D policy and practice. Proponents for the large-scale use of mobiles to support education in LMICs carry the responsibility to present proportional evidence claims to justify their position. Given the clear results of my systematic review that the current evidence-base does not support claims to mobile learning's positioning as a tool for development, policy-makers and practitioners should exercise caution in the adoption of ML4D programmes.

The review further clearly implies the need for an explicit theory of change for how mobile learning programmes can be designed to support education and development outcomes. My thesis presented a first attempt to construct such a theory of change, which could serve as a tool for policy-makers and practitioners to assess and think-through the adoption of ML4D programmes. The theory of change illustrates the different pathways through which the use of mobiles in an educational context could influence development outcomes. It also indicates the different contextual factors and mechanisms at play in each of these pathways. The theory of change thereby could allow policy-makers and practitioners to interrogate their own assumptions of how mobile learning can be used as a tool for education and development in their respective contexts, and what pathway might be the most relevant and effective approach when implementing such mobile learning programmes.

Moreover, the systematic review and its theory of change make some direct recommendations for the design and implementation of ML4D programmes. Having appraised and synthesised all the available evidence on existing mobile learning programmes in LMICs three key design and implementation implications stand out. First, programmes using mobile technologies to support education in LMICs need to place greater emphasis on the pedagogical considerations underpinning programme design. A detailed integration of mobile learning pedagogies into programme design has the potential to enhance programme effects as well as to expand the range of applications of mobiles to support human development. For example, programme designs could draw on Kearney and colleagues' framework for the facilitation of pedagogical rich mobile learning pedagogies identified in the thesis refer to game-based learning, context-crossing, authentic learning, and collaborative learning.

Second, mobile learning programmes need to be embedded within the existing formal education system. My review found no evidence that mobile learning programmes outside the formal education system present a viable and effective long-term intervention approach. It is therefore crucial to deepen the current integration of mobile learning programmes into the existing education system. Such an integration would require a greater focus on how mobile technologies can be used to support school management and administration as well as how they relate to developing educational leadership and management structures. This is

an area of ML4D that remains under-explored despite seemingly possessing potential to affect more systemic changes in education in LMICs. This systemic angle on using mobiles to support school management and governance could further incentivise national governments in LMICs to play an increased role in driving investment into educational mobile technologies rather than leaving this drive to private sectors companies, research institutes, and NGOs.

Third, my systematic review findings constitute a strong recommendation to ML4D programmes to focus more on teachers during intervention design and implementation. The reviewed empirical evidence points towards the centrality of teachers in most pathways from the educational use of mobiles to development outcomes. Teachers are central to the effective adoption and implementation of ML4D programmes; and, further, in their own right present a major constituency that could benefit from the introduction of mobile learning programmes. Alas, the vast majority of ML4D programmes currently bypasses teachers to focus largely on learners, a practice which my research results suggest is neither sustainably in the long-term nor beneficial for the results of mobile learning programmes in the short term. Using mobile learning to support teacher training and development in LMICs stands out as a particular promising programme approach.

10.3.2 Implications of my capabilities conception of ML4D

In addition to my systematic review findings, my capabilities conception of ML4D has a range of implications for ML4D policy design and practice too. Applying a capabilities perspective to conceptualise ML4D shifts mobile learning's role and contribution in international development in line with an opportunity-, people-, education-, and transformation-focused understanding of using technologies as an educational tool in LMICs. This shift implies a conceptual focus away from a techno-centric, access-based view of mobile technologies towards a view that prioritises how educational actors can apply these technologies to enhance their capability and agency to live a life they have reason to value. However, in order to achieve this enhancement of capabilities and agency of educational actors, mobile learning programs need to be carefully designed for this particular objective. It is by no means an inherent property of ML4D interventions to exercise an expansion of human capabilities and agency.

The explicit integration of the CA into mobile learning programme design is currently nonexistent in the literature on ML4D. However, there are a range of examples from the broader ICT4D literature in which the CA is applied directly to inform programme design⁸³. For instance, Poveda and Roberts (2017) report on two examples of designing ICT4D interventions through a capabilities lens. This refers to the Asikana Network, an NGO formed by women working in Zambia's ICT sector, which uses technologies to provide training, mentoring, and networking activities to support women in the ICT sector experiencing gender discrimination. The second example is provided by CDI-Campinas in Brazil, a digital inclusion organisation providing ICT skills education with a particular focus on using ICTs to further social justice and empowerment. In both examples the authors highlight in detail how the CA was directly applied to shape an ICT4D intervention design in which participants can use ICTs to self-determine and self-actualise their own development.

This work resonates strongly with the implications of my capabilities conception for the design of future ML4D programmes. First, my thematic synthesis as well as my qualitative case study identified changes in self-efficacy, agency, and identities to be key mechanisms through which the use of mobiles by educational actors was linked to development outcomes. For example, teachers reported a range of valued opportunities related to personal and professional growth as well as related to addressing historical injustices, which were directly associated with their well-being. However, these mechanisms around personal empowerment are currently not integrated into mobile learning programme designs and largely emerge from qualitative studies of teachers' and learners' perceptions on using mobile devices.

Applying my CA conception of ML4D with its primary objective of enhancing the capabilities and agency of educational actors allows for a closer integration of these mechanisms into ML4D programme design. Rather than pre-specifying programme outcomes, a CA-inspired design would call for a more open-ended and participatory lens on programme outcomes. This would give rise to design aspects such as personal ownership of devices, constant connectivity, opportunities to use the devices across contexts, and more room for collaboration—each of which was highly valued by technology users and facilitated a range of valued opportunities. The key aspect here is however not the specific design criteria that a CA-inspired programme approach could bring to light, but the more over-arching design principle of a more open-ended and participatory understanding of programme processes and outcomes. Above all, a ML4D programme based on my capabilities conception would

⁸³ This further acknowledges that this thesis's positioning of the CA as a useful conceptual framework to assess and unpack ML4D interventions reflect parallel arguments made in the wider ICT4D literature (see more in chapter 3, sections 3.2–3.4.

allow technology users more power to self-determine technology usages and room to explore how best the technologies can be integrated with their own pursuits of a good life.

Such a more open-ended and participatory programme design then would have further implications on a related design aspect of ML4D: the risk that mobile learning programmes reproduce—rather than counter—structural inequalities by overlooking the most disadvantaged groups in society. For example, my systematic review found a strong bias towards urban areas and more affluent educational environments within ML4D intervention designs. Applying my capabilities conception, the emphasis in ML4D intervention design would shift from the provision of resources (i.e. mobile technologies) to the conversion factors of individuals to apply these resources to an effective use to further their own valued beings and doings. This capabilities lens thus highlights the discrepancy between the conversation factors of more disadvantaged groups, the technology and programme design provided, and the assumed development impacts. The suggested capabilities conception of ML4D should therefore reduce the risk of ML4D reproducing structural inequalities as to who is included and can benefit from ML4D interventions.

A last implication of my capabilities conception of ML4D relates to the integration of Freire's critical pedagogy within mobile learning programme in LMICs (Freire 1972). This particular type of pedagogy seems to hold large promise to guide the design of learning and teaching with mobiles to further development outcomes. Freire proposed an educational approach that supports marginalised groups to develop a 'critical conscious' allowing them to identify the root causes of their oppression and to formulate strategies to overcome this oppression. This pedagogical approach seems to provide a useful tool to operationalise my CA conception of ML4D which calls for an enhancement of the capabilities and agency of educational agents in LMICs to live a life they have reason to value. There is a small body of work that has explored a potential synergy between the CA and Freire's critical pedagogy (Unterhalter 2017; Roberts 2016; Petit & Santos 2014; Gigler 2015). However, none of this work provides an empirical or conceptual investigation of how mobile learning programmes in LMICs could support critical pedagogies in order to enhance the capabilities of educational actors to support their human development and agency.

Such exploration was unfortunately beyond the scope of my thesis, but seems to present a rich area for future research. For example, applying a critical pedagogy lens to the ICT4RED programme would have highlighted the lack of opportunities within the programme design for teachers to challenge existing narratives around their professional role and contribution. Poveda and Roberts show how the explicit incorporation of critical pedagogy techniques

such as critical dialogue and problem-posing methods into ICT4D programme design leads to a use of ICTs that allows people to develop a critical agency and conscious required to take control over their own development and well-being. Mobile learning programmes such as ICT4RED could apply a number of designs that incorporate elements of a critical pedagogy to support educational actors to enhance their capabilities and valued functionings.

10.4 Implications of the thesis for future research

The findings of my thesis have implications for a range of areas for future research. Below, I briefly describe four main areas that could be explored by future research.

First, having confirmed the applicability and relevance of the CA as a conceptual tool to assess ML4D programmes, I would argue for an increased empirical research effort into the CA and ML4D. To my knowledge, this thesis presents the first empirical investigation of a ML4D project using the CA as an analytical lens. Research and practice on the use of mobiles to support teaching and learning in LMICs has so far not been based on the CA. Given the proposed capabilities conception of ML4D, further research applying the CA to investigate the use of mobiles to support teaching and learning in LMICs is required. This thesis has outlined how the use of the CA's analytical devices can support a rich empirical analysis of mobile learning in LMICs and proposed a conceptualisation of ML4D based on this investigation. However, so far, this conceptualisation is informed by a single case study and further research is needed to validate and refine my proposed conception. Such research should investigate applied ML4D programmes evaluating programme effects using a CA lens. This would allow for additional data sets on what valued capabilities and functionings educational actors in LMICs explored following the provision of mobiles, and why. A particular focus on learners' capabilities and functionings would be welcome as my case study research did not explore this angle.

Second, in addition to calling for an enhanced application of the CA to investigate ML4D, I would also advocate for an increased use of explicit mobile learning frameworks to inform future research on ML4D. For example, research could apply Kearney and colleagues' (2012) framework for mobile learning to investigate the pedagogical contribution of

programmes using mobile technologies to support education in LMICs. This thesis has shown that while the pedagogical underpinnings of mobile learning programs are crucial for their educational impact, this pedagogical aspect is often under-explored in programme design and implementation. Incorporating educational research on mobile learning, in particular Kearney and colleagues' framework, could support mobile learning practice in LMICs. However, this requires a sound evidence-base comprised of primary research applying explicit mobile learning frameworks as well as research synthesis unpacking in more detail the extent to which existing ML4D interventions have adopted explicit pedagogical principles of mobile learning⁸⁴.

Third, further research also is required in relation to the CA's methodological individualism. In order to support an investigation of the structural and political factors mitigating individual's human development and agency, additional conceptual frameworks and tools are required. A rich body of work in this regard, which should be explored further in relation to my capabilities conception of ML4D, relates to the integration and combination of social justice theories and critical theories with the CA (e.g. Tilky & Barrett 2011; Venkatapuram 2011; Winters et al 2017; Zheng & Stahl 2011; 2013). A particular strong overlap for future research in this regard seems to rest in the incorporation of one particular type of critical theory with my CA conceptualisation of ML4D: Freire's critical pedagogy as flagged above (Freire 1972).

Fourth and last, a strong area for future research on the role of mobile learning in international development is to investigate the implications of the SDGs and new EFA targets and the shift in educational and developmental policy priorities that these reflect. Much of the core work on ML4D at a policy level has focused on mobile learning's role in the area of the MDGs (c.f. UNESCO 2012a; GSMA 2010; Unwin 2015). But, these policy priorities have now changed in the era of the SDGs, which were announced in September 2015 and provide a framework for sustainable global development from 2016 to 2030. From a mobile learning perspective, this shift in global development priorities should be a positive signal. As argued in this thesis, an access-based, techno-centric understanding of mobile learning and teaching with mobiles in LMICs. A policy environment that is more focused on quality rather than access to education thus supports a research and practice agenda on ML4D that is more focused on the pedagogical value and contribution of mobile technologies in an educational context.

⁸⁴ An example of such an investigation in the field of mHealth in LMICs is provided by Winters and colleagues (2017).

Chapter 10: Conclusion

Conclusion

In this thesis I have proposed a rethink of ML4D. When I started my research, I was motivated by a surprising lack of rigorous evidence and conceptual models on what was (and is) a growing range of public and private programmes and projects on mobile learning in LMICs. The educational use of mobile technologies continues to enjoy high-level development policy support as much as high-level commercial private sector interest. Unfortunately, the need for a rethink of ML4D remains as acute now as it was in 2014 when I started my research. Despite more primary research being produced on the effects of mobile learning in LMICs, the highly optimistic narrative on mobile technologies' revolutionary potential to change education in LMICs has not been altered.

In this context, my thesis could be seen as a pessimistic contribution, challenging many of the claims laid to ML4D's impact as well as the conceptual arguments made in its favour. However, while I certainly would want to pose a strong challenge to current development policy thinking on ML4D, I would suggest a more reconciliatory reading of my thesis. Above all, I regard my thesis as a call for humility and curiosity among the ML4D community. Humility in terms of how we advocate for and portray the impacts of mobile learning in LMICs; and curiosity in terms of how we develop and extend our conceptual models of the range of applications in which the educational use of mobile technologies in LMICs can support people's lives. My empirical research in rural South Africa and the developed capabilities conception of ML4D illustrate a much larger range of valued beings and doings and opportunities that could be explored through the use of mobiles than what current understandings of ML4D incorporate.

Mobile learning can be a powerful tool to support education in LMICs; the rich pedagogical applications observed in my case study and systematic review pay testimony to this. Likewise, mobile learning can support individuals to enhance their opportunities to live a life they have reason to value; it can carry the potential to contribute to an expansion of human capabilities and agency. Though, this contribution is more adequately defined as a series of step-wise endogenous changes in educational practices and processes driven by educational actors' own definitions of valued beings and doings. By no means can and should ML4D be attributed revolutionary impacts. Mobile learning can only be effective alongside a healthy and functioning formal education system, which provides the educational structures required for mobile learning to make its distinct contribution to education and development in LMICs. Therefore, I would advocate strongly for an understanding of ML4D that focuses on an endogenous transformation of education in

LMICs to which the carefully designed educational use of mobile technologies can make but one contribution.

References

- Agence Française de Développement (AFD). (2015). *Digital Services for Education in Africa*. AFD: Paris, France.
- Adera, E. O., Waema, T. M., & May, J. D. (Eds.). (2014). *ICT pathways to poverty reduction: Empirical evidence from East and Southern Africa*. IDRC: Ottawa, Canada.
- Aker, J. C., & Mbiti, I. M. (2010). Mobile Phones and in Africa Economic Development. *The Journal of Economin Perspectives*, 24(3), 207–232.
- Alderson, P., & Morrow, V. (2011). The ethics of research with children and young people: A practical handbook. SAGE: London, UK.
- Alkire, S. (2005). *Valuing freedoms: Sen's capability approach and poverty reduction*. Oxford University Press: Oxford, UK.
- Ally, M. (Ed.). (2009). *Mobile learning: Transforming the delivery of education and training.* Athabasca University Press: Alberta, Canada.
- Ananny, M., & Winters, N. (2007). Designing for development: Understanding one laptop per child in its historical context. 2007 International Conference on Information and Communication Technologies and Development, ICTD 2007.
- Andersson, A., Grönlund, Å., & Wicander, G. (2012). Development as freedom how the Capability Approach can be used in ICT4D research and practice. *Information Technology for Development*, *18*(1), 1–4.
- Asabere, N. Y. (2013). Benefits and challenges of mobile learning implementation: Story of developing nations. *International Journal of Computer Applications*, 73(1).
- Alsop, R., & Heinsohn, N. (2005). *Measuring empowerment in practice: Structuring analysis and framing indicators*. World Bank: Washington DC, US.
- Avgerou, C. (2010). Discourses on ICT and Development. *Information Technologies & International Development*, 6(3), 1–18.
- Balasubramanian, K., Thamizoli, P., Umar, A., & Kanwar, A. (2010). Using mobile phones to promote lifelong learning among rural women in Southern India. *Distance Education*, *31*(2), 193-209.
- Banerjee, A. V., & Duflo, E. (2011). *Poor economics: A radical rethinking of the way to fight global poverty*. Public Affairs: Boston, US.
- Banks, K. (2014). Mobile learning in the last mile. *Prospects*, 44(1), 5-11.
- Baxter, P., & Jack, S. (2008). Qualitative case study methodology: Study design and implementation for novice researchers. *The qualitative report*, *13*(4), 544-559.
- Batchelor, J. (2007). *From zero to hero–is the mobile phone a viable learning tool for Africa?* CSIR: Pretoria, South Africa.
- Beatty, A., and Pritchett, L. (2012). From Schooling Goals to Learning Goals: How Fast Can Student Learning Improve? CGD Policy Paper 012. Accessed 20 October 2017: http://www.cgdev.org/files/1426531 file Beatty Pritchett Time to MLG FINAL.pdf.
- Becker, L. A., & Oxman, A. D. (2008). Overviews of reviews. *Cochrane handbook for systematic reviews of interventions: Cochrane book series*, 607-631.
- Bell, F., 2010, May. Network theories for technology-enabled learning and social change: Connectivism and actor network theory. In *Networked learning conference 2010: Seventh international conference on networked learning*. Accessed 20 October 2017: <u>http://usir.salford.ac.uk/9270/?utm_source=twitterfeed&utm_medium=twitter</u>
- Bello-Bravo, J., Tam??, M., Dannon, E. A., & Pittendrigh, B. R. (2017). An assessment of

learning gains from educational animated videos versus traditional extension presentations among farmers in Benin*. *Information Technology for Development*, *1102*(April), 1–21.

- Bentham, J. (1789). *An Introduction to the Principles of Morals and Legislation*. Printed in the Year 1780, and Now First Published. T. Payne, and Son.
- Beuermann, D. W., Cristia, J., Cueto, S., Malamud, O., & Cruz-Aguayo, Y. (2015). One laptop per child at home: Short-term impacts from a randomized experiment in Peru. *American Economic Journal: Applied Economics*, 7(2), 53-80.
- Botha, A., Herselman, M., Ford, M. (2014). Gamification beyond Badges. In: IST-Africa Conference Proceedings, 2014 (pp. 1-10). IEEE.
- Botha, A., & Hersleman, M. (2015a). Supporting Teacher Professional Development to use Tablets in Resource Constrained Schools : A Case Study of Cofimvaba schools , Eastern Cape Province , South Africa, *13*(January 2016), 20–27.

Botha, A., & Herselman, M. (2015b). ICTs in rural education : Let the game begin. *Proceedings of the 2015 Annual Symposium on Computing for Development*, 105–113. https://doi.org/10.1145/2830629.2830646

- Borenstein, M., Hedges, L. V., Higgins, J., & Rothstein, H. R. (2009). *References* (pp. 409-414). John Wiley & Sons, Ltd.
- Booyse, J. J., Le Roux, C. S., Seroto, J., & Wolhuter, C. C. (2013). *A history of schooling in South Africa: Method and context*. Van Schaik Publishers: Pretoria, South Africa.
- Brown, T. H. (2005). Towards a Model for m-Learning in Africa. *International Journal on E-Learning*, *4*(3), 299–315. Accessed 20 October 2017: <u>http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=psyc4&NEWS=N&AN=2007-01608-002</u>
- Brown, A.N., & Skelly, H. (2017). ICT4D: Does the evidence match the hype. Poster presented at the Global Evidence Summit, 13–16 September 2017, Cape Town.
- Buckler, A. (2015). Quality teaching in rural Sub-Saharan Africa: Different perspectives, values and capabilities. *International Journal of Educational Development*, 40, 126–133. https://doi.org/10.1016/j.ijedudev.2014.10.008
- Buckler, A. (2016). Teachers' Professional Capabilities and the Pursuit of Quality in Sub-Saharan African Education Systems: Demonstrating and Debating a Method of Capability Selection and Analysis. *Journal of Human Development and Capabilities*, 17(2), 161–177.
- Buckner, E., & Kim, P. (2014). Integrating technology and pedagogy for inquiry-based learning: The Stanford Mobile Inquiry-based Learning Environment (SMILE). *Prospects*, *44*(1), 99-118.
- Cameron, D. B., Mishra, A., & Brown, A. N. (2016). The growth of impact evaluation for international development: how much have we learned? *Journal of Development Effectiveness*, *8*(1), 1-21.
- Castells, M., Himanen, P. (2014). Reconceptualizing Development in the Global Information Age, Oxford University Press: Oxford, UK.
- Carr-Hill, R., Ndalichako, J. (2005). Tanzania Education Sector Situation Analysis. Dares-Salaam. Economic and Social Research Foundation.
- Carr-Hill, R., Rolleston, C., Pherali, T., & Schendel, R. (2014). The Effects of School-Based Decision Making on Educational Outcomes in Low- and Middle-Income Contexts: A Systematic Review . *Campbell Systematic Reviews*, (July 2014). Accessed 20 October 2017: http://www.campbellcollaboration.org/lib/project/325/
- Chalmers, I., Hedges, L. V., & Cooper, H. (2002). A brief history of research

synthesis. Evaluation & the health professions, 25(1), 12-37.

- Chaudhury, N., Hammer, J., Kremer, M., Muralidharan, K., & Rogers, F. H. (2006). Missing in action: teacher and health worker absence in developing countries. *Journal of Economic Literature*, *20*(1), 91–116.
- Chen, X., Liu, C., Zhang, L., Shi, Y., & Rozelle, S. (2010). Does taking one step back get you two steps forward? Grade retention and school performance in poor areas in rural China. *International Journal of Educational Development*, *30*(6), 544-559.
- Cin, F. M., & Walker, M. (2013). Context and history: Using a capabilities-based social justice perspective to explore three generations of western Turkish female teachers' lives. *International Journal of Educational Development*, 33(4), 394-404.
- Cirera, Xavier, Filipe Lage, and Leo Sabetti. 2015. "ICT Adoption, Innovation and Productivity in Africa." Background paper for the World Development Report 2016, World Bank, Washington, DC
- Cristia, J., Ibarrarán, P., Cueto, S., Santiago, A., & Severín, E. (2012). Technology and child development: Evidence from the one laptop per child program.
- Critical Appraisal Skills Programme (CASP). (2006). 10 questions to help you make sense of qualitative research. CASP. Accessed 20 October 2017: http://www.cfkr.dk/images/file/CASP%20instrumentet.pdf
- CSIR. (2012). *Support for rural education. ICT4RED Baseline survey*. Council for Scientific and Industrial Research: Pretoria.
- De Buck, E., Van Remoortel, H., Hannes, K., Govender, T., Naidoo, S., Avau, B., ... & Vandekerckhove, P. (2017). Approaches to promote handwashing and sanitation behaviour change in low-and middle income countries: a mixed method systematic review. *Campbell Systematic Reviews*, *7*, 1-447.
- Department of Basic Education (2015). Speech delivered by Basic Education Angie Motshekga at the release of NSC Examination Results on Monday 05 January 2015. Accessed 31 July 2015:
- http://www.education.gov.za/Newsroom/Speeches/tabid/298/ctl/Details/mid/1749/ItemID/306 4/Default.asp
- De Melo, G., Machado, A., & Miranda, A. (2014). The impact of a one laptop per child program on learning: Evidence from Uruguay.
- Deeks, J. J., Altman, D. G., & Bradburn, M. J. (2008). Statistical methods for examining heterogeneity and combining results from several studies in meta-analysis. *Systematic Reviews in Health Care: Meta-Analysis in Context, Second Edition*, 285-312.
- Deneulin, S. (Ed.). (2009). An introduction to the human development and capability approach: Freedom and agency. Routledge.
- Deriquito, M. and Domingo, Z. 2012. Mobile Learning for Teachers in Asia: Exploring the Potential of Mobile Technologies to Support Teachers and Improve Practice. Paris, UNESCO. Accessed 20 October 2017:

http://unesdoc.unesco.org/images/0021/002162/216284E.pdf

- DFID (1999) Sustainable Livelihood Guidance Sheet Section 2 . London: Department for International Development.
- Dixon-Woods, M., Agarwal, S., Jones, D., Young, B., & Sutton, A. (2005). Synthesising qualitative and quantitative evidence: a review of possible methods. *Journal of health services research & policy*, *10*(1), 45-53.
- Donner, J. (2008). Research approaches to mobile use in the developing world: A review of the literature. *Information Society*, 24(3), 140–159.
- Donner, J. (2016). After access: Inclusion, development, and a more mobile Internet. MIT

Press: Boston, US.

- Duflo, E., Dupas, P., & Kremer, M. (2015). School governance, teacher incentives, and pupil–teacher ratios: Experimental evidence from Kenyan primary schools. *Journal of Public Economics*, *123*, 92-110.
- Duncombe, R. A. (2014). Understanding the Impact of Mobile Phones on Livelihoods in Developing Countries. *Development Policy Review*, 32(5), 567-588.
- Education for all (EFA). (2014) Global Monitoring Report 2014: Teaching and learning: achieving quality for all. UNESCO: Paris, France.
- Education for all (EFA). (2015) Global Monitoring Report 2015: Education for all 2000–2015: Achievements and Challenges. UNESCO: Paris, France.
- Elliott, A., & Urry, J. (2010). Mobile Lives (International Library of Sociology) 1st Edition. Routledge: London, UK.
- Engeström, Y. (2001). Expansive learning at work: Toward an activity theoretical reconceptualization. *Journal of education and work*, *14*(1), 133-156.
- Evans, D. (2017). Even if technology improves literacy, is it worth the cost? Blog post. Accessed 20 October 2017: <u>https://davidevans.blog/2017/06/29/even-if-technology-improves-literacy-is-it-worth-the-cost/</u>
- Evans, D. K., & Yuan, F. (2017). The economic returns to interventions that increase learning. Background paper, World Bank, Washington, DC.
- Ford, M., Botra, A., & Rerselman, M. (2014). ICT4RED 12—Component implementation framework: A conceptual framework for integrating mobile technology into resourceconstrained rural schools. In *IST-Africa Conference Proceedings, 2014* (pp. 1-11). IEEE.
- Fraser, N., 2008. Scales of Justice: Reimagining Political Space in a Globalizing World. Polity Press, Cambridge
- Freire, P. (1972). Pedagogy of the oppressed. Penguin Books.: London, UK.
- Frohberg, D., Göth, C., & Schwabe, G. (2009). Mobile Learning projects a critical analysis of the state of the art: Original article. *Journal of Computer Assisted Learning*, 25(4), 307–331. h
- Fukuda-Parr, S. (2003). The human development paradigm: operationalizing Sen's ideas on capabilities. *Feminist economics*, 9(2-3), 301-317.
- Galperin, H., & Viecens, F. (2017). Connected for Development? Theory and evidence about the impact of Internet technologies on poverty alleviation. *Development Policy Review*, *35*(3), 315–336.
- Garnham, N. (1997). Amartya Sen's "capabilities" approach to the evaluation of welfare: Its application to communications. *Javnost-The Public*, *4*(4), 25-34.
- Gasper, D. (2007). What is the capability approach?: Its core, rationale, partners and dangers. *The Journal of Socio-Economics*, *36*(3), 335-359.
- Geldof, M., Grimshaw, D. J., Kleine, D., & Unwin, T. (2011). What are the key lessons of ICT4D partnerships for poverty reduction? *Systematic Review Report*. DFID: London, UK. http://r4d.dfid.gov.uk/pdf/outputs/systematicreviews/dfid_ict_sr_final_report_r5.pdf
- Gigler, B.-S. (2011). "Informational Capabilities" The missing link for the impact of ICT on development. *Working Paper Series No. 1,*
- Gigler, B.S. (2015). Development as Freedom in a Digital Age Experiences of the Rural Poor in Bolivia. World Bank, Washington, D.C.
- GIZ. (2015). Landscape Review Mobile Education for Numeracy Evidence from interventions in low-income countries. GIZ: Bonn/Berlin, Germany.
- Gough, D., Oliver, S., & Thomas, J. (2012). Introducing systematic reviews. SAGE: London,

UK.

- Gough, D., Oliver, S., & Thomas, J. (2017). Introducing systematic reviews. 2nd edition. SAGE: London, UK.
- Gough, D., & Thomas, J. (2016). Systematic reviews of research in education: aims, myths and multiple methods. *Review of Education*, *4*(1), 84-102.
- Greenhalgh, T., & Peacock, R. (2005). Effectiveness and efficiency of search methods in systematic reviews of complex evidence: audit of primary sources. *Bmj*, *331*(7524), 1064-1065.
- Grimus, M., & Ebner, M. (2013). M-Learning in Sub Saharan Africa Context What is it about? *Word Conference on Educatinal Multimedia, Hypermedia and Telecommunications*.
- Grimus, M., Ebner, M., & Holzinger, A. (2012). Mobile learning as a chance to enhance education in developing countries On the example of Ghana. *CEUR Workshop Proceedings*.
- GSMA (2010). mLearning: A Platform for Educational Opportunities at the Base of the Pyramid. London: GSMA.
- GSMA (2011). The Mobile Proposition for Education. London: GSMA.
- GSMA (2012a). Transforming learning through mEducation. London: GSMA.
- GSMA (2012b) Shaping the Future Realising the potential of informal learning through mobile. London: GSMA.
- GSMA. (2017). The mobile economy. London: GSMA.
- Gulati, S. (2008). Technology-enhanced learning in developing nations: A review. *The International Review of Research in Open and Distributed Learning*, 9(1).
- Haenssgen, M. J., & Ariana, P. (2017). The place of technology in the Capability Approach. *Oxford Development Studies*, *818*(May), 1–15.
- Haßler, B., Hennessy, S., & Lubasi, B. (2011). Changing classroom practice using a schoolbased professional development approach to introducing digital resources in Zambia.
- Haßler, B., Major, L., & Hennessy, S. (2016). Tablet use in schools: A critical review of the evidence for learning outcomes. *Journal of Computer Assisted Learning*, *32*(2), 139–156.
- Hedges, L. (2007). Effect sizes in cluster-randomised designs. *Journal of Educational and Behavioural Statistics*, 32: 341-370.
- Hedges, L. V., Olkin, I. (1985). *Statistical methods for meta-analysis*. Orlando, FL.: Academic Press.
- Hedges, L., Pigott, T. (2004). The power of statistical tests for moderators in meta-analysis. *Psychological Methods,* 9(4): 426-445.
- Heeks, R. (2008). ICT4D 2.0: The next phase of applying ICT for international development. *Computer*, *41*(6), 26–31.
- Heeks, R., & Molla, A. (2009). Compendium on impact assessment of ICT-for-development projects.
- Heeks, R. (2010). Do information and communication technologies (ICTs) contribute to development?. *Journal of International Development*, 22(5), 625-640.
- Heeks, R., & Stanforth, C. (2015). Technological change in developing countries: opening the black box of process using actor–network theory. *Development Studies Research*, *2*(1), 33-50.
- Hennessy, S., Haßler, B., & Hofmann, R. (2015). Challenges and opportunities for teacher professional development in interactive use of technology in African schools. *Technology, Pedagogy and Education*, *24*(5), 1–28.

- Hennessy, S., Haßler, B., & Hofmann, R. (2016). Pedagogic change by Zambian primary school teachers participating in the OER4Schools professional development programme for one year. *Research Papers in Education*, *31*(4), 399-427.
- Hennessy, S., Onguko, B., Harrison, D., Kiforo, E., Namalefe, S., & Naseem, A. (2010).
 Developing the use of information and communication technology to enhance teaching and learning in East African schools: Review of the literature. *Department for International Development*, (May). Accessed 20 October 2017
 http://www.educ.cam.ac.uk/centres/cce/publications/CCE_Report1_LitRevJune0210
- Herselman, M., & Botha, A. (2015). Designing and implementing an Information Communication Technology for Rural Edcuation Development (ICT4RED) in a reservuce constraint environment: Nciba school district, Eastern Cape South Africa.
 CSIR: Pretoria, South Africa.
- Higgins, J. P. T., Thompson, S. G. (2002). Quantifying heterogeneity in a meta-analysis. *Statistics in Medicine*, 21: 1539-1558.
- Higgins J. P. T., Thompson, S. G., Deeks, J. J., Altman, D.G. (2003). Measuring inconsistency in meta-analyses. *British Medical Journal*, 327: 557-560.
- Higgins, J. P., & Green, S. (Eds.). (2011). Cochrane handbook for systematic reviews of *interventions* (Vol. 4). John Wiley & Sons.
- infoDev [Crandall, A., Otieno, A., Mutuku, L., Colaço, J., Grosskurth, J., & Otieno, P.] (2012). Mobile Phone Usage at the Kenyan Base of the Pyramid. *InfoDev*, (November), 1–59.
- Ibrahim, S. S. (2006). From individual to collective capabilities: The capability approach as a conceptual framework for self-help. Journal of Human Development, 7(3), 397–416.
- Ibrahim, S. S. (2009). Self-help: a catalyst for human agency and collective capabilities. In: Martinetti, E. C. (Ed.). *Debating global society: reach and limits of the capability approach*. Fondazione Giangiacomo Feltrinelli: Milan, Italy.
- Iqbal, S., & Qureshi, I. A. (2012). M-learning adoption: A perspective from a developing country. *The International Review of Research in Open and Distributed Learning*, *13*(3), 147-164.
- Isaacs, S. (2012). Turning on mobile learning in Africa and the Middle East: Illustrative initiatives and policy implications. UNSESO: Paris, France.
- International Telecommunications Union (ITU). (2015). m-Powering Development Initiative. ITU: Geneva, Switzerland.
- International Telecommunications Union (ITU). (2017). ICT facts and figures. ITU: Geneva, Switzerland.
- James, J. (2010). New technology in developing countries: A critique of the one-laptop-perchild program. *Social Science Computer Review*, *28*(3), 381-390.
- Jensen, R., 2007. The digital provide: Information (technology), market performance, and welfare in the South Indian fisheries sector. *The quarterly journal of economics*, *122*(3), pp.879-924.
- Kabeer, N. (2010). Women's empowerment, development interventions and the management of information flows. *IDS Bulletin*, *41*(6), 105-113.
- Kaleebu, N., Gee, A., Jones, R., Watson, A.H.A. (2013). SMS story impact assessment report. VSO, Papua New Guinea.
- Kay, A.C. (1972) A personal computer for children of all ages. http://www.mprove.de/diplom/gui/Kay72a.pdf
- Kearney, M., Schuck, S., Burden, K., & Aubusson, P. (2012). Viewing mobile learning from a pedagogical perspective. *Research in Learning Technology*, *20*(1), 1–17.
- Kim, P., Miranda, T., & Olaciregui, C. (2008). Pocket School: Exploring mobile technology as
a sustainable literacy education option for underserved indigenous children in Latin America. *International Journal of Educational Development*, 28(4), 435–445.

- Kinuthia, W., & Marshall, S. (eds) (2013). *On the move: Mobile learning for development*, Charlotte, US: Information Age Publishing.
- Kleine, D., & Unwin, T. (2009). Technological revolution, evolution and new dependencies: What's new about ICT4D? *Third World Quarterly*, *30*(5), 1045–1067.

Kleine, D. (2010). ICT4WHAT?—Using the choice framework to operationalise the capability approach to development. *Journal of International Development*, 22(5), 674-692.

- Kleine, D. (2013). Technologies of choice: ICTs, Development, and the Capabilities Approach. MIT Press: Cambridge, MA.
- Koole, M., McQuilkin, J. L., & Ally, M. (2010). Mobile learning in distance education: Utility or futility. *International Journal of E-Learning & Distance Education*, 24(2).
- Kremer, M., Brannen, C., & Glennerster, R. (2013). The Challenge of Education and Learning in the Developing World. *Science*, *340*(6130), 297–300.
- Kullman, K., & Lee, N. (2012). Liberation from/liberation within: Examining one laptop per child with Amartya Sen and Bruno Latour. In *The Capability Approach, Technology and Design* (pp. 39-55). Springer Netherlands.
- Kukulska-Hulme, A., Evans, D. and Traxler, J. (2005), Landscape study in wireless and mobile learning in the post-16 sector. JISC Technology and Standards Watch. [Online]. Accessed 20 October 2017

http://www.jisc.ac.uk/whatwedo/programmes/elearning_innovation/eli_outcomes.aspx

- Kukulska-Hulme, A. (2009). Will mobile learning change language learning? *ReCALL*, 21(2), 157.
- Langer, L., Winters, N., & Stewart, R. (2014). Mobile Learning for Development: Ready to Randomise?. In *International Conference on Mobile and Contextual Learning* (pp. 156-167). Springer International Publishing.
- Langer, L., & Stewart, R. (2014). What have we learned from the application of systematic review methodology in international development?—a thematic overview. *Journal of Development Effectiveness*, 6(3), 236-248.
- Langer, L., Stewart, R., Erasmus, Y., & de Wet, T. (2015). Walking the last mile on the long road to evidence-informed development: building capacity to use research evidence. *Journal of Development Effectiveness*, 7(4), 462-470.
- Latour, B. (1993). We have never been modern . Cambridge: Harvard University Press.
- Laurillard, D. (2002). Rethinking University Education: A conversational framework for the effective use of learning technologies. *RoutledgeFalmer, London*.
- Leach, J. (2008). Do new information and communications technologies have a role to play in the achievement of education for all? *British Educational Research Journal*, 34(6), 783–805.
- Lee, Jong-Wha, and Hanol Lee. 2016. "Human Capital in the Long Run." *Journal of Development Economics*.

http://www.sciencedirect.com/science/article/pii/S0304387816300396

Lester, R.T., Ritvo, P., Mills, E.J., Kariri, A., Karanja, S., Chung, M.H., Jack, W., Habyarimana, J., Sadatsafavi, M., Najafzadeh, M. and Marra, C.A., 2010. Effects of a mobile phone short message service on antiretroviral treatment adherence in Kenya (WelTel Kenya1): a randomised trial. *The Lancet*, 376(9755), pp.1838-1845.

Lewin, Keith M. "Access to education in sub-Saharan Africa: Patterns, problems and possibilities." *Comparative Education* 45.2 (2009): 151-174.

Lipsey, M.W. & Wilson, D. B. (2001). Practical meta-analysis(Vol. 49). Thousand Oaks, CA:

Sage publications.

- Martinetti, E. C. (Ed.). (2009). *Debating global society: reach and limits of the capability approach*. Fondazione Giangiacomo Feltrinelli: Milan, Italy.
- Martinetti, E. C., & Roche, J.M. (2009). Operationalisation of the Capability Approach, from theory to practice: a review of techniques and empirical applications. In: Martinetti, E. C. (Ed.). *Debating global society: reach and limits of the capability approach*. Fondazione Giangiacomo Feltrinelli: Milan, Italy.
- Matthee, M., & Liebenberg, J. (2007). Mathematics on the Move: Supporting Mathematics Learners through Mobile Technology in South Africa. *The 6th International Conference on Mobile Learning.*, 156–161.
- McCowan, T. (2011). Human rights, capabilities and the normative basis of "Education for All." *School Field*, 9(3), 283–298.
- McCowan, T. and Unterhalter, E. eds., 2015. *Education and international development: An introduction*. Bloomsbury Publishing.
- McEwan, P. J. (2015). Improving Learning in Primary Schools of Developing Countries: A Meta-Analysis of Randomized Experiments. *Review of Educational Research*, *85*(3), 353–394.
- mEducation Alliance. (2017). Objectives. Accessed 20 October 2017: http://www.meducationalliance.org/
- Merriam, S. B. (1988). *Case study research in education: A qualitative approach*. Jossey Bass.
- Mill, J. S. (1861). Utilitarianism (Vol. 7). Bobbs-Merrill.
- Misra, P. K. (2012). Each-one-teach-one mobile networks: An innovative strategy for knowledge access in Asian countries. *Educational Media International*, 49(2), 109–122.
- Mitra S., Rana, V. (2001). 'Children and the Internet: Experiments with minimally invasive education in India', British Journal of Educational Technology 32 (2), 221-232.
- Mitra, S. (2003). Minimally invasive education: a progress report on the "hole-in-the-wall" experiments. *British journal of educational technology*, *34*(3), 367-371.
- Motala, S., Dieltiens, V., Sayed, Y. (Eds) (2012). Finding place and keeping pace. Exploring Meaningful and Equitable Learning in South African Schools. HSRC Press: Pretoria, South Africa.
- Mo, D., Swinnen, J., Zhang, L., Yi, H., & Qu, Q. (2012). Can one laptop per child reduce the digital divide and educational gap? Evidence from a randomized experiment in migrant schools in Beijing.
- Nkula, K., Krauss, K.E.M. (2014) The integration of ICTs in marginalized schools in South Africa: Considerations for understanding the perceptions of in-service teachers and the role of training. In: Steyn, J., Van Greunen, D. (Eds). (2014). ICTs for inclusive communities in developing societies. Proceedings of the 8th International Development Informatics Association Conference, held in Port Elizabeth, South Africa.
- Noblit, G. W., & Hare, R. D. (1988). *Meta-ethnography: Synthesizing qualitative studies* (Vol. 11). sage.
- Nsibande, R. N., & Modiba, M. M. (2012). 'I just do as expected'. Teachers' implementation of Continuous Assessment and challenges to curriculum literacy. *Research Papers in Education*, *27*(5), 629-645.
- Nussbaum, M.C. (2011). Creating capabilities. The human development approach. Harvard University Press: Cambridge, MA.
- Oakley, A. (2000). Experiments in knowing: Gender and method in the social sciences.
- Ogata, H., & Yano, Y. (2004). Context-aware support for computer-supported ubiquitous

learning. In Wireless and Mobile Technologies in Education, 2004. Proceedings. The 2nd IEEE International Workshop on (pp. 27-34). IEEE.

- Oliver, M. (2005). The Problem with Affordance. *E-Learning and Digital Media*, 2(4), 402–413.
- Oliver, S., Dickson, K., Newman, M. (2012). Getting started with a review. In: Gough, D., Oliver, S., & Thomas, J. Introducing systematic reviews. SAGE: London, UK.
- Oliver, S., & Sutcliffe, K. (2012). Describing and analysing studies. In: Gough, D., Oliver, S., & Thomas, J. Introducing systematic reviews. SAGE: London, UK.
- Oliver, S. (2014). Advantages of concurrent preparation and reporting of systematic reviews of quantitative and qualitative evidence. *Journal of the Royal Society of Medicine*, *108*(3), 108-111.
- One Laptop Per Child (OLPC). (2017). About. Accessed 20 Ocotber 2017: http://one.laptop.org/about/mission
- Oosterlaken, I. (2015). Technology and human development. Routledge: London, UK.
- Oosterlaken, I. van den Hoven, J. (eds) (2012). The capability approach, technology and design: Taking stock and looking ahead. Springer: London, UK.
- Oya, C., Schaefer, F., Skalidou, D., McCosker, C., & Langer, L. (2017). Effects of certification schemes for agricultural production on socio-economic outcomes in lowand middle-income countries: a systematic review.
- O'Mara-Eves, A., & Thomas, J. (2016). Ongoing developments in meta-analytic and quantitative synthesis methods: Broadening the types of research questions that can be addressed. *Review of Education*, *4*(1), 5-27.
- Pachler, N., & Seipold, J. (2009). Mobile Learning Cultures Across Education, Work and Leisure. *Learning Education & Social Research*
- Pachler, N., Bachmair, B., & Cook, J. (2010). Mobile learning: structures, agency, practices. Springer Science & Business Media.
- Patton, M. (2002). Qualitative evaluation and research methods (2nd ed.). Newbury Park, CA: Sage.
- Pawson, R. (2006). Evidence-based policy: a realist perspective. Sage.
- Petit, T., & Santos, G. L. (2014, November). Mobile Learning: An Ergonomic Alternative for Long-Awaited Educational Changes?. In *International Conference on Mobile and Contextual Learning* (pp. 1-14). Springer, Cham.
- Petrosino, A. (2012). Interventions in Developing Nations for Improving Primary and Secondary School Enrollment of Children: A Systematic Review. https://doi.org/10.4073/csr.2012.19
- Petticrew, M., & Roberts, H. (2006). Systematic reviews in the social sciences: A practical guide. John Wiley & Sons.
- Phiri, A. C., Foko, T., & Mahwai, N. (2014). Evaluation of a pilot project on information and communication technology for rural education development : A Cofimvaba case study on the educational use of tablets. *International Journal of Education and Development Using Information and Communication Technology*, *10*(4), 60–79.
- Pimmer, C., Pachler, N., Attwell, G. (2012). Towards Work-Based Mobile Learning: What We Can Learn from the Fields of. *Refining Current Practices in Mobile and Blended Learning: New Applications: New Applications*, 210.
- Piper, B., Jepkemei, E., Kwayumba, D., & Kibukho, K. (2015). Kenya's ICT policy in practice: The effectiveness of tablets and e-readers in improving student outcomes. In *FIRE: Forum for International Research in Education* (Vol. 2, No. 1, p. 2).
- Pluye, P., Robert, E., Cargo, M., Bartlett, G., O'cathain, A., Griffiths, F., ... & Rousseau, M.

C. (2011). Proposal: A mixed methods appraisal tool for systematic mixed studies reviews. *Montréal: McGill University*, 1-8.

- Pluye, P., & Hong, Q. N. (2014). Combining the power of stories and the power of numbers: mixed methods research and mixed studies reviews. *Annual review of public health*, 35.
- Porter, G., Hampshire, K., Abane, A., Munthali, A., Robson, E., Mashiri, M., & Tanle, A. (2012). Youth, mobility and mobile phones in Africa: findings from a three-country study. *Information Technology for Development*, *18*(2), 145-162.
- Poveda, S., & Roberts, T. (2017). Critical agency and development: applying Freire and Sen to ICT4D in Zambia and Brazil. *Information Technology for Development*, *0*(0), 1–19.
- Powell, C., 2014. *Rethinking marginality in South Africa: Mobile phones and the concept of belonging in Langa Township*. Langaa Rpcig.
- Pritchett, Lant, Rukmini Banerji, and Charles Kenny. 2013. "Schooling Is Not Education! Using Assessment to Change the Politics of Non-Learning." *Center for Global Development Report*. Accessed 20 October 2017:

http://www.cgdev.org/sites/default/files/schooling-is-not-learning-WEB.pdf.

- Qiang, C. Z. W., Rossotto, C. M., & Kimura, K. (2009). Economic Impacts of Broadband.ll In Information and Communications for Development 2009: Extending Reach and Increasing Impact, 35–50. *Washington, DC. World Bank*.
- Quinn, C. (2000) mLearning: Mobile, Wireless, in your Pocket Learning. LineZine, http://www.linezine.com/2.1/features/cqmmwiyp.htm
- Rawls, J. (1972). A theory of justice. London. Cambridge Univerity Press: Cambridge, UK.
- Roberts, T. (2016). Critical-agency in ICT4D: A case study of Zambian women's use of participatory video technology to challenge gender inequality (Unpublished PhD). Royal Holloway University of London, London.
- Robeyns, I. (2005). The Capability Approach: a theoretical survey. *Journal of Human Development*, 6(1), 93–117.
- Robeyns, I. (2011). The capability approach. Handbook of economics and ethics, 39.
- Robinson, J.P. and Winthrop, R., 2016. Millions Learning: Scaling up Quality Education in Developing Countries. *Center for Universal Education at The Brookings Institution*.
- Roschelle, J. (2003). Unlocking the learning value of wireless mobile devices. *Journal of computer assisted learning*, *19*(3), 260-272.
- Rostow, W.W. (1960) The Stages of Economic Growth: A Non-Communist Manifesto. Cambridge University Press, Cambridge, MA.
- Saldaña, J. (2013). The coding manual for qualitative researchers. Sage: London, UK.
- Sandelowski, M., Voils, C. I., Leeman, J., & Crandell, J. L. (2011). Mapping the mixed methods–mixed research synthesis terrain. *Journal of mixed methods research*, 6(4), 317-331.
- Selwyn, N. (2016). *Education and technology: Key issues and debates*. Bloomsbury Publishing.
- Sen, A. (Ed.). (1980). Equality of what? Cambridge: Cambridge University Press.
- Sen, A. (1985a). Commodities and capabilities. Amsterdam: North-Holland.
- Sen, A. (1985b). Well-being, agency and freedom. The Journal of Philosophy, LXXXII(4), 169-221.
- Sen, A. (1987). The standard of living. Cambridge University Press: Cambridge, UK.
- Sen, A. (1992). Inequality reexamined. Oxford University Press: Oxford, UK.
- Sen, A. (1999). Development as freedom. Oxford University Press: Oxford, UK.
- Sen, A. (2009). The idea of justice. Harvard University Press: Boston: US.

- Sen, A. (2010). The mobile and the world. *Information Technologies & International Development*, 6(SE): 1-3.
- Sharmin, S., Faith, B., Prieto Martín, P., & Ramalingam, B. (2017). *The Contribution of Digital Technologies to Service Delivery: An Evidence Review* (No. IDS Evidence Report; 221). IDS.
- Sharples, M., Taylor, J., & Vavoula, G. (2007). Towards a Theory of Mobile Learning. *Proceedings of mLearn*, *1*(1), 1–9.
- Shohel, M. M. C., & Kirkwood, A. (2012). Using technology for enhancing teaching and learning in Bangladesh: challenges and consequences. *Learning, Media and Technology*, *37*(4), 414-428.
- Smith, M. L., Spence, R., & Rashid, A. T. (2011). Mobile Phones and Expanding Human Capabilities. *Information Technologies & International Development*, 7(3), 77–88.
- Snilstveit, B. (2012). Systematic reviews: from 'bare bones' reviews to policy relevance. *Journal of development effectiveness*, *4*(3), 388-408.
- Snilstveit, B., Gallagher, E., Phillips, D., Vojtkova, M., Eyers, J., Stevenson, J., ... Davies, P. (2015). Interventions for improving learning outcomes and access to education in lowand middle-income countries: A systematic review, (January).
- Snilstveit, B., Vojtkova, M., Bhavsar, A., Stevenson, J., & Gaarder, M. (2016). Evidence & Gap Maps: A tool for promoting evidence informed policy and strategic research agendas. *Journal of clinical epidemiology*, *79*, 120-129.

Snilstveit, B. & Langer, L. (2016) Mixed-methods systematic reviews in international development. Panel at *What Works Global Summit*, 10 November 2016.

- SolarClassRoom. (2017). About. Accessed 20 October 2017: <u>http://www.aleutia.com/solarclassroom/</u>
- Srinivasan, J., & Burrell, J. (2013, December). Revisiting the fishers of Kerala, India. In Proceedings of the Sixth International Conference on Information and Communication Technologies and Development: Full Papers-Volume 1 (pp. 56-66). ACM.

Stake, R. E. (1995). The art of case study research. Thousand Oaks, CA: Sage

Stark S, Torrance H. (2005). Case study. In Somekh, B., & Lewin, C. (Eds.).

(2005). Research methods in the social sciences. Sage: London, UK.

- Statistics SA. (2015). *Methodological report on rebasing of national poverty lines and development of pilot provincial poverty lines*. Technical Report No. 03-10-11. Statistics South Africa: Pretoria.
- Sterne, J., Higgins, J., & Reeves, B. C. (2013, September). Extending the risk of bias tool to allow for assessment of non-randomised studies, cluster-randomised trials and cross-over trials: a Cochrane methods innovation fund project (Workshop).
- Stewart, F., & Deneulin, S. (2002). Amartya Sen's contribution to development thinking. *Studies in Comparative International Development*, *37*(2), 61-70.
- Stewart, F. (2005). Groups and capabilities. *Journal of human development*, 6(2), 185-204.
- Stewart, R. (2014). Changing the world one systematic review at a time: A new development methodology for making a difference. *Development Southern Africa*, *31*(4), 581-590.
- Tamim, R.M., Borokhovski, E., Pickup, D., Bernard, R.M. (2015). Large-Scale, Government-Supported Educational Tablet Initiatives, Common Wealth of Learning: Burnaby, Canada.
- Tao, S. (2013a). Rethinking teacher quality: Using the capability approach and critical realism to provide causal explanations for teacher practice in Tanzania (Doctoral dissertation, Institute of Education, University of London).

- Tao, S. (2013b). Why are teachers absent? Utilising the Capability Approach and Critical Realism to explain teacher performance in Tanzania. *International Journal of Educational Development*, *33*(1), 2–14.
- Thomas, J., & Harden, A. (2008). Methods for the thematic synthesis of qualitative research in systematic reviews. *BMC medical research methodology*, *8*(1), 45.
- Thomas, J., Sutcliffe, K., Harden, A., Oakley, A., Oliver, S., Rees, R., ... & Kavanagh, J. (2003). Children and healthy eating: a systematic review of barriers and facilitators.
- Thomas, J., Newman, M., & Oliver, S. (2013). Rapid evidence assessments of research to inform social policy: taking stock and moving forward. *Evidence & Policy: A Journal of Research, Debate and Practice*, 9(1), 5-27.
- Tikly, L., & Barrett, A. M. (2011). Social justice, capabilities and the quality of education in low income countries. *International Journal of Educational Development*, *31*(1), 3–14.
- Toyama, K. (2015). *Geek heresy: Rescuing social change from the cult of technology*. PublicAffairs.
- Traxler, J., Leach, J. (2006). A Model of Mobile Learning in Developing Countries, (Jisc 2005), 94–100.
- Traxler, J., & Kukulska-Hume, A. (2005). Mobile Learning in Developing Countries knowledge series. *Commonwealth of Learning*, 1–6.
- Traxler J (2013a) 'mLearning solutions for international development–rethinking the thinking'. *Digital Culture & Education*, 5(2), 74-85.
- Traxler J (2013b) 'Mobile learning in Africa: the elephants in the room', In Kinuthia W & Marshall S (eds) *On the move: Mobile learning for development*, Charlotte, US: Information Age Publishing.
- Traxler (2013c) Tsinakos, A. and Ally, M., 2013. Global mobile learning implementation and trends.
- Traxler, J., & Vosloo, S. (2014). Introduction: The prospects for mobile learning. *Prospects*, *44*(1), 13–28.
- Trucano, M. (2013). Surveying ICT use in education. Blog post. Accessed 20 October 2017: <u>https://blogs.worldbank.org/edutech/category/tags/survey</u>
- Trucano, M. (2015). 'Tablets in education'. Accessed 31 July 2015: http://blogs.worldbank.org/edutech/tablets-education
- Tsinakos, A., & Ally, M. (2013). *Global Mobile Learning Implementations and Trends*. Accessed 20 October 2017:
 - http://oasis.col.org/bitstream/handle/11599/558/pub_Mobile_Learning_web.pdf
- Turkle, S. (2016). *Reclaiming conversation: The power of talk in a digital age*. Penguin: London, UK.
- United Nations Millennium Goals. 2016. Goal 2: Achieve Universal Primary Education". Accessed 20 October 2017: http://www.un.org/millenniumgoals/education.shtml.
- UNESCO Institute for Statistics. (2009). *Guide to Measuring Information and Communication Technologies (ICT) in Education*. <u>https://doi.org/10.15220/978-92-9189-078-1-en</u>
- UNESCO. (2010). Mobiles, learning, and development. Accessed 14 May 2014: <u>http://www.unesco.org/new/en/unesco/themes/icts/m4ed/mobile-learning-resources/unescomobilelearningseries/</u>
- UNESCO. (2012a) Turning on mobile learning. Global themes. Paris: UNESCO.
- UNESCO. (2012b) Mobile learning for teachers. Global themes. Paris: UNESCO.
- UNESCO. (2012c) The future of mobile leanring. Implications for policy-makers and planners. Paris: UNESCO.
- UNESCO. (2012d) UNESCO policy guiidlnes for mobile learning. Paris: UNESCO.

- UNESCO (2014) Reading in the mobile era: A study of mobile reading in developing countries. Paris: UNESCO.
- UNESCO/UNICEF. (2017) Stories of innovation. Accessed 20 October 2017: <u>http://unicefstories.org/2014/07/17/unicef-brings-together-bright-minds-in-uganda-to-tackle-tough-problems/</u>
- Unterhalter, E. (2005). 'Global inequality, capabilities, social justice: The millennium development goal for gender equality in education', International Journal of Educational Development 25(2), 111-122.
- Unterhalter, E. (2015). Analysing inequalities in education. In: McCowan, T. and Unterhalter, E. (eds.). *Education and international development: An introduction*. Bloomsbury Publishing.
- Unterhalter, E., Poole, N., Wells, K., & Winters, N. (2015). Education, Information & Knowledge. Accessed 20 Ocotber 2017:

http://www.oapen.org/download?type=document&docid=560103#page=68

- Unterhalter E. (2017). Balancing pessimism of the intellect and optimism of the will: Some reflections on the capability approach, gender, empowerment, and education. In: D. A. Clark, M. Biggeri and A. Frediani (eds). The Capability Approach, Empowerment and Participation: Concepts, Methods and Applications. Forthcoming.
- Unwin, T. (Ed.). (2009). ICT4D: Information and communication technology for development. Cambridge University Press, Cambridge, MA.
- Unwin, T. (2015). Evolution and prospects for the use of mobile technologies to improve education access and learning outcomes, Background paper prepared for the Education for All Global Monitoring Report 2015. Paris: UNESCO.
- USAID. (2010). Mobiles for education for development Accessed 14 May 2014: <u>http://www.worldbank.org/en/topic/edutech/brief/mobile-phones-in-education</u>
- USAID. (2014). The 2013 Mobiles for Education (mEducation) Alliance International Symposium: Commit Fair for Project Scaling. USAID: Washington DC, US.
- van der Knaap, L. M., Leeuw, F. L., Bogaerts, S., & Nijssen, L. T. (2008). Combining Campbell Standards and the Realist Evaluation Approach The Best of Two Worlds?. *American journal of evaluation*, 29(1), 48-57.
- Valters, C. (2014). Theories of Change in International Development: Communication, Learning, or Accountability? *Justice and Security Research Programme Paper*, <u>https://doi.org/10.1177/109821400302400102</u>
- Valk, J., Rashid, A. T., & Elder, L. (2010). Using Mobile Phones to Improve Educational Outcomes : An Analysis of Evidence from Asia, *11*(1), 13–14.
- Vaughan, R. P., & Walker, M. (2012). Capabilities, values and education policy. *Journal of Human Development and Capabilities*, *13*(3), 495-512.
- Vavoula, G., & Sharples, M. (2009). Meeting the challenges in evaluating mobile learning: A 3-level evaluation framework. *International Journal of Mobile and Blended Learning*, *1*(2), 54–75.
- Velghe, F. (2014). "I wanna go in the phone": literacy acquisition, informal learning processes, "voice" and mobile phone appropriation in a South African township. *Ethnography and Education*, *9*(1), 111–126.
- Venkatapuram, S. (2013). *Health justice: An argument from the capabilities approach*. John Wiley & Sons.
- Vogel, I. (2012). Review of the use of 'Theory of Change'in international development. UK: Department for International Development (DFID).
- Voils, C. I., Sandelowski, M., Barroso, J., & Hasselblad, V. (2008). Making sense of

qualitative and quantitative findings in mixed research synthesis studies. *Field methods*, 20(1), 3-25.

- Vosloo, S., Deumert, A., & Town, C. (2008). m4Lit : a teen m-novel project in South Africa, 25–26. Accessed 20 October 2017: <u>http://www.gsma.com/mobilefordevelopment/wp-content/uploads/2012/04/m4litateenmnovelprojectinsouthafrica.pdf</u>
- Vrasidas, C., Zembylas, M., & Glass, G. V. (Eds.). (2009). *ICT for education, development, and social justice*. IAP.
- Vu Henry, J., Winters, N. Lakati, A., Oliver, M., Geniets, A., Mbae, S. and Wanjiru, H. (2016) Enhancing the supervision of community health workers with mobile messaging: Implementing WhatsApp in low resource settings in Kenya, *Global Health: Science & Practice*. DOI: <u>https://dx.doi.org/10.9745%2FGHSP-D-15-00386</u>
- Wagner, D. (2014). Mobiles for Reading: A Landscape Research Review. Accessed 20 October 2017:

http://www.meducationalliance.org/sites/default/files/usaid wagner report finalforweb 14jun25 1.pdf

- Wagner, D. A., Castillo, N. M., Murphy, K. M., Crofton, M., & Zahra, F. T. (2014). Mobiles for literacy in developing countries: An effectiveness framework. *Prospects*, 44(1), 119– 132.
- Wagner, H. R. (1962). Displacement of scope: A problem of the relationship between smallscale and large-scale sociological theories. *American Journal of Sociology*, 69(6), 571-584.
- Wali, E., Winters, N., & Oliver, M. (2008). Maintaining, changing and crossing contexts: an activity theoretic reinterpretation of mobile learning. *Alt-J*, *16*(1), 41–57.
- Walker, M. (2005). Amartya Sen's capability approach and education. *Educational Action Research*, *13*(1), 103–110. <u>https://doi.org/10.1080/09650790500200279</u>
- Walker, M. (2006). Towards a capability-based theory of social justice for education policymaking. *Journal of Education Policy*, *21*(2), 163-185.
- Walker, M., Unterhalter, E. (Eds.) (2007). Amartya Sen's Capability Approach and Social Justice in Education. PalgraveMacmillan, New York, US.
- Watkins, C., & Mortimore, P. (1999). Pedagogy: What do we know. Understanding pedagogy and its impact on learning, 1-19.
- Westbrook, J., Durrani, N., Brown, R., Orr, D., Pryor, J., Boddy, J., & Salvi, F. (2013). Pedagogy, Curriculum, Teaching Practices and Teacher Education in Developing Countries. Accessed 20 October 2017 <u>http://eppi.ioe.ac.uk/cms/LinkClick.aspx?fileticket=XHbLclohFq0=&tabid=3433&mid=68</u> 28
- White, H. (2009). Theory-based impact evaluation: principles and practice. *Journal of development effectiveness*, *1*(3), 271-284.
- White, H., & Waddington, H. (2012). Why do we care about evidence synthesis? An introduction to the special issue on systematic reviews. *Journal of Development Effectiveness*, *4*(3), 351-358.
- Winters, N. (2007). What is mobile learning. In Sharples, M. *Big issues in mobile learning*, pp 7-11.
- Winters, N. (2010). Mobile learning for development. Accessed 24 August 2014: http://www.ml4d.org
- Winters, N. (2013). Mobile learning in the majority world: A critique of the GSMA position. In: Price, S., Jewitt, C., Brown, B. (eds.) The SAGE Handbook of Digital Technology Research. Sage, London.

Winters, N., Africa, S., Bocock, P., President, U. S., & Clinton, B. (2013). How teachers in Africa are failed by mobile learning. *mCHW: A Mobile Learning Intervention for Community Health Workers*. Accessed 20 October 2017: http://www.mchw.org/?p=426

- Winters, N (2015) "E-Learning for development: the role of the teacher", In: AP Hwa, R Mansell (eds.) Encyclopedia of Digital Communication and Society. New York: Wiley.
- Winters, N., Oliver, M., & Langer, L. (2017). Can mobile health training meet the challenge of 'measuring better'?. *Comparative Education*, *53*(1), 115-131.
- Wolff, J., & De-Shalit, A. (2007). *Disadvantage*. Oxford University Press on Demand.
- World Bank. (2012). Maximizing Mobile. World Bank, Washington D.C.
- World Bank. (2016). World Development Report 2016: Digital Dividends, Washington DC: World Bank.
- World Bank. (2017). Country Classification. Accessed 20 October 2017: https://blogs.worldbank.org/opendata/new-country-classifications-2016
- Worldreader. (2017). About. Accessed 20 October 2017: https://www.worldreader.org/
- World Economic Forum. (2012). Accelerating the Adoption of mLearning : A Call for Collective and Collaborative Action, 1–16.
- World Education Forum (2016) Education 2030: Incheon Declaration and Framework for Action Towards Inclusive and Equitable Quality Education and Lifelong Learning for All', UNESCO. Accessed 13 October 2016:

http://unesdoc.unesco.org/images/0023/002338/233813M.pdf

- Yin, R. K. (2003). Case study research: Design and methods (3rd ed.). Thousand Oaks,CA: Sage.
- Yin, R.K. (2014) Case study research design and methods. Thousand Oaks, CA: SAGE
- Zelezny-Green, R. (2014). She called, she Googled, she knew: girls' secondary education, interrupted school attendance, and educational use of mobile phones in Nairobi. *Gender & Development*, *22*(1), 63-74.
- Zelezny-Green, R. (2017). The Role of Girls' Mobile Phone Use to Increase Access to Educational Content After School: A Capabilities-Based Evaluation in Nairobi. Unpublished PhD thesis. <u>Royal Holloway</u>, <u>University of London</u>
- Zheng, Y. (2007). Exploring the Value of the Capability Approach for e-Development. 9th International Conference on Social Implications of Computers in Developing Countries, (May 2007), 14. <u>https://doi.org/10.1002/itdj</u>
- Zheng, Y. (2009). Different spaces for e-development: What can we learn from the capability approach?. *Information Technology for Development*, *15*(2), 66-82.
- Zheng, Y., & Stahl, B. C. (2011). Technology, capabilities and critical perspectives: What can critical theory contribute to Sen's capability approach? *Ethics and Information Technology*, *13*(2), 69–80.
- Zheng, Y., & Stahl, B. C. (2012). Evaluating emerging ICTs: A critical capability approach of technology. In *The capability approach, technology and design* (pp. 57-76). Springer Netherlands.
- Zurita, G., & Nussbaum, M. (2004). A constructivist mobile learning environment supported by a wireless handheld network. *Journal of Computer Assisted Learning*, *20*(4), 235-243.

Appendices

The below appendices are organised by chapter.

Appendix 4.1: Search strategy and report

Summary:

- An exhaustive search strategy, that is a search strategy that aims to identify and collect all relevant studies, was applied. This neglected the alternative options of purposeful or iterative searching.
- This exhaustive search was deliberately over-inclusive, and as broad as possible in design, to create a diversity of search sources, which overlap each other to some extent. A large number of search hits entailing a considerable number of duplicates was the result.
- The rationale behind this exhaustive and over-inclusive search strategy is three fold:
 - (1) The literature on mobile learning for development is limited, and not structurally archived
 - ightarrow need for more diverse search sources and flexible search terms.
 - (2) Bias and limitation of being a single reviewer
 → need to control for bias such as personal condition (fatigue), or context (previous inclusions / exclusions), or administrative mistakes (not saving the pdf) by exposure to the same study multiple times under different conditions.
 - (3) Methodological limitations of alternative search strategies \rightarrow risk of missing important studies.

Search strategy

The search strategy developed a master search string applied in a range of academic and Grey literature sources.

Search terms: A master search string is presented below. A master string serves as an example of how the search will be designed for the most sophisticated search engines. In most cases however search terms will have to be adapted, or key word searches will apply.

Terms:

(1) Intervention terms:

("mobile learn* ") OR mlearning OR m-learning OR ("mobile educat*") OR meducation OR m-education OR ("mobile teach*") OR "portable interactive learning technology" OR ("mobile phon*" AND learn*) OR ("mobile phon*" AND educat*) OR ("mobile phon*" AND teach*) OR ("smart phon*" AND learn*) OR ("smart phon* AND educat*) OR ("smart phon*" AND teach*) OR ("mobile devic*" AND learn*) OR ("mobile devic*" AND educat*) OR ("mobile devic*"

AND teach*) OR ("mobile technolog*" AND learn*) OR ("mobile technolog*" AND educat*) OR ("mobile technolog*" AND teach*) OR "mobile and contextual learning"

AND

(2) Outcomes terms

development OR "international development" OR "social development" OR poverty OR inequality OR "social change" OR livelihood OR empower* OR educat* OR learn*OR school* OR read* OR literate OR literac* OR numerac*

AND

(3) Country terms

A full list of countries based on the World Bank's list LMICs was used combined with generic terms for LMICs

Africa OR Asia OR Caribbean OR "West Indies" OR "South America" OR "Latin America" OR "Central America" or Afghanistan OR Albania OR Algeria OR Angola OR Antigua OR Barbuda OR Argentina OR Armenia OR Aruba OR Azerbaijan OR Bahrain OR Bangladesh OR Barbados OR Benin OR Byelarus OR Byelorussian OR Belarus OR Belorussian OR Belorussia OR Belize OR Bhutan OR Bolivia OR Bosnia OR Herzegovina OR Hercegovina OR Botswana OR Brasil OR Brazil OR Bulgaria OR "Burkina Faso" OR "Burkina Faso" OR "Upper Volta" OR Burundi OR Urundi OR Cambodia OR "Khmer Republic" OR Kampuchea OR Cameroon OR Cameroons OR Cameron OR Camerons OR "Cape Verde" OR "Central African Republic" OR CAR OR Chad OR Chile OR China OR Colombia OR Comoros OR "Comoro Islands" OR Comores OR Mayotte OR Congo OR Zaire OR "Costa Rica" OR "Cote d'Ivoire" OR "Ivory Coast" OR Croatia OR Cuba OR Cyprus OR Czechoslovakia OR "Czech Republic" OR Slovakia OR "Slovak Republic" OR Djibouti OR "French Somaliland" OR Dominica OR "Dominican Republic" OR "East Timor" OR "East Timur" OR "Timor Leste" OR Ecuador OR Egypt OR "United Arab Republic" OR "El Salvador" OR Eritrea OR Estonia OR Ethiopia OR Fiji OR Gabon OR "Gabonese Republic" OR Gambia OR Gaza OR Georgia OR Ghana OR "Gold Coast" OR Greece OR Grenada OR Guatemala OR Guinea OR Guam OR Guiana OR Guyana OR Haiti OR Honduras OR Hungary OR India OR Maldives OR Indonesia OR Iran OR Iraq OR Jamaica OR Jordan OR Kazakhstan OR Kazakh OR Kenya OR Kiribati OR Korea OR Kosovo OR Kyrgyzstan OR Kirghizia OR "Kyrgyz Republic" OR Kirghiz OR Kirgizstan OR "Lao PDR" OR Laos OR Latvia OR Lebanon OR Lesotho OR Basutoland OR Liberia OR Libva OR Lithuania OR Macedonia OR Madagascar OR "Malagasy Republic" OR Malaysia OR Malaya OR Malay OR Sabah OR Sarawak OR Malawi OR Nyasaland OR Mali OR Malta OR "Marshall Islands" OR Mauritania OR Mauritius OR "Agalega Islands" OR Mexico OR Micronesia OR "Middle East" OR Moldova OR Moldovia OR Mongolia OR Montenegro OR Morocco OR Mozambique OR Mocambique OR Myanmar OR Myanma OR Burma OR Namibia OR Nepal OR "Netherlands Antilles" OR "New Caledonia" OR Nicaragua OR Niger OR Nigeria OR "Northern Mariana Islands" OR Oman OR Muscat OR Pakistan OR Palau OR Palestine OR Panama OR Paraguay OR Peru OR Philippines OR Philipines OR Philipines OR Phillippines OR Poland OR Portugal OR "Puerto Rico" OR Romania OR Rumania OR Roumania OR Russia OR Russian OR Rwanda OR Ruanda OR "Saint Kitts" OR "St Kitts" OR Nevis OR "Saint Lucia" OR "St Lucia" OR "Saint Vincent" OR "St Vincent" OR Grenadines OR Samoa OR "Samoan Islands" OR "Navigator Island" OR "Navigator Islands" OR "Sao Tome" OR "Saudi Arabia" OR Senegal OR Serbia OR Montenegro OR Seychelles OR "Sierra Leone" OR Slovenia OR "Sri Lanka" OR Ceylon OR "Solomon Islands" OR

Somalia OR Sudan OR Suriname OR Surinam OR Swaziland OR Syria OR Tajikistan OR Tadzhikistan OR Tadjikistan OR Tadzhik OR Tanzania OR Thailand OR Togo OR "Togolese Republic" OR Tonga OR Trinidad OR Tobago OR Tunisia OR Turkey OR Turkmenistan OR Turkmen OR Uganda OR Ukraine OR Uruguay OR "USSR" OR "Soviet Union" OR "Union of Soviet Socialist Republics" OR Uzbekistan OR Uzbek OR Vanuatu OR "New Hebrides" OR Venezuela OR Vietnam OR "Viet Nam" OR "West Bank" OR Yemen OR Yugoslavia OR Zambia OR Zimbabwe OR "developing country" OR "developing countries" OR "developing nation" OR "developing nations" OR "developing world" OR "less-developed countr*" OR "less developed countr*" OR "less-developed world" OR "less-developed world" OR "lesser-developed countr*" OR "lesser developed countr*" OR "lesser-developed nation" OR "lesser developed nation*" OR "lesser developed world" OR "lesser-developed world" OR "under-developed countr*" OR "under developed countr*" OR "under-developed nation*" OR "under developed nation*" OR "under-developed world" OR "underdeveloped world" OR "under developed world" OR "underdeveloped countr*" OR "under-developed countr*" OR "Under developed countr*" OR "under developed nation*" OR "under-developed nation*" OR "underdeveloped nation*" OR "lower middle income countr*" OR "lower middleincome countr*" OR "lower middle income nation*" OR "lower middle-income nation*" OR "upper middle-income countr*" OR "upper middle income countr*" OR "upper middle-income nation*" OR "upper middle income nation*" OR "low-income countr*" OR "low income countr*" OR "low-income nation*" OR "low income nation*" OR "lower income countr*" OR "lower-income countr*" OR "lower income nation*" OR "lower-income nation*" OR "Lowand Middle- Income countr*" OR "Low and Middle Income Countr*" OR "underserved country" OR "underserved countries" OR "underserved nation" OR "underserved nations" OR "underserved world" OR "under served country" OR "under served countries" OR "under served nation" OR "under served nations" OR "under served world" OR "deprived country" OR "deprived countries" OR "deprived nation" OR "deprived nations" OR "deprived world" OR "poor country" OR "poor countries" OR "poor nation" OR "poor nations" OR "poor world" OR "poorer country" OR "poorer countries" OR "poorer nation" OR "poorer nations" OR "poorer world" OR "developing economy" OR "developing economies" OR "less developed economy" OR "less developed economies" OR "lesser developed economy" OR "lesser developed economies" OR "under developed economy" OR "under developed economies" OR "underdeveloped economy" OR "underdeveloped economies" OR "middle income economy" OR "middle income economies" OR "low income economy" OR "low income economies" OR "lower income economy" OR "lower income economies" OR Imic OR Imics OR "third world" OR "lami country" OR "lami countries" OR "transitional country" OR "transitional countries" LMIC OR LMICs OR LIC OR LICs OR LMICs OR LMIC OR UMICs OR UMIC) OR ("khmer" AND "republic") OR ("cape" AND "verde") OR ("central" AND "african" AND "republic")

(4) NOT medication

Combining these terms, the *default search* can be simplified as:

(1) AND (2) AND (3) NOT (4)

As stated above, the default search requires a sophisticated search engine to be applicable. This is usually only the case for major academic databases. Since those present only a minority of search sources (see below), search terms had to be adapted individually. In such instances, only key terms from (1) were applied ensuring the search strategy remains as broad as possible and no studies are overlooked. If the identified numbers appeared too large, a term of (2) was connected with the 'AND' boolean operator to reduce search hits. In general, a flexible approach to search terms was applied, carefully weighting their value

against possible predefined key word categories in databases, and redefining the terms for most sensible usage in each individual database.

Search Sources: Studies connecting the ideas of mobile technologies, education, and international development will come from a variety of sources, including the academic as well as grey literature, and most likely published and unpublished reports. I therefore compiled a list of search sources comprising academic literature and Grey literature sources.

Academic sources

(1) Databases:

- EBSCO all databases including among other (Academic Search Complete, ERIC, EconLit, Education FullTexts, Teacher Reference Centre);
- EdITLib, (Digital Library for Education and Information technology),
- Ingenta Connect;
- JSTOR;
- SabiNet;
- SAGE Journals Online;
- Science Direct;
- Taylor&Francis Online;
- ISI Web of Science;
- ELDIS;
- Institute of education library guides, OER guide and DOAJ guide and International Education Guide

(2) Journals:

Most of the journals below are included in at least one of the database above. However, to double check that the databases search parameters did not miss relevant publications, individual key word searches will be conducted for the period of 2004-2014 in the journals listed below. Individual key word searches address the redundancy of applying the search terms rigidly; for example, it does make little sense to search for the term 'mobile learning' in a journal dedicated to the subject of mobile learning. Journals listed with an asterisk (*) are assumed to be key journals relevant to the type of publications this review seeks to identify, and will therefore be hand searched exhaustively on title and abstract for the same period.

Development related:

- Economic Development and Cultural Change
- Journal of Development Economics
- *Journal of Development Effectiveness
- *Journal of International Development
- Journal of Sustainable Development
- *World Development
- World Bank Research Observer

Education related:

- British Journal of Education
- Educational Researcher
- European Journal of Open and Distance Learning
- *International Journal of Educational Development
- Journal of the Learning Sciences

ICT related:

- African Journal of Information and Communication
- Computers and Education
- Computers and Human behavior
- Electronic Journal of Information Systems in Developing Countries
- Information, Communication & Society
- Information Technologies and International Development
- Information Technology for Development
- International Journal of Education and Development Using Information and Communication Technology
- International Journal of Mobile and Blended Learning
- International Journal of Education and Development Using Information and Communication Technology
- International Journal of Information and Communication Technologies for Human Development
- International Journal of Mobile learning and Organisation
- International Journal on Advances in ICT for Emerging Regions
- Journal of Information Technology in Social Change
- Journal of Computer Assisted Learning
- Learning Media and Technology
- South African Journal of Information Management
- International Journal of Learning Technology.
- International Journal of Interactive Mobile Technologies (iJIM)
- International Journal of ICT Research and Development in Africa
- Journal of Health Informatics in Developing Countries
- International Journal of Information Systems and Social Change
- International Review of Research in Open and Distance Learning

(3) Thesis/Dissertation search:

- ProQuest
- Ethos

Grey literature

- (1) Google, Google Scholar
- (2) Websites of key organisations

AusAID; Bill&Melinda Gates Foundation; Centre for Development Informatics; Consortium for Research on Educational Access, Transitions and Equity (CREATE);DFID; GSMA; MIT Poverty Action Lab; ICT4D blog (Heeks); IICD; IDRC; Innovations for Poverty Center; International Association for Mobile Learning; London Knowledge Lab; Network for Policy Research, Review and Advice on Education and Training (NORRAG); Network of Networks Impact Evaluation Initiative (NONIE); OECD; ODI; Research Consortium on Educational Outcomes and Poverty (RECOUP); UNESDOC; USAID (mEducation alliance); World Bank (especially their impact evaluation section); World Bank (EduTech blog); Word Bank (infoDev); 3ie.

(3) Conference proceedings:

elearning Africa; IADIS International Conference Mobile Learning (2005-2011), mEducation symposium (2011, 2012, 2013), mLearn,(2002-2011; UNESCO mobile learning weeks (2011; 2013;2014), Wireless, Mobile and Ubiquitous Technologies in Education (WMUTE); INTEL education summit (2011; 2012) ICTD (2006-2013).

Snowballing

(1) Expert contacts:

Mohamed Ally; John Cook; Jonathan Donner; Richard Heeks; Agnes Kukulska-Hulme; Dorothea Kleine; Tim Unwin; Mike Sharples; John Traxler; Michael Trucano; Niall Winters; Maggie Verster; Steve Vosloo

(2) Twitter search:

m4d
ict4d
ict4e
mlearning

♯ Edchat

(3) Reference/Citation searches of key publications:

- Ally (2009) Mobile Learning: Transforming the Delivery of Education and Training
- **Donner** (2008) Research Approaches to Mobile Use in the Developing World: A Review of the Literature.
- Frohberg (2009) Mobile learning projects a critical analysis of the state of the art
- Kinuthia & Marshal (eds) (2013) On the move: Mobile Learning for Development
- Kukulska-Hulme & Traxler (2005) M-learning: A Handbook for Educators and Trainers.
- Naismith et al (2004) Literature Review in Mobile Technologies and Learning.
- **Pachler** et al (2010) Mobile learning: structures, agency, practices.
- Sharples et al (2007) A Theory of Learning for the Mobile Age.
- Traxler (2006) Mobile learning in developing countries
- Traxler (2009) Learning in a mobile age
- Worldreader (2012) iRead Ghana study. Final evaluation report.

- **IICD** Mobile Learning reports
- **GSMA** Mobile Learning Reports
- UNESCO Mobile Learning Reports turning on 8
- WB Maximizing Mobile

(4) Previous research synthesis (SRs, MAs) and their included studies:

| Key words | Author | Type of review | |
|--|--------------------|--------------------|--|
| | | | |
| Combined | · | | |
| All interventions on enrollment; | 3ie (2013) | 3ie: Working Paper | |
| attendance; drop-outs; learning outcomes | | | |
| All interventions on enrollment | Petrosino (2012) | 3ie: SR | |
| All interventions that improve learning at | McEwans (2013) | WB: SR & meta- | |
| primary school level | | analysis | |
| Synthesis: Evidence-based education in | JPAL (2013) | JPAL: Brief | |
| Africa | | Synthesis | |
| Synthesis: Overview of RCTs in | Kremer (2009) | WB: Overview | |
| developing-country education | Kremer (2013) | ScienceDirect | |
| Effect of school resources on educational | Glewwe (2011) | WB: SR | |
| outcomes | | | |
| | | | |
| Teachers | | | |
| Effectiveness of rise in teacher salaries | Carr (2011) | EPPI-centre: SR | |
| Interventions to increase teacher | Guerrero (2012) | EPPI-centre: SR | |
| attendance | | | |
| Strategies to improve performance of | Orr (2013) | EPPI-centre: SR | |
| under-trained? | | | |
| Effectiveness of pedagogic approaches | Westbrook (2013) | EPPI: literature | |
| | | review | |
| | | | |
| Health | | | |
| Female toilets on girl's enrollment and | Birdthishle (2011) | SR: (LSHTM/IOE) | |
| attendance | | | |
| School feeding on nutritional outcomes | Kirstjansson | EPPI-centre: SR | |
| | (2006) | | |
| Deworming and attendance & nutrition | TRobinson | EPPI-centre: SR | |
| | (2012) | | |
| Effects of assessment programs on | Best (2013) | EPPI-centre: SR | |
| educational policy | | | |
| | | | |
| ICTs | 1 | 1 | |
| ICT4D partnerships | Geldorf (2011) | DFID: SR | |
| M4D approaches | Donner (2007) | Literature review | |
| ICTs and other technology (no mobile) | Waxman (2003) | MA | |
| Lit review of evaluations of OLPC | Nguhro (2010) | Literature review | |

| Mobile computer-supported collaborative | Hsu (2013) | Review |
|--|------------------|-----------------|
| learning. A review of experimental | | |
| research | | |
| | | |
| Single interventions | | |
| CCT/UCTs on enrollment; attendance; | Baird (2013) | Campbell: SR |
| test scores | | |
| Effectiveness of school vouchers | Morgan (2013) | EPPI-centre: SR |
| Effectiveness of School-based monitoring | BarreraOsorio | WB: review |
| | (2012) | |
| Effectiveness of school-based decision | Carr-Hill (2016) | Campbell: SR |
| making | | |
| | | |
| Other | | |
| Effectiveness of higher education | Clifford (2013) | EPPI-centre: SR |
| Education and economic growth | Hawkes (2012) | EPPI-centre: SR |
| Access to education for people with | Bakhshi (2013) | EPPI-centre: SR |
| disability | | |
| | | |

(5) Website search of mobile learning for development projects:

List of projects:

Road to reading; World Reader; Project ABC; Paje Nieta; EIA; Shaquodon; Yoza; MILLE; Programa Nacional de Alfabetización; Cambridge to Africa (deaf); PSU Movile learning assessment through mobile; Educational bridges; roots of mobile learning ; SMILE; Jokko intiative; Fire & Gold; eSchool 360; Impact network; mUbuntu; eLife; Tangerine; Total Reading Approach for Children; Cocoa Link; Global Literacy Professional Development Network; Ustad Mobile; FATIH project turkey; LISTA; 1001 across radio; AlfabeTIC; Bangladesh Virtual Interactive Classrooms; BridgeIT; Bridges to the future; Broad Class; Bunyad Mobile; Dr Math; eEGRA; FunDza; Interactive Radio Instruction; International Children's Digital Library; Women Mobile Literacy Afghanistan; SIRIP Somali Interactive Radio Instruction Program; Shellbooks; Mobiliteracy Uganda; MoMath; Senmobil; Nokia life tools; Earth Institute; Sesam; Teacher mate; Pratham; PIEQ; PAJEF; OLPC; Twaweza; Tessa; IFADEM

Overview of search results

| Academic hits combined | 24,010 | Individual sources: 37/37 | | | |
|---|--|---|--|--|--|
| Grey literature | 7,992 | Individual sources: 71/71 | | | |
| ACADEMIC LITERA | TURE | | | | |
| Total studies | 32,002 | Individual sources: 108/108 | | | |
| Databases | Hits (of interest) | Search terms / comments | | | |
| keywoitor,aayostract | 681 (4) | "mobile learning" | | | |
| | 23 (1) | "mobile education" | | | |
| | 0 | "mobile teaching" | | | |
| | 296 (3) | 'developing countries' key word | | | |
| EBCSO including: Academic Search Complete, ERIC, EconLit, | 1144 (71) | (1) AND (2) ['LIC' and 'MIC' terms not working] | | | |
| Education FullTexts, Teacher Reference Centre | 3 9 (2) 5 (1) 65 5302 (82) | "mobile learning for development" "mobile education for development" "mobile learning" AND ("international development" OR poverty) mobile learning AND "develop* countr*" OR "develop* nation" OR "low income countr* " OR poor "mobile learning" OR "mobile education" | | | |
| EdITLib | 1234 (36) | "mobile learning" | | | |
| (Digital Library for Education and Information technology) | 109 (8) | "mobile education" | | | |
| ELDIS | 244 | Mobile OR phone OR tablet OR device in | | | |
| | | 'ICT for development category' | | | |
| IOE library guides | 0 | "mobile learning" OR "mobile education" | | | |
| International Education Guide | 0 | "mobile learning" OR "mobile education" | | | |

| OER Guide | 0 | |
|---|--------------------------------------|---|
| JSTOR | 3393 (22) | ("mobile learn* ") OR mlearning OR m- learning OR ("mobile educat*") OR meducation OR m-education OR ("mobile phone" AND learning) OR ("mobile phone" AND education) OR ("mobile phone" AND teaching) |
| SabiNet | 2324 (12) | "mobile learning" |
| | 153 | "mobile education" |
| Science Direct* Saved search and weekly update via search alerts | 670 (12) 0 66 (4) | "mobile learning" "mobile learning for development" "mobile education" |
| | 828 (18) | (((mobile learn* ") OR (mobile AND learn*) OR mlearning OR ("mobile educat*") OR medication OR (mobile AND education) OR (mobile AND teach*") OR ("mobile technolog*" AND learn*)) and development OR "international development" OR "social development" OR poverty OR inequality OR "social change" OR ("develop* countr*") OR ("develop* nation") OR ("develop* world") OR ("low income countr*") OR ("middle income nation"))) |
| Taylor&Francis Saved search and weekly update via search alerts | 230 (12) 62 (4) 0 2245 (43) | "Mobile learning" "Mobile education" "mobile learning for development" in Abstract: ("mobile learn* ") OR mlearning OR m-learning OR ("mobile educat*") OR meducation OR m-education OR ("mobile teach*") OR "portable interactive learning technology" OR ("mobile phon*" AND learn*) OR ("mobile phon*" AND teach*) OR ("smart phon*" AND educat*) OR ("smart phon*" AND teach*) OR |

| | | ("mobile devic*" AND learn*) OR ("mobile devic*" AND educat*) OR ("mobile devic*" AND teach*) OR ("mobile technolog*" AND learn*) OR ("mobile technolog*" AND educat*) OR ("mobile technolog*" AND teach*) OR (mobile and contextual learning" |
|--|----------------------|--|
| ISI web of science Saved search and weekly update via search alerts | 1272 (18) | "mobile learn*" OR "mobile educat*" |
| Database hits (screened on title, key word and abstract for relevant) | 21,014 | Not controlled for duplicated |
| | | |
| Journals | HITS (OT INTEREST) | Search terms / comments |
| Educational Technology | 417 (9) | Mobile |
| Educational researcher | 212 (0) | Mobil* OR tablet* OR phone* OR device* |
| European Journal of Open, Distance, and e-Learning | 7 (1) | Keyword: Mobiles |
| Computers & Education | 178 (4) | "Mobile learn*" |
| Journal of Sustainable Development | 354 (0) | (mobile) OR (technologies) OR (ICT) OR (mobile learning) |
| Journal of Economic Development | 270 (0) | (mobile) |
| Economic Development and Cultural Change | 134 (1) | (mobile OR learning OR technology OR ICT) |
| South African Journal of Information Management | 16 (2) 7 10 | "mobile" "mobile learning" "mobile education" |
| International Journal of Interactive Mobile Technologies (iJIM) | 65 (3) 0 2 (1) | "development" "poverty" "social change" |
| Electronic Journal of | 0 | "Mobile learning" |

| Information Systems | 10 (1) | "Mobile education" |
|-----------------------|--------------------|--|
| in Developing | 18 | "Mobile phones" |
| Countries | 201 (2) | "Development" |
| | 18 | "Povertv" |
| | | |
| Computers and | 49 | "Mobile learning" OR "mobile education" |
| Human behavior | | |
| | | |
| Information, | 181 | Mobile AND learning |
| Communication & | | J J |
| Society | | |
| Journal of the | 0 | "Mobile learning" |
| learning sciences | 29 | Mobile AND learning |
| learning bolenoes | 5 | |
| | 26 | Phonos |
| | 20 | FIDLES |
| Learning Media and | 3 (2) | "Mobile learning" |
| Technology | 1 (1) | "Mobile education" |
| roomology | 162 (6) | Mobile AND learning |
| | 102 (0) | Nosile And Ioanning |
| World Bank | 37 (0) | Mobile |
| Research Observer | 2(0) | ICT |
| | 115 (0) | mobile AND technologies |
| | | |
| World Development | 468 (1) | (mobile learning) OR (mobile technologies) |
| | | OR (mobile education) |
| | | · · · · · · · · · · · · · · · · · · · |
| Journal hits | 2996 | Not controlled for duplicated |
| (screened on title, | | |
| key word and | | |
| abstract for relevant | | |
| | | |
| Journals (hand- | Hits (of interest) | Search terms / comments |
| searched) | | |
| African Journal of | Hand-searched | N/A |
| Information and | (0) | |
| Communication | | |
| Computers and | Hand-searched | N/A |
| Human behavior | 24 | |
| | | |
| Information | Hand-searched | N/A |
| Technologies and | 36 | |
| International | | |
| Development | | |
| | | |
| Information | Hand-searched | N/A |
| Technology for | 0 | |

| Development | | |
|--|------------------------------------|-----------------------------------|
| International Journal on Advances in ICT for Emerging Regions | Hand-searched (2) | N/A |
| International Journal of Educational Development | Hand-searched 2004-2014 (37) | N/A |
| International Journal of Education and Development Using Information and Communication Technology | Hand-searched (27) | N/A |
| Journal of Computer Assisted Learning (JCAL) | Hand-searched 2004-2014 (20) | N/A |
| Journal of Development Effectiveness | Hand-searched 2004-2014 (4) | N/A |
| Journal of Health Informatics in Developing Countries | Hand-searched (2) | N/A |
| Journal of International Development | Hand-searched 2004-2014 (7) | N/A |
| Academic hits combined | 24,010 | Individual sources: 37/37 |
| GREY LITERATURE | | |
| Online Sources | Hits (of interest) | Search terms / comments |
| AusAID | 67 2 | "education category" "mobile" |
| Bill & Melinda Gates Foundation | 142 | "mobile" |
| Centre for education Innovations | 56 | "educational technology category" |
| DIFD | 98 (32) | 'ICTs in education category' |

| | 10 (8) | ('mobile learning' OR 'mobile education') | | | | | |
|---|--------------------------|--|--|--|--|--|--|
| EduTech blog | Hand-searched | N/A | | | | | |
| | | Blog printed out as PDF, to be screened | | | | | |
| ICT4D blog | Hand-searched | N/A | | | | | |
| Innovation for poverty action | 116 (6) | (mobile OR learning OR technology OR ICT) | | | | | |
| | 78 | 'education category' | | | | | |
| JPAL | Hand-searched 466 (7) | N/A | | | | | |
| Google | 2000 | "mobile learning" AND development | | | | | |
| | | "mobile learning for development" (exact term) | | | | | |
| Google scholar | 1000 (14) | "mobile learning" AND development | | | | | |
| | 17 (4) | "mobile learning for development" (exact term) | | | | | |
| GSMA | Hand-searched 238 (6) | N/A | | | | | |
| IICD | 33 (12) | 'education category' | | | | | |
| IDRC (IDL) | 127 (2) | "mobile learning" OR "mobile education" OR "mobile for development" OR "mobile learning for development" | | | | | |
| International Association for Mobile Learning | 148 (1) | Mobile learning bibliography | | | | | |
| LKL | 312 23 | "Mobile learning" Publications category | | | | | |
| NORAG | 7 65 | 'mobile' 'ict' | | | | | |
| OECD | 16 191 | ("mobile learning" OR "mobile education") 'ICT AND Development' | | | | | |
| ODI | 6 4 16 (1) 190 | "mobile learning" "mobile education" "mobile technologies" "mobile phones" | | | | | |

| RECOUP | 4 | 'mobile' | | | | | |
|---|-----------------------|--------------------|----------------------------|----------------|--|--|--|
| Conference proceedings | | Hits (of interest) | Search terms / | | | | |
| | | | | comments | | | |
| eletersidg Alfrica | 12 | | Hand-searched | N/A | | | |
| | | | | | | | |
| MAXBIBIN protect national Co | nfeterence Mobile | 'educa | ti dancategan ohied | N/A | | | |
| Escaluation (2005-2011 | | | | | | | |
| Datab(23006-2013) | | | Hand-searched | N/A | | | |
| | 532 | 'educa | tion category' | | | | |
| WBEDU beliete activation sum | mûfo(2011; 2012 | 'ICT c | a tegndy 'searched | N/A | | | |
| | 838 | "mobil | e phones" | | | | |
| mEducation symposit | n\$ (2011, 2012, 20 | 13mobil | eHerbreterithed | N/A | | | |
| | | | | | | | |
| WEeiafo, (22002-2011 | Hand-searched | N/A | Hand-searched | N/A | | | |
| | 52 (4) | | | | | | |
| Binesco mobile lear | niag7w25ks (2011; | 'educa | tigandetsearched | N/A | | | |
| 2013;2014 | | | | | | | |
| Wateless, Mobile and | ปอิจิญี่ใช่งาน Techno | ogies | Hand-searched | N/A | | | |
| in Education (WMUTE | E);) | | | | | | |
| | | | | | | | |
| Citation Searches / R | eference Searches | | Hits (of interest) | Search terms / | | | |
| | | | | comments | | | |
| Ally (2009) Mobile Learning: Transforming the | | | 205 | N/A | | | |
| Delivery of Education and Training | | | | | | | |
| Donner (2008) Research Approaches to | | | 381 | N/A | | | |
| Mobile Use in the Developing World: A Review | | | | | | | |
| of the Literature. | | | | | | | |
| Frohberg (2009) Mobile learning projects – a | | | 162 | N/A | | | |

| critical analysis of the state of the art | | |
|--|-------------------|--------------------|
| Kinuthia & Marshal (eds) (2013) On the | 0 – Interesting, | N/A |
| move: Mobile Learning for Development | shows term is not | |
| | widely used | |
| Kukulska-Hulme & Traxler (2005) M- | 401 | N/A |
| learning: A Handbook for Educators and | | |
| Trainers. | | |
| Naismith et al (2004) Literature Review in | 629 | N/A |
| Mobile Technologies and Learning. | | |
| Pachler et al (2010) Mobile learning: | 177 | N/A |
| structures, agency, practices. | | |
| Sharples et al (2007) A Theory of Learning for | 407 | N/A |
| the Mobile Age. | | |
| Traxler (2006) Mobile learning in developing | 25 – interesting, | N/A |
| countries | shows term is not | |
| | widely used | |
| Traxler (2009) Learning in a mobile age | 129 | N/A |
| | | |
| Worldreader (2012) iRead Ghana study. Final | 0 – interesting, | N/A |
| evaluation report. | shows term is not | |
| | widely used | |
| Total | 2516 | |
| | | |
| Contact experts | Date contacted | Response |
| Mohamed Ally | See separate | |
| Marie-Lise Bourcier | sheet | |
| John Cook | | |
| Jonathan Donner | | |
| Richard Heeks | | |
| Agnes Kukulska-Hulme | | |
| Dorothea Kleine | | |
| Tim Unwin | | |
| Mike Sharples | | |
| John Traxler | | |
| Michael Trucano | | |
| Niall Winters | | |
| Maggie Verster | | |
| | 11:4- | First we ad |
| I WITTER | HIts | First used |
| ⊅ m4a | 43 | |
| i int4o | 10 | |
| i → ICl4e | 10 | |
| | 52 | |
| | | |
| Grev literature combined | 7 992 | Individual |
| | ., | sources: 71/71 |

Appendix 4.2: Mixed-methods critical appraisal tool

| Study type | Methodological appraisal criteria | Response | | |
|---|--|----------|--------|-------------------------------|
| | | Yes | No | Comment |
| Screening questions: assessing 'fatal flaws' (Dixon-Woods 2005) | Aggregative assessment: ✓ Study reports primary data and applied methods ✓ Study reports before and after data ¹ ✓ Study features an intervention and control group | | | |
| Aggregative 'fatal flaws' based on Stewart et al (2014) Configurative 'fatal flaws' based on Pawson (2003) TAPUS framework | <u>Configurative assessment:</u> ✓ Study reports primary data and applied methods ✓ Study states clear research questions and objectives ✓ Study states clear research design, which is appropriate to address the stated research question and objectives (<i>Purposivity</i>) ✓ The findings of the study are based on collected data, which justify the knowledge claims (<i>Accuracy</i>) Screening question based on abstract and/or superficial reading of full-text: | | | |
| | Further appraisal is not feasible or appropriate when the answer is 'No' to any | of the | e abov | /e screening questions! |
| Study type | Methodological appraisal criteria | Rose | oneo | |
| | | Yes | No | Comment / Confidence judgment |
| Qualitative e.g. (A) Ethnography (B) Phenomenology (C) Narrative (D) Grounded theory (E) Case study | I. RESEARCH IS DEFENSIBLE IN DESIGN (providing a research strategy that addresses the question) <u>Appraisal indicators:</u> ✓ Is the research design clearly specified and appropriate for aims and objectives of the research? <u>Consider whether</u> | | | |
| | <i>i.</i> there is a discussion of the rationale for the study design | | | |

| ii. the resear | <i>ii.</i> the research question is clear, and suited to qualitative inquiry | | | | | |
|--|---|---|---------------------------|--------------------|--|--|
| iii. there are o | iii. there are convincing arguments for different features of the study design | | | | | |
| iv. limitations evidence a | iv. limitations of the research design and implications for the research evidence are discussed | | | | | |
| Defensible | Defensible Arguable Critical Not defensible | | | | | |
| | | | - | 1 1 | | |
| II. RESEARC (following) | an adequate strategy | for selection of participa | = nts) | | | |
| Appraisal indicato | <u>rs:</u> | | | | | |
| Consider wheth | er | | | | | |
| i. there is a | description of study lo | cation and how/why it w | as chosen | | | |
| ii. the resear | cher has explained ho | w the participants were | selected | | | |
| iii. the selected | ed participants were a | ppropriate to collect rich | and relevant data | | | |
| iv. reasons a | e given why potential | participants chose not i | ake part in study | | | |
| Appropriate sample | Functional sample | Critical sample | Flawed sample | Worth to continue: | | |
| | | CONDUCT | | | | |
| (providing | a systematic and tra | nsparent account of the | research process) | | | |
| Appraisal indicato | <u>rs:</u> | | | | | |
| Consider whether | | | | | | |
| i. researche data was o interviews data?) | <i>i.</i> researchers provide a clear account/description of the process by which data was collected (e.g. for interview method, is there an indication of how interviews were conducted?/procedures for collection or recording of data?) | | | | | |
| ii. researche richness o | rs demonstrate that da f information (e.g. inte | ata collection targeted d rview/observation sche | epth, detail and dule) | | | |

| | iii. there is evid etc. have be | ence of how descrip en generated and u | ntive analytical categoi sed | ies, classes, labels, | | | | | |
|------------------------|---|---|---|-----------------------------|--------------------|---------|---------------------------------------|--|--|
| | iv. presentation frame used, | of data distinguishe and the interpretation | | | | | | | |
| | v. methods we explained ho | re modified during tl ow and why? | he study; and if so, ha | s the researcher | | | | | |
| | Rigorous conduct | Considerate conduct | Critical conduct | Flawed conduct | Worth to continue: | | | | |
| (p ge <u>A</u> j | IV. RESEARCH roviding well-found enerated) opraisal indicators | FINDINGS ARE CI ded and plausible ar | REDIBLE IN CLAIM/E rguments based on the | BASED ON DATA e evidence | | | | | |
| C | onsider whether <i>i. there is a cle</i> | ear description of the | lata | | | | | | |
| | ii. sufficient am findings/con | nount of data are pre clusions | | | | | | | |
| | iii. the research original sam cited data re repeated de | ners explain how the ple to feed into the a late; there is an ana scription; is there an | | | | | | | |
| | iv. there is a cle findings/con | ear and transparent clusion | | | | | | | |
| | v. there is evid etc. | ence (of attempts) to | | | | | | | |
| C | Credible claims | Arguable claims | Doubtful claims | Not credible | lf fina | lings n | not credible, can data still be used? | | |
| | V. REASEARC (describing t | H ATTENDS TO CO | DNTEXTS ticulars of the study) | | | | | | |

| | Appraisal indicators | | | | | | | |
|---|--|--|---|---------------------------|---------------|---|--|--|
| | | <u>.</u> | | | | | | |
| | Consider whether | | | | | | | |
| | i. there is an a they are ret | adequate descriptic ained and portraye | a sources and how | | | | | |
| | ii. participants | ' perspectives/obse | | | | | | |
| | iii. appropriate (how finding | consideration is given is given by a consideration is given by a constant of the constant of t | ven to how findings rela v or influence the contex | te to the contexts (t) | | | | |
| | iv. the study m (if yes, com | akes any claims (in ment on appropriat | nplicit or explicit) that in eness) | fer generalisation | | | | |
| | Context central | Context considered | Context | No context attention | | | | |
| | | | | | 1 | | | |
| | VI. RESEARCH (assessing wha <u>Appraisal indicators</u> Consider whether | t factors might have | | | | | | |
| | I. appropriate | consideration is giv | en to how findings relation of dot | e to researchers' | | | | |
| | ii. researchers have attempted to validate the credibility of findings (e.g. triangulation, respondent validation, more than one analyst) | | | | | | | |
| | iii. researchers study | explain their reacti | t occurred during the | | | | | |
| | iv. researchers impact on th (implicit/exp | discuss ideologica ne methodological c licit) | | | | | | |
| | Reflection | Consideration | Acknowledgement | Unreflective research | NB: Can o | override previous exclusion! | | |
| OVERALL DECISCO (study generates new | N – EXLUDE / INCL knowledge relevant | UDE to the review quest | ion and complies with m | ninimum criteria to ens | ure reliabili | ity and empirical grounding of knowledge) | | |
| | | | | | | | | |

Sources used in this section (in alphabetical order); Campbell et al (2003); CASP (2006); CRD (2009); Dixon-Woods et al (2004); Dixon-Woods et al (2006)^{cited} ^{in Gough 2012}; Greenhalgh & Brown (2014); Harden et al (2004)^{cited in SCIE & Gough 2012}; Harden et al (2009); Harden & Gough (2012); Mays & Pope (1995); Pluye et al (2011); Spencer et al 2006; Thomas et al (2003); SCIE (2010).

| Study type | Methodological appraisal criteria | | | | | | Response | | | | | |
|---------------------------------------|-----------------------------------|----------------------------|-----------------------------|---------------------------|--------------------|----|---------------------------------|--|--|--|--|--|
| | | | | | Yes | No | Comment / risk of bias judgment | | | | | |
| 2. Quantitative | I. Selection bi | as: | | | | | | | | | | |
| | (Are participants rec | ruited in a way the | at minimizes selection | bias?) | | | | | | | | |
| (non-randomised; | | , | | , | | | | | | | | |
| Randomised- | Appraisal indicators: | | | | | | | | | | | |
| Controlled) | F | | | | | | | | | | | |
| | Consider whether | | | | | | | | | | | |
| | i. there is a cle | ar description of <i>l</i> | now and why sample w | as chosen | | | | | | | | |
| Common non- | | • | | | | | | | | | | |
| random design | ii. there is adeq | uate sample size | to allow for representation | tive and/or statistically | | | | | | | | |
| include: | significant co | onclusions | | | | | | | | | | |
| | iii. participants ı | ecruited in the co | ntrol group were samp | led from the same | | | | | | | | |
| (A) Non-randomised | population as | s that of the treatn | nent | | | | | | | | | |
| СТ | iv. group allocat | tion process atter | npted to control for pote | ential risk of bias | | | | | | | | |
| (B) Cohort studies | - . | • | | | | | | | | | | |
| (C) Case-control | Low risk of bias | Risk of bias | High risk of bias | Critical risk of bias | Worth to continue: | | | | | | | |
| (D) Cross-sectional | | | | | | | | | | | | |
| analytical studies | | | | | • | r | | | | | | |
| | II. Bias due to base | eline confoundin | ig: | | | | | | | | | |
| Most common ways | (Is confounding pote | entially controllable | e in the context of this | study?) | | | | | | | | |
| of controlling for bias | | | | | | | | | | | | |
| due to baseline | Appraisal indicators: | | | | | | | | | | | |
| confounding: | | | | | | | | | | | | |
| U U | Consider whether | | | | | | | | | | | |
| Matching attempts | i. the treatme | nt and control gro | up are comparable at l | baseline | | | | | | | | |
| to emulate | ii matahing wa | a applied and in | and footured oufficie | at aritaria | | | | | | | | |
| randomization | II. matching wa | is applied, and in | | | | | | | | | | |
| • Propensity score | iii. the authors | conducted an app | propriate analysis that o | controlled for all | | | | | | | | |
| matching and | potential crit | ical confounding o | domains | | | | | | | | | |

| methods | iv. the authors | avoided to adjust i | | | | | | | | | |
|---|---|--|-------------------------|--------------------------|---|--|-----------------------------|--|--|--|--|
| sub-groups have | Low risk of bias | Risk of bias | High risk of bias | Critical risk of bias | Worth to continue: | | | | | | |
| been compared • Regression analysis where covariates are adjusted for | IF RANDOMISED C Bias due to ineffec (Is allocation of treat Appraisal indicators | CONTROL TRIAL, tive randomisation tment status truly r | | | | | | | | | |
| Randomised | Consider whether | ear description of t | he randomisation proc | iess | | | | | | | |
| designs: Randomised Control | | | | | | | | | | | |
| Trial (RCT) | ii. the unit of ra (pay special | andomisation and l attention to treatn | | | | | | | | | |
| | iii. eligibility crit | eria for study entr | | | | | | | | | |
| | <i>iv.</i> characteristics of baseline and endline sample are provided ¹ | | | | | | Preferable condition, see 1 | | | | |
| | Low risk of bias | Risk of bias | High risk of bias | Critical risk of bias | If critical risk of bias, treat as non-random study | | | | | | |
| | | | | | | | | | | | |
| | (Was the interventio | n implemented as | | | | | | | | | |
| | Appraisal indicators | <u>.</u> | | | | | | | | | |
| | Consider whether | | | | | | | | | | |
| | <i>i.</i> the critical co-interventions were balanced across intervention groups | | | | | | | | | | |
| | ii. treatment sw estimated ef | ritches were low en fect of intervention | nough to not threaten t | he validity of the | | | | | | | |
| | iii. implementati outcome esti | ion failure was mir imate | | | | | | | | | |

| iv. it is possible possible cros | that intervention wa sing-over)* | | *whilst challenging in terms of estimating impact, spill-overs might be an important finding in itself (eg teachers read to pupils/village/family members) | | |
|---|---|---|--|-------|--|
| v. it is possible study groups | that knowledge of th are treated in cours | ne intervention group a se of follow-up by inves | ffects how the two stigators?** | | **consider only in extreme cases in which preferential treatment is clearly evident; blinding in general not expected in social interventions |
| Low risk of bias | Risk of bias | High risk of bias | Critical risk of bias | Worth | to continue: |
| IV. Bias due to miss (Are the intervention data?) <u>Appraisal indicators:</u> <u>Consider whether</u> <i>i. outcome da</i> <i>ii. If 'no', are r</i> <i>iii. If missing da</i> <i>data similar</i> <i>iv. If missing da</i> | sing data (attrition groups free of critic ta are reasonably co missing data reporte ata: are proportion c across groups? ata: Were appropria |) cal differences in partic complete (80% or above ed? of participants and reas te statistical methods u | ipants with missing a) cons for missing used to account for | | |
| v. If not possib excluded fro | ala. were appropria a? (e.g. sensitivity a le to control for mis om analysis? | | | | |
| Low risk of bias | Risk of bias | Worth | to continue: | | |
| V. Outcome reporti (Are measurements | ng bias appropriate, e.g. cle | ear origin, or validity kn | own?) | | |

| Appraisal indicators | <u>.</u> | | | | |
|---|---|--|---|--------------|---|
| Consider whethe | r | | | | |
| i. there was a | n adequate period | for follow up*** | | | ***in many social science interventions, follow-up is not required to coincide with the start of the treatment; further, longer period of follow up are often required to measure changes. In the context of education, the question of retention – in particular when dealing with short intervention periods –(< 1 month) is of major interest. |
| ii. the outcome | e measure was clea | arly defined and objectiv | re | | |
| iii. outcomes w | ere assessed using | g standardised instrume | ents and indicators | | |
| iv. outcome me | easurements reflec | t what the experiment s | | | |
| v. the methods groups | s of outcome asses | sment were comparable | e across experiential | | |
| Low risk of bias | Risk of bias | High risk of bias | Worth | to continue: | |
| | | | | 1 | |
| VI. Bias in selectic (Are the reported or stage?) | on of results repo utcomes consistent | r ted t with the proposed outc | omes at the protocol | | |
| Appraisal indicators | <u>::</u> | | | | |
| Consider whethe | r | | | | |
| i. it is unlikely it was a not | that the reported e able finding among | effect estimate is availat numerous exploratory | le primarily because analyses | | |
| ii. it is unlikely from among | that the reported end | ffect estimate is prone measurements within th | to selective reporting ne outcome domain | | |

| | iii. it is unlikely from among iv. the analysis appropriate data?)**** Low risk of bias | that the re multiple a includes and were Risk of | reported effect estimate is prone to selective reporting analyses of the outcome measurementsan intention to treat analysis? (If so, was this e appropriate methods used to account for missingbiasHigh risk of biasCritical risk of | | | | | n clini ce: on | cal RCTs, rare in ly rate if conducted | |
|---|---|--|--|---|---|------------------------------------|------------------------------------|-------------------|---|--|
| | | | | | bias | | | | | |
| | | | | | | | | | | |
| Sources used in this s Pluye et al (2011); Go | section (in weighted o bugh et al (2007) ^{Weight} | rder): Co of evidence thin | chrane (201 ^{ngi} | 14); Stewart et al (20 | 14); Stewart et al (201 | 2); Higgins et al (2 | 011); G | Greent | algh & Brown (2014); | |
| Study type | | | Methodol | ogical appraisal crit | eria | | Response | | | |
| | | | | | Yes | No | Comment /confidence judgment | | | |
| 3. Mixed-methods ² <u>Sequential explanator</u> The quantitative comp qualitative. The purpo results using qualitativ quantitative results gu qualitative data source the qualitative findings interpretation of quant | r <u>y design</u> bonent is followed by se is to explain quant ve findings. E.g., the uide the selection of es and data collectior s contribute to the titative results. | the litative n, and | I. RE (as Applied mi o Se o Se o Tri o Em | SEARCH INTEGRA | TION/SYNTHESIS OF dded of the mixed-met : design lesign | • METHODS hods approach) | | | | |
| Sequential exploratory The qualitative compo quantitative. The purp and test an instrumen conceptual framework the qualitative findings | <u>y design</u> onent is followed by th oose is to explore, dev t (or taxonomy), or a ((or theoretical mode s inform the quantitat | ie velop I). E.g., | Appraisal i Consider v <i>i. the</i> to a | indicators: whether e rationale for integrat answer the research | ing qualitative and quadra and qua | antitative methods | | | | |

| collection, and the quantitative results allow a generalization of the qualitative findings. <u>Triangulation designs</u> The qualitative and quantitative components are | the mixed-methods research design is relevan qualitative and quantitative research questions qualitative and quantitative aspects of the mixe research question IDEFENSIBLE1 | t to address the , or the ed methods |
|---|--|---|
| concomitant. The purpose is to examine the same phenomenon by interpreting qualitative and quantitative results (bringing data analysis together at the interpretation stage), or by integrating qualitative and quantitative datasets (e.g., data on | iii. there is evidence that data gathered by both re was brought together to inform new findings to mixed-methods research question (e.g. form a picture, synthesise findings, configuration) [CREDIBLE] | esearch methods answer the complete |
| same cases), or by transforming data (e.g., quantization of qualitative data). <u>Embedded/convergent design</u> | iv. the approach to data integration is transparent considering all findings from both the qualitativ quantitative module (danger of cherry-picking) [RIGOROUS] | and rigorous in e and |
| The qualitative and quantitative components are concomitant. The purpose is to support a qualitative study with a quantitative sub-study (measures), or to better understand a specific issue of a quantitative study using a qualitative sub-study, e.g., the efficacy or the implementation of an intervention based on the views of participants. | v. appropriate consideration is given to the limitation with this integration, e.g., the divergence of qu quantitative data (or results)? [REFLEXIVE] | tions associated alitative and |
| For mixed-methods research studies, each compone no combined risk of bias assessment is facilitated, an research. | ent undergoes its individual critical appraisal first. Since nd the assigned risk of bias from the quantitative compo | qualitative studies are either included or excluded, onent similarly holds for the mixed-methods |
| The above appraisal indicators only refer to the appli appraisal criteria below, then the quantitative/qualitat | ed mixed-methods design. If this design is not found to tive components will individually be included in the revie | comply with each of the four mixed-methods ew: |
| Mixed-methods critical appraisal: | Qualitative critical appraisal: | Quantitative critical appraisal: |
| 1. Research is defensible in design | Include / Exclude | 1. Low risk of bias |
| 2. Research is rigorous in conduct | | 2. Risk of bias |
| 3. Research is credible in claim | | 3. High risk of bias |
| 4. Research is reflective | | Critical risk of bias |
| Combined appraisal: | | | | | | | |
|--|------------------|--|--|--|--|--|--|
| Include / Exclude mixed-methods findings judged wit | h risk of bias | | | | | | |
| | | | | | | | |
| Section based on Pluye et al (2011). Further sources consulted (in alphabetical order): Creswell & Clark (2007); Crow (2013); Long (2005); O'Cathain et al | | | | | | | |
| (2008): O'Cathain (2010): Pluve & Hong (2014): Sirri | veh et al (2011) | | | | | | |

¹Two theoretical exceptions to this rule apply:

- i) A RCT with appropriate randomization procedure can be included without showing baseline data, as both experimental groups can be assumed to be equal at baseline by design.
- ii) A sophisticated quasi-experimental design such as PSM or RDD in theory could make the same claim to not require baseline data.

In both cases, the advise of an evaluation specialist will be thought as the researcher does not have the capacity to make an informed judgment in such specialist cases.

² The mixed-methods Critical Appraisal is facilitated for studies applying an explicit mixed-methods approach. The component is applied in addition to criteria for the qualitative component (I to VI), and appropriate criteria for the quantitative component (I to VI).

Appendix 4.3: Effect size calculations

Corrected SMD and corrected Standard Errors (SE) will be estimated as follows:

$$SMD_{corrected} = SMD_{uncorrected} * \left[1 - \frac{3}{4 * (n_t + n_c - 2) - 1} \right]$$
$$SE(SMD)_{corrected} = SE(SMD)_{uncorrected} * \left[1 - \frac{3}{4 * (n_t + n_c - 2) - 1} \right]$$

For regression-based estimates, we will follow Keef and Roberts (2004: p100-101; p129, equation A9) to correct for potential sample bias in the effect sizes.

Many of the impact evaluation designs that we expect to see in this review are likely to use complex statistical analyses, and there is a lack of standard methods for computing effect sizes from these designs. In most cases we expect our approach for computing effect sizes to be as follows.

Calculating standardised mean differences

For studies reporting matching-based estimates, the following formulae to compute g and its standard error will be used, where Y_t and Y_c are the post-intervention mean outcome in the treatment group and control group respectively.

$$SMD = \frac{Y_t - Y_c}{S_p}$$

To calculate S_p , the pooled standard deviation (the standard deviation of the outcome variable for both treated (S_t) and control (S_c) individuals), we will use the Hedges' approach described in Lipsey and Wilson (2001), where n_t and n_c are the sample size of the treatment and the control group.

$$S_p = \sqrt{\frac{(n_t - 1) * S_t^2 + (n_c - 1) * S_c^2}{n_t + n_c - 2}}$$

The standard error of *g* will be computed using the following formula.

$$SE (SMD) = \sqrt{\frac{n_t + n_c}{n_t * n_c}} + \frac{SMD^2}{2 * (n_c + n_t)}$$

Alternatively, in the event that the necessary information for calculating SE is not available, we will approximate it as follows, where *t* is the *t* statistic of the treatment effect.

$$SE(SMD) = \frac{SMD}{t}$$

For regression-based studies, we intend to use the following formula described in Keef and Roberts (2004), where β is the coefficient of interest (i.e. yielding the impact of the intervention) and σ is the standard deviation of the error term in a regression.

$$SMD = \frac{\beta}{\sigma}$$

Where σ is not reported by the study authors (highly likely), we will use the following formula to compute an equivalent using the sample standard deviation of the dependent variable and the sample size for both treatment and control groups.

$$S_{p} = \sqrt{\frac{(SD_{y}^{2} * (n_{t} + n_{c} - 1) - \frac{(\beta^{2} * (n_{t} * n_{c}))}{n_{t} + n_{c}}}{n_{t} + n_{c}}}$$

Standard errors will be approximated using the following formula where *t* is the *t* statistic for the regression coefficient.

$$SE(SMD) = \frac{SMD}{t}$$

Conversion of SMDs into percentages

In general, an effect size of 0.10 reflects 1/10 standard deviation improvement for the treatment group (i.e. learners engaged in the mobile learning intervention) compared to the control group (Petrosino et al 2010). In an attempt to make this measurement scale less abstract, Rosenthal and Rubin (1982) convert the SDM to a percentage expression of the change between experimental groups. Based on this approach, a more intuitive translation of effect sizes is presented below. For example, a SDM of 0.10 represents and improvement of 5 per cent in the intervention group.

| Effect size (cohen's <i>d</i>) | Percentage success Treatment | Percentage Success Control | Difference in success rates |
|------------------------------------|---------------------------------|-------------------------------|--------------------------------|
| 0.00 | 50.0 | 50.0 | 0 |
| 0.10 | 52.5 | 47.5 | 5% |
| 0.20 | 55.0 | 45.0 | 10% |
| 0.30 | 57.4 | 42.6 | 14.8% |
| 0.40 | 59.8 | 40.2 | 19.6% |
| 0.50 | 62.1 | 37.9 | 24,2% |
| 0.60 | 64.4 | 35.6 | 28.8% |

Translation of effect sizes Table adapted and updated from Petrosino et al 2012, based on calculation formula by Rosenthal and Rubin (1982). Note calculation in the meta-analysis are based on g, a sample size correct version of d. The correction has marginal implications for the above translation.

Appendix 4.4: Interview guide

1. Introduction

- Provide information sheet and give verbal introduction to myself and research
- Provide consent form and talk through if needed
- Survey monkey on basic demographic information together on tablet

2. Start semi-structured conversation

- 2.1 Basic information around ICT4RED (≈5 mins) Pointers:
 - How long have you been a teacher?
 - Why did you become a teacher; what do you value about it?
 - How long have you been involved in the ICT4RED project?
 - How were people chosen for ICT4RED?
 - Why did you choose project involvement?
 - How has it been received at your school/community?

2.2 Teachers' experiences of using the tablets (≈10 mins)

Pointers:

- In general, what do you value about mobile technology in your normal life?
- What was your experience of learning how to teach with the tablet?
- Do you feel like you can use ICTs now? And if so, do you value this, don't you value this?
- What do you value about using mTech in your daily teaching; what don't you value?
- Can you describe how you teach with the tablets?
- Main functions of using it? What do you do with the mtech?
- Where do you use it most?
- Do you think it is important / not important for a teacher to be able to use and teach with mtech?
 - Would you recommend it?
- 2.3 Teachers' personal development (≈10 mins)

Pointers:

- Has the experience of using mtech changed from when you started?
- Has experience of using mtech changed you? Do you value these changes; don't you value these changes?
- What was your biggest success/nicest memory of using the tablets?
 - If you feel comfortable sharing it with me, what was a disappointment in using the tablets?
- What do you think other people (colleagues, family, community) think about you using tablets to teach?
- Do you use the tablet at home and with your family?
 - Do you use it for your own or your family's education?
- 2.4 Teacher's educational experience (≈10 mins)

Pointers:

- Recap on this: Do you think it is important / not important for a teacher to be able to use and teach with mtech?
- Can you elaborate on what are things you are able to do now in your teaching that you weren't able to before? Do you value/not value these changes?
- Do you think that the tablets have helped you reach your teaching goals (reformulate as: *reach your goals as a teacher*?) Or have they hindered you?
- In your opinion, is mtech like the tablets an educational tool, or not?
 - What is the difference between using mobile technology vs. normal ICTs?
- Can you describe the perfect teacher to me?

3. Debrief

- Is there anything else you want to tell me?
- Explain what happens next.

Appendix 4.5: Observation schedule⁸⁵

| Classroom set up: | Organisation and state of furniture; Classroom atmosphere (lighting, pictures on the wall, temperature, etc) |
|----------------------|--|
| | Number of learners: |
| Educational material | |
| | |
| ICT material | What devices in use? |
| | What educational content is on the tablets? |
| Lesson content | |
| | |
| Class atmosphere | Greetings, etc |
| | |
| Class interaction | Questions? Encouragement for questions? Praises? |
| | |
| Class participation | Who? On what terms? Who's in control? |
| | |
| What teaching | |
| methods ? | |
| Teachers' behaviour | |
| | |
| Learners' behaviour | |
| | |

⁸⁵ Formatting adapted for publication in the thesis.

Technology event teacher:

| Event | Perception/Usage | Traditional (substitute) | Unique affordance of technology that supported the event |
|---|---|---------------------------------------|--|
| Event 1: Taking pictures of the school garden to teach a class on botany | E.g. Tech literate; Eagerness to use them; Handle with care | Taking notes / sketches of the plants | Taking pictures and videos |
| | | | |

Technology event learner:

| | Perception/Usage | Traditional (substitute) | Unique affordance (independent) |
|---|---|---|---------------------------------|
| Event 1: Taking pictures of the school garden to teach a class on botany | E.g. Tech literate; Eagerness to use them; Handle with care | Taking notes / sketches of the plants Presenting about the plants to the rest of the class using pictures and videos of the plant | Taking pictures and videos |
| | | | |

Learner-centred teaching:

| General | Participatory | Interactive | Authenticity | Personalisation | Collaboration | Who's in power? | Learn on your own? |
|---------|---------------|-------------|--------------|-----------------|---------------|--------------------|-----------------------|
| | | | | | | | |
| | | | | | | | |

Appendix 5.1: List of included studies

- Aker, J. C., Ksoll, C., & Lybbert, T. J. (2012). Can mobile phones improve learning? Evidence from a field experiment in Niger. *American Economic Journal: Applied Economics*, 4(4), 94-120.
 - Aker, J. C., Ksoll, C., & Lybbert, T. J. (2010). ABC, 123: The impact of a mobile phone literacy program on educational outcomes. CGD Working Paper 223. Washington, D.C.: Center for Global Development.
- Aker, J. C., & Ksoll, C. (2015). Call Me Educated: Evidence from a Mobile Monitoring Experiment in Niger. CGD Working Paper 406. Washington, DC: Center for Global Development.
- Aladejana, F. Mobile Phones in Higher Education Institutions: A Case Study of a Nigerian University. Currently unavailable online.
- Ale, K., & Chib, A. (2011). Community factors in technology adoption in primary education: Perspectives from rural India. *Information Technologies & International Development*, 7(4), 53-68.
- Andersson, A., & Hatakka, M. (2010). Increasing interactivity in distance educations: Case studies Bangladesh and Sri Lanka. *Information Technology for Development*, *16*(1), 16-33.
- Balasubramanian, K., Thamizoli, P., Umar, A., & Kanwar, A. (2010). Using mobile phones to promote lifelong learning among rural women in Southern India. *Distance Education*, *31*(2), 193-209.
- Bello-Bravo, J., & Baoua, I. (2012). Animated videos as a learning tool in developing nations: A pilot study of three animations in Maradi and surrounding areas in Niger. *The Electronic Journal of Information Systems in Developing Countries*, 55.
- Beltramo, T., & Levine, D. (2012). Do SMS Text Messaging and SMS Community Forums improve outcomes of adult and adolescent literacy programs? Accessed 13 October 2017: <u>http://escholarship.org/uc/item/1c31c2m4</u>
- Binsaleh, S., & Binsaleh, M. (2013). Mobile learning: What guidelines should we produce in the context of mobile learning implementation in the conflict area of the four southernmost provinces of Thailand. *Asian Social Science*, 9(13), 270-281.
- Bridge International Academies. (2013. The Bridge Effect: Comparison of Bridge Pupils to Peers at Nearby Schools EGRA-EGMA Evaluation Programme. Accessed 13 October 2017: <u>http://www.bridgeinternationalacademies.com/wp-</u> <u>content/uploads/2013/01/Bridge-International-Academies White-Paper The-Bridge-</u> Effect Nov-2014 Website.pdf
- Chang, A. Y., Ghose, S., Littman-Quinn, R., Anolik, R. B., Kyer, A., Mazhani, L., ... & Kovarik, C. L. (2012). Use of mobile learning by resident physicians in Botswana. *Telemedicine and e-Health*, 18(1), 11-13.
- Chen, C. M., & Li, Y. L. (2010). Personalised context-aware ubiquitous learning system for supporting effective English vocabulary learning. *Interactive Learning Environments*, 18(4), 341-364.
- Cilliers, J., Kasirye, I., Leaver, C., Serneels, P., & Zeitlin, A. (2014). Improving teacher attendance using a locally managed monitoring scheme: Evidence from Ugandan Primary Schools. Policy note14/0189. London, UK: International Growth Centre Accessed 13 October 2017: <u>https://www.theigc.org/wp-content/uploads/2014/09/Cilliers-Et-Al-2014-Policy-Brief.pdf</u>
- Cole, S., & Fernando, A. N. (2012). The value of advice: Evidence from mobile phone-based agricultural extension. Accessed 13 October 2017: <u>http://hbswk.hbs.edu/item/the-value-of-advice-evidence-from-mobile-phone-based-agricultural-extension</u>
- Daud, N.M., & Husin, Z. (2013) Mobile Learning: A Quasi-Experiment on Using SMS to Support Reading Comprehension Programme. In Embi, M. A., & Nordin, N. M. (eds).

Mobile learning: Malaysian initiatives and research findings. Malaysia: Centre for Academic Advancement, Universiti Kebangsaan Malaysia.

- English in Action. (2011). A qualitative observation study of primary and secondary lessons conducted by teachers participating in EIA. Report 1 synopsis. Dhaka, Bangladesh: EIA.
 - English in Action. (2011). Perceptions of English language learning and teaching among primary and secondary school teachers and students participating in EIA. Dhaka, Bangladesh: EIA
 - English in Action. (2012). The Classroom Practices of Primary and Secondary School Teachers Participating in English in Action (Study 2a2). Dhaka, Bangladesh: EIA.
- Ekanayake, T. M. S. S. K. Y., & Wishart, J. M. (2014). Developing teachers' pedagogical practice in teaching science lessons with mobile phones. *Technology, Pedagogy and Education*, 23(2), 131-150.
 - Ekanayake, S. Y., & Wishart, J. (2014). Mobile phone images and video in science teaching and learning. *Learning, Media and Technology*, 39(2), 229-249.
- Enge, K. (2011) Elimu kwa teknolojia. a BridgeIT programme. Summative evaluation. Currently unavailable online.
- Frias-Martinez, V., Virseda, J., & Gomero, A. (2012). Mobilizing education: evaluation of a mobile learning tool in a low-income school. In Proceedings of the 14th international conference on Human-computer interaction with mobile devices and services (pp. 441-450). ACM.
- GSMA. (2012b). Shaping The Future: Realising the potential of informal learning through mobile. Accessed 13 October 2017: <u>https://www.gsma.com/mobilefordevelopment/wp-content/uploads/2012/05/mLearning_Report_230512_V2.pdf</u>
- Haßler, B., Hennessy, S., Lord, T., Cross, A., Simpson, M., & Jackson, A. (2011). An investigation of appropriate new technologies to support interactive teaching in Zambian schools (ANTSIT). Final report to DfID. Accessed 13 October: <u>http://www.educ.cam.ac.uk/centres/archive/cce/publications/CCE_Report_No6_%20Dfl_D_ANTSIT.pdf</u>
- He, F., Linden, L., & MacLeod, M. (2008). How to teach English in India: Testing the relative productivity of instruction methods within the Pratham English language education program. New York, United States: Columbia University. Mimeographed document.
- Hwang, G. J., Wu, P. H., Zhuang, Y. Y., & Huang, Y. M. (2013). Effects of the inquiry-based mobile learning model on the cognitive load and learning achievement of students. *Interactive Learning Environments*, *21*(4), 338-354.
- Iqbal, S., & Qureshi, I. A. (2012). June 2012 M-Learning Adoption : A Perspective from a Developing Country. *The International Review of Research in Open and Distance Learning*, 13(3), 1–10.
- Ismail, I., Bokhare, S. F., Azizan, S. N., & Azman, N. (2013). Teaching via Mobile Phone: a Case Study on Malaysian Teachers' Technology Acceptance and Readiness. *Journal of Educators Online*, 10, 1–38.
- Jantjies M, M. J. (2015). Mobile enhanced learning in a South African context. *Journal of Educational Technology & Society*, *18*(1), 308.
- Jere-Folotiya, J., Chansa-Kabali, T., Munachaka, J. C., Sampa, F., Yalukanda, C., Westerholm, J., ... & Lyytinen, H. (2014). The effect of using a mobile literacy game to improve literacy levels of grade one students in Zambian schools. *Educational Technology Research and Development*, 62(4), 417-436.
- Kafyulilo, A. (2014). Access, use and perceptions of teachers and students towards mobile phones as a tool for teaching and learning in Tanzania. *Education and Information Technologies*, 19(1), 115–127
- Kaleebu, N., Gee, A., Jones, R., & Watson, A. H. A. (2013). SMS story impact assessment report. VSO, Papua New Guinea.
 - Kaleebu, N., Gee, A., Maybanks, N., Jones, R., Jauk, M., & Watson, A. H. (2013). SMS story: Early results of an innovative education trial. Contemporary PNG Studies, 19, 50.

- Kim, P. H. (2009). Action research approach on mobile learning design for the underserved. *Educational Technology Research and Development*, *57*(3), 415-435.
- Kim, P., Hagashi, T., Carillo, L., Gonzales, I., Makany, T., Lee, B., & Garate, A. (2011). Socioeconomic strata, mobile technology, and education: A comparative analysis. *Educational Technology Research and Development*, 59(4), 465-486.
- Kim, P., Buckner, E., Kim, H., Makany, T., Taleja, N., & Parikh, V. (2012). A comparative analysis of a game-based mobile learning model in low-socioeconomic communities of India. *International Journal of Educational Development*, 32(2), 329–340.
- Korkmaz, H. (2010). The effectiveness of mobile assisted language learning as a supplementary material for English language teaching coursebooks (MA dissertation, Bilkent University).
- Kumar, A., Reddy, P., Tewari, A., Agrawal, R., & Kam, M. (2012). Improving literacy in developing countries using speech recognition-supported games on mobile devices. In Proceedings of the SIGCHI Conference on Human Factors in Computing Systems (pp. 1149-1158). ACM.
- Kumar, A., Tewari, A., Shroff, G., Chittamuru, D., Kam, M., & Canny, J. (2010). An exploratory study of unsupervised mobile learning in rural India. Proceedings of the 28th International Conference on Human Factors in Computing Systems CHI '10, 743.
- Lai, C. H., Yang, J. C., Chen, F. C., Ho, C. W., & Chan, T. W. (2007). Affordances of mobile technologies for experiential learning: the interplay of technology and pedagogical practices. *Journal of Computer Assisted Learning*, 23(4), 326-337.
- Leach, J., Ahmed, A., Makalima, S., & Power, T. (2005). DEEP IMPACT: an investigation of the use of information and communication technologies for teacher education in the global south. London, UK: DFID.
 - Leach, J., Patel, R., Peters, A., Power, T., Ahmed, A., & Makalima, S. (2004).
 Deep impact: a study of the use of hand-held computers for teacher professional development in primary schools in the Global South. *European Journal of Teacher Education*, 27(1), 5-28.
 - Leach, J., & Makalima, S. (2006). 4D technologies for teachers: investigating the use of ICT by the rural poor in Eastern Province, South Africa, paper presented to the American Educational Research Association Conference, San Francisco, 7–11 April.
- Liu, C. C., Tao, S. Y., & Nee, J. N. (2008). Bridging the gap between students and computers: supporting activity awareness for network collaborative learning with GSM network. *Behaviour & Information Technology*, 27(2), 127-137.
- Liu, Y., Li, H., & Carlsson, C. (2010). Factors driving the adoption of m-learning: An empirical study. *Computers and Education*, 55(3), 1211–1219.
- Liu, T. Y., Tan, T. H., & Chu, Y. L. (2009). Outdoor natural science learning with an RFIDsupported immersive ubiquitous learning environment. Journal of Educational *Technology & Society*, 12(4).
- Lu, M. (2008). Effectiveness of vocabulary learning via mobile phone. *Journal of computer assisted learning*, 24(6), 515-525.
- Majid, F.A. (2013). Adult Learners and Mobile Learning: A Case Study. In Embi, M. A., & Nordin, N. M. (eds). Mobile learning: Malaysian initiatives and research findings. Malaysia: Centre for Academic Advancement, Universiti Kebangsaan Malaysia.
 - Almaiah, M. A., & Jalil, M. A. (2014). Investigating Students' Perceptions on Mobile Learning Services. *iJIM*, 8(4), 31-36.
- Makoe, M. (2010). Exploring the use of MXit: a cell-phone social network to facilitate learning in distance education. *Open Learning*, 25(3), 251-257.
- Masika, M. M., Omondi, G. B., Natembeya, D. S., Mugane, E. M., Bosire, K. O., & Kibwage, I. O. (2015). Use of mobile learning technology among final year medical students in Kenya. *Pan African Medical Journal*, 21, 1–12.

- Masperi, P., Hollow, D. (2008). The Potential of portable interactive learning technology within basic education in Malawi. Accessed 13 October 2017: https://eurotalk.com/malawi/eurotalk_malawi_report.pdf
- Mruz, K. (2011). Rwanda TeacherMate report. Accessed 13 October 2017: <u>http://www.heart-resources.org/doc_lib/rwanda-teachermate-report/</u>
- Onguko, B., Jepchumba, L., & Gaceri, P. (2013). "For us it was a learning experience." *European Journal of Training and Development*, 37(7), 615–634.
- Osman, M., & Chung, P. W. H. (2011). Language learning using texting and wiki: a Malaysian context. In e-CASE & e-Tech International Conference. January 18-20, 2011, Toshi Center Hotel, Tokyo, Japan
- Pamuk, S., Çakır, R., Ergun, M., Yılmaz, H. B., & Ayas, C. (2013). The use of tablet PC and interactive board from the perspectives of teachers and students: Evaluation of the FATİH Project. *Educational Sciences: Theory & Practice*,13(3), 1815-1822.
- Pimmer, C., Brysiewicz, P., Linxen, S., Walters, F., Chipps, J., & Gröhbiel, U. (2014). Informal mobile learning in nurse education and practice in remote areas-A case study from rural South Africa. *Nurse Education Today*, 34(11), 1398–1404.
- Pimmer, C., Linxen, S., Gröhbiel, U., Jha, A. K., & Burg, G. (2013). Mobile learning in resource-constrained environments: A case study of medical education. *Medical Teacher*, 35(5), e1157–e1165.
 - Pimmer, C., Linxen, S., & Gröhbiel, U. (2012). Facebook as a learning tool? A case study on the appropriation of social network sites from mobile phones in developing countries. *British Journal of Educational Technology*, 43(5), 726-738.
- Piper, B., & Kwayumba, D. (2015). USAID/Kenya Primary Math and Reading (PRIMR) Initiative: Kisumu information and communication technology (ICT) intervention. Endline Report. Research Triangle Institute (RTI), USA. Accessed 13 October 2017: <u>http://pdf.usaid.gov/pdf_docs/PA00K282.pdf</u>
 - Piper, B., Jepkemei, E., Kwayumba, D., & Kibukho, K. (2015). Kenya's ICT policy in practice: The effectiveness of tablets and e-readers in improving student outcomes. In *FIRE: Forum for International Research in Education* (Vol. 2, No. 1, p. 2).
- Pitchford, N. (2014). Unlocking talent: Evaluation of a tablet-based masamu intervention in a Malawian primary school. University of Nottingham Report: United Kingdom. Accessed 13 October 2017: https://onebillion.org/downloads/unlocking-talent-final-report.pdf
 - Pitchford, N. J. (2015). Development of early mathematical skills with a tablet intervention: a randomized control trial in Malawi. *Frontiers in psychology*, 6.
- Porter, G., Hampshire, K., Milner, J., Munthali, A., Robson, E., Lannoy, A., ... & Abane, A. (2016). Mobile Phones and Education in Sub-Saharan Africa: From Youth Practice to Public Policy. *Journal of International Development*, 28(1), 22-39.
- Ramos, A.J.O. & Triñona, J.P. (2009). Mobile Technology in Nonformal Distance Education. In Baggaley, J., & Belawati, T. (eds) Distance Education Technology in Asia. PANdora: Distance and Open Resource Access. Accessed 13 October 2017: <u>http://www.pandora-asia.org/downloads/Book-2/PANdora-book2_v6.pdf</u>
- Roberts, N., & Vänskä, R. (2011). Challenging assumptions: Mobile learning for mathematics project in South Africa. *Distance Education*, 32(2), 243–259.
- Rosas, R., Nussbaum, M., Cumsille, P., Marianov, V., Correa, M., Flores, P., ... & Rodriguez, P. (2003). Beyond Nintendo: design and assessment of educational video games for first and second grade students. *Computers & Education*, 40(1), 71-94.
- Sahni, U., Gupta, R., Hull, G., Javid, P., Setia, T., Toyama, K., & Wang, R. (2008). Using digital video in rural Indian schools: A study of teacher development and student achievement. In annual meeting of the American Educational Research Association, New York. Accessed 13 October 2017:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.161.9827&rep=rep1&type=pdf

Shohel, M. M. C., & Kirkwood, A. (2012). Using technology for enhancing teaching and learning in Bangladesh: challenges and consequences. *Learning, Media and Technology*, 37(4), 414-428.

- Shohel, M. M. C., & Power, T. (2010). Introducing mobile technology for enhancing teaching and learning in Bangladesh: teacher perspectives. *Open learning*, 25(3), 201-215.
- Shraim, K. Y. (2014). A Case Study of Mobile Technology-enabled English Language Learning: the Amazon Kindle e-Reader Initiative in Palestine. *International Journal of Interactive Mobile Technologies (iJIM)*, 8(3), 25-31.
- Suki, N. M., & Suki, N. M. (2011). Using Mobile Device for Learning: From Students' Perspective. *Education Review*, A1, 44-53.
- Sung, H. Y., Hwang, G. J., & Chang, Y. C. (2013). Development of a mobile learning system based on a collaborative problem-posing strategy. *Interactive Learning Environments*, 24(3), 456-471.
- Taleb, Z., & Sohrabi, A. (2012). Learning on the move: the use of mobile technology to support learning for university students. *Procedia-Social and Behavioral Sciences*, 69, 1102-1109.
- Turtiainen, E., Blignaut, S., Els, C., Laine, T. H., & Sutinen, E. (2009). Story-based ufractions mobile game in South Africa: Contextualization process and multidimensional playing experiences. CEUR Workshop Proceedings, 498.
- UNESCO. (2014). Reading in the mobile era: A study of mobile reading in developing countries. Paris, France: UNESCO.
- USAID. (2010). Somalia Interactive Radio Instruction Project (SIRIP): Mid-term Evaluation. Accessed 13 October 2017:

http://m4ed4dev.linhost1.jbsinternational.com/content/somalia-interactive-radioinstruction-project-sirip

- USAID. 2007. Somali Interactive Radio Instruction Program. Baseline report: Grade 1 Evaluation 2006/2007.
- Valderrama, E. D. C., & Schmidt, A. (2010). A survey to assess the potential of mobile phones as a learning platform for panama. In *CHI'10 Extended Abstracts on Human Factors in Computing Systems* (pp. 3667-3672). ACM.
- Velghe, F. (2014). "I wanna go in the phone": literacy acquisition, informal learning processes, "voice" and mobile phone appropriation in a South African township. *Ethnography and Education*, 9(1), 111–126.
- Voigt, T., & Matthee, M. (2012). Tablets with restricted mobility: Investigating user acceptance in a South African mathematics mobile learning project. CEUR Workshop Proceedings, 955(September), 172–179.

Walton, M. (2009). Mobile literacies & South African teens: Leisure reading, writing, and MXit chatting for teens in Langa and Guguletu. Shuttleworth Foundation

- Deumert, A. (2010). Klk cc... Supporting Indigenous Literacies in the Digital Space. University of Capetown. Accessed 13 October 2017: <u>https://m4lit.files.wordpress.com/2010/03/m4lit_indigenous_literacies_adeumert_</u> 2010.pdf
- Vosloo, S. (2010). M4Lit Project report. Shuttleworth Foundation. Accessed 13 October 2017:

https://m4lit.files.wordpress.com/2010/03/m4lit_project_report_svosloo_20102.pd f

Wennersten, M., & Qureshy, Z. B. (2012). Summative evaluation of the Nokia & Pearson Foundation BridgeIT India Project. Year 1. 2011/2012 school year. EZ Vidya, Delhi. Accessed 13 October 2017:

http://www.educationinnovations.org/sites/default/files/BridgeIT%20India%20Endline%2 0Report%202011-12.pdf

- Witt, R. E., Kebaetse, M. B., Holmes, J. H., Ryan, L. Q., Ketshogileng, D., Antwi, C., ... Nkomazana, O. (2016). The role of tablets in accessing information throughout undergraduate medical education in Botswana. *International Journal of Medical Informatics*, 88, 71–77.
- Worldreader. (2012). iRead Ghana study. Final evaluation report. Accessed 13 October 2017: <u>http://pdf.usaid.gov/pdf_docs/pnadz402.pdf</u>

- Worldreader. (2014). Ghana iREAD Study 2012-2014. Final evaluation. Accessed 13 October 2017: <u>https://comms.worldreader.org/wp-content/uploads/2017/01/iREAD-Full.pdf?x80235</u>
- Worldreader. (2013). iRead2 Midterm study results. All children reading in Ghana. Currently unavailable online.
- Worldreader. (2015). Project Leap. Evaluating the Effects of E-readers in Libraries in Kenya. Final report. Accessed 13 October 2017: <u>https://comms.worldreader.org/wp-</u> content/uploads/2015/04/PROJECT_LEAP.pdf?x80235
 - Worldreader. (2014). Project LEAP Libraries, E-reading, Activities, Partnership Baseline Report.
 - Worldreader. (2014). Project LEAP Libraries, E-reading, Activities, Partnership Midterm Report.
- Wu, P. H., Hwang, G. J., Tsai, C. C., Chen, Y. C., & Huang, Y. M. (2011). A pilot study on conducting mobile learning activities for clinical nursing courses based on the repertory grid approach. *Nurse Education Today*, 31(8), e8-e15.
- Yang, C. C., Hwang, G. J., Hung, C. M., & Tseng, S. S. (2013). An evaluation of the learning effectiveness of concept map-based science book reading via mobile devices. *Journal* of Educational Technology & Society, 16(3).
- Zelezny-Green, R. (2014). She called, she Googled, she knew: Girls' secondary education, interrupted school attendance, and educational use of mobile phones in Nairobi. *Gender and Development*, 22(1), 63–74.
- Zurita, G., & Nussbaum, M. (2004). Computer supported collaborative learning using wirelessly interconnected handheld computers. *Computers & education*, 42(3), 289-314.
- Zurita, G., & Nussbaum, M. (2004). A constructivist mobile learning environment supported by a wireless handheld network. *Journal of Computer Assisted Learning*, 20(4), 235-243.
- Zurita, G., & Nussbaum, M. (2007). A conceptual framework based on Activity Theory for mobile CSCL. *British Journal of Educational Technology*, 38(2), 211–235.

Appendix 5.2: Overview of included studies

| Region | Intervention details | | | | | etails | Outcome details | |
|---|---|--|---|---|--|---|--|---|
| | Technology/ application | Subject/ educational feature | Focus/ Access | Pedagogy | N | Setting | Mobile context | Outcome & Measures |
| Aker 2010 Africa: Niger | Basic/feature phone: SMS | Literacy & numeracy: community based lessons enhanced by SMS quizzes & messages MALL/MAML | Learner- centered 1:group | To learn: Additional learning by extending learning to informal context | 5014 Adults Education program | Informal: Rural Community Groups | Yes: Community centers & home | Literacy & Numeracy Test Scores Student & Teacher effort Attendance & hotline |
| Aker 2015 Africa: Niger | Basic/feature phone: Audio: Calls | Literacy & numeracy: community based lessons enhanced by Mobile Monitoring | Teacher- centered 1:1 | To monitor: Additional teaching and motivation through teacher attendance monitoring | 1926 Adults Education program | Informal: Rural Community groups | Yes: Teachers being called in formal and informal contexts | Literacy & Numeracy (test scores) Self-esteem Self-efficacy Teacher attendance Teacher motivation Student attendance Student motivation |
| Aladejana 2007 Africa: Nigeria | Basic/feature phone: SMS | General perceptions | Learner- centered | To access/record information: Anywhere, any time | 413 Tertiary | Formal Urban students | Yes: Anywhere, anytime | Perceptions: Factors of adoption |
| Ale 2011 Asia: India | Mobile PCs: Educational content (quizzes etc) | All subjects: Educational gaming software | Learner/ teacher- centered | To learn: Game based; more interactive | 22 Primary | Formal Rural Community | No: Only usage in classroom | Perceptions: Factors of adoption |
| Andersson 2010 Asia Bangladesh | Basic/feature phone: SMS, camera | English (FL): Curriculum & content embeds mobiles to make education more interactive MALL | Learner/ teacher- centered 1:1 | To learn & administer: Interactive learning to yield to social transformation | 45 Tertiary | Formal: Distance education | Yes: anytime, anywhere | Changes in pedagogy & learning experience: Qualitative |
| Balasubramanian 2010 Asia: India | Basic/feature phone: SMS/Audio messages | Extension (Agric): Educational messages send to | Learner- centered | To learn & connect: In-context learning for women to | 320 Adults Livestock | Informal: Women groups | Yes: In-the field; at home; anywhere, | Empowerment & Social Capital |

| | | newly-owned cellphone | 1:1 Females | increase lifelong learning & empowerment, social transformation | extension program | | anytime | |
|---|---|---|---|--|--|--|--|--|
| Bello-Bravo 2012 Africa: Niger | Smartphones: Educational 3D animations | Extension (Agric/health) Educational 3D animations shared among rural communities | Learner- centered 1:group | To learn: In-context learning, more interactive/less literacy requirements | 48 Adults Livelihood Extension program | Informal: Rural communities at own will | Yes: Anytime, anywhere, in- context (potential) | Knowledge gains Perceptions: Factors of adoption |
| Beltramo 2012 Africa: Niger | Basic/feature phone: SMS – quizzes, educational messages, networks | Literacy: community based lessons enhanced by SMS quizzes & messages MALL | Learner- centered 1:1 | To learn & to connect: Additional learning by extending learning to informal context | 800 Adults Education program | Informal: Rural Community Groups | Yes: Anywhere, anytime: community centers & home | Literacy: Test scores |
| Binsaleh 2013 Asia: Thailand | Tablets: with pre- loaded educational content: e-books, exercises | General educational tool: lessons, content, gaming | Learner/ teacher- centered 1:1 | To teach & to learn General educational support by tablets; more student- centered | 18 High school | Formal: Rural Community | Yes: Anywhere, anytime; no restrictions | Perceptions: factors of adoption |
| Bridge International Academies 2013 Africa: Kenya | Tablets: with pre- loaded lesson plans and teacher monitoring | English (FL) / math: scripted lessons for teachers read off + monitoring of teacher practice MALL/MAML | Teacher- centered 1:1 | To teach: reading off scripted lessons & being monitored | 1359 Primary | Formal: urban students | No: restricted to classroom | Literacy & math: Test scores |
| Chang 2012 Africa: Botswana | Smartphones: mhealth application | mHealth: in- context practice tools (drug reference, disease diagnostic) | Learner- centered 1:1 | To learn: In-context learning for health trainees | 10 Tertiary | Formal: Urban students – hospital | Yes: Anywhere, anytime; no restrictions | Perceptions: factors of adoption |
| Chen 2010 Asia: Taiwan | PDAs: personalised context-aware ubiquitous learning system | English (FL): context-aware vocabulary lessons & quizzes MALL | Learner- centered 1:1 | To learn: in-context learning for English vocabulary | 36 High school | Formal: Urban students | Yes: context restricted to device recognition (GPS) | English literacy: Test scores |
| Cilliers 2014 Africa: Uganda | Basic/feature phone: SMS system to monitor teacher attendance | EMIS: monitoring teacher attendance | Teacher- centered 1:1 | To monitor: increased attendance leads to better education | 180 schools High/ Primary | Formal Rural/urban teachers | Yes: parents act as monitors too | Teacher attendance Quality of monitoring reports |

| Cole 2012 Asia: India | Basic/feature phone: SMS & voice; extension hotline | Extension (agric): contextualised information services via audio & SMS | Learner- centered 1:1 | To learn & access information: in- context/in-the-field learning for agric workers | 766 Adults Extension program | Informal: Rural Community Groups | Yes: Anywhere, anytime; in particular in the field | Knowledge gains: Test scores |
|--|---|---|---|--|---------------------------------------|---|--|--|
| Daud 2013 Asia: Malaysia | Basic/feature phone: SMS | English (FL): reminder and information SSM MALL | Learner- centered 1:1 | To access information: additional learning events created by mobiles | 615 Tertiary | Formal: Urban students | Yes: Anywhere, anytime | English literacy: Test scores |
| EIA 2011 Asia: Bangladesh | i-Pod: audio features, video lessons speakers | English (FL): model lessons, more interactivity & student involvement through mobiles MALL | Teacher- centered 1:1 1:class | To teach: lessons to be more student- centered & to encourage interactivity | 102 Primary & High School | Formal: Mixed rural/urban teachers | Yes: Anywhere, anytime. Domestic, enroute & in the classroom | Teaching practice: Observation |
| EIA [Sohel] 2012 Asia: Bangladesh | i-Pod: audio features, video lessons speakers | English (FL): model lessons, more interactivity & student involvement through mobiles MALL | Teacher- centered 1:1 1:class | To teach: lessons to be more student- centered & to encourage interactivity | 20 High school | Formal Urban teachers | Yes: Anywhere, anytime. Domestic, enroute & in the classroom | Teaching practice: Interviews |
| Ekanayake 2013 Asia: Sri- Lanka | Smart-phones: Graphics, animations, camera | Outdoor learning: learning through more practical examples shown by mobiles | Learner/ teacher- centered 1:1 | To learn & to teach: Contextualise information; more practical/ experimental | 18 Primary | Formal Urban teachers | Mixed: No as restricted to classroom; yes for potential field trip usage | Learning practice Perceptions & Feasibility |
| Enge 2011 Africa: Tanzania | Smart-phones: Graphics, animations, multimedia TV | Science & Math: phone-content & TV delivery to increase quality of content/more interactive MAML | Teacher- centered 1:1 1:class | To teach: mobile content to make education more interactive & student- centered | 18 Primary & High School | Formal Rural teachers | No: Usage restricted to class- room | Science & Math: Test scores Teaching practice Qualitative |
| Friaz-Martinez 2012 Latin America: Peru | Feature/smart phone: Java-application educational game | Math: math mobile game that teacher creates & administers to | Learner- centered 1:1 | To learn: game- based approach to math learning | 60 Primary | Formal Urban students | Yes: anytime, anywhere; encouraged to use in schools | Math: Test scores |

| | | students | | | | | | |
|--|---|--|---|--|------------------------------------|---|--|--|
| GSMA 2012b Africa (numerous) | Mobile phones (smart/basic) | General perceptions | Learner- centered 1:1 | To learn: mobiles assumed to support education | 23 Primary/H igh School | Mixed Formal /Informal Urban/rural | Not assessed | Perceptions |
| Hassler 2011 Africa Zambia | Tablet; e-Book: wiki readers: content, and software | Teacher development: math & science MAML | Teacher- centered Different models | To teach: mobiles to support teacher to have more participator, interactive and collaborative approach | 6 teachers Primary school | Formal: Mixed Student and teachers | No: usage restricted to classroom | Changes in pedagogy & learning experience: Qualitative |
| He 2008 Asia: India | PicTalk machine: educational software, audio | English (FL): a mobile device with interactive vocab features & quizzes MALL | Learner- centered 1:1 | To learn: game- based | 5317 Primary | Formal: Urban Student | No: usage restricted to classroom | English literacy: test-scores Attendance |
| Hwang 2013 Asia: Taiwan | PDAs: Voice/camera | Outdoor learning: Inquiry-based mobile learning during outdoor visit | Learner- centered 1:1 | To learn: contextualised learning; more practical/ experimental | 26 High School | Formal: Urban Students | Mixed: semi- formal context during field trips | Learning gains: Test scores Perceptions |
| lqubal 2012 Asia: Pakistan | Mobile phones (Basic/Smart) | General perceptions | Learner- centered | To learn: potential of mobile technologies | 250 Tertiary | Formal: Urban students | Not assessed | Perceptions |
| Ismail 2013 Asia: Malaysia | Mobile phones (Basic/Smart) | General perceptions | Teacher- centered | To teach: potential of mobile technologies | 38 Tertiary | Formal: Urban teachers | Not assessed | Perceptions |
| Jantjies 2015 Africa: South Africa | Smart-phone: educational gaming | English (FL)/math: using mobile phone game-based learning MALL/MAMM | Learner- centered 1:1 | To learn: game- based | 90 High- school | Formal: Urban Students | No: usage restricted to classroom | Learning gains: Test scores Perceptions |
| Jere-Folotiya 2014 Africa: Zambia | Feature/smart phone: educational games | Literacy: game- based learning | Learner- centered | To learn: game- based | 573 Primary | Formal: Urban Students | No: usage restricted to classroom | Literacy gains: test scores |

| Kafyulio 2012 Africa: TanzaniaKelbile phones (Basic/Smart)General perceptionsLearner/ teacher- centeredTo learn & teach: teacher- centered85 borntaid of mologiesFormal: teachers & learnersNot assessedPerceptionsKaleebu 2013 Oceania: Papua New GuineaBasic/feature phones: SNSEnglish (FL): lesson plans & content delaviered to teachers via teachers via teacher via teacher via teacher via teacher via teacher via teacher via violavideaEnglish (FL): lesson teacher via teacher via tea | | | | 1:1 | | | | | |
|---|---|---|---|--|---|---|---|-------------------------------------|---|
| Kaleebu 2013 Oceania: Papua New Guinea Basic/feature phones: English (FL): lesson plans & content plans & content basic Meature SMS Teacher- plans & content plans & content basic plans & sontent plans & content basic plans & content basic plans & content to interactivity & quality of lessons softwarefagening: voice/video Teacher- plans & content plans & content basic plans & content to interactivity & quality of lessons softwarefagening: voice/video Teacher- plans & content plans & content to interactivity & quality of lessons plans & content to interactivity & quality of lessons poor - social transformation 2274 Pinnary Plans & formal: Primary Formal: Primary Plans & content to informal: Rural / peni- informal: rural me- plans & content to interactive features & books with audio English ilteracy: teachers via SMS English ilteracy: teachers English ilteracy: teachers English ilteracy: teachers English ilteracy: teachers English ilteracy: teachers English ilteracy: teachers En | Kafyulio 2012 Africa: Tanzania | Mobile phones (Basic/Smart) | General perceptions | Learner/ teacher- centered | To learn & teach: potential of mobile technologies | 85 High School | Formal: Urban Teachers & learners | Not assessed | Perceptions |
| Kim 2009 Latin AmericaPDA (teacher mate): Educational software/gaming; voice/videoLiteracy: mobile to support language learning with interactive features MALLLearner- centeredTo learn: contextualised learning to support poor - social transformation270Informal: mural peri- urban urban & ruralYes: anywhere, anytimePerceptions/ FeasibilityKim 2011 Latin America: MexicoPDA (teacher mate): educational games & books with audioLiteracy: gaming features & audio e- booksLearner- centeredTo learn: contextualised transformationTo learn: contextualised transformationNo: Restricted to class-room useLiteracy gains: Test scoresKim 2012 Asia: IndiaPDA (teacher mate): educational games & books with audioLiteracy: gaming features & audio e- booksLearner- centeredTo learn: Game- based learning to become more interactive & student centered80Formal: muralNo: Restricted to class-room useLiteracy gains: Test scoresKorkmaz 2010 Asia: IndiaBasic phone/smart phone:Math: gaming featureLearner- centeredTo learn: additional learning into informal contexts50Formal: muralVes: anytime, anywhereEnglish literacy: Test scoresKumar 2010 Asia: IndiaSmart phone: featureEnglish: educational gameLearner- centeredTo learn: educational game20Informal: muralYes: anytime, anywhereEnglish literacy: Test scoresKumar 2010 Asia: IndiaSmart phone: feature | Kaleebu 2013 Oceania: Papua New Guinea | Basic/feature phones: SMS | English (FL): lesson plans & content delivered to teachers via SMS MALL | Teacher- centered 1:1 1:class | To teach & learn: lesson plans & content to increase interactivity & quality of lessons | 2274 Primary | Formal: Rural teachers | Yes: anywhere, anytime | English literacy: test-scores Teaching practice |
| Kim 2011 Latin America: MexicoPDA (teacher mate): educational games & books with audioLiteracy: gaming features & audio e- booksLearner- centeredTo learn: Game- based learning to become more interactive & student centered80Formal: Mixed - urban & ruralNo: Restricted to class-room useLiteracy gains: Test scoresKim 2012 Asia: IndiaPDA (teacher mate): educational games & books with audioMath: gaming features & audio e- booksLearner- centeredTo learn: Game- based learning to based learning to based learning to based learning to interactive & student centeredMath gains: Test scoresKim 2012 Asia: IndiaPDA (teacher mate): educational games & books with audioMath: gaming features & audio e- booksLearner- centeredTo learn: Game- based learning to interactive & student centeredMath gains: Test scoresKorkmaz 2010 Asia: TurkeyBasic phone/smart phone: SMS quizzes & remindersEnglish: educational tearLearner- centeredTo learn: additional learning by extending learning into informal contextsSoFormal: urban communityYes: anytime, anywhereEnglish literacy: Test scoresKumar 2010 Asia: IndiaSmart phone: educational gaming featureEnglish: educational gameLearner- centeredTo learn: educational games to extend & improve learning in to informalInformal: centeredYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational gameLearner- <td>Kim 2009 Latin America</td> <td>PDA (teacher mate): Educational software/gaming; voice/video</td> <td>Literacy: mobile to support language learning with interactive features MALL</td> <td>Learner- centered 1:1</td> <td>To learn: contextualised learning to support poor – social transformation</td> <td>270 Informal education program (Children)</td> <td>Informal: Rural/ peri- urban community</td> <td>Yes: anywhere, anytime</td> <td>Perceptions/ Feasibility</td> | Kim 2009 Latin America | PDA (teacher mate): Educational software/gaming; voice/video | Literacy: mobile to support language learning with interactive features MALL | Learner- centered 1:1 | To learn: contextualised learning to support poor – social transformation | 270 Informal education program (Children) | Informal: Rural/ peri- urban community | Yes: anywhere, anytime | Perceptions/ Feasibility |
| Kim 2012 Asia: IndiaPDA (teacher mate): educational games & books with audioMath: gaming features & audio e- booksLearner- centeredTo learn: Game- based learning to become more interactive & student centered40Informal: Rural/ peri- urban communityYes: anytime, anywhereMath gains: Test scoresKorkmaz 2010 Asia: TurkeyBasic phone/smart phone: SMS quizzes & remindersEnglish: educational messages, quizzes, remindersLearner- centeredTo learn: additional learning into informal centered50Formal: Urban studentsYes: anytime, anywhereEnglish literacy: Test scoresKumar 2010 Asia: IndiaSmart phone: educational gaming featureEnglish: educational gameLearner- centeredTo learn: educational games to extend & improve learning in rural areas20Informal: Urban studentsYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone: enteredEnglish: educational tenteredLearner- centeredTo learn: educational games to extend & improve learning in rural areas20Informal: Rural education childrenYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational tenteredLearner- tenteredTo learn: educational games to extend & improve learning in rural areas20Informal: Rural education toildrenYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational tenter | Kim 2011 Latin America: Mexico | PDA (teacher mate): educational games & books with audio | Literacy: gaming features & audio e- books MALL | Learner- centered 1:1 | To learn: Game- based learning to become more interactive & student centered | 80 Primary | Formal: Mixed – urban & rural | No: Restricted to class-room use | Literacy gains: Test scores |
| Korkmaz 2010 Asia: TurkeyBasic phone/smart phone: SMS quizzes & remindersEnglish: educational nessages, quizzes, remindersLearner- centeredTo learn: additional learning by extending learning into informal contexts50Formal: Urban studentsYes: anytime, anywhereEnglish literacy: Test scores PerceptionsKumar 2010 Asia: IndiaSmart phone: educational gaming featureEnglish: educational gameLearner- centeredTo learn: educational games to extend & improve learning in rural areas20Informal: Rural education 10-14Yes: anytime, anywhereEnglish literacy: Test scores PerceptionsKumar 2012Smart phone:English: educational to ne:Learner- centeredTo learn: educational games to extend & improve learning in rural areas20Informal: Rural education 10-14Yes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational tenglish: educational tenglish: educational tenglish: educational tenglish: educational tenglish: educational tenglish: educationalLearner- to learn: educational tenglish: educationalSolutional tenglish literacy:21Formal:No: restricted toEnglish literacy: | Kim 2012 Asia: India | PDA (teacher mate): educational games & books with audio | Math: gaming features & audio e- books MAML | Learner- centered 1:1 | To learn: Game- based learning to become more interactive & student centered | 40 Informal education program (Children) | Informal: Rural/ peri- urban community | Yes: anytime, anywhere | Math gains: Test scores Perceptions |
| Kumar 2010 Asia: IndiaSmart phone: educational gaming featureEnglish: educational gameLearner- centeredTo learn: educational games to extend & improve learning in rural areas20 Informal education thildrenInformal: anywhereYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational Learner-Learner- To learn: educational To learn: educational20 Informal education thildrenInformal: Rural thildrenYes: anytime, anywhereEnglish literacy: Test scoresKumar 2012Smart phone:English: educational Learner-Learner- To learn: educationalZ1Formal:No: restricted toEnglish literacy: Test scores | Korkmaz 2010 Asia: Turkey | Basic phone/smart phone: SMS quizzes & reminders | English: educational messages, quizzes, reminders MALL | Learner- centered 1:1 | To learn: additional learning by extending learning into informal contexts | 50 Tertiary | Formal: Urban students | Yes: anytime, anywhere | English literacy: Test scores Perceptions |
| Kumar 2012 Smart phone: English: educational Learner- To learn: educational 21 Formal: No: restricted to English literacy: | Kumar 2010 Asia: India | Smart phone: educational gaming feature | English: educational game MALL | Learner- centered 1:1 | To learn: educational games to extend & improve learning in rural areas | 20 Informal education 10-14 | Informal: Rural children | Yes: anytime, anywhere | English literacy: Test scores |
| | Kumar 2012 | Smart phone: | English: educational | Learner- | To learn: educational | 21 | Formal: | No: restricted to | English literacy: |

| Asia: India | Educational gaming feature | game, including speech recognition | centered 1:1 | games to extend & improve learning in rural areas | Primary | Rural students | class-room setting | Test scores |
|--|--|---|---|--|----------------------|----------------------------------|--|--|
| Lai 2007 Asia: Taiwan | PDAs: Camera, wikis, internet, note taking | Outdoor learning: embedding mobiles into an experimental learning flow | Learner- centered 1:1 | To learn & access, record information: in context; experimental leaning while in the field | 68 Primary | Formal: Urban students | Mixed: semi- formal context during field trips | Science learning: Test scores Perceptions |
| Leach 2005 Africa: South Africa & Egypt | Netbooks & PDA: far reaching educational support – lexica, camera, e-books, etc. | General educational support for teachers | Teacher- centered 1:1 PDAs 1:2 PCs | To teach: improve teaching in general | 52 Primary | Formal Rural teachers | Yes: anytime, anywhere | Teaching practice: Qualitative Perceptions |
| Liu 2010 Asia: China | Basic/feature phones | General perceptions | Learner- centered | To learn: potential of mobile technologies | 230 Tertiary | Formal: Urban students | Not assessed | Perceptions |
| Liu 2009 Asia: Tawian | PDAs: camera, connectivity, connected with context-aware system | Outdoor learning: immersive ubiquitous learning environments for outdoor teaching | Learner- centered 1:1 | To learn & access information: contextualised, ubiquitous learning environment | 72 High School | Formal: Urban students | Mixed: semi- formal context during field trips | Knowledge gains Perceptions |
| Lui 2008 Asia: Taiwan | Feature/Smart Phone: SMS reminders | General support: SMS reminders for collaborative programme | Learner- centered 1:1 | To administer & connect: SMS to support collaboration | 48 Tertiary | Formal: Urban students | Yes: anytime, anywhere | Knowledge gains |
| Lu 2008 Asia: Taiwan | Feature/Smart Phone: SMS messages | English (FL): SMS messages to learn new words and test knowledge MALI | Learner- centered 1:1 | To access information: additional learning events to support retention | 30 High school | Formal Urban students | Yes: anytime, anywhere* | English gains: Test scores Perceptions |
| Majid 2013 Asia: Malaysia | Feature/Smart phone: internet access | General perceptions | Learner- centered | To learn: potential of mobile technologies | 3 Tertiary | Formal Urban students | Not assessed | Perceptions |
| Makoe 2010 Africa: South Africa | Basic/feature phone: MxIT chats | General distance education support: collaborative online study/chat groups | Learner- centered 1:1 | To learn & connect: Collaborative learning via mXIT to support distance education | 12 Tertiary | Formal: Mixed: Rural/Urban | Yes: anytime, anywhere | Perceptions |

| Masika 2015 Africa: Kenya | Smart-phone; tablets | General perceptions | Learner- | To learn: mobiles | 292 | Formal | Not assessed | Perceptions |
|--------------------------------|---|---|-----------------------------|---|----------------|-----------------------------------|---|-----------------------------------|
| , inioal rionja | | | 1:1 | education | Tertiary | Under- graduate students | | |
| Masperi 2008 Africa: Malawi | PDAs (TeacherMate): | Literacy & English Quizzes, e-books, | Learner- centered | To learn: gaming- features to make | 112 | Formal: | No: restricted to class-room setting | Learning outcomes: Test scores |
| | educational software, audio/video | lesson design MALL | 1:1 | education more interactive | Primary | Urban students | | Perceptions |
| Mruz 2011 Africa: South | PDAs (teacherMate): | English (FL): Literacy & English | Learner- centered | To learn: gaming- features to make | 250 | Formal: | No: restricted to class-room setting | English literacy: Test scores |
| Africa | educational software, audio/video | Quizzes, e-books, lesson design MALL | 1:1 | education more interactive | Primary | Urban students & teachers | | Teaching practice |
| Ongoku 2013 Africa: Kenya | Tablets: educational software & recording | Outdoor learning: to design apps that | Teacher- centered | To teach: Use mobile apps to facilitate | 10 | Formal: | Mixed: semi- formal context | Perceptions |
| | features | allow for a blended learning experience | 1:1 | Blended learning for learners | Primary | Rural teachers | during field trips | |
| Osman 2011 Asia: Malaysia | Basic/feature phone: SMS & online wiki | English (FL): SMS messages with | Learner- centered | To learn & administer: Additional | 61 | Formal | Yes: anytime, anywhere | English literacy: Test scores |
| | | reminders & quizzes MALL | 1:1 | learning through receiving info in informal context | Tertiary | Urban students | | |
| Pamuk 2013 Asia: Turkey | Tablets: e-books | General educational support: textbooks & | Learner- | To learn: General educational support | Provincial | Formal: Mixed – rural | No: restricted to class-room setting | Learning gains: Test scores |
| , iolai rantoj | | content | 1:1 | through tablet devices | High School | Urban students | olabo room oolahiy | |
| Pimmer 2013 Asia: Nepal | Smartphones/ netbooks: general | mHealth : Google, Facebook, | Learner- centered | To learn: general support by mobile | 43 | Formal: | Mixed: use of device in work | Perceptions |
| • | educational support | communication, note taking | 1:1 | technologies | Tertiary | Urban students | setting only | |
| Piper 2015 Africa: Kenya | Tablets & eReaders: New pedagogies + | Reading: Structured, learner-centered | Teacher- centered | To teacher: more learner-centered | 1560 | Formal: | Mixed: use of device in work | Reading gains: Test scores |
| | more access to books | lessons MALL | Learner- centered 1:1 | lesson plans & materials To learn: access to | Primary | Mixed: Urban/rural students | setting only | Cost-effectiveness |

| | | | | more e-books | | | | |
|----------------------------------|---------------------------------------|---|-----------------------|---|-------------------------------|---------------------------------------|---|-----------------------------------|
| Pitchford 2014 Africa: Malawi | Tablets (apple): Euro talk | Math: educational games, quizzes, | Learner- centered | To learn: game- based learning in a | 88 | Formal: | No: restricted to class-room setting | Math gains: Test scores |
| | educational software | lessons, etc. MAML | 1:1 | resource poor context | Primary | Urban students | | |
| Porter 2016 Africa: Ghana, | Feature/smart phone: internet; | General perception | Learner- centered | To access information & | 3085 | Formal | Not assessed | Perceptions |
| Malawi and South Africa | camera & video | | 1:1 | communicate: potential of mobile technologies | Primary & high school | Mixed: Urban/rural students | | |
| Ramos 2009 Asia: Mongolia | Basic/feature phone: SMS | English & Math: SMS quizzes and | Learner- centered | To learn & administer: Additional | 52/73 | Informal: | Yes: anytime, anywhere | English & math gains |
| & Philippines | | reminders MALL/MAML | 1:1 | learning through receiving info in informal context | Adult education program | Mixed: Urban/rural students | | Test scores |
| Roberts 2011 Africa: South | Basic/feature phone: SMS via MxIT | Math: Quizzes & game based | Learner- centered | To learn: Collaborative & game | 512 | Formal: | Yes: anytime, anywhere | Math gains: Self reports |
| Africa | | activities over chat platform MAML | 1:1 | make math more relevant & fun | Hign school | Mixed: Urban/rural students | | Perceptions |
| Rosas 2002 Latin America: | Nintendo device: gaming features | Math & Literacy: educational game to | Learner- centered | To learn: game- based learning to | 1274 | Formal: Mixed: | No: restricted to class-room setting | Learning gains: Test scores |
| Chile | | support interaction MALL/MAML | 1:1 | support disadvantaged learners | Primary | Urban/rural Students & teachers | | Teaching practice: Observation |
| Sahni 2008 Asia: India | DVD&TV: multi- media video ability | English: model lessons of peer- | Teacher- centered | To teach: peer-model lessons to support | 54 | Formal: | No: restricted to class-room setting | English gains: Test scores |
| | | teachers streamed to more classes via video MALL | 1:1 1:class | pedagogic innovation | Primary | Mixed: Urban/rural teachers | | Teaching practice: Observation |
| Shraim 2014 Asia: Palestine | e-Reader (kindle): e-books | English: e-books to make for more | Learner- centered: | To learn: assumed benefits of e-readers | 114 | Formal: | No: restricted to class-room setting | English gains: test scores |
| | | interactive & authentic learning experience MALL | 1:1 | | Primary | Urban students | | |

_

| Suki 2011 Asia: Malaysia | Basic/feature phones | General perceptions | Learner- centered: 1:1 | To learn: potential of mobile technologies | 20 Tertiary | Formal: Urban students | Not assessed | Perceptions |
|--|--|--|---|--|---|--|--|---|
| Sung 2013 Asia: Taiwan | PDA: audio/camera, internet note-taking | Outdoor learning: collaborative & contextual affordance of PDA | Learner- centered: 1:1 | To learn: collaborative & in- context learning experience | 53 Primary | Formal: Urban students | Mixed: semi- formal context during field trips | Learning gains: Test scores |
| Taleb 2012 Asia: Iran | Basic/feature phones | General perceptions | Learner- centered: 1:1 | To learn: potential of mobile technologies | 289 Tertiary | Formal: Urban students | Not assessed | Perceptions |
| Turtianinen 2012 Africa: South Africa | Smart phones: UFraction mobile game | Math: an educational game with interactive & relevant features MAML | Learner- centered: 1:class | To learn: game- based interactive & relevant learning experience. | 34 High school | Formal: Urban students | No: restricted to class-room setting | Perceptions |
| UNESCO 2014 Africa (various) | Smartphone: World reader app | Literacy: access to a range of e-books MALL | Learner- centered 1:1 | To access: more access to reading materials | Online survey Youth | Informal: Urban | Yes: anytime, anywhere | Perceptions |
| USAID 2010 Africa: Somalia | Radio: Interactive Radio programme | Literacy, Numeracy & Teacher development: interactive lesson transmitted via radio | Teacher/ learner- centered 1:class | To learn: more access to education & more interactive lessons | 122 Primary | Formal Rural teachers | No: restricted to class-room setting | Learning gains: Test scores |
| Valderama 2010 Latin America: Panama | Basic/feature phone; Netbooks | General perceptions | Learner/ teacher- centered: 1:1 | To learn & teach: potential of mobile technologies | 435 High school | Formal Mixed: teachers & students | Not assessed | Perceptions |
| Velghe 2014 Africa: South Africa | Basic/feature phone: SMS; SNs | Literacy: sending messages and becoming phone- literate MALL | Learner- centered 1:1 | To learn & connect: learning extended into informal contexts to increase learning events | 3 (Adults) Informal Education Program | Informal: Women's groups | Yes: anytime, anywhere | Perceptions Empowerment: Ethnography NOTE: No pedagogies at all |

| Voigt 2014 Africa: South | Tablets: mobimath software for learners | Math: educational software with | Learner/ teacher- | To learn: game- based all-round math | 31 | Formal: | No: restricted to class-room setting | Perceptions |
|------------------------------|---|---|----------------------|---|----------|---------------|---|--------------------------------|
| Africa | & parents &teachers | learning support & | centered: | software; more | High | Urban | - | |
| | | parental M&E | | practical learning | school | teachers & | | |
| | | MAML | 1:1 | | | students | | |
| Walton 2009 Africa: South | Smart-phones: | Literacy: allowing learners to self- | Learner- | To learn: using social networks for | 111 | Formal: | Yes: anytime, anywhere | Literacy gains: |
| Africa | media. SMS | create a mobile | | teenagers to create a | Hiah | Urban | | indigenous |
| | | story | 1:1 | short story | school | students | | languages |
| Wennersten 2012 | Smart-phones: Graphics, | English (FL) & Science: phone- | Teacher- centered | To teach: mobile content to make | 1562 | Formal | No: restricted to class-room setting | Learning gains: Test scores |
| Asia: India | animations, | content & TV | | education more | High | Mixed: | | |
| | multimedia | delivery to increase | 1:1 | interactive & student- | school | Teachers & | | |
| | | quality of | 1:class | centered | | students | | |
| | IV | content/more | | | | | | |
| | | | | | | | | |
| | | | | | | | | |
| Witt 2016 | Tablets: Internet, | mHealth: | Learner- | To learn: additional | 82 | Formal: | Yes: anytime, | Perceptions |
| Africa: | content; apps | educational software | centered | and contextualised | | | anywhere | |
| Botswana | | & access to medical | | access to information | Tertiary | Urban | | |
| | | information | 1:1 | | | under- | | |
| | | | | | | graduate | | |
| Wordreader | Tablets: e-books & | English (EL); ebooks | l earner- | To learn: more | 374 | Formal: | Yes: anytime. | Learning gains: |
| 2012 | working books | & educational | centered | resources for | | | anywhere (with | Test scores |
| Africa: Ghana | 0 | materials on tablets | | students to gain | Mixed | Rural | temporary | |
| | | to give more | 1:1 | literacy | Primary | students | restrictions) | |
| | | educational access | | exposure/learning | High | | | |
| | | to learners | | events due to tablets | school | | | |
| Worldreader | Tablets: e-books & | English (FL): ebooks | l earner- | To learn: more | | Informal: | No: restricted to | Perceptions and |
| 2015 | working books | & educational | centered | resources for | | internal. | library facilities | adoption |
| Africa: Kenva | | materials on tablets | | students to gain | | Mixed: | ····· | |
| , | | to give more | Provision | literacy | | Rural/ urban | | |
| | | educational access | to | exposure/learning | | Library users | | |
| | | to learners | libraries | events due to tablets | | - | | |
| Wu 2011 | PDAs: GPS, | mHealth: context- | Learner- | To learn & access | 48 | Formal: | No: restricted to | Learning gains: |
| Asia: Taiwan | internets, graphics | aware mobile | centered | info: more practical | | | class-room setting | Test scores |
| | | learning system to | | and experiential | Tertiary | Urban | | |

| | diagnose minor diseases | 1:1 | learning to increase relevance of education | | students | | Perceptions |
|--|--|---|--|---|--|--|--|
| Smartphone: QR scanner, internet, graphics | Natural Science: digital concept map to enhance book | Learner- centered | To learn & access info: Digital concepts maps to | 59 Primarv | Formal: Urban | No: restricted to class-room setting | Learning gains: Test scores |
| | reading through diagrams etc | 1:1 | contextualise & visualize learning concepts | | students | | Perceptions |
| Basic/smart phone | General perception | Learner- centered | To access information & | 36 | Formal: | Mixed: use not permitted at | Perceptions |
| | | 1:1 | communicate: mobile tech has potential for girls actively excluded from education | Hign school | Urban students | SCHOOL | |
| PDAs: educational game, guizzes | Math & literacy : a collaborative | Learner- centered | To learn & connect: collaborative & | 48 | Formal: | No: restricted to class-room setting | Math & literacy gains: |
| | computer-assisted learning game | 1:1 | game-based learning facilitated by mobiles | Primary | Urban students | C C | Test scores |
| | MALL/MAML | | | | | | Perceptions |
| PDAs: educational game, quizzes | Literacy : a collaborative | Learner- centered | To learn & connect: constructivist | 24 | Formal: | No: restricted to class-room setting | Literacy gains: Test scores |
| | computer-assisted learning game MALL | 1:1 | learning facilitated by collaborative mtech learning event | Primary | Urban students | | |
| PDAs: educational game, quizzes | Math: a collaborative computer-assisted | Learner- centered | To learn & connect: collaborative & | 24 | Formal: | No: restricted to class-room setting | Math gains: Test scores |
| | math learning game MALL | 1:1 | game-based learning facilitated by mobiles | Primary | Urban students | | Perceptions |
| | Smartphone: QR scanner, internet, graphics Basic/smart phone Basic/smart phone PDAs: educational game, quizzes PDAs: educational game, quizzes | diagnose minor diseasesSmartphone: QR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcBasic/smart phoneGeneral perceptionPDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning game MALL/MAMLPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALL | diagnose minor diseases1:1Smartphone: QR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcLearner- centeredBasic/smart phoneGeneral perceptionLearner- centeredBasic/smart phoneGeneral perceptionLearner- centeredPDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning game MALL/MAMLLearner- centeredPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLLearner- centeredPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLLearner- centeredPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLLearner- centeredPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLLearner- centeredPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLLearner- centeredPDAs: educational game, quizzesMath: a collaborative computer-assisted math learning game MALLLearner- centered | diagnose minor diseases1:1learning to increase relevance of educationSmartphone: QR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcLearner- centeredTo learn & access info: Digital concepts maps to contextualise & visualize learning conceptsBasic/smart phoneGeneral perceptionLearner- centeredTo access information & contextualise & visualize learning conceptsPDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning game MALL/MAMLLearner- centeredTo learn & connect: collaborative & game-based learning facilitated by mobilesPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: collaborative & game-based learning facilitated by mobilesPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: constructivist learning facilitated by collaborative mtech learning eventPDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: constructivist learning facilitated by collaborative & game-based learning facilitated by collaborative & game-based learning facilitated by mobiles | diagnose minor diseases1:1learning to increase relevance of educationSmartphone: QR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcLearner- centeredTo learn & access info: Digital concepts maps to contextualise & visualize learning concepts59Basic/smart phoneGeneral perceptionLearner- centeredTo access information & communicate: mobile tech has potential for girls actively excluded from education36PDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning game MALL/MAMLLearner- centeredTo learn & connect: collaborative & game-based learning facilitated by mobiles48PDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: constructivist learning facilitated by collaborative tech learning revent24PDAs: educational game, quizzesMath: a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: constructivist learning facilitated by collaborative metch learning event24PDAs: educational game, quizzesMath: a collaborative computer-assisted math learning game MALLTo learn & connect: constructivist learning facilitated by collaborative & game-based learning facilitated by mobiles24 | diagnose minor diseases1:1learning to increase relevance of educationstudentsSmartphone: QR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcLearner- centeredTo learn & access info: Digital concepts maps to contextualise & visualize learning concepts59Formal: studentsBasic/smart phoneGeneral perception diagrams etcLearner- centeredTo access information & contextualise during concepts36Formal: studentsPDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning gameLearner- centeredTo learn & connect: collaborative & game-based learning facilitated by mobiles48Formal: PrimaryPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted learning gameLearner- centeredTo learn & connect: constructivist learning facilitated by collaborative metch learning facilitated by collaborative metch learning facilitated by collaborative & game-based learning facilitated by collaborative metch learning facilitated by mobiles24Formal: PrimaryPDAs: educational game, quizzesMAtLMath: a collaborative computer-assisted math learning game tiltLearner- | Image: studentsdiagnose minor diseases1:1learning to increase relevance of educationstudentsSmartphone: OR scanner, internet, graphicsNatural Science: digital concept map to enhance book reading through diagrams etcLearner- centeredTo learn & access info: Digital concepts maps to concepts59Formal: class-room settingBasic/smart phoneGeneral perception diagrams etcLearner- centeredTo access contextualise & visualize learning concepts36Formal: permitted at schoolPDAs: educational game, quizzesMath & literacy : a collaborative computer-assisted learning game MALLLearner- centeredTo learn & connect: constructive & game-based learning facilitated by mobiles48Formal: tubentsNo: restricted to class-room settingPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted math learning game MALLLearner- centeredTo learn & connect: constructivist constructivist constructivist collaborative & game-based learning game, quizzesNo: restricted to class-room settingPDAs: educational game, quizzesLiteracy : a collaborative computer-assisted math learning game MALLLearner- centeredTo learn & connect: constructivist constructivist game-based learning game-based learning facilitated by mobiles24Formal: constructivist |

Appendix 5.3: Detailed overview of studies included in the meta-analysis

| Study | Design | Intervention details | | Population | Outcome | Findings & Effect size |
|---------------------|---|---|--|--|---|---|
| | Risk of n Follow-up Bias | Programme & technology | Context | Age/schoolin g Gender | Outcome (Measures) | ES (SE); [Confidence interval] |
| Aker 2010 Niger | Low 5014 12 months Randomised Controlled Trial (cluster adjusted) | SMS-based adult literacy programme Basic/feature phone | Rural Anywhere, anytime | Adults; NGO- adult literacy programme Mixed | Literacy (test scores) Numeracy (test scores) Student effort (attendance) Teacher effort (attendance) Student motivation (paid hotline calls) Cost-effectiveness (cost per student attaining Level 1 proficiency) | +0.18 (0.03); [0.13, 0.24] +0.19 (0.03); [0.13, 0.25] No change No change Increased interest in education For \$6.50 per student, 4% more students reach Level 1 |
| Aker 2015 Niger | Low 1926 6 months Randomised Controlled Trial | Mobile teacher monitoring Basic/feature phone | Rural Monitoring conducted in formal and informal settings | Adults; NGO- adult literacy programme Mixed | Literacy (test scores) Numeracy (test scores) Self-esteem Self-efficacy Teacher attendance Teacher motivation Student attendance Student motivation | +0.15 (0.03); [0.07, 0.23] +0.30 (0.04); [0.22, 0.38] 2% Decrease in self-esteem 3% Decrease in self-efficacy 1.27 extra days over 6 months Self-reported motivation increase No change Self-reported motivation increase For \$6.50 per village, 0.2 SD in learning |
| Chen 2010 Taiwan | Moderate 36 2 weeks Quasi-experimental (random assignment) | A personalised context-aware ubiquitous learning system (PCULS) PDAs with GPS capability | Urban Anywhere, anytime | High school Mixed | English vocabulary (test scores) User satisfaction | +0.22 (0.33); [-0.43, 0.88] Not statically significant Not relevant for this module |
| Cole 2012 India | Low 766 4 months | SMS-based agaric extension programme | Rural Anywhere, anytime | Adults; NGO- agric extension programme | Agricultural knowledge | +0.04 (0.07); [-0.10, 0.19] Not statistically significant Learning outcomes are greater for better-educated |

| | | Basic/feature phone | | | | famers |
|---|---|---|--|-----------------------------|--|--|
| | Randomised Controlled Trial (cluster adjusted) | | | Mixed | Cost-effectiveness (comparison of input costs) | Programme cost: \$ 1.13 per famer compared to \$8.50 for traditional extension service |
| Cilliers (2014) Uganda | High 180 schools 14 months | Mobile monitoring of teacher attendance | Urban/ Rural | Primary/ High Schools | Teacher attendance Quality of reporting | 11% increase in attendance Large underreporting remains |
| ogundu | Randomized controlled trial | SMS-based | Formal: Monitoring took place during school hours | Mixed | | |
| He 2008 India | Low 5317 10 months Randomised Controlled Trial | PicTalk English & math learning gaming device Educational software/gaming device | Urban Formal: Usage restricted to classroom | Primary school; Mixed | English (test scores) Math (test scores) Student attendance (Observation) Cost-effectiveness (input cost per tenth of SD) | +0.35 SD gain (Experiment 1) +0.36 SD gain (Experiment 2) +0.052 SD gains (Year 1) Not statistically significant +0.36 SD gains (Year 2) Teachers included in implementation in year 2 – found more effective No change Cost-effective compared to other education interventions |
| Kaleebu 2013 Papua New Guinea | Low 2274 7 months Quasi-experimental (random assignment; matched) | SMS-based, in- practice teacher development programme Basic/feature phone | Rural Anywhere, anytime | Primary school Mixed | English (test scores) 1 Oral fluency* 2 Invented words 3 Sight word fluency 4 Decodable Words Teaching practice (self reports & observation) Cost-effectiveness (input cost per student) | +0.29 (0.05); [0.20, 0.38] +0.21 (0.05); [0.12, 0.30] +0.14 (0.04); [0.06, 0.23] +0.15 (0.04) [0.07, 0.24] Teacher use more interactive teaching approaches Low input cost, but no comparison made |
| Kumar 2012 India | High 21 2 weeks Quasi-experimental | Literacy development Using Speech Recognition- | Rural Formal: Usage restricted | Primary school Mixed | English Design 1* (test scores) English Design 2 (test scores) | +1.31 (0.49); [0.35, 2.28] +0.73 (0.45); [-0.16, 1.62] Not statically significant |

| Lai 2007 Taiwan | Moderate 68 1 week Quasi-experimental | Supported Games on Mobile Devices Smart phone Experimental learning through embedding learners via tech into contexts PDAs | to classroom Urban Mixed: semi- formal use during field trips | Primary school Mixed | Knowledge gains (test scores) Knowledge acquisition (Text mining) Knowledge created (Text mining) User satisfaction | +0.46 (0.25); [-0.03, 1.00] Not statically significant Critical Risk of Bias - instrument not adequate Critical Risk of Bias - instrument not adequate Not relevant for this module |
|--------------------|---|---|--|----------------------------|---|---|
| Liu 2009 Tawian | Moderate 72 1 week Quasi-experimental | Outdoor Natural Science Learning with an RFID- Supported Immersive Ubiquitous Learning Environment PDAs | Urban Mixed: semi- formal use during field trips | High school Mixed | Knowledge gains (test scores) User satisfaction | +1.61 (0.27); [1.08, 2.14] Not relevant for this module |
| Liu 2008 Taiwan | High 48 2 weeks Quasi-experimental | Web-based collaborative learning scheme based on activity awareness carried out through mobile phones Feature/Smart Phone | Urban Anywhere, anytime | Tertiary Mixed | Knowledge gains (test scores) | +0.54 (0.30); [-0.05, 1.12] Not statically significant |
| Lu 2008 Taiwan | Moderate 30 2 weeks Quasi-experimental (random assignment) | SMS-based English vocabulary programme Basic/feature phone | Urban Anywhere, anytime | High school Mixed | English vocabulary (test scores) User satisfaction | +0.68 (0.38); [-0.06, 1.42] Not statically significant Not relevant for this module |

| Piper 2015 Kenya | Moderate 1560 10 months Randomised Controlled Trial (cluster adjusted) | Tablet based PRIMR interventions + e-Reader provision Tablets/e-Readers | Urban/ Rural Formal: Use during work hours at school | Primary Mixed | Tutor tablets: reading Teacher tablets: reading e-Reader: reading Cost-effectiveness Cost of e-reader per student: \$40 Cost of teacher tablet per student: \$3 Cost of tutor tablet per student: \$10 | +0.44 (0.06); [0.32, 0.56] +0.47 (0.06); [0.35, 0.59] +0.35 (0.04); [0.27, 0.43] 6 times less cost-effective than no ICT \$100 leads to 0.5 students reaching full reading fluency Cost-effective compared to no ICT but not to tutor tablet \$100 leads to 4.3 students reaching full reading fluency 2 times more cost-effective than no ICT and teacher tablet; 10 times to e-readers \$100 leads to 7.8 students reaching full reading fluency |
|-----------------------------|---|---|--|----------------------------|---|---|
| Pitchford 2014 Malawi | Low 88 2 months Randomised Controlled Trial | Tablet-based math educational gaming programme Tablets (iPads) | Urban Formal: Usage restricted to classroom | Primary school Mixed | Math (test scores) 1 Mathematical concepts 2 Math curriculum knowledge* 3 Math curriculum knowledge generalisation Empowerment (life ambitions & attitudes) | +0.16 (0.21); [-0.26, 0.58] +1.68 (0.25); [1.19, 2.18] +0.81 (0.22); [0.37, 1.25] Critical Risk of Bias - instrument not adequate |
| Rosas 2002 Chile | High 1274 3 months Quasi-experimental | Game-based math & literacy learning on Nintendo Gaming device/ educational software | Rural/ Urban Formal: Usage restricted to classroom | Primary school Mixed | Literacy (test scores) 1 Reading comprehension 2 Spelling Math (test scores) Motivation to use video Teacher practice (Observation & Interviews) | Statistical significant impact. ANOVA estimate +0.06 (0.21) Statistical significant impact ANOVA estimate -0.27 (0.18) Statistical significant impact ANOVA estimate +0.51 (0.43) Critical Risk of Bias - instrument not adequate Critical Risk of Bias - not in both experimental groups |

| Sung 2013 Taiwan | Moderate 53 2 weeks | Mobile learning system based on a collaborative problem-posing strategy | Urban Mixed: semi- formal use | Primary school Mixed | Knowledge gains (test scores) User satisfaction | +0.62 (0.28); [0.06, 1.17] Not relevant for this module |
|----------------------|--|---|---|----------------------------|---|--|
| | | PDAs | trips | | | |
| Wu 2011 Taiwan | High 48 Instantly | Context-aware mobile learning | Urban | Tertiary | Knowledge gains (test scores) | +1.04 (0.31); [0.44, 1.65] |
| | Quasi-experimental | system for nurse students | Formal: Usage restricted to classroom | Mixed | Cognitive load Learner perception User satisfaction | +0.58 (0.30); [-0.00, 1.16] Critical Risk of bias: Leading questions Not relevant for this module |
| Yang 2013 Taiwan | High 59 1 week | concept map- oriented ubiquitous | Urban | Primary school | Knowledge gains (test scores) | +0.45 (0.26); [-0.07, 0.96] Not statistically significant |
| | Quasi-experimental | learning approach for supporting printed science book reading | Formal: Usage restricted to classroom | Mixed | User satisfaction | Not relevant for this module |
| 7 11 0004 | | | | | | |
| Chile | Algn 48 2 months Quasi-experimental | A Mobile Computer Supported Collaborative | Urban Formal: | schools | Math (test scores) | +0.74 (0.32); [0.12, 1.37] +0.51 (0.28); [-0.04, 1.05] Not statistically significant |
| | | Learning math & literacy game | Usage restricted to | Mixed | Educational events (Observation) | Critical Risk of Bias - not in both experimental groups |
| | | FDAS | Classicon | | | |
| Zurita 2004 Chile | High 24 2 months | A Mobile Computer Supported | Urban | Primary schools | Literacy (test scores) | +0.91 (0.43); [0.06, 1.76] |
| | Quasi-experimental | Collaborative Learning math & literacy game | Formal: Usage restricted to | Mixed | User satisfaction | Not relevant for this module |
| | | PDAs | classroom | | | |

_

Following the risk of bias assessment, I calculated 29 effect sizes of the impact of mobile learning programmes on educational outcomes in LMICs. As explained in chapter 4, section 4.3.4, these effect sizes are expressed as standardised mean differences. To ensure the comparability of the different measurement scales of mobile learning's impact, each individual impact measure is standardised and presented in terms of standard deviations. This allows for a meaningful comparison of the direction and magnitude of impact across all 16 programmes. The 29 effect sizes were derived from 17 studies only. Two studies, He and colleagues (2008), Rosas and peers (2002) applied a regression analysis and only reported regression, or the sample standard deviation, or the treatment and control standard deviations. I was since unable to calculate SDM for these two studies, but integrate their findings into the analysis narratively where appropriate. Cilliers and peers (2014) did not report any statically information apart from percentage change between experimental groups and I was therefore unable to calculate effect sizes for this study.

I did not calculate any synthetic effect sizes in which I would have statistically pooled multiple treatment or control groups into a single treatment or control. A number of studies featured multiple control or treatment arms (Kumar 2012; Pitchford 2014; Sung 2012; Yang 2013: Piper 2014; Cilliers 2014). Yet, in each case the groups were not regarded as homogenous enough in order to combine their results. I since used the results of the strongest treatment and the plainest control group respectively. Similarly, I did not combine different treatment effects measured at different periods of follow-up. The longest period of follow-up was assumed as the most reliable outcomes indicator and used for effect size calculation. This rule applied to only a single study (Aker 2012). Two studies applied multiple outcome measures to assess the same effect (Kaleebu et al 2013; Pitchford et al 2014). For example, Kaleebu and colleagues used four different indicators of literacy. In both cases, we decided on the outcome measure most closely related to the measured phenomenon. This translated into the usage of oral fluency (Kaleebu 2013) and mathematical curriculum knowledge (Pitchford 2014) as relevant outcomes measures. Lastly, I adjusted standard errors and sample sizes from two cluster-randomisation trials as suggested by Higgins and Green (2011).

I calculated effect sizes for a number of different outcome constructs. All effect sizes were based on the means and standard deviations of test scores of two experimental groups. Test scores assessed knowledge gains in six pedagogical mobile learning experiments; learners' mathematical proficiency in five studies; literacy acquisition of participants in four mobile learning programmes; the effectiveness of mobile-assisted English as a foreign language learning in four studies; and lastly, in one study, test scores assessed changes in agricultural knowledge. The calculated effect sizes since all measured the impact of mobile learning programmes on learning outcomes in developing countries.

Appendix 5.4: More detailed analysis of mobile learning pedagogies

Mobile learning pedagogy as a moderator

The last potential moderator I analysed referred to the specific pedagogies which mobile devices were assumed to produce. Of the different underlying pedagogies described in section 5.1.3, only three types were prevailing in the included studies for meta-analysis (game-based n=4; context-aware/experiential n=7; collaborative n=4). First, I found that pedagogy served as an effective moderating variable that highly reduced heterogeneity across the studies combined in each sub-group. Each of the three pedagogic sub-groups presents a homogenous sample (see Table 5.4) and I since describe each sub-group in more detail before formally comparing their effect sizes.

Collaborative learning was a pedagogic approach in four mobile learning programmes implemented in diverse settings. Two programmes were conducted in urban Chile using a learning game in which learners had to solve a collaborative task (Zurita & Nussbaum 2004a; Zurita & Nussbaum 2004b). Learners were allocated into groups and each learner in the group had to contribute to the collaborative learning task. The mobile technology assumed a guiding role ensuring each player contributed equally to the group task as it was found that during a traditional collaborative learning game individual learners dominated group work. The mobile technology served as a mechanism for each learner in the group to express her consent with the group's solution before engaging in the next task. Both studies found this approach to be effective in improving true collaboration (i.e. involving all learners in the group task) and improving learning outcomes, arguably through the inclusion of all learners in the group work.



Figure: Effectiveness of collaborative mobile learning approaches Heterogeneity: Q = 0.581; df = 3; p = 0.901; *I*-squared = 0%; tau-squared = 0

Two studies were conducted in urban Taiwan. Liu and colleagues (2008) added a SMS module to an existing online collaborative learning task. Learner groups had to collaborate in a competition with other groups to create a Wiki on a given topic. Liu and peers assessed whether using SMS technology to inform group members about each other's online activities would motivate them to improve their own contribution to the Wiki. This information was delivered via SMS and each learner was informed if a group member had updated the Wiki or if a competing group had successfully completed a task on their Wiki. The experiment found positive results on learners' activities and subsequent grading of the Wiki when informed about each other's progress through SMS. Due to the small sample size, however, this finding was not statistically significant. The combined meta-analysis was then able to

show that despite the effect's statistical insignificance it seemed, nevertheless, to present an adequate reflection of the true impact of the mobile learning module.

Sung and colleagues (2013) assessed the effectiveness of mobile collaborative learning during a local heritage project. Learners were equipped with PDAs during the visit of a local temple and guided by the device to explore and learn about the connection of the temple to their heritage. The device produced animations and extra information based on a GPS and QR-code system to determine the learner's position. The experiment was delivered in a traditional mobile learning approach in which learners individually explored the temple using their device; and in a collaborative mobile learning approach, in which learners collaboratively explored the temple as each learner's device only had access to a limited set of information. Both mobile learning approaches were found to be more effective than a conventional learning approach (guided tour through the temple by a teacher). The collaborative approach in particular had the largest impact on learning outcomes.

Game-based mobile learning presented the programme approach in four mobile learning interventions. In the above-mentioned studies from Chile (Zurita & Nussbaum 2004a; Zurita & Nussbaum 2004b), the educational game was paired with a collaborative learning approach. The learning game was designed on a Nintendo device and featured a number of syllabi that learners had to group together in order to form words. The device facilitated the correct matching of syllabi and ensured instant feedback and hints to learners. The evaluation found that children learning with the mobile devices had significantly larger learning gains and language acquisition than learners in the control group.



Figure: Effectiveness of game-based mobile learning programmes Heterogeneity: Q = 6.12; df = 3; p = 0.106; I-squared = 51%; tau-squared = 0.13.

Kumar and colleagues (2012) used a related programme approach in rural India. The research team designed an English learning game based on PDA speech-recognition. Children in rural schools were exposed in the game to familiar situations, e.g. purchasing groceries at the local market, and had to correctly identify and pronounce English words in order to proceed in the game. Learning English with the mobile technology increased posttest English scores significantly.

Lastly, Pitchford and peers (2014) applied a math learning application delivered on iPads to learners in urban Malawi. The application presented a sophisticated learning game with numerous levels and variations. It was found to improve learning outcomes by 1.20 standard deviations (0.70-1.70). This effect size was surprisingly large and would have translated into a difference in success rates of 52.2 per cent—or more simply put: almost every learner using the tablets will achieve a higher test score in math than learners in the control group.

Despite this unconventional effect, the study was based on a randomised-controlled design judged of low risk of bias. The applied mobile application has further been found effective in a HIC setting in the United Kingdom (Guardian 2014). However, I caution to assume that such a large effect size can be replicated across contexts and our understanding of the programme would gain from a replication of the study and application of the intervention in additional schools.

Context-aware mobile learning programmes were evaluated by seven studies. Again, these studies presented a homogenous set and each exploited the ability of mobile technologies to facilitate learning across contexts. Context-aware mobile learning programmes applied the most sophisticated interplay of mobile technologies' particular affordances and educational approaches. Chen and colleagues 2010, for example, designed a personalised context-aware ubiquitous learning system to support effective English vocabulary learning. Using PDAs with GPS capabilities, they determined learners' positions and current social context (e.g. at the train station). The mobile system then supplied the learners via the PDAs with context-relevant English vocabulary and exercises. This situated learning approach was found effective to improve English literacies among learners. This applied in respect to two control groups—one using traditional mobile learning. Traditional mobile learning was similarly identified as superior to classroom learning. Alas, due to a small sample size the primary findings are not statistically significant—again, a shortcoming the meta-analysis was able to rectify.



Figure: Effectiveness of context-aware/experiential mobile learning programmes *Heterogeneity:* Q = 1.18; df = 5; p = 0.947; *I*-squared = 0%; tau-squared = 0.

A direct combination of conventional learning practice and mobile learning was investigated by Yang and colleagues (2013). The researchers complemented traditional science textbooks with digital concept maps of complex systems such as the solar systems. Relevant sections in the textbook were linked via QR codes and PDAs to digital concept maps. These concepts maps were interactive and learners could since experience and investigate the solar system (e.g. move planets around, experience a 3D vision) more authentically. This mobile learning approach was found effective in improving learning outcomes, albeit not at a statistical significant level.

Each of the seven context-aware mobile learning programmes was evaluated on a small scale. This came as a result to the high level of control the sophisticated mobile learning

systems required. For example, the studies using GPS technology to determine learners' context needed to keep the GPS maps up-to date. Studies assessing context-aware mobile learning in informal contexts (e.g. cultural visits) further had to facilitate these visits and synchronize the learning content. As a result, this set of studies could best be described as feasibility experiments, which aimed to generate proof of concept rather than statistical significance. This conception explained that five out of seven studies reported positive but statically non-significant findings. Meta-analysis, as a tool designed for such cases, was able to show that the true effect of studies in fact was robust and statistically significant.

Appendix 6.1: Overview of mechanism and context themes

| Context themes | Mechanism themes |
|--|--|
| 1. Mobile devices' particular affordances nurture a general | 1. The usage of mobile technologies in LMICs can change |
| positive perception of mobile technologies among | the way education is practiced, leading to a number of |
| educational stakeholders in LMICs. | more learner-centered educational approaches. |
| [Positive perception] | [Pedagogy and practice] |
| 1.1 Mobile devices' ubiquity supports the familiarity of the | 1.1 The affordances of mobiles (connectivity, information sharing |
| technology among educators, students, and parents. | & collection, visualisation, gamification) can support a more |
| | interactive and collaborative educational approach. |
| 1.2 Mobile devices conveniently blend in the daily routines and | 1.2 The affordances of mobiles (portability, connectivity, multi- |
| existing practices of educators and students in LMICs. | media content; Camera, GPS) can support a more authentic |
| | educational approach by facilitating contextualisation, real life |
| | situated learning, inquiry-based learning, and real life application |
| | of learning. However, the extension of learning to informal |
| | contexts often was limited due to contexts. |
| 1.3 Mobile devices ability to produce new and support existing | 1.3 The affordances of mobiles (access to information; learning |
| forms of communication is a key feature of mobiles' attraction | software; data collection and analysis; multi-media and multi- |
| among educators and students in LMICs. | format content) can support more personalised and targeted |
| | educational approaches. An emerging theme is a small number |
| | of more recent studies (n=3) highlights the ability of mobiles to |
| | track learners' progress as well as the mobile-enabled matching |
| | of content levels and students' proficiency to support learning. |
| 1.4 The affordability of mobile devices and services shapes the | 1.4 A key mechanism through which mobiles influence |
| students in LMICs | information |
| | 1.5 Mobile technologies increase the range of educational tools |
| | thereby increasing variety in teaching and learning activities |
| | |
| | |
| 2. Mobile devices are technically able to serve as | 2. The usage of mobile technologies in LMICs can change |
| educational tools in LMICs. [Feasibility] | the way education is experienced by learners facilitating a more engaging, relevant, and open educational experience. |
|--|---|
| 2.1 Educators and students rapidly develop device literacies. (ease of use) | 2.1 Students perceive mobile technologies to support a more engaging and enjoyable educational experience. |
| 2.2 Technical queries relate to breakage and size of the device screen, battery life, and memory capacity. 2.3 Interface design is central to support the use of mobiles including features that allow for the personalisation of devices. | 2.2 Students perceive technologies to allow for a more personalised and independent educational experience. 2.3 Students perceive technologies to enhance the relevance of their educational experience by allowing for a greater variety and more up-to-date content; more control over the learning process; ability to connect and debate content with peers. 2.4 Students perceive mobile learning to facilitate a more open educational experience by supporting more participation and interaction in lessons; more self-direction and control over the learning process; and opportunities to share and debate content/information with peers. |
| While mobile technology as an educational tool is acceptable to teachers, learners, and parents in LMICs, supporting factors have to be in place to nurture its adoption as an educational tool. [Factors of adoption] | Mobile technologies can support pedagogical and non- pedagogical changes in teaching. [Teaching practice] |
| 3.1 User training is an essential input and requisite for the responsible and adequate use of mobile devices' applications for learners in LMICs (safety, instructions) (partly to reassure communities and partly to indicate to learners the changed function of the phone). Teacher training , too, is an essential input and requisite for the use of mobile devices in LMICs. | 3.1 Change towards a more learner-centered approach, i.e. less frontal delivery of content/lecture style and more learner-led activities, more interaction (e.g. asking questions/debate), wider range of activities. 3.2 Increased subject knowledge through access to more information, greater variety of content and greater collaboration and peer-support. |
| 3.2 The introduction of mobile devices and mobile content needs to be in line with and integrated in the existing curriculum . | 3.3 More opportunities for collaboration, peer-support, and social learning in general (linked to content, pedagogy, and administration) through connectivity, communication, and networking abilities of mobiles. |

| 3.3 Power issues on domination and control of mobile technology that shape group dynamics and technology usage among both educators and students in LMICs. 3.4 In deprived settings, an intrinsic economic motivation supports the adoption of mobile devices as an educational tool. | 3.4 Technologies often reinforce or trigger existing practices rather than establishing new practices. |
|--|---|
| Mobile devices are intrinsically associated with status and professionalism leading to higher perceptions of schools/teachers who possess the devices. [Tech as a social and professional status symbol] | 4. Mobile devices facilitate the act of teaching leading to a better perception of, and satisfaction with, the teaching profession (in return increasing the motivation to teach). [Teaching experience] |
| 4.1 Teachers who master mobile technologies receive respect and recognition from peers and function as peer educators . | 4.1 Teachers report teaching as more enjoyable and less stressful as learners are easier to handle, more attentive, and motivated. In addition, they have greater variety of ready-to-go content to choose from. |
| 4.2 Parents and community members experience pride of being part of mobile learning programmes and access educational services more strongly that incorporate mobile devices. They also have higher perceptions of teachers and schools that use mobile technologies. | 4.2 Teachers feel more efficient and better organised |
| 4.3 Teachers feel more professional and valued in their role in society. Their self-perception and job satisfaction increases. | 4.3 Teacher feel more connected and, in rural areas, less isolated . |
| 4.4 Teachers express higher opinions of students who are learning with the mobile devices. | 4.4 Teachers feel more recognised and valued in their work and taken serious as professionals with an increased social status and respect from the community and learners |
| 5. The mobile and contextualised use of technologies is determined and limited by formal educational actors and social structures. [Informal use limited for learners] | Mastery of technology is large motivation for learning with technology and a source of pride, confidence, self- efficiency (both teachers/adult learners) [Self-efficacy] |
| 5.1 Teachers report a range of mobile and contextualised technology usages. | 5.1 Mastering technology use improved teachers and learners' self-perceptions (pride). |
| 5.2 Students rarely are allowed to use mobile technologies unsupervised during formal educational activities. | 5.2 Acquiring a skill considered as advanced, increased teachers and learners' self-efficacy and aspiration . |

| 5.3 Teachers and parents are reluctant to allow the possession of school-owned devices outside school hours (i.e. taking devices home) for social and financial reasons. | 5.3 Community and peer recognition and status reinforced the above and in return led teachers and student to educate their peers on the use of technology. |
|---|---|
| | 5.4 However, proficiency of using and controlling mobiles in education also served as an exercise of power for educators and learners |
| A range of systemic issue negatively affected the implementation of ML4D programmes in LMICs. [Systemic challenges] | 6. The usage of mobile technologies in LMICs can change the relationship and interaction between students and teachers [Relationships and interactions] |
| 6.1 The practice of constant teacher transfers to different schools challenged the consistent and sustainable use of technologies. | 6.1 There was an overall shift towards teachers assuming a role of facilitating knowledge rather than being the holder and transmitter of knowledge leading to more student-led interaction and open teacher-student relationships but: |
| 6.2 Frequent strikes disrupted mobile learning programmes leaving less time for learners and teachers to apply the devices. | 6.2 Teachers preferred learner-centered methods that they were still in control of , and often adopted learner-centered methods for the efficacy in reducing their work burden rather than their pedagogical grounding. |
| | 6.3 While students generally welcomed a more interactive and participatory approach, there was also a theme which d id not agree with a shift away from teacher's powers. |
| 6.3 Absenteeism and teachers and learners were forced by economic reasons to cut short or miss their educational commitments. | 6.4 Technology itself can become an instrument of power and both teacher and students used tech to dominate and exercise power. |
| | 6.5 The shift towards more interaction and open relations is not intrinsically connected to the use of technology. |
| 6.4 Learners and teachers rarely spend as much time with devices as assumed by programme design. | 7. Teacher training is key to ensure technology adoption and its pedagogic use during teaching [Teacher training] |

| 7.1 In order to feel in control of technology use , teachers need |
|---|
| to be explicitly trained for it; and receive the technology before |
| learners. This makes teacher training (or professional |
| development) the ideal vehicle to deliver ML interventions. |
| 7.2 Training on technology use to focus on pedagogy not ICT |
| skills; and about subject expertise (i.e. show how tech can (i) |
| support the teaching of the content and (ii) give teacher |
| additional subject knowledge). |
| 7.3 Collaboration, peer-support, and role models are in |
| particular supportive to increase teachers' educational use of |
| mobile technologies. |
| 7.4 Training delivered via mobile tech can reduce teachers time |
| away at professional development courses |
| 8. The educational use of mobile technology can support the |
| accumulation of social capital |
| [Social capital] |
| 8.1 In case device use is not limited, mobile technology is shared |
| widely and leads to large spill-over effects . |
| 8.2 The access to information facilitated by mobile devices is |
| the second descendence of the second second second terms and a |
| valued beyond the educational sector and is used to support a |
| range of social and cultural activities. |
| valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, |
| valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. |
| valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of mobile technologies increasing social learning, networking, and |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of mobile technologies increasing social learning, networking, and communities of practice (groups of learning) |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of mobile technologies increasing social learning, networking, and communities of practice (groups of learning) 8.5 Possession of technology and mastery of use is associated |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of mobile technologies increasing social learning, networking, and communities of practice (groups of learning) 8.5 Possession of technology and mastery of use is associated with social standing and allows teachers in particular to serve |
| Valued beyond the educational sector and is used to support a range of social and cultural activities. 8.3 The ability to connect with mobile devices through friends, colleagues, and relatives was highly valued for social reasons. 8.4 There is a strong theme running across the above that collaboration and peer-to-peer support is a key value of mobile technologies increasing social learning, networking, and communities of practice (groups of learning) 8.5 Possession of technology and mastery of use is associated with social standing and allows teachers in particular to serve as role models and local leaders. |

| 8.6 Parents and community members experience pride of being |
|---|
| part of mobile learning programmes and access educational |
| services more strongly that incorporate mobile devices. They |
| also have higher perceptions of teachers and schools that use |
| mobile technologies. |
| 9. The educational use of mobile technology can support the |
| accumulation of economic capital |
| [Economic capital] |
| 9.1 Teachers and learners assume a direct economic benefit |
| from their acquisition of IT skills (e.g. better employment |
| opportunities; abilities to take part in the information society) |
| 9.2 Teachers assume that learners can use mobile devices to |
| look for jobs online. |
| 9.3 In rural communities, parents and community members |
| hoped that the provision of mobile technologies would support |
| the development of the local community and slow the urban |
| migration of youth. |
| 9.4 In the context of adult learning, there was an intrinsic |
| economic motivation to use mobile phones in order to obtain |
| business and save costs and phone calls. |
| |
| 10. A theme of using mobile technologies to support |
| empowerment ran through a sub-set of studies, but was |
| not focused on the particular role of mobile learning in this |
| process. |
| [Empowerment] |
| 10.1 General belief that access to information can empower |
| communities and people, but marginal ideas that pedagogies can |
| be built around that to make them aware of their oppression and |
| overcome it akin to Freire's critical pedagogy. |
| 10.2 Women empowerment through literacy acquisition |
| mentioned in a small sub-set of studies. |
| 10.3 There were constant suggestions but few empirical |
| examples of using mobile technologies to support the co- |

| | construction of knowledge (e.g. learners generating content; teacher designing new exercises; local knowledge fed into the educational process)10.4 Mastery of mobile technology to support self-worth and aspiration |
|--|--|
| NON-Themes: | NON-themes: |
| There is no systematic empirical evidence that different groups or actors in education are able to exploit or access mobile devices differently in formal education; gender emerges as a determinant of access in | There was no systematic empirical evidence to indicate that mobile technology presents a mechanism to access otherwise inaccessible education services. |
| unstructured informal education programmes. | 2. There was no systematic empirical evidence that access to educational materials is a key mechanism |
| There is no systematic empirical evidence that educators and learners develop dependencies on mobile technologies in formal education; illiterate learners in informal programmes, however, are at risk of developing device-based literacies. | 3. There was no systematic empirical evidence that mobile devices fostered life-long learning. |
| There is no systematic empirical evidence that mobile technologies distract learners and promote illicit social behaviour. | There was no systematic empirical evidence to indicate that mobile technologies caused harm to educational actors in LMICs. |
| There is no evidence of unsustainable cognitive loads created due to the usage of mobile devices. | There was no systematic empirical evidence on mobile learning's effects on systemic change in education. |
| There is no empirical evidence that the effect and usage of mobile technologies wears-off over time. (novelty effect) | 6. There was no systematic empirical evidence on the theme of leadership in ML4D. 7. There was no systematic empirical evidence on the theme of management in ML4D. |

Appendix 6.2: Example of analysis in thematic synthesis

This appendix serves to provide an illustration of the detailed process underlying the thematic synthesis. It supplements the description of the methods for thematic synthesis in chapter 4, section 4.3.4 and the recap of these linked to the thematic synthesis findings in chapter 6, section 6.1.3. The below illustration is linked to the analytical theme on mechanism 1: Pedagogy and practice.



Research findings: Ekanayake 2013 Mobile phone images supported bringing the outside world into the classroom making use of authentic contexts and students' science learning personally relevant.

Research findings: Pimmer 2013 The interviewees reported accessing information in situ as needed in hospitals to understand a particular case or when they were studying for exams and were unable to find relevant information in their textbooks. In this sense, mobile devices supported learning and sense-making that arose within the immediacy of a situation by linking codified knowledge to situated cognition.

Research findings: Leach 2005 It improves language skills for those [Xhosa speakers] who do not speak English. At the beginning they are stiff, we play games to facilitate their ICT skills! Now they love working and learning. We have made isiXhosa stories that are related to their experience. **Stage 2:** Configuration of line-by-line codes into descriptive themes

Descriptive theme 1

• interactive and collaborative learning

Descriptive theme 2

• authentic educational approach

Descriptive theme 3

personalised and targeted learning

Descriptive theme 4

increased access to information

Descriptive theme 5

 variety in teaching and learning activities **Stage 3**: Configuration of descriptive themes into analytical themes that are unit of analysis

Analytical theme

Mechanism 1: Pedagogy and practice

Mobile learning in LMICs can change the way education is practiced, leading to a number of more learner-centred educational approaches

Research findings: Onguko 2013

We find that PD offered through blended learning on appropriate technologies, provides teachers with personalized learning. Studying through self-directed approaches, teachers are able to pace their study. They adjust their self-directed study to suit their own schedules determined by their workload and other communal engagements.

Research findings: Kim 2011

This enrichment activity (i.e., consists of reading with visual effects, listening narrations and own recorded voices, and reflecting, etc.) was personal and at the individual level because each student freely selected own materials and interacted at own pace.

Stage 2: Configuration of line-by-line codes into descriptive themes

Descriptive theme 1

• interactive and collaborative learning

Descriptive theme 2

• authentic educational approach

Descriptive theme 3

personalised and targeted learning

Descriptive theme 4

increased access to information

Descriptive theme 5

 variety in teaching and learning activities

Stage 3: Configuration of descriptive themes into analytical themes that are unit of analysis

Analytical theme

Mechanism 1: Pedagogy and practice

Mobile learning in LMICs can change the way education is practiced, leading to a number of more learner-centred educational approaches

Two aspects are evident from the above analysis: selfdirected learning and gender dimensions. The community came together, defined the learning goals based on individual needs, identified the resources and strategies for

Research findings: Balasubramanian (2010)

learning, and are in the process of evaluating the outcomes. The blending of vertical and horizontal transfer of knowledge helped individual learners to learn in their own time and at their own pace.

Research findings: Wu 2011 So it's really helpful, not just for us, but also for the doctors. Because most times during ward rounds, they ask us to check on our tablets to look up information to make sure what they are doing is right, so it helps them as well in managing patients.

In summary, undergraduate medical student participants in the focus group discussions felt that the tablets benefitted their medical education as well as patient care during clinical rotations. In particular, they appreciated the constantly available source of information and the opportunities for communication.

Research findings: Zelesny-Green 2014

In other instances of afterschool mobile phone use for education and learning, the NDSS learners employed the devices for informationgathering purposes related to both their informal and formal education: I also use the phone to Google and search for various information about the topics...that I learn...I use

the mobile phone to access the internet so that I can get information about my education.

Research findings: Bello-Bravo 2012

As for women and girls it is more difficult to obtain information about controversial topics like forced marriage, divorce, the animation approach providing this information could help to solve that gap. **Stage 2:** Configuration of line-by-line codes into descriptive themes

Descriptive theme 1

• interactive and collaborative learning

Descriptive theme 2

authentic educational approach

Descriptive theme 3

personalised and targeted learning

Descriptive theme 4

• increased access to information

Descriptive theme 5

 variety in teaching and learning activities

Stage 3: Configuration of descriptive themes into analytical themes that are unit of analysis

Analytical theme

Mechanism 1: Pedagogy and practice

Mobile learning in LMICs can change the way education is practiced, leading to a number of more learner-centred educational approaches

Research findings: Sahni 2008

Additionally, we noticed that the instructors over time used a greater variety of representational tools to facilitate learning, such as drawing pictures on the blackboard, or writing verbally spoken questions on the blackboard. The teachers also showed increases in requests that students think critically about the subject material, approach the blackboard. solve equations, answer questions, and assist other students in solving a difficult question through group-work and group-activities.

Research findings: Leach 2005 ICT use extended the range of teachers' existing pedagogic practices.

Research findings: Masperi 2008

The focus on audio and video material and interactive learning techniques offered pedagogical advantages in combining learner-centred and outcome-based activities with continuous assessment. It was also noted that the flexible and diversity in learning approaches allowed for better class management and small group work, supporting the development of social, leadership and interaction skills in the learners. Teachers reported that shy and slower learners were positively affected by the direct interaction with technology and were more willing to engage in learning activities.

Stage 2: Configuration of line-by-line codes into descriptive themes

Descriptive theme 1

• interactive and collaborative learning

Descriptive theme 2

authentic educational approach

Descriptive theme 3

personalised and targeted learning

Descriptive theme 4

increased access to information

Descriptive theme 5

 variety in teaching and learning activities

Stage 3: Configuration of descriptive themes into analytical themes that are unit of analysis

Analytical theme

Mechanism 1: Pedagogy and practice

Mobile learning in LMICs can change the way education is practiced, leading to a number of more learner-centred educational approaches

Appendices

Appendix 6.3: A narrative summary of the best available aggregative evidence

The meta-analysis generated a robust positive effect of mobile learning, which is based on the statistical synthesis of the evaluations of 17 different ML4D programmes. It aggregated the diverse finding of these evaluations into one combined pooled effect size. Having established ML4D overall effectiveness through this analysis, I outline briefly in the following how some of these ML4D interventions achieved their positive impact in practice. This narrative summary is based on the seven studies judged of the lowest risk of bias, which therefore were assumed to present the best available evidence included in the aggregative module. It serves to give the reader a better understanding of mobile learning's effectiveness that is not merely based on a numerical number.

Four of the seven studies were conducted as part of a wider development programme. Aker and colleagues (2012; 2015) added a SMS module to an existing adult literacy campaign (Project ABC) in rural areas in Niger. Similarly, Cole and peers (2012) used mobile phone technology in a rural setting in India. The Avaaj Otalo programme presented a mobile phone-based agricultural extension service aiming to reduce pesticide use and to encourage the adoption of more effective agricultural practices. He and colleagues (2008) worked with an English language training program developed by Pratham, an Indian network of NGOs that aims to change the way that English is taught in Indian classrooms. They introduced PicTalk a mobile gaming device into the ongoing programme approach to facilitate more interactive learning.

The PicTalk device is based on Interactive Paper Technology and rolled out in urban primary school on a 1:1 model of ownership. The device was applied in English and math classes. Learners could access visual and verbal hints on the machine while engaging in the learning games and the device was programmed in local dialect, Marathi. Apart from the mobile learning game the devices also provided additional features to create picture diaries and poems. He and colleagues' (2012) evaluation of PicTalk found gains of about 0.25-0.35 standard deviations in students' knowledge of English. They further identified two important mechanisms that moderate the programmes effectiveness. When the mobile learning intervention was implemented by teacher, who directed the usages of the devices actively, lower performing students benefited more from the mobile learning. On the other hand, in case device use was self-directed, leaving learner at their own will to access the intervention, higher performing students were found to benefit more. Teaches and learners identified the device as an acceptable educational tool and the overall programme appraisal was since positive.

The Avaaj Otalo extension programme allowed farmers access to locally relevant agricultural information through a toll-free hotline, daily SMS messages, and a peer-to-peer support facility. The Avaaj Otalo interface features a touch-tone navigation system with local language prompts, developed specifically for ease of use by semi-literate farmers. The programme in particular encourages farmers to access the service while at work in their fields to allow for a context-relevant learning experience. The programme aimed at improving farming practice and output, as well as improvements in agricultural knowledge.

Avaaj Otalo aimed to not merely tell farmer what to do (e.g. which pesticide not to use) but rather aimed to generate an improved understanding of farming processes and their interaction with the local eco-system. Cole and colleagues (2012) evaluation found a small positive but statistically non-significant impact on agricultural knowledge. As at the same time farming output improved significantly they conclude that farmers trusted and adopted the information received by the mobile phone-based extension. However, learning did not seem to have taken place. The trust placed in mobile phones though is a notable finding. Farmers expressed a preference for mobile extension services over the local extension officers, citing reliability, timeliness/in-context, and quality as main reasons. Avaaj Otalo was since highly accessed and farmers receiving the service turned less often to other farmers and input sellers for agricultural advice.

Jenny Aker and colleagues (2012; 2015) evaluations of the Project ABC programmes arguably presents the most well-known assessment of a ML4D intervention. An existing adult literacy and numeracy campaign was supplemented by a SMS module that entailed training on how to use a mobile phone as well as a subsidised supply of basic phones. As the use of mobile phones, e.g. reading and writing SMS, requires a certain amount of functional literacy, mobile phones training and ownership was assumed to reinforce the existing literacy campaign. This assumption was validated in two large RCTs. The evaluation since showed that mere mobile phone usage supported literacy and numeracy acquisition. These learning gains further were more likely to be retained one year after the programme had ended.

The studies generate a wealth of rigorous evidence on mobile phones impact on informal learning. They identified, among other, that younger learner showed higher learning gains and that the overall impact was not driven by differences in teacher quality or in teacher and student attendance. One study also tried to assess whether learners' overall perception and value of education had increased. This assessment could have potentially served as an indicator of empowerment but due to the poor outcome instrument, it was judged at critical risk of bias in this review. Most importantly, however, it revealed how economic factors provided an intrinsic incentive to obtain functional literacy in order to be able to operate mobile phones. Since SMS in Niger were much more affordable than voice calls, illiterate rural learners were excluded from the financial reward the proficiency of SMS technology offered. This served as a large incentive to obtain the literacy skills required to exploit the financial savings associated with SMS. All in all, the studies produced highly rigorous evidence that simple and relatively cheap mobile technology can serve as an effective vehicle for learning among rural populations.

The studies by Pitchford and colleagues (2014) and Kaleebu and peers (2013) and Piper and peers (2015) were embedded in a formal educational setting rather than in conjunction with an ongoing development programme. They were designed to support formal education systems challenged by widespread poverty in Malawi, Kenya, and Papua New Guinean respectively. Pitchford and colleagues (2014) and Piper and peers (2015) implemented the only tablet-based programmes included in the meta-analysis. Likewise, Kaleebu and peers (2013) and Piper and peers (2015) implemented the only teacher-centered ML4D intervention in the included sample.

The innovative SMS Story project in Papua New Guinean aimed to improve the teaching of

reading to learner through the delivery of daily short stories and complementary lesson plans via SMS technology to primary teachers (Kaleebu et al 2013). The programme aimed to increase interactivity and learner inclusion in the learning process. It assumed that teachers' applied pedagogies could be changed through the provision of teaching content and support on mobile phones. The programme was positively received by teacher, who actively took part in its iterations, e.g. calling for a need to deliver SMS a day in advance rather than in the morning prior to the lesson. The intervention was highly effective in increasing English test scores for the involved learners. The evaluation since provided evidence that mobile technologies targeted at teachers could similarly improve learning outcomes. The learners, in this case, experienced academic benefits of the introduction of mobile technologies even though the devices were introduced at teacher level.

Piper and peers (2013) evaluated the Primary Math and Reading Initiative (PRIMR) Kisumu Information and Communication Technology (ICT) intervention in Kenya. The programme aimed to assess different types of technology provision to assess their impact and cost-effectiveness. Three programme approaches were tested: (i) tablet provision with pre-loaded basic educational content to teaching assistants; (ii) tablet provision to teachers themselves with advanced educational content and learning analytics abilities; and (iii) ereaders provided to learners at a 1:1 ratio. The outcomes of the three different treatments were positive on learning gains as follows: The effect size for e-readers, the teacher tablets, and the teaching assistant tablets was .35 SD, .47 SD, and .44 SD, respectively. These effects were not statistically significant from each other, but taking into consideration the cost-effectiveness of each programme arm, large differences are revealed. The teaching assistant tablet programme is by far the most cost-effective approach being almost two times more cost-effective than the teacher tablets, and 10 times more cost-effective than the e-reader learner group. The latter is not cost-effective at all being six times less costeffective than the control group despite the larger gains in learning outcomes. All in all, this study, too, finds that providing mobile technologies to teachers is an effective approach to achieve positive outcomes at a learner level. What is more, it is also a more cost-effective approach than focusing technology provisions to learners.

Lastly, Pitchford and colleagues' (2014) evaluation of the tablet-based Masamu intervention focused on a tablet-based mobile learning programme in Malawi. As a note of caution it needs to be highlighted that the study was funded and commissioned by EuroTalk, the commercial company who developed the applied mobile learning application. The study also generated an outlying effect size of 1.68, which diverted more than a complete standard deviation from the pooled standardised effect size. Having said that, the study used a highly sophisticated pedagogic application on a high-end device (iPad mini), which might offer an explanation for its large effect size. EuroTalk designed a mobile math learning game that since has been commercially offered in HICs. Evaluations of the application, Pitchford and colleagues, moreover, tried to link the improvements in learning gains to increased level of ambition and progressive attitudes, which then could be described as empowerment. However, the outcome instrument was of critical risk of bias and investigating the life ambitions of primary learners, arguably, is unlikely to present a valid indicator of empowerment.

In sum, this short narrative summary underlines the heterogeneity in context and programme

approach of the reviewed ML4D interventions. The interventions' impacts are far more nuanced than a single effect size could reflect. From this short juxtaposition of the best available evidence a number of themes and questions arise. For example, He and colleagues found a greater benefit for weaker students if the technology is applied by teachers, whereas Aker and peers establish a larger effect for younger students. However, none of these themes are based on a systematic investigation of the evidence and merely present individual observations. To fully understand these contexts and mechanisms explained the identified effectiveness of ML4D, we require an equally rigorous review and transparent synthesis of a different type of evidence. Understanding contexts and mechanisms subscribes to a configurative mode of analysis rather than an aggregative approach as applied in this meta-analysis.

Appendix 7.1: On the absence of negative case study findings

Lastly, there is also a question whether desirability bias has influenced some of the teachers' responses. There is a curious absence of negative themes in my qualitative case study research and even when probed for what they did not value about using the tablets, teachers reported little feedback that could be seen as a criticism of the ICT4RED.

First, the ICT4RED programme did present a large-scale intervention in teachers' professional lives and the functioning of their schools. It needs to be kept in mind that the rural Eastern Cape is by many measures South Africa's most impoverished region and some of the ICT4RED schools lack basic infrastructure such as built classroom structures. In this context, at an individual level, the ICT4RED provided each educator with a high-end 10-inch tablet, a monthly voice and data allocation, and access to alternative educational content and lesson plans. This was complemented by a certification of becoming an 'ICT champion', bi-weekly trips to the administrative capital including hotel accommodation, travels to ICTrelated summits such as the Google summit, and visits by national and district education policy-makers. Professionally as well as socially, ICT4RED therefore did present a major input into teachers' lives. In addition, at a school level, the programme also provided major inputs and changes in infrastructures. The schools gained teaching equipment such as projectors and whiteboards, laptops, and tablets for their pupils, as well as WIFI-connectivity and charging and storage facilities. Schools not connected to the electricity grid gained preferred access to the electricity roll-out in their communities. It could therefore be argued that the ICT4RED presented such a major stimulus in teachers' professional and social contexts to justify the weight of positive data collected. Though, this line of thought assumes a rather naïve conception of technology provision and its application and transformation into social and professional well-being.

Second, the above narrative of ICT4RED as a major intervention into teachers' lives and schools' functioning can be used to set up a situation in which teachers feel compelled to provide a positive account of the programme. This could be in particular the case as the schools' acquisition of technological inputs depended directly on teachers' participation and completion of the professional development course. Teachers therefore had a major responsibility (as well as social reward) for the schools' development of IT capabilities. In this context, teachers might have perceived me as an evaluator of the programme whose findings and recommendations might have affected their own and the schools' continued access to technological inputs. I attempted to counter this assumption by stressing the independence of my research from both the internal and the external evaluation verbally and in the information sheet provided as part of the data collection. However, the introduction of myself to principals was conducted through formal channels involving education district officials and I can therefore not rule out that teachers perceived me and my research as an evaluation of the ICT4RED on which a future access to the programme might have depended.

Third, the manner in which the interview schedule focused around what teachers valued about the technology usage might have influenced the nature of the data. To be clear,

exploring valued beings and doings presents an analytical device of the CA and I could not have avoided probing themes in this direction. However, I paired each interview question exploring valued beings and doings with a primer for participants to report beings and doings they did *not* value. I further can cross-reference valued educational and informational beings and doings with the data collected during the classroom observations⁸⁶.

In summary, I would argue that all three reasons possibly influenced the content of the data collected. For reasons two and three, I implemented quality assurance measures to the best of my ability, while the scale and nature of the ICT4RED intervention in the context of the Cofimvaba school district is outside my control. When interpreting the results of my qualitative case study, these three caveats should be kept in mind, however.

⁸⁶ This is not possible for the functionings related to economic and societal capabilities as these referred to beings and doings that took place outside the school environment.

Appendix 7.2: Exhaustive list of all capabilities and functionings

| Capabilities (opportunities) | Valued beings and doings (functionings) |
|--|--|
| | |
| Informational capabilities | |
| To be able to use ICT effectively (be tech champion) [ICT usage] | operate a tablet device with its particular apps and affordances (e.g. camera) connect to and use the internet achievement and pride for tech mastery independent usage and curiosity to 'keep trying new things' differentiation of devices and educational application of affordances |
| | |
| 2) To be able to find, evaluate, use, and process information [Information literacy] | access information instantly, more conveniently access information independently and tailored to personal preferences (choice—get anything I want) stay up-to date translate access to information into teaching relevant knowledge translate access to information into socially relevant knowledge (bible app at church) |
| 2) To be able to compare and | |
| [Connectivity] | 11. communicate instantly and personal 12. increase and maintain professional networks 13. stay in touch with family and friends 14. stay in touch with learners to follow their path 15. reach out to the world (less isolated/agency) |
| 4) To be able to engage in one's professional identity and practice anywhere, anytime [Mobility] 5) To be able to extend | 16. access to device anywhere, anytime 17. more convenient than traditional teaching tools and more dynamic and professional in appearance 18. prepare/gather teaching content anywhere, anytime 19. teach anywhere, anytime (remote instructions, two classes at once) 20. collection of teaching materials all in one place 21. to possess and identify with a personalised and centralized teaching tool (value of tablets/personalisation; cellphone analogy) |
| 5) To be able to extend | 22. collect teaching ideas and contexts anywhere |

| | professional being/realities | (take pictures at home eg) |
|--------------------------|------------------------------|--|
| | across contexts | 23. provide education across time & space (record |
| | | lesson for sick children; let children take home |
| | [Merging realities/contexts] | videos to study further/revise at night) |
| | | 24. contextualise students learning experiences |
| | | across space and time |
| | | 25. be regarded as a professional in occupational |
| | | and social domain |
| | | 26. transfer valued beings across time and space |
| | | (apartheid deprivation; showing of to former |
| | | teacher colleagues visiting) |
| | | |
| | To be able to create, share | This capability while realised was not valued as it did |
| | and use local content | against the highly top-down fashion of teaching in SA in |
| | | which teachers did not deem it appropriate to deviate |
| | [Content generation] | from the Departmental Syllabus. |
| Educational capabilities | | |
| 6) | To be able to improve | 27. learners are more motivated |
| | learning achievements | 28. learners have more fun |
| | | 29. learners succeed in tests |
| | [Student success/support] | 30. teachers achieve teaching goals |
| | | |
| () | I o be able to change and | 31. Increase interaction and participation (IC14RED |
| | explore teaching methods | methods, group work) |
| | | 32. provide instant and personal feedback |
| | [Pedagogies] | 33. access different and more appropriate content |
| | | 34. Increase subject knowledge |
| | | 35. teach outside (fieldwork) |
| | | So. choose from a variety of methods (creativity) |
| 8) | To be able to become a | 37 co-produce produce knowledge (they are |
| 0) | facilitator of knowledge | teaching me/do their own research/bring their |
| | | own information) |
| | [Role of teacher] | 38. create room for discovery (learn by themselves |
| | | /bring their own information/become problem |
| | | solvers) |
| | | 39. teach learner-centred |
| | | 40. deliver more relevant teaching |
| | | J |
| 9) | To be able to enjoy | 41. organise administration/limits my workload (no |
| | teaching more | need for paper; marking; books; all in one place; |
| | | content stays) |
| | [Occupational satisfaction] | 42. easier to control (tool of power; better behaviour) |
| | | 43. boost moral |
| | | 44. pleasure in using tech (ease of use) |
| | | |
| 10 |) To be able to innovate | 45. changed view about technology as an |

| | educational tool | |
|----------------------------------|---|--|
| [Innovation] | 46. know how to teach with technology | |
| | 47. learning by doing | |
| | 48. make mistakes together/tolerance for failure | |
| | 49. life-long learning | |
| | 5 5 | |
| 11) To be able to work in | 50. exchange teaching content with other teachers | |
| collaboration | 51. gain technological skills through working | |
| | together | |
| [Collaboration] | 52. extend teaching horizon with international | |
| | teachers on the internet | |
| | 53. Observation of after-hours class | |
| | | |
| Economic capabilities | | |
| 12) To be able to be employed | 54. build IT skills that are in demand | |
| | 55. train other teachers in IT skills | |
| [Employment] | 56. stay employable in a changing time | |
| | 57. transfer to a different post | |
| | 58. make learners more employable | |
| | | |
| 13) To be able to build social | 59. to be able to guide learners | |
| capital | 60. to be able to access & share up-to-date job | |
| | information | |
| [Social Capital] | 61. to be able to market oneself online | |
| | | |
| 14) To be able to build the | 62. Increased enrolment of learners at IC14RED | |
| brand | Schools | |
| [Markating] | 63. Increase the returns to education at ICT4RED | |
| [warkeung] | SCHOOIS 64. train other teachers in IT skills | |
| | | |
| 15) To be able to extend | 65, further teaching gualifications | |
| educational opportunities | 66. provide additional teaching services in the | |
| | community | |
| [Online education] | 67. support own children's education | |
| | | |
| 16) To be able to access digital | 68. save costs of going to town | |
| services | 69. use online banking | |
| | 70. make online appointments | |
| [Digital services] | 71. use online call rates | |
| | | |
| | Societal capabilities | |
| 17) To be able to be impress | 72. being respected by the learners | |
| | 73. being respected by other teachers | |
| [Respectability] | 74. being respected by the community | |
| | 75. impress people | |
| | | |

| 18) To be able to be 'on the | 76. have access to same educational |
|-------------------------------|---|
| same page' | inputs/resources |
| | 77. attract learners/teachers back to rural schools |
| [Recognition] | 78. engage on par with 'reputable' educational |
| | institutions |
| | 79. be the first to use technology |
| | |
| 19) To be able to create a | 80. appear more professional |
| professional teaching | 81. increase the reputation of the teaching |
| identity | profession |
| | 82. attract learners to become teachers |
| [Professional identity] | 83. be appreciated by the government |
| | 84. merging professional and social |
| | realities/identities |
| | |
| 20) To be able to 'grow' as a | 85. being empowered/enlightened/uplifted |
| person | 86. become a role model |
| | 87. gain confidence and pride (over tech mastery) |
| [Self-actualisation] | 88. contribute one's part to society |
| | |
| 21) To be able to support the | 89. community excitement (feel important and on the |
| local community | same page) |
| | 90. increase community cohesion (no more break- |
| [Community development] | ins) |
| | 91. create opportunities (community development; |
| | learners take up good jobs) |
| | |